University of Southampton

1.000

Faculty of Engineering, Science and Mathematics

School of Mathematics

Epimorphic images of simplicial Coxeter groups and some associated hyperbolic manifolds

by

Cormac Diarmuid Long

Thesis for the degree of Doctor of Philosophy

June 2007

UNIVERSITY OF SOUTHAMPTON

ABSTRACT

FACULTY OF ENGINEERING, SCIENCE AND MATHEMATICS SCHOOL OF MATHEMATICS

Doctor of Philosophy

EPIMORPHIC IMAGES OF SIMPLICIAL COXETER GROUPS AND SOME ASSOCIATED HYPERBOLIC MANIFOLDS.

by Cormac Diarmuid Long

Abstract

In this work torsion-free normal subgroups of the Lannér groups Γ_i are studied. All torsion-free normal subgroups of the orientation-preserving subgroups Γ_i^+ whose factor groups have the form $L_2(q)$, where $q = p^n$ and p is a prime, are classified. In the case of each group, some examples of manifolds are constructed and their homology is computed. Minimal index torsion free subgroups of each Lannér group are also constructed.

Computational techniques are developed to construct complete lists of conjugacy classes of subgroups of low index in these groups. These lists are then used to test the theoretical results proved in this thesis and also to search for specific subgroups. Computational techniques are also developed to calculate the action of the isometries of these manifolds \mathcal{M} on their homology groups. This gives $H_1(\mathcal{M})$ the structure of an Isom(\mathcal{M})module, which allows for the construction of arbitrarily large manifolds exhibiting a high degree of symmetry.

The computational techniques developed in this work are applied to the 4-dimensional Coxeter group [5, 3, 3, 3] and a detailed study of the low index subgroups of this group has been implemented. The existence of torsion free subgroups of index 115200 is established and a possible approach towards determining the minimal index torsion-free subgroups of this group is outlined.

Contents

1	Int	roduct	ion	18
	1.1	Introc	luction	18
	1.2	Aims	of this work	20
	1.3	Struct	ture of this Thesis	21
2	Bac	kgrou	nd	23
	2.1	Introd	luction to Coxeter groups	23
	2.2	Conju	gacy classes in finite Coxeter groups	28
		2.2.1	Conjugacy classes in $A_1 \times A_1 \times A_1$	29
		2.2.2	Conjugacy classes in $A_1 \times I_2(3)$	30
		2.2.3	Conjugacy classes in $A_1 \times I_2(4)$	30
		2.2.4	Conjugacy classes in $A_1 \times I_2(5)$	30
		2.2.5	Conjugacy classes in A_3	31
		2.2.6	Conjugacy classes in B_3	31
		2.2.7	Conjugacy classes in II_3	32

/

	2.3	Discrete extensions of the 3-dimensional simplicial Coxeter groups .	32
	2.4	Determining low index subgroups of finitely presented groups $\$	34
3	$\mathbf{E}\mathbf{p}$	imorphic images of the $[5, 3, 5]$ Coxeter group	36
	3.1	The Coxeter group $[5, 3, 5]$	36
	3.2	Quotients of Γ^+ isomorphic to A_5	38
	3.3	Some facts about the simple groups $L_2(q)$	41
	3.4	Quotients of Γ^+ isomorphic to $L_2(q), p \notin \{2, 5\}$	42
		3.4.1 The restriction to Γ_0^+	42
		3.4.2 Extension to Γ^+	43
	3.5	The cases $p = 2$ and $p = 5$	46
		$3.5.1 p = 5 \dots \dots$	46
		$3.5.2 p = 2 \dots \dots$	46
	3.6	The main result	47
	3.7	Extending to Γ	48
4	Fur	ther maps to linear groups	50
	4.1	Preliminaries	51
		4.1.1 Coxeter schemes containing [5, 3]	51
		4.1.2 Coxeter schemes containing [4,3]	52
		4.1.3 Extending $\Gamma^+ \twoheadrightarrow L_2(q)$ to $\Gamma \twoheadrightarrow L_2(q)$	53
		4.1.4 A note on the order of elements in $L_2(2^n)$:	54
	4.2	$\Gamma = T_1[2, 2, 3; 3, 5, 2] \dots$	55
			-

	4.2.1	Γ^+ to $L_2(q)$	55
	4.2.2	Extending to $\Gamma \twoheadrightarrow L_2(q) \ldots \ldots \ldots \ldots \ldots \ldots$	57
4.5	$\Gamma = 7$	$\Gamma_2[2, 2, 3; 2, 5, 3]$	58
4.4	$\Gamma = 7$	$\Gamma_3[2, 2, 4; 2, 3, 5]$	58
	4.4.1	Γ^+ to $L_2(q)$	58
	4.4.2	Extending to $\Gamma \twoheadrightarrow L_2(q) \ldots \ldots \ldots \ldots \ldots \ldots$	61
4.5	$\Gamma = 7$	$T_4[2, 2, 5; 2, 3, 5]$	61
4.6	$\Gamma = T$	$\Gamma_5[2,3,3;2,4,3]$	62
	4.6.1	Maps from Γ^+ to $L_2(q)$	62
	4.6.2	The exceptional case $p = 7 \dots \dots \dots \dots \dots \dots \dots$	64
	4.6.3	Extending to $\Gamma \twoheadrightarrow L_2(q)$	65
4.7	$\Gamma = T$	$G_6[2,3,4;2,3,4]$	66
	4.7.1	Normal subgroups with quotient S_4	66
	4.7.2	$\Gamma^+ \twoheadrightarrow L_2(q) \dots \dots \dots \dots \dots \dots \dots \dots \dots $	66
	4.7.3	Extending to $\Gamma \twoheadrightarrow L_2(q)$	69
4.8	$\Gamma = T$	$_{7}[2,3,3;2,5,3]$	69
	4.8.1	$\Gamma^+ \twoheadrightarrow L_2(q) \ldots \ldots$	69
	4.8.2	Case $p = 2 \dots \dots$	72
	4.8.3	Extending to $\Gamma \twoheadrightarrow L_2(q)$	72
4.9	$\Gamma = T_{\delta}$	[2, 4, 3; 2, 5, 3]	74
	4.9.1	$\Gamma^+ \twoheadrightarrow L_2(q) \dots \dots \dots \dots \dots \dots \dots \dots \dots $	74

		4.9.2 Extending to $\Gamma \twoheadrightarrow L_2(q) \ldots \ldots \ldots \ldots \ldots \ldots$	76
	4.10	$\Gamma = T_9[2, 3, 5; 2, 3, 5] \dots$	77
		4.10.1 $\Gamma^+ \twoheadrightarrow L_2(q)$	77
		4.10.2 Exceptional cases	80
		4.10.3 Extending to $\Gamma \rightarrow L_2(q)$	80
	4.11	Conclusion	81
5	Mai	nifolds from the $[5, 3, 5]$ group	82
	5.1	Introduction	82
	5.2	The quotient manifolds M_i	82
	5.3	The tessellations \mathcal{D}/N_i' by 125 dodecahedra	87
	5.4	The tessellation \mathcal{D}/N by 60 dodecahedra $\ldots \ldots \ldots \ldots \ldots \ldots$	91
	5.5	Subgroups of direct squares	92
	. 5.6	The structure of the tessellation \mathcal{D}/N	92
	5.7	The Poincaré dodecahedral spaces	98
	5.8	The structure of the 120-cell	101
	5.9	The 120-cell and \mathcal{D}/N	102
	5.10	The tessellation \mathcal{D}/K	106
	5.11	Quotients of Γ^+ isomorphic to $L_2(19)$	108
		5.11.1 The three normal quotients and their extensions $\ldots \ldots \ldots$.08
		5.11.2 Factorisation in $L_2(19)$.10
		5.11.3 A note on the above method $\ldots \ldots \ldots$.11

		5.11.4	Structure of the quotient tessellations for $L_2(19)$	111
		5.11.5	The three tessellations	113
6	Ma	nifolds	from the other groups 1	18
	6.1	$\Gamma = T$	$\Gamma_1[2, 2, 3; 3, 5, 2]$	119
		6.1.1	Torsion in Γ and in Γ^+	119
		6.1.2	Minimal index torsion free subgroups	120
	6.2	$\Gamma = T_{\rm f}$	$T_2[2, 2, 3; 2, 5, 3]$	123
	6.3	$\Gamma = T_{i}$	$G_3[2, 2, 4; 2, 3, 5]$	124
		6.3.1	Torsion in Γ and in Γ^+	125
		6.3.2	Minimal index torsion free subgroups	126
		6.3.3	Minimal index torsion free normal subgroups	128
		6.3.4	Manifold structure	132
		6.3.5	Other manifolds and minimal index torsion free subgroups .	133
	6.4	$\Gamma = T_4$	[4[2,2,5;2,3,5]	133
	6.5	$\Gamma = T_5$	[2,3,3;2,4,3]	134
		6.5.1	Torsion in Γ and in Γ^+	34
		6.5.2	The action of $L_2(7)$ on $H_1(\mathcal{M})$.37
		6.5.3	Computational results	.38
	6.6	$\Gamma = T_6$	[2, 3, 4; 2, 3, 4]	.39
		6.6.1	Torsion in Γ and in Γ^+	.39
		6.6.2	Maps to S_4	40

		6.6.3	Geometrical description of the two manifolds	141
		6.6.4	Maps to $L_2(7)$	142
		6.6.5	Maps to $A_6 \cong L_2(9)$	143
	6.7	$\Gamma = T$	$f_7[2,3,3;2,5,3]$	144
		6.7.1	Torsion in Γ and in Γ^+	144
		6.7.2	Maps to A_5	145
		6.7.3	Presentations for K_i	146
	6.8	$\Gamma = T_8$	$_{3}[2,4,3;2,5,3]$	148
		6.8.1	Torsion in Γ and in Γ^+	148
		6.8.2	Maps to finite groups	149
	6.9	$\Gamma = T_{\mathfrak{g}}$	[2, 3, 5; 2, 3, 5]	153
		6.9.1	Torsion in Γ and in Γ^+	153
		6.9.2	Maps to $L_2(5) \cong A_5$	154
		6.9.3	Further interesting subgroups	157
		6.9.4	Maps to M_{12}	160
	6.10	Conclu	uding remarks	172
7	The	4-dim	ensional compact simplicial Covotor groups	173
•	THE	unit	ensional compact simplicial Coxeter groups	
	7.1	Motiva	tion	173
	7.2	Conjug	cacy class representatives	175
	7.3	Subgro	ups of the $[5, 3, 3, 3]$ Coxeter group	178
		7.3.1	Low index subgroups of Γ	179

		7.3.2 Maps to $S_4(4)$	81
		7.3.3 Maps to $S_4(5)$	84
	7.4	Characterising maps to simple groups	84
8	Cor	clusion and further work 18	86
	8.1	Conclusion	86
	8.2	Further work	88
Aj	ppen	dix 18	38
A	Tab	les of low index subgroups for the simplicial groups 18	3 9
	A.1	$\Gamma = T_1[2, 2, 3; 3, 5, 2] \dots $	89
	A.2	$\Gamma = T_2[2, 2, 3; 2, 5, 3] \dots $	96
	A.3	$\Gamma = T_3[2, 2, 4; 2, 3, 5] \dots $	99
	A.4	$\Gamma = T_4[2, 2, 5; 2, 3, 5] \dots $	07
	A.5	$\Gamma = T_5[2, 3, 3; 2, 4, 3] \dots $	10
	A.6	$\Gamma = T_6[2, 3, 4; 2, 3, 4] \dots $	12
	A.7	$\Gamma = T_7[2, 3, 3; 2, 5, 3] \dots $	16
	A.8	$\Gamma = T_8[2, 4, 3; 2, 5, 3] \dots $	17
	A.9	$\Gamma = T_9[2, 3, 5; 2, 3, 5] \dots $	19

List of Tables

2.1	Finite irreducible Coxeter groups	25
2.2	Parabolic irreducible Coxeter groups	26
2.3	The nine Lannér groups	28
2.4	Finite Coxeter groups whose conjugacy class structure is required .	29
2.5	Conjugacy classes in $A_1 \times A_1 \times A_1$ and $(A_1 \times A_1 \times A_1)^+$	29
2.6	Conjugacy classes in $A_1 \times I_2(3)$ and $(A_1 \times I_2(3))^+ \ldots \ldots$	30
2.7	Conjugacy classes in $A_1 \times I_2(4)$ and $(A_1 \times I_2(4))^+ \ldots \ldots$	30
2.8	Conjugacy classes in $A_1 \times I_2(3)$ and $(A_1 \times I_2(3))^+$	31
2.9	Conjugacy classes in A_3 and A_3^+	31
2.10	Conjugacy classes in B_3 and B_3^+	32
2.11	Conjugacy classes in H_3 and H_3^+	32
3.1	A presentation for N_1 as an abstract group	40
3.2	A presentation for N_2 as an abstract group	40
4.1	The nine Lannér groups	50

4.2	Subgroup structure in Γ
6.1	Conjugacy class representatives for elements of finite order in Γ_{-} 119
6.2	Conjugacy class representatives for elements of finite order in Γ^+ 119
6.3	Conjugacy class representatives for elements of finite order in Γ 125
6.4	Conjugacy class representatives for elements of finite order in Γ^+ 125
6.5	The generating sets for K_1 and K_2 in Theorem 6.3
6.6	Conjugacy class representatives for elements of finite order in Γ . . 134
6.7	Conjugacy class representatives for elements of finite order in Γ^+ . . 134
6.8	Conjugacy class representatives for elements of finite order in Γ . . 139
6.9	Conjugacy class representatives for elements of finite order in Γ^+ 139
6.10	Presentation for K_1
6.11	Presentation for K_2
6.12	Presentation for K_3
6.13	Generators for K_3 in Γ^+
6.14	Conjugacy class representatives for elements of finite order in Γ 144
6.15	Conjugacy class representatives for elements of finite order in Γ^+ . . 144
6.16	Presentation for the group K_1 of Lemma 6.7.2
6.17	Representating the generators of K_1 as elements of Γ^+
6.18	Presentation for the group K_2 of Lemma 6.7.2
6.19	Representing the generators of K_2 as elements of Γ^+
6.20	Conjugacy class representatives for elements of finite order in Γ_{-} 149

6.21	Conjugacy class representatives for elements of finite order in Γ^+ 149
6.22	Conjugacy class representatives for elements of finite order in Γ . . 153
6.23	Conjugacy class representatives for elements of finite order in Γ^+ 153
6.24	A generating set for $K_3 \ldots \ldots$
6.25	A presentation for K_3
6.26	The images of the generators of K_3 as a subgroup of Γ^+ in H 156
7.1	The 5 co-compact simplicial Coxeter groups with a hyperbolic 4- simplex as a fundamental region
7.2	Simplicial Coxeter group with Non-co-compact simplex Δ^4 in \mathbb{H}^4 174
7.3	Maximal finite subgroups of Γ_1
7.4	Conjugacy classes of elements of prime order in $G_1 = [5, 3, 3]$ 176
7.5	Conjugacy classes of elements of prime order in $G_3 \cong D_{10} \times D_6$ 176
7.6	Conjugacy classes of elements of prime order in $G_2 = [5,3] \times C_2 \cong$ $A_5 \times C_2 \times C_2 \ldots \ldots$
7.7	Conjugacy classes of elements of prime order in $G_4 \cong C_2 \times S_4$ 177
7.8	Conjugacy classes of elements of prime order in $G_5 \cong S_5$
7.9	Number of classes of subgroups of index ≤ 720 in Γ

List of Figures

3.1	A hyperbolic tetrahedron T	36
5.1	Identification of two faces of a dodecahedron	85
5.2	$+\frac{3\pi}{5}$ identification with Petrie polygon marked (dashed line)	86
5.3	Right-handed petrie path	87
5.4	Left-handed petrie path	87
5.5	Local adjacency about v	95
5.6	Partial schematic at v	95
5.7	Vertex structure at $v: I \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	97
5.8	Vertex structure at v : II $\ldots \ldots \ldots$	97
5.9	Vertex structure at v : III	98
5.10	Coxeter's $\{5, 3, 3\}_3$	101
5.11	two linked tori with meridian path of one marked \ldots	103
5.12	Two hundred faces of $\{5, 3, 3\}$ forming the boundary torus of a do- decahedral tower. Meridian path bounded by dotted lines	104
5.13	Vertex identification for \mathcal{M}_0	15

5.14	Edge pairings for \mathcal{M}_0
5.15	Edge pairings for \mathcal{M}_2
5.16	Vertex identification for \mathcal{M}_1
6.1	Tessellation of \mathbb{H}^3 by right-angled dodecahedra
6.2	Standard hyperbolic cube
6.3	Pinched hyperbolic cube
6.4	Identifications for $\gamma = (1, 3, 2, 4) \dots $
6.5	Identifications for $\gamma = (1, 3, 4, 2) \dots $
6.6	Higman diagrams for A_5 acting on 5, 6 and 10 points
6.7	Higman diagram for A_5 acting on 12 points
6.8	Representation of the maps $\Gamma^+ \to A_5$ constructed in Lemma 6.9.1. Black dotted lines correspond to the action of x on points, and red dotted lines correspond to the action of z on points
6.9	A possible transitive action of Γ^+ on 12 points
6.10	Images of S under the action of $(xy^2z)^2$
6.11	Partial subscheme containing the $\{5, 6, 1\}$ Higman diagram 166
6.12	Partial subscheme containing the $\{5, 6, 1\}$ Higman diagram 166
6.13	Diagram 3
6.14	Diagram \mathcal{D} : A_5 action of type $\{6, 6\}$ on 12 points $\ldots \ldots \ldots$
6.15	Action of Γ^+ on 12 points $\ldots \ldots \ldots$
6.16	Diagram 5
6.17	Possible diagrams

6.18	Admissible diagrams	169
6.19	Transitive action of A_5 on 12 points with generators x , y and z	170
6.20	Transitive action of A_5 on 12 points with generators x , y and z	170
6.21	Transitive action of A_5 on 12 points with generators x, y and $z \dots$	170
6.22	Transitive action of $L_2(11)$ on 12 points with generators x, y and $z = 1$	171
6.23	Transitive action of M_{12} on 12 points with generators x, y and z	.71
6.24	Transitive action of M_{12} on 12 points with generators x, y and z	.71

Acknowledgements

First, I would like to thank my Ph.D. supervisor, Prof. Gareth Jones, for providing me this opportunity to be his student and to work on this Coxeter group project. His professional guidance, support and understanding throughout my PhD project has provided me with a good opportunity to start my professional career, as well as a very pleasant working environment within the Pure Mathematics Group during the past three and half years.

Many thanks goes to Prof. David Singerman for his professional advice, valuable theories and lots of interesting discussions. Thanks for his advice and encouragement. I would also like to thank Dr. Jim Anderson for his advice and support.

I would like to thank Dr. Ian Hinder for his technical assistance in setting up the computing cluster, without which the computational results obtained in this thesis would be very much poorer.

I would also like to thank my fellow PhD students, Jon Barker, Gemma Holloway and Jon Williams for providing a willing sounding board whenever I needed to talk through some mathematics: thanks for all the support and encouragement. Thanks also to Dr. Ken Shackleton for many interesting hours spent talking about Coxeter groups and the Curve Complex.

Also to my group of friends: Dr. Jos Akhtman, Stuart Boden, Dr. Sarah Cambell, Dr. Claudia Cerrina, Chris Froud, Dr. Maaten Furlong, Dr. Enrico Gill, Katerina Gargaroni, Dr. Dominik Kunz, Sophie Monje-Dumas, Dr. Fiona Oakley and Tristan Temple. Thanks for supporting me throughout my PhD.

I would also like to thank my sister Rathnait for all her support and advice. A special thanks goes to my parents, without whose generous support and tireless effort I would not be where I am today.

Finally I would like to thank my girlfriend, Dr. Wen Zhang, for all her support and patience while I was writing my thesis. Thank you for believing in me. Chapter 1

Introduction

1.1 Introduction

The classification and characterisation of topological 3-manifolds has had a powerful influence on mathematical research over the past hundred years. Starting with the work of Henri Poincaré [Po] and continuing up to the present-day work of William Thurston [Th1], Richard Hamilton [Ha] and Grigory Perelman [Pe1] [Pe2], the question of classifying all compact 3-dimensional manifolds has yielded many fruitful avenues of research.

In 1982, Thurston proposed his *Geometrization conjecture*. In the case of closed oriented 3-manifolds, the conjecture can be stated as follows:

GEOMETRIZATION CONJECTURE [Th1]

Let M be a closed, oriented, 3-manifold. Then there is a finite collection of disjoint, embedded tori T_i^2 in M, and a finite collection of disjoint, embedded spheres S_i^2 in M, such that each component of the complement $(M \setminus \bigcup S_i^2) \setminus \bigcup T_i^2$ admits a geometric structure, i.e., a complete, locally homogeneous Riemannian metric.

There are a total of eight 3-dimensional model geometries and they are listed as

follows:

- 1) Spherical geometry \mathbb{S}^3 , with constant positive curvature +1.
- 2) Euclidean geometry \mathbb{R}^3 , with constant curvature 0.
- 3) Hyperbolic geometry \mathbb{H}^3 , with constant negative curvature -1.
- 4) The geometry of $\mathbb{S}^2 \times \mathbb{R}$.
- 5) The geometry of $\mathbb{H}^2 \times \mathbb{R}$.
- 6) The geometry of the universal cover $\widetilde{SL}_2(\mathbb{R})$ of the Lie group $SL_2(\mathbb{R})$.
- 7) Nil geometry a left invariant metric on the Heisenberg group

$$\left\{ \begin{pmatrix} 1 & x & y \\ 0 & 1 & z \\ 0 & 0 & 1 \end{pmatrix} \mid x, y, z \in \mathbb{R} \right\}$$

8) Sol geometry - the split extension $\mathbb{R} \ltimes (\mathbb{R} \times \mathbb{R})$.

Six of these eight geometries (1, 2, 4, 5, 6 and 7) are now well understood geometric manifolds modelled on these geometries are Seifert fibered. Geometric manifolds modelled on Sol geometry are torus fibered over the circle. Since most 3-manifolds do not admit such fibrations, it is reasonable to say that "most" 3manifolds are hyperbolic [An2].

In this work, one particular class of hyperbolic manifolds will be investigated in detail. These are the manifolds that arise as covers of the hyperbolic orbifolds associated to the Lannér groups. A Lannér group is a Coxeter group acting on \mathbb{H}^3 whose fundamental region is a compact simplex. The manifolds constructed have many interesting and useful geometric properties and can often be constructed so as to admit a high degree of symmetry. Furthermore, the manifolds inherit several possible tessellations derived from the original group. These tessellations can be used to provide explicit combinatorial constructions for the manifolds.

1.2 Aims of this work

This thesis aims to investigate simplicial Coxeter groups, in particular the Lannér groups. All maps from these groups to the classical family of simple groups $L_2(q)$ are classified and a comprehensive study of the low index subgroups of some of these groups is provided. Results on some simplicial Coxeter groups acting on \mathbb{H}^4 are presented as well.

As a consequence of Selberg's Lemma, since Coxeter groups Γ admit faithful representations into $GL(n, \mathbb{C})$ ([HW]) for a suitable value of n, they contain finite index torsion free subgroups. Let $\mathcal{L}(\Gamma)$ be the least common multiple of the orders of all the finite subgroups of Γ . Then a minimal index torsion free subgroup of Γ must have index divisible by $\mathcal{L}(\Gamma)$ [CFJR]. Jones and Reid have shown in [JR] that for any k there exist Kleinian groups whose minimal index torsion free subgroups has index greater than $k \times \mathcal{L}(\Gamma)$. It will be shown in this thesis that Lannér groups, by contrast, contain a torsion free subgroup of index $\mathcal{L}(\Gamma)$ or $2 \times \mathcal{L}(\Gamma)$.

Using techniques from group theory and the classification of finite simple groups, this work aims to provide a construction for families of manifolds exhibiting a high degree of symmetry. In some interesting cases, further results will be deduced by considering the action of the isometry group of the manifold on the first homology.

Additionally, in the last ten years advances in computational resources have enabled an explosion in the computational-aided research of low index subgroups of discrete groups acting on spaces. This, coupled with a resurgent interest in Coxeter groups, has led to several interesting publications and significant progress in the classification of interesting families of Coxeter groups ([CM], [E2], [ERT]). A computing cluster was designed in this work and sophisticated algorithms were used to classify all subgroups of the Lannér groups, up to a given index. A summary of these results is also provided in this thesis. This adds significantly to the previous work carried out by B. Everitt ([EMc], [E1], [E2]).

1.3 Structure of this Thesis

Chapter 2 introduces definitions and previously known results important for this thesis. Abstract properties of Coxeter groups are introduced. Key properties of these groups, including the structure of conjugacy classes of torsion elements in the Lannér groups and the discrete extensions of these groups in $\text{Isom}(\mathbb{H}^3)$, as classified in [DM2], are introduced. The computational methods used in this thesis are also discussed in this chapter.

Chapter 3 focuses on the construction of a family of torsion free normal subgroups of the orientation preserving subgroup of the [5, 3, 5] Coxeter group whose factor groups are projective special linear groups $L_2(q)$ over a field of $q = p^n$ elements. The extensions of these results to the full Coxeter group are also characterised. In particular, the case where p = 5 is discussed due to the fact that it yields a chiral pair of Seifert - Weber manifolds.

Chapter 4 analyses the other Lannér groups using the same techniques as used in Chapter 3. The case $\Gamma = [3, 5, 3]$ is omitted since it has already been studied by Anna Torstensson in her PhD thesis [To]. In each case all normal subgroups N of the orientation preserving subgroup Γ^+ of Γ with $\Gamma^+/N \cong L_2(q)$ are classified.

Chapter 5 investigates some of the manifolds arising from the maps constructed in chapter 3. A chiral pair of Seifert-Weber manifolds are investigated and all symmetric abelian covers (covering manifolds whose deck transformations form an abelian group and on which the factor group acts by automorphisms) are characterised. Other manifolds whose isometry group is $L_2(19)$ are also constructed.

In Chapter 6, some of the manifolds arising from the epimorphisms described in chapter 4 are constructed and their properties are investigated. In a few special cases, the action of the isometries on homology is characterised. This work unifies and extends many previously known constructions, such as Zimmerman's manifolds, as described in [Z], as well as creating many new and interesting examples.

Generalizing to higher dimensions, Chapter 7 focuses on the [5, 3, 3, 3] Coxeter group. This is a simplicial Coxeter group acting on \mathbb{H}^4 . The construction of a torsion-free subgroup of index 14400 in this group will provide an example of a smallest volume hyperbolic 4-manifold. Subgroups of index 8×14400 are constructed using a combination of computational and group theoretic techniques. Some interesting results on the existence of subgroups of this group are proved.

In chapter 8 the main results of this work are restated, some additional concluding remarks are provided and futher avenues of study are suggested. A summary of the low index subgroups of the Lannér groups is provided in Appendix A.

Chapter 2

Background

This chapter introduces the relevant background material required for this thesis. Coxeter groups are introduced in both an algebraic and a geometric setting and the equivalence of the definitions is shown. Special properties of Coxeter groups are defined, as well as some general terminology to be used throughout the thesis. The classification theorem for finite Coxeter groups is stated [Hu] and the conjugacy classes of elements in certain finite Coxeter groups are listed. The classification of all finite extensions in $Isom(\mathbb{H}^3)$ of certain hyperbolic Coxeter groups (the Lannér groups), due to Derevnin and Mednykh [DM2], is stated. A general overview of some special computational techniques in determining conjugacy classes of subgroups in given finitely presented groups is also discussed in this chapter.

2.1 Introduction to Coxeter groups

Definition 2.1 A Coxeter group Γ is a group with generating set $S = \{s_1, \ldots, s_n\}$, together with a set of defining relators $(s_i s_j)^{m_{ij}}$ where $m_{ii} = 1$, $m_{ij} \ge 2$ if $i \ne j$ and $m_{ij} = m_{ji}$ for all $i \ne j$.

Associated with such a group is a graph called the *Coxeter graph* or *Coxeter* scheme for the group. It consists of a set of vertices v_i (one for each generator $s_i \in S$), connected by (labeled) edges under the following conditions:

- For $i \neq j$, if $m_{ij} > 3$ then v_i is connected to v_j by an edge labeled by m_{ij}
- For $i \neq j$, if $m_{ij} = 3$ then v_i is connected to v_j by an unlabeled edge
- If i = j or $m_{ij} = 2$ then no edge is drawn.

Definition 2.2 A Coxeter group Γ with a connected Coxeter scheme is called irreducible.

For any subset $S' \subseteq S$, the special subgroup of Γ generated by the $s \in S'$ is also a Coxeter group with Coxeter scheme obtained from that of Γ by deleting the vertices whose corresponding generators lie in $S \setminus S'$. If a Coxeter scheme is disconnected, then $\Gamma = \Gamma_1 \times \ldots \times \Gamma_k$, where each Γ_i is an irreducible Coxeter group.

Definition 2.3 Let Γ be a Coxeter group with generating set S. Then a maximal special subgroup is a subgroup generated by $S' = S \setminus \{s\}$ for any $s \in S$.

In order to determine the torsion elements in a Coxeter group, some information is needed on the finite Coxeter groups. The finite irreducible Coxeter groups are the Weyl groups of simple Lie algebras over \mathbb{C} , the dihedral groups, the group of symmetries of a regular dodecahedron and the group of symmetries of the 120-cell. In the Killing-Cartan notation, these are the three infinite classical families, A_n with $n \geq 1$, B_n with $n \geq 2$ and D_n with $n \geq 4$, as well as the five exceptional groups $G_2 \cong I_2(6)$, F_4 , E_6 , E_7 and E_8 . The non-Weyl groups are the dihedral groups $I_2(n)$ for $n \geq 3$, the symmetry group of the dodecahedron II_3 and the symmetry group of the 120-cell, H_4 . The Coxeter scheme for any finite Coxeter group is a disjoint union of the connected Coxeter schemes listed in Table 2.1. In the cases of A_n , B_n and D_n , the subscript n is the number of generators of the group.

The irreducible parabolic Coxeter groups (affine groups) are listed in Table 2.2. These are the affine Weyl groups and they form four infinite families and five exceptional groups. In the Killing-Cartan notation, the four infinite families are $\widetilde{A_n}$ with $n \ge 1$, $\widetilde{B_n}$ with $n \ge 3$, $\widetilde{C_n}$ with $n \ge 3$ and $\widetilde{D_n}$ with $n \ge 4$. The five exceptional groups are $\widetilde{G_2}$, $\widetilde{F_4}$, $\widetilde{E_6}$, $\widetilde{E_7}$ and $\widetilde{E_8}$. A Coxeter group is parabolic if and only if it is the direct product of finite and parabolic irreducible Coxeter groups. In the cases of $\widetilde{A_n}$, $\widetilde{B_n}$, $\widetilde{C_n}$ and $\widetilde{D_n}$, the number of generators of the group is n+1.



Table 2.1: Finite irreducible Coxeter groups



Table 2.2: Parabolic irreducible Coxeter groups

A convex polytope \mathcal{P} in \mathbb{H}^n is the intersection $\bigcap_{s \in S} \overline{H_s}$ of closed halfspaces bounded by hyperplanes H_s . A Coxeter group is hyperbolic of dimension n if and only if there is some polyhedron $\mathcal{P} \in \mathbb{H}^n$ such that, assigning each $s_i \in S$ to the reflection in the hyperplane H_{s_i} , an isomorphism is induced between Γ and the group generated by reflections in the faces of \mathcal{P} . \mathcal{P} is called a *fundamental region* for Γ .

The Gram matrix $G(\Gamma)$ for a Coxeter group Γ is a matrix with entries $g_{ij} = -\cos(\frac{\pi}{m_{ij}})$, where the m_{ij} are as above. The *signature* of a symmetric matrix is a pair (p,q) where p is the number of negative eigenvalues and q is the number of positive eigenvalues. The following theorem is due to Vinberg [V2]

Theorem 2.1 Let $G(\Gamma)$ be an indecomposable symmetric matrix of signature (n, 1) with 1's along the diagonal and non-positive entries off it. Then there is a convex polytope P in \mathbb{H}^n whose Gram matrix is G. The polytope is uniquely determined up to a motion in \mathbb{H}^n .

Let Γ be a hyperbolic Coxeter group. Then Γ is co-compact if it has a compact fundamental polyhedron \mathcal{P} . Otherwise Γ is non-co-compact. Γ is co-finite if it has a finite volume fundamental polyhedron \mathcal{P} .

Let \mathcal{F} be the collection of finite subgroups of Γ generated by subsets $S' \subset S$. Partially order \mathcal{F} by inclusion. Similarly, let $\overline{\mathcal{F}}$ be the partially ordered collection of finite subgroups or parabolic subgroups of Γ generated by subsets $S' \subset S$. Then the following result is due to Vinberg [V2]

Proposition 2.1.1 An n-dimensional hyperbolic Coxeter group Γ is co-compact (co-finite) if and only if $\mathcal{F}(\overline{\mathcal{F}})$ is isomorphic as a partially ordered set to the poset of some n-dimensional abstract polytope.

In particular, if Γ is a hyperbolic Coxeter group, then Γ is co-compact if and only if every maximal special subgroup is finite, and Γ is co-finite if and only if, for every maximal special subgroup, every factor is either finite or is an irreducible parabolic Coxeter group.

2.2 Conjugacy classes in finite Coxeter groups

The main focus of this thesis is on the nine co-compact simplicial hyperbolic Coxeter groups, first described by Lannér [La] and listed here in Table 2.3:



Table 2.3: The nine Lannér groups

These nine groups form a complete list of all Coxeter groups which have a 3dimensional compact hyperbolic simplex as a fundamental region. Each group is described by a 6-tuple $[\lambda_1, \lambda_2, \lambda_3; \mu_1, \mu_2, \mu_3]$, of numbers, where π/λ_i are the angles between a chosen base face and the other three faces of T, and π/μ_i is the dihedral angle of the edge opposite that labelled with π/λ_i .

The finite Coxeter groups play a key role in the construction of manifolds arising as the covers of the fundamental regions of the simplicial Coxeter groups. If Γ is a Coxeter group acting discretely on \mathbb{H}^3 with the natural action and if \mathcal{M} is an orbifold of the form $\mathcal{M} = \mathbb{H}^3/H$, for some $H < \Gamma$, then \mathcal{M} is a manifold precisely when H is torsion free. It remains to determine all representatives for torsion elements in Γ . The following theorem is well known. It is exercise V.4.2 in [B], and is proved in [BH].

Theorem 2.2 Let Γ be a Coxeter group with generating set S. Then any element of finite order in Γ is conjugate to an element of a finite subgroup generated by some $S' \subseteq S$.

The maximal special subgroups of the Lannér groups are all of types $A_1 \times A_1 \times$

 $A_1, A_1 \times A_2 \cong A_1 \times I_2(3), A_3, A_1 \times B_2 \cong A_1 \times I_2(4), A_1 \times I_2(5), B_3$ or H_3 . Hence, to understand the torsion elements in each of the Lannér groups, the conjugacy classes of torsion elements in each of these finite groups are needed. A Coxeter scheme and presentation for each of these seven groups is provided in Table 2.4

Group	Coxeter Scheme	Presentation
$A_1 \times A_1 \times A_1$	$\begin{array}{ccc} a & b & c \\ \bullet \times \bullet \times \bullet \end{array}$	$\langle a, b, c \mid a^2 = b^2 = c^2 = (ab)^2 = (ac)^2 = (bc)^2 = 1 \rangle$
$A_1 \times I_2(3)$	a b c ● × ●	$\langle a, b, c \mid a^2 = b^2 = c^2 = (ab)^2 = (ac)^2 = (bc)^3 = 1 \rangle$
$A_1 \times I_2(4)$	a b 4 c • × • •	$\langle a, b, c \mid a^2 = b^2 = c^2 = (ab)^2 = (ac)^2 = (bc)^4 = 1 \rangle$
$A_1 \times I_2(5)$	$a \\ \bullet \times \bullet 5 \\ \bullet $	$\langle a, b, c \mid a^2 = b^2 = c^2 = (ab)^2 = (ac)^2 = (bc)^5 = 1 \rangle$
A_3	a b c ●──●	$\langle a, b, c \mid a^2 = b^2 = c^2 = (ab)^3 = (ac)^2 = (bc)^3 = 1 \rangle$
B_3	a 4 b c	$\langle a, b, c \mid a^2 = b^2 = c^2 = (ab)^4 = (ac)^2 = (bc)^3 = 1 \rangle$
H_3	a5b c ● ●	$\langle a, b, c \mid a^2 = b^2 = c^2 = (ab)^5 = (ac)^2 = (bc)^3 = 1 \rangle$

Table 2.4: Finite Coxeter groups whose conjugacy class structure is required

2.2.1 Conjugacy classes in $A_1 \times A_1 \times A_1$

The Coxeter group $A_1 \times A_1 \times A_1$ has presentation $\langle a, b, c | a^2 = b^2 = c^2 = (ab)^2 = (ac)^2 = (bc)^2 = 1 \rangle \cong C_2 \times C_2 \times C_2$. The orientation preserving subgroup $(A_1 \times A_1 \times A_1)^+$ has presentation $\langle \alpha, \beta | \alpha^2 = \beta^2 = (\alpha\beta)^2 = 1 \rangle$, where $\alpha = ab$ and $\beta = bc$ and is isomorphic to the Klein 4-group V_4 . Representatives for the conjugacy classes of non-identity elements of $A_1 \times A_1 \times A_1$ and $(A_1 \times A_1 \times A_1)^+$ are given in Table 2.5:

$A_1 \times A_1 \times A_1$		$(A_1 \times A_1 \times A_1)^+$	
Order 2:	a, b, c, ab, ac, bc	Order 2:	$\alpha, \beta, \alpha\beta$

Table 2.5: Conjugacy classes in $A_1 \times A_1 \times A_1$ and $(A_1 \times A_1 \times A_1)^+$

2.2.2 Conjugacy classes in $A_1 \times I_2(3)$

The Coxeter group $A_1 \times A_2$ has presentation $\langle a, b, c \mid a^2 = b^2 = c^2 = (ab)^2 = (ac)^2 = (bc)^3 = 1 \rangle \cong C_2 \times D_3$. The orientation preserving subgroup $(C_2 \times I_2(3))^+$ has presentation $\langle \alpha, \beta \mid \alpha^2 = \beta^3 = (\alpha\beta)^2 = 1 \rangle$, where $\alpha = ab$ and $\beta = bc$ and is isomorphic to the symmetric group S_3 . Representatives for the conjugacy classes of non-identity elements of $A_1 \times I_2(3)$ and $(A_1 \times I_2(3))^+$ are given in Table 2.6:

$A_1 \times I_2(3)$		$(A_1 \times I_2(3))^+$	
Order 2:	a, b, ab	Order 2:	α
Order 3:	bc	Order 3:	β
Order 6:	abc		

Table 2.6: Conjugacy classes in $A_1 \times I_2(3)$ and $(A_1 \times I_2(3))^+$

2.2.3 Conjugacy classes in $A_1 \times I_2(4)$

The Coxeter group $A_1 \times I_2(4)$ has presentation $\langle a, b, c \mid a^2 = b^2 = c^2 = (ab)^2 = (ac)^2 = (bc)^3 = 1 \rangle \cong C_2 \times D_4$. The orientation preserving subgroup $(C_2 \times I_2(4))^+$ has presentation $\langle \alpha, \beta \mid \alpha^2 = \beta^4 = (\alpha\beta)^2 = 1 \rangle$, where $\alpha = ab$ and $\beta = bc$ and is isomorphic to the dihedral group D_4 of order 8. Representatives for the conjugacy classes of non-identity elements of $A_1 \times I_2(4)$ and $(A_1 \times I_2(4))^+$ are given in Table 2.7:

$A_1 \times I_2(4)$		$(A_1 \times \overline{I_2(4)})^+$	
Order 2:	$a, b, c, ab, ac, (bc)^2$	Order 2:	$\alpha, \alpha\beta, \beta^2$
Order 4:	bc, abc	Order 4:	β

Table 2.7: Conjugacy classes in $A_1 \times I_2(4)$ and $(A_1 \times I_2(4))^+$

2.2.4 Conjugacy classes in $A_1 \times I_2(5)$

The Coxeter group $A_1 \times I_2(5)$ has presentation $\langle a, b, c \mid a^2 = b^2 = c^2 = (ab)^2 = (ac)^2 = (bc)^3 = 1 \rangle \cong C_2 \times D_5$. The orientation preserving subgroup $(C_2 \times I_2(5))^+$ has presentation $\langle \alpha, \beta \mid \alpha^2 = \beta^4 = (\alpha\beta)^2 = 1 \rangle$, where $\alpha = ab$ and $\beta = bc$

and is isomorphic to the dihedral group D_5 of order 10. Representatives for the conjugacy classes of non-identity elements of $A_1 \times I_2(5)$ and $(A_1 \times I_2(5))^+$ are given in Table 2.8:

$A_1 \times I_2(5)$		$(A_1 \times I_2(5))^+$	
Order 2:	a, b, ac	Order 2:	lphaeta
Order 5:	$bc, (bc)^2$	Order 5:	β, β^2
Order 10:	$abc, a(bc)^2$		

Table 2.8: Conjugacy classes in $A_1 \times I_2(3)$ and $(A_1 \times I_2(3))^+$

2.2.5 Conjugacy classes in A_3

The Coxeter group A_3 has presentation $\langle a, b, c | a^2 = b^2 = c^2 = (ab)^3 = (bc)^3 = (ac)^2 = 1 \rangle$. That this is a presentation for the symmetric group Sym_4 can be shown by the identifications: $a \leftrightarrow (1, 2), b \leftrightarrow (2, 3)$ and $c \leftrightarrow (3, 4)$. The orientation preserving subgroup A_3^+ has presentation $\langle \alpha, \beta | \alpha^3 = \beta^3 = (\alpha \beta)^2 = 1 \rangle$, with $\alpha = ab$ and $\beta = bc$, and is isomorphic to the alternating group Alt_4 . Representatives for the conjugacy classes of non-identity elements of A_3 and A_3^+ are as shown in Table 2.9:

A_3		A_3^+	-
Order 2:	a, ac	Order 2:	lphaeta
Order 3:	bc	Order 3:	β, β^{-1}
Order 4:	abc		

Table 2.9: Conjugacy classes in A_3 and A_3^+

2.2.6 Conjugacy classes in B_3

 B_3 has presentation $\langle a, b, c | a^2 = b^2 = c^2 = (ab)^4 = (bc)^3 = (ac)^2 = 1 \rangle \cong S_4 \times C_2$. The orientation preserving subgroup B_3^+ has presentation $\langle \alpha, \beta | \alpha^4 = \beta^3 = (\alpha \beta)^2 = 1 \rangle$, where $\alpha = ab$ and $\beta = bc$, and is isomorphic to S_4 . Representatives for the conjugacy classes of non-identity elements of B_3 and B_3^+ are as shown in Table 2.10:

B_3		B_{3}^{+}	
Order 2:	$a, c, ac, (ab)^2, (abc)^3$	Order 2:	$\alpha^2, \alpha\beta$
Order 3:	bc	Order 3:	β
Order 4:	$ab, c(ab)^2$	Order 4:	α
Order 6:	abc		

Table 2.10: Conjugacy classes in B_3 and B_3^+

2.2.7 Conjugacy classes in H_3

 H_3 has presentation $\langle a, b, c | a^2 = b^2 = c^2 = (ab)^5 = (bc)^3 = (ac)^2 = 1 \rangle \cong Alt_5 \times C_2$. The orientation preserving subgroup H_3^+ has presentation $\langle \alpha, \beta | \alpha^5 = \beta^3 = (\alpha \beta)^2 = 1 \rangle$, where $\alpha = ab$ and $\beta = bc$, and is isomorphic to the alternating group Alt_5 . Then representatives for the conjugacy classes of non-identity elements of H_3 and H_3^+ are as shown in Table 2.11:

H_3		H_3^+	
Order 2:	$a, ac, (abc)^5$	Order 2:	lphaeta
Order 3:	bc	Order 3:	β
Order 5:	$ab, (ab)^2$	Order 5:	α, α^2
Order 6:	$c(ab)^2$		
Order 10:	$abc, (abc)^3$		

Table 2.11: Conjugacy classes in H_3 and H_3^+

2.3 Discrete extensions of the 3-dimensional simplicial Coxeter groups

In [La] Lannér listed all the tetrahedra in \mathbb{H}^3 whose dihedral angles are of the form π/n , for some integer n. Associated with these tetrahedra are the reflection groups generated by reflections in the faces of the tetrahedra and referred to either as Coxeter groups or as Lannér groups. The angle between any pair of faces is an integer submultiple of π , hence the group generated by reflections across the faces of the tetrahedron forms a discrete subgroup of Isom(\mathbb{H}^3).

Much of this thesis is devoted to the classification of torsion free normal subgroups N of the orientation preserving subgroup Γ_i^+ of the Lannér groups Γ_i , as listed in Table 2.3. In particular a complete classification of all $N \triangleleft \Gamma_i^+$ satisfying $\Gamma_i^+/N \cong L_2(p^n)$, where p is a prime, is given. Hyperbolic manifolds with a large number of symmetries are also constructed. A manifold \mathcal{M} is said to have a large number of symmetries if $\operatorname{vol}(\mathcal{M}/\operatorname{Isom}(\mathcal{M}))$ is small. Let $\tilde{\Gamma}_i = N_{\operatorname{Isom}(\mathbb{H}^3)}(\Gamma_i)$ be the normalizer of Γ_i in $\operatorname{Isom}(\mathbb{H}^3)$ and $\mathcal{M} = \mathbb{H}^3/K$, where K is a torsion free subgroup of Γ_i . Then $\operatorname{Isom}(\mathcal{M})$ is the normalizer $N_{\tilde{\Gamma}_i}(K)$ of K in $\tilde{\Gamma}_i$. Hence it is important to know the normaliser $N_{\operatorname{Isom}(\mathbb{H}^3)}(\Gamma_i)$ of Γ_i in $\operatorname{Isom}(\mathbb{H}^3)$. In [DM2] Derevnin and Mednykh classified all extensions of the Lannér groups:

Theorem 2.3 ([DM2]) Let Γ_i be a one of the nine Lannér groups listed in Table 2.3 and let Γ_i^+ be the orientation preserving subgroup of Γ_i . If Γ_i^+ is a subgroup of a discrete group $G \subset Ison(\mathbb{H}^3)$, then Γ_i^+ is normal in G.

As a consequence of this theorem, $N_{\text{Isom}(\mathbb{H}^3)}(\Gamma_i)$ can be identified with $\text{Aut}(\Gamma_i^+)$ (since the Γ_i^+ have trivial center) and $\text{Inn}(\Gamma_i^+) = \Gamma_i^+$. The outer automorphism group of Γ_i^+ is then one of the following types:

If $i \in \{6, 9\}$, then $\operatorname{Aut}(\Gamma_i^+)/\operatorname{Inn}(\Gamma_i^+) \cong \mathbb{Z}_2 \times \mathbb{Z}_2 \times \mathbb{Z}_2$. If $i \in \{1, 2, 4, 5, 7, 8\}$, then $\operatorname{Aut}(\Gamma_i^+)/\operatorname{Inn}(\Gamma_i^+) \cong \mathbb{Z}_2 \times \mathbb{Z}_2$.

If i = 3, then Γ_3 is the unique non-trivial discrete extension of Γ_3^+ in Isom(\mathbb{H}^3) and Aut(Γ_3^+)/Inn(Γ_3^+) $\cong \mathbb{Z}_2$.

The outer automorphisms of Γ_i can also be considered as the graph automorphisms of the Coxeter scheme for Γ_i , acting on the generators s_i corresponding to the vertices v_i . This induces an action on Γ_i^+ , and together with conjugation by elements of $\Gamma_i \setminus \Gamma_i^+$ they generate the outer automorphism group described in the remarks following the statement of Theorem 2.3.

2.4 Determining low index subgroups of finitely presented groups

As is mentioned in the previous section, much of this thesis is spent characterizing families of normal subgroups of the Lannér groups. To better understand these groups, computer-aided constructions were employed to classify all conjugacy classes of low index subgroups. In order to construct complete lists of low index subgroups of G the program Lowx [Lx] was used. All conjugacy classes of subgroups up to index 60 were computed for Γ_1 , Γ_2 , Γ_3 and Γ_4 . All conjugacy classes of subgroups up to index 20 were computed for Γ_5 , Γ_6 , Γ_7 and Γ_8 , while all conjugacy classes of subgroups up to index 24 were computed for Γ_9 .

Using this data, further subgroups could then be determined by letting a Lannér group Γ_i^+ act on the cosets of a representative H of a conjugacy class. The action induces a permutation representation $\phi : \Gamma_i^+ \to S_n$ of Γ_i in some symmetric group S_n , where n is the index of H in Γ_i . The core of a representative, K(H), is the kernel of this representation. In many cases, provided that n is not too big, it is possible to determine a presentation for K(H). Various invariants of the associated manifold, $\mathcal{M}_{\phi} = \mathbb{H}^3/K(H)$, such as its fundamental group and its homology, can then be determined.

Given a permutation representation $\phi(\Gamma_i)$, the combinatorial structure of the associated manifold can then be deduced by careful analysis of the group structure. As a result, a complete combinatorial description of the manifold \mathcal{M}_{ϕ} , the action of the isometry group on \mathcal{M}_{ϕ} , a presentation for its fundamental group, a complete description of its homology and its volume can all be provided.

The method of determining all conjugacy classes of subgroups up to a given index is a modification of the *coset enumeration* algorithm ([HEB], [Ne]). The basic coset enumeration algorithm constructs, for a given finitely generated subgroup Hof a finitely presented group G, a coset table for H in G. One of the most important facts about the algorithm is that, for any finitely presented G and H of finite index in G, the algorithm will eventually terminate.

The low index search algorithm works as follows: In order to find all subgroups

of index $\leq N$ in a group G, the algorithm starts with the trivial group and proceeds to enumerate cosets of it using the Todd-Coxeter (TC) algorithm. The number of cosets allowed is bounded by some function f(N). It is generally assumed that $N < f(N) \leq 2N$. The TC algorithm then either returns the statement that Ghas some order $\leq f(N)$ or halts with an incomplete coset table. If G has order less than f(N) then its coset table (modulo the unit subgroup) is printed as a first table. If, however, G has order bigger than f(N), the TC algorithm stops with an incomplete coset table. In this case, pairs of forced coincidences (i_l, i_k) are then systematically introduced between the incomplete coset rows i_l and i_k in the table. After applying a coincidence, the TC procedure is then restarted on the modified coset table. In theory this procedure will continue until there are no further incomplete tables. In practise, however, if the initial bound N is even moderately large, the procedure is limited by machine constraints.

The above algorithm describes a method for finding all subgroups H of index less than N in G. In general it is more efficient to find conjugacy class representatives of subgroups. This is how the algorithm is implemented in both GAP and Lowx. Furthermore, the TC algorithm does not provide a generating set for each representative. There are a number of methods available to compute a presentation for H. One such method is to use the completed coset table for H in G to construct a set of *Schreier generators* for H([HEB] §2.5). Once a set of generators for H has been constructed, a modified Todd-Coxeter procedure can be used to find a presentation for H as an abstract group. The resulting presentation is generally rather cumbersome and can be simplified using a sequence of Tietze transformations.

Chapter V_____

Epimorphic images of the [5, 3, 5] Coxeter group

3.1 The Coxeter group [5,3,5]

Let T be a tetrahedron in \mathbb{H}^3 with vertices A, B, C, D, and with dihedral angles $\pi/5, \pi/3, \pi/5, \pi/2, \pi/2$ and $\pi/2$ along its edges CD, AD, AB, BD, BC, AC, as shown in Figure 3.1. The standard generators a, b, c and d of the Coxeter group $\Gamma = [5, 3, 5]$ are the reflections of \mathbb{H}^3 in the sides of T opposite A, B, C and D.



Figure 3.1: A hyperbolic tetrahedron ${\cal T}$

Thus Γ is a discrete group of isometries of \mathbb{H}^3 , and this space carries a tessellation \mathcal{T} by the images of T, permuted regularly by Γ . The normaliser Ω of Γ in Isom \mathbb{H}^3 is a semidirect product of Γ by a cyclic group of order 2, generated by a half-turn r of \mathbb{H}^3 about an axis through the mid-points of the edges AD and BC of T; this preserves the tessellation \mathcal{T} , and acts by conjugation on Γ as $a^r = d$, $b^r = c$, $c^r = b$ and $d^r = a$.

The subgroup $\Gamma_0 = \langle a, b, c \rangle$ of Γ is the [5,3] Coxeter group. This group is the isometry group of a dodecahedron and is isomorphic to $A_5 \times C_2$. The 120 images of T under Γ_0 form a hyperbolic dodecahedron \mathbb{D} , with dihedral angles $2\pi/5$, and the images of \mathbb{D} under Γ form a dodecahedral tessellation \mathcal{D} of \mathbb{H}^3 ; there are five dodecahedra around every edge, so the vertex figure of \mathcal{D} is an icosahedron. The isometry group of \mathcal{D} is Γ , while r sends \mathcal{D} to its dual tessellation \mathcal{D}^* , which is isomorphic to but distinct from \mathcal{D} .

The orientation-preserving subgroup Γ^+ of Γ has a presentation

$$\Gamma^{+} = \langle \alpha, \beta, \gamma \mid \alpha^{5} = \beta^{3} = \gamma^{5} = (\alpha\beta)^{2} = (\alpha\beta\gamma)^{2} = (\beta\gamma)^{2} = 1 \rangle.$$

where $\alpha = ab$, $\beta = bc$ and $\gamma = cd$ are rotations of \mathbb{H}^3 through $2\pi/5, 2\pi/3$ and $2\pi/5$ about the edges CD, AD and AB of T. It has an associated fundamental region $T' = T \cap a(T)$ formed by doubling T across the face BCD.

Lemma 3.1.1 The torsion elements of Γ^+ are the conjugates in Γ^+ of the powers of α , β , γ and $\alpha\beta$.

Proof: Any isometry g of \mathbb{H}^3 of finite order has a fixed point $p \in \mathbb{H}^3$. Let $g \in \Gamma^+$ be an element of finite order. Then, since T is a fundamental region for Γ , conjugating g by a suitable element of Γ , it may be assumed that $p \in T$. Suppose that $g \neq 1$, so p is in the boundary ∂T of T. If p is in the interior of an edge or face of ∂T then g fixes that entire edge or face, including any incident vertices; it can therefore be assumed that p is a vertex of T.

The stabiliser in Γ of the vertex D of T is the rank three Coxeter group $\Gamma_0 = \langle a, b, c \rangle$, so if g fixes D then g lies in the orientation-preserving subgroup $\Gamma_0^+ = [5, 3]^+ = \langle \alpha, \beta \rangle \cong A_5$ of Γ_0 ; the non-identity elements of this group, of orders 2, 3 or 5, are conjugate to the powers of $\alpha\beta$, β or α respectively, as required. Similarly
if g fixes A then it lies in the orientation-preserving subgroup $\Gamma_1^+ = \langle \beta, \gamma \rangle \cong A_5$ of $\Gamma_1 = \langle b, c, d \rangle$, and is conjugate to a power of $\beta\gamma$, β or γ . If g fixes C then it lies in the orientation-preserving subgroup $\langle \alpha, \beta\gamma \rangle \cong D_5$ of $\langle a, b, d \rangle \cong D_5 \times C_2$, and is therefore conjugate to a power of α or $\beta\gamma$; similarly if g fixes B it is conjugate in $\langle \alpha\beta, \gamma \rangle \cong D_5$ to a power of γ or $\beta\gamma$. The involution $\alpha\beta\gamma$ is conjugate in $\langle \alpha\beta, \gamma \rangle$ to $\alpha\beta$, and in $\langle \alpha, \beta\gamma \rangle$ to $\beta\gamma$, so these three involutions are all conjugate in Γ^+ .

Note: Lemma 3.1.1 can be extended to determine all the conjugacy classes of torsion elements of Ω .

Corollary 3.1.1 Each proper normal subgroup of Γ^+ is torsion-free.

Proof: It follows easily from the presentation of Γ^+ that if any non-identity power of α , β , γ or $\alpha\beta$ is mapped to the identity then the group collapses to the trivial group. Thus the only normal subgroup containing non-identity torsion elements is Γ^+ .

Hence, if N is a proper normal subgroup of Γ^+ , then the corresponding quotient space \mathbb{H}^3/N of \mathbb{H}^3 is a manifold, and it is compact if and only if N has finite index in Γ^+ . The presentation given above shows that Γ^+ is perfect, so its maximal normal subgroups have nonabelian simple groups as quotients. The family $L_2(q)$ of finite simple groups will be considered as possible quotients since in a certain sense "most" nonabelian finite simple groups have the form $L_2(q)$. For technical reasons it is necessary to deal first with quotients isomorphic to the alternating group $A_5 \cong L_2(4) \cong L_2(5)$.

3.2 Quotients of Γ^+ isomorphic to A_5

Lemma 3.2.1

(i) There are two normal subgroups N_1 and N_2 of Γ^+ with $\Gamma^+/N_i \cong A_5$, namely the kernels of the epimorphisms $\theta_i : \Gamma^+ \to A_5$ (i = 1, 2) given by

$$\alpha \mapsto (13524), \ \beta \mapsto (123), \ \gamma \mapsto (14352) \text{ or } (13425).$$

(ii) N_1 and N_2 are conjugate in Γ .

(iii) Each N_i is normal in the orientation preserving subgroup Ω^+ of Ω , with Ω^+/N_i isomorphic to the symmetric group S_5 .

Proof: (i) The normal subgroups with quotient A_5 are the kernels of epimorphisms $\Gamma^+ \to A_5$. Two epimorphisms have the same kernel if and only if they differ by an automorphism of A_5 . Any epimorphism θ must map α and β to elements of order 5 and 3 in A_5 and any pair of elements of this type generate A_5 . By composing θ with a unique automorphism of A_5 , it may be assumed that θ maps α and β to (13524) and (123), since their product is an involution. It remains to find all possible images of γ in A_5 , preserving the relations of Γ^+ involving γ . It is easily seen that the only possibilities are (14352) and (13425), both of which are conjugate to α^2 . This gives two inequivalent epimorphisms θ_1 and θ_2 . Define $N_i = \ker \theta_i$ for i = 1, 2.

(ii) Since Γ normalises Γ^+ , its action by conjugation permutes the normal subgroups of Γ^+ with a given quotient, so N_1 and N_2 must be either normal in Γ or conjugate in Γ . If either N_i were normal in Γ , then b would induce an automorphism of $\Gamma^+/N_i \cong A_5$ imitating the effect it has by conjugation on Γ^+ . This automorphism must invert the images (13524) and (123) of α and β , and the only element of Aut $A_5 \cong S_5$ doing this is the permutation (13)(45). Mapping b to (13)(45), then $c = b\beta$ is mapped to (23)(45). Then $d = c\gamma$ is mapped to (14253) or (13524) for i = 1, 2, which is clearly not to an involution in either case. Thus the subgroups N_i are not normal in Γ , so they are conjugate to each other. (Alternatively, note that the element $(abc)^5$ of Γ , which represents the antipodal involution in the dodecahedral group $\Gamma_0 \cong A_5 \times C_2$ and hence commutes with α and β , transposes θ_1 and θ_2 .)

(iii) The action of r by conjugation on Γ^+ is to transpose α and γ^{-1} and to invert β , and for i = 1, 2 the permutations (23) and (13) in S_5 respectively have the same effect on the images of these generators under θ_i ; this shows that each θ_i extends to an epimorphism $\Omega^+ = \langle \Gamma^+, r \rangle \to S_5$, so each N_i is normal in Ω^+ with $\Omega^+/N_i \cong S_5$.

The quotient manifolds \mathbb{H}^3/N_i form a chiral pair of Seifert-Weber spaces, so it is useful to obtain presentations for their fundamental groups N_i . These were obtained by using GAP to implement the Reidemeister-Schreier process. The epimorphism θ_1 sends *abcd* to (13524)(14352) = (15)(23). Adjoining the relation $(abcd)^2 = 1$ to the presentation for Γ^+ gives a presentation for A_5 . Hence N_1 is the normal closure of $(abcd)^2$ in Γ^+ ([C1, pp 23-33]). Using GAP a presentation for N_1 as an abstract group was found. This presentation is given in Table 3.1.

$$\begin{array}{lll} \langle F_1, \ F_2, \ F_3, \ F_4 & \mid & F_1F_4^{-1}F_2F_3^{-1}F_1F_2F_3F_4^{-1}F_1^{-1}F_3F_1^{-1}F_2 = 1, \\ & F_3^{-1}F_1F_4^{-1}F_2F_3^{-1}F_1F_4F_3^{-1}F_2^{-1}F_4F_1^{-1}F_2F_1^{-1} = 1, \\ & F_1^{-1}F_2F_1^{-1}F_2^{-1}F_4F_1^{-1}F_2F_1^{-1}F_3^{-1}F_2^{-1}F_4^{-1}F_1^{-1}F_3 = 1, \\ & F_1F_4^{-1}F_2F_3^{-1}F_1F_4^{-1}F_1^{-1}F_3F_1^{-1}F_2F_1^{-1}F_3^{-1}F_1F_4^{-1} = 1 \end{array}$$

Table 3.1: A presentation for N_1 as an abstract group

The generators for N_1 as a subgroup of Γ^+ can be described by writing F_i as follows: $F_1 = abacbdcd$, $F_2 = ababacbdcdba$, $F_3^{-1}F_1 = acbadcdb$ and $F_4^{-1}F_1^{-1}F_3 =$ bcbabdcdca. Rewriting these gives $F_1 = \alpha^2 \beta^2 \gamma^2$, $F_2 = \alpha(\alpha \gamma)^2 \alpha^{-1}$, $F_3 = \alpha \gamma^2 \alpha^3 \beta$ and $F_4 = \alpha^3 \gamma^3 \beta$. Similarly the epimorphism θ_2 sends abcd to (13524)(13425) = (143). Adjoining the relation $(abcd)^3 = 1$ to the presentation for Γ^+ gives a presentation for A_5 , so N_2 is the normal closure of $(abcd)^3$ in Γ^+ ([C1, pp 23-33]). Using GAP a presentation for N_1 as an abstract group was found. This presentation is given in Table 3.2.

$$\langle F_1, F_2, F_3, F_4 \mid F_1 F_3^{-1} F_2^{-1} F_4^{-1} F_1 F_4^{-1} F_1 F_3^{-2} F_4 F_2^{-1} = 1,$$

$$F_2 F_1 F_3 F_2^{-1} F_1 F_3^{-1} F_4 F_2 F_3^{-1} F_4 F_1^{-1} F_4 = 1,$$

$$F_2 F_4^{-1} F_3 F_1^{-1} F_2 F_3^{-1} F_1 F_3^{-1} F_2^{-1} F_4^{-1} F_1 = 1,$$

$$F_2 F_3^{-1} F_4 F_3^{-1} F_1^{-1} F_2 F_3^{-1} F_4 F_1^{-1} F_4 F_2 F_4 = 1 \rangle$$

Table 3.2: A presentation for N_2 as an abstract group

The generators for N_2 as a subgroup of Γ^+ can be described by writing F_i as follows: $F_1 = abadcdbc$, $F_3F_2^{-1}F_1F_3^{-1}F_4 = babcbcdcda$, $F_3F_2^{-1}F_1F_3^{-1}F_4F_2^{-1} = ababacbcdcdb$ and $F_1F_3^{-1} = ababadcdbcba$. Rewriting these gives $F_1 = (\alpha\gamma^{-1})^2$, $F_2 = (\alpha^{-1}\gamma^2)^2$, $F_3 = \alpha\beta^2\alpha\gamma^3$ and $F_4 = (\alpha\gamma)^2\gamma(\beta\alpha^{-1}\beta\alpha\beta)^2[\gamma,\beta]\alpha^{-1}$.

3.3 Some facts about the simple groups $L_2(q)$

In order to study normal subgroups of Γ or Γ^+ with quotients isomorphic to $L_2(q)$, some useful facts about these groups will now be presented. A more extensive treatment can be found in [D], Chapter XII.

The groups $L_2(q)$ are simple for all prime powers $q = p^n$, with the exception of $L_2(2) \cong S_3$ and $L_2(3) \cong A_4$. They are mutually non-isomorphic, with the exception of $L_2(4)$ and $L_2(5)$, both of which are isomorphic to A_5 .

By Corollary 3.1.1, any nontrivial homomorphism $\Gamma^+ \to L_2(q)$ must send the non-identity torsion elements of Γ^+ , as classified in Lemma 3.1.1, to elements of order 2,3 or 5, so it is useful to be able to identify the elements of these orders in $L_2(q)$. Regarding an element of $L_2(q) = SL_2(q)/\{\pm I\}$ as a pair $\pm A$ of matrices in $SL_2(q)$, then its trace is well-defined up to multiplication by -1. For each q the elements of order 2 form a single conjugacy class, consisting of the non-identity elements of trace 0. The elements of order 3 also form a single class, and these are the non-identity elements of trace ± 1 . There are elements of order 5 in $L_2(q)$ if and only if p = 5 or $q \equiv \pm 1 \pmod{5}$, in which case they form two conjugacy classes, each inverse-closed and consisting of the squares of the elements of the other class; these classes consist of the non-identity elements of traces $\pm(-1 \pm \sqrt{5})/2$, where $\sqrt{5}$ lies in a field \mathbb{F}_p iff $p \cong \pm 1 \mod 5$ or else $q = p^2$.

Since Γ^+ is perfect, the image of any nontrivial homomorphism $\Gamma^+ \to L_2(q)$ must be a perfect subgroup of $L_2(q)$. These are all isomorphic to A_5 (icosahedral subgroups) or to $L_2(q')$ where $F_{q'}$ is a subfield of F_q , so that $q' = p^m$ for some mdividing n. If p = 2 there is a single conjugacy class of icosahedral subgroups when n is even, and there are none when n is odd. If p = 5 there are two classes of such subgroups if n is even, or one class if n is odd. For odd $q \equiv \pm 1 \pmod{5}$ there are two conjugacy classes, merging to form a single class in $PGL_2(q)$, whereas for odd $q \equiv \pm 2 \pmod{5}$ there are none.

3.4 Quotients of Γ^+ isomorphic to $L_2(q), p \notin \{2, 5\}$

Let \overline{F}_p be the algebraic closure of the field F_p of order p, where p is prime. This is the union of the finite fields F_q for all powers q of p, with the natural inclusions, so the group $\overline{L} := L_2(\overline{F}_p)$ is the union of the corresponding groups $L_2(F_q) =$ $L_2(q)$, with the induced inclusions. It follows that any epimorphism $\Gamma^+ \to L_2(q)$ can be regarded as a homomorphism $\Gamma^+ \to \overline{L}$, by composition with the natural embedding $L_2(q) \to \overline{L}$. Conversely, since Γ^+ is finitely generated, the image of any homomorphism $\Gamma^+ \to \overline{L}$ is contained in a subgroup $L_2(K)$ for some finite subfield K of \overline{F}_p ; since Γ^+ is perfect, the image (if nontrivial) must be isomorphic to A_5 or to $L_2(q)$ for some power q of p. Thus we can find all quotients of Γ^+ isomorphic to $L_2(q)$ by considering all nontrivial homomorphisms $\Gamma^+ \to \overline{L}$, and excluding those with image isomorphic to A_5 if $q \neq 4, 5$. For technical reasons, it is assumed here that $p \neq 2$ or 5. These exceptional primes will be considered in § 3.5 of this chapter.

3.4.1 The restriction to Γ_0^+

If $\theta : \Gamma^+ \to \overline{L}$ is any nontrivial homomorphism, then its restriction ψ to $\Gamma_0^+ = \langle \alpha, \beta \rangle \cong A_5$ must be an isomorphism with a subgroup $G \cong A_5$ of \overline{L} . There is a single conjugacy class of such subgroups G in \overline{L} . This is because, being finite, any pair of such subgroups are both contained in a subgroup $L_2(K)$ for some finite subfield K of \overline{F}_p ; now $L_2(K)$ has two conjugacy classes of such subgroups (as previously shown in § 3.3), and these are all conjugate in the subgroup $PGL_2(K) \leq L_2(\tilde{K}) \leq \overline{L}$, where \tilde{K} is the quadratic extension of K in \overline{F}_p . Let us define two embeddings $\Gamma_0^+ \to \overline{L}$ to be equivalent if they differ by an inner automorphism of \overline{L} . Given any pair of embeddings ψ_i (i = 1, 2), the conjugacy of their images implies that ψ_2 is equivalent to an embedding ψ'_2 with the same image as ψ_1 , so ψ'_2 differs from ψ_1 by an automorphism of A_5 . Since $|\operatorname{Out} A_5| = 2$ it follows that there are at most two equivalence classes of embeddings. Indeed, since the outer automorphisms of A_5 transpose its two conjugacy classes of elements of order 5, and these are distinguished by the traces $(-1 \pm \sqrt{5})/2$ of their images in \overline{L} , which are invariant under conjugation, it follows that there are exactly two equivalence classes of

embeddings $\Gamma_0^+ \to \overline{L}$.

Representatives of these two classes can be constructed as follows. Define $F = \mathbb{F}_p$ or \mathbb{F}_{p^2} , depending on whether $p \equiv \pm 1$ or $\pm 2 \mod (5)$, so F is the smallest subfield of \overline{F}_p containing a square root of 5, or equivalently for which $L := L_2(F)$ contains elements of order 5. A simple counting argument shows that in any finite field, each element is a sum of two squares, so for each $t = t_i = (-1 \pm \sqrt{5})/2$ (i = 1, 2)in F, elements $e, f \in F$ (depending on i) can be chosen such that $e^2 + f^2 + 3 = t^2$, or equivalently

$$e^2 + f^2 + t + 2 = 0 \tag{3.1}$$

since $t^2 = 1 - t$. For instance, if -3 is a square in F, then taking $e = \sqrt{-3}$ and f = -t (or vice versa), Equation 3.1 is satisfied. The case e = f = 0 can not happen, since this gives t = -2 and so $4 = t^2 = 1 - t = 3$, which is impossible; thus, without loss of generality, it can be assumed that $e \neq 0$. Now define $\psi_i : \Gamma_0^+ \to L \leq \overline{L}$, for i = 1, 2, by taking $t = t_i$ and sending

$$\alpha \mapsto \frac{1}{2} \begin{pmatrix} t-f & e+1\\ e-1 & t+f \end{pmatrix}, \ \beta \mapsto \frac{1}{2} \begin{pmatrix} e+1 & t+f\\ f-t & 1-e \end{pmatrix}$$
(3.2)

so $\alpha\beta \mapsto \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix}$. Since α, β and $\alpha\beta$ are sent to elements with traces $\pm t, \pm 1$ and 0 it follows that each ψ_i is a homomorphism and hence an embedding by the simplicity of Γ_0^+ . Since the image of α under ψ_i has trace $\pm t_i$, the embeddings ψ_1 and ψ_2 are not equivalent, so every embedding is equivalent to precisely one of them.

3.4.2 Extension to Γ^+

A mapping

$$\gamma \mapsto \begin{pmatrix} w & x \\ y & z \end{pmatrix} \in \overline{L}$$

with wz - xy = 1 extends ψ_i to a homomorphism $\Gamma^+ \to \overline{L}$ if and only if it preserves the relations $\gamma^5 = (\beta \gamma)^2 = (\alpha \beta \gamma)^2 = 1$. Multiplying the above matrix by -1 if necessary, the first condition can be rewritten as $u = w + z = t_1$ or t_2 . The second and third conditions are equivalent to (e+1)w + (t+f)y + (f-t)x + (1-e)z =y - x = 0, giving y = x and 2ew + 2fx + (1-e)u = 0, that is:

$$2ew = (e-1)u - 2fx (3.3)$$

and so 2ez = 2eu - (e - 1)u + 2fx = (e + 1)u + 2fx. The equation wz - xy = 1 then gives:

$$4e^{2} = 2ew.2ez - 4e^{2}xy$$

= $((e-1)u - 2fx)((e+1)u + 2fx) - 4e^{2}x^{2}$
= $e^{2}u^{2} - (u + 2fx)^{2} - 4e^{2}x^{2}$,

so

$$4(e^{2} + f^{2})x^{2} + 4fux + 4e^{2} - (e^{2} - 1)u^{2} = 0$$
(3.4)

This is a quadratic equation for x, and extensions $\Gamma^+ \to \overline{L}$ of ψ_i are in one-to-one correspondence with its solutions in \overline{F}_p . Thus, for each possible choice of the pair t and u from $\{t_1, t_2\}$, there are at most two such homomorphisms. We will denote such an extension of ψ_i by θ_{ij} if $t = t_i$ and $u = t_j$. From § 3.2 it is known that any isomorphism $\psi : \Gamma_0^+ \to A_5$ has two extensions to epimorphisms $\theta : \Gamma^+ \to A_5$, with the image of γ in each case conjugate in A_5 to that of α^2 . It follows that if $i \neq j$ then the extensions $\theta_{ij} : \Gamma^+ \to \overline{L}$ of ψ_i have images isomorphic to A_5 , and hence not isomorphic to $L_2(q)$ for any power q of p, since $p \neq 2, 5$.

From now on it will be assumed that i = j. Thus t = u, so Equation 3.4 can be rewritten as:

$$(e^{2} + f^{2})x^{2} + ftx + e^{2} - \frac{1}{4}(e^{2} - 1)t^{2} = 0$$
(3.5)

a quadratic equation with discriminant

$$D = f^{2}t^{2} + (e^{2} + f^{2})((e^{2} - 1)t^{2} - 4e^{2})$$

$$= (e^{2} + t + 2)(t - 1) - (t + 2)((e^{2} - 1)(1 - t) - 4e^{2})$$

$$= e^{2}(t - 1) + (t + 2)(t - 1) + (e^{2} - 1)(t + 2)(t - 1) + 4e^{2}(t + 2)$$

$$= e^{2}(t - 1) + e^{2}(t + 2)(t - 1) + 4e^{2}(t + 2)$$

$$= e^{2}(5t + 6).$$
(3.6)

Now $e \neq 0$, so D is a square in F if and only if 5t + 6 is. If D is a square, then provided $D \neq 0$ there are two solutions for x in F, giving two homomorphisms $\theta : \Gamma^+ \to L < \overline{L}$. Since t = u their images cannot be isomorphic to A_5 , so these must be epimorphisms onto L. The exceptional case D = 0 corresponds to t = -6/5; then $36/25 = t^2 = 1 - t = 1 + 6/5$, that is, 36 = 25 + 30 = 55, so p = 19 and t = -5; in this case, there are two epimorphisms $\Gamma^+ \to L = L_2(19)$ for t = u = 4 (since $5t + 6 = 8^2$), and one for t = u = -5 corresponding to the unique solution x of Equation 3.5. For the moment, let us assume that $p \neq 19$, so that $D \neq 0$.

For $t = u = t_i$ (i = 1, 2), with corresponding values $e = e_i$, the discriminants $D_i = e_i^2(5t_i + 6) = e_i^2(7 \pm 5\sqrt{5})/2$ satisfy $D_1D_2 = -19(e_1e_2)^2$. Suppose first that -19 is a square in F; equivalently, either $p \equiv \pm 2 \mod (5)$, so $|F| = p^2$, or $p \equiv \pm 1$ mod (5) and $p \equiv 1, 4, 5, 6, 7, 9, 11, 16$ or 17 mod (19). Then either both or neither of D_1, D_2 are squares, giving four epimorphisms onto L or none respectively. Suppose first that D_1 and D_2 are both squares, so that there are four epimorphisms; since $PGL_2(F)$ preserves traces, and only its identity element centralises an A_5 in L, it acts trivially on these four epimorphisms. It follows that when $p \equiv \pm 1 \mod (5)$, so |F| = p and hence Aut $L = PGL_2(F)$, there are four normal subgroups in Γ^+ with quotient $L = L_2(p)$; when $p \equiv \pm 2 \mod (5)$, however, so that $|F| = p^2$, the Galois group of the field F transposes t_1 and t_2 . In this case Aut $L = P\Gamma L_2(F)$ has two orbits on the four epimorphisms and hence there are just two normal subgroups with quotient $L = L_2(p^2)$. If neither D_1 nor D_2 is a square, so that there are no epimorphisms onto L, then, since D_1 and D_2 are both squares in the quadratic extension K of F of order $|F|^2$, four epimorphisms onto $L_2(K)$ are obtained. If $p \equiv \pm 1 \mod (5)$, so that |F| = p, then the Galois group $\operatorname{Gal} K/F$ of this extension transposes the two epimorphisms corresponding to each discriminant, so there are two distinct normal subgroups with quotient $L_2(K) = L_2(p^2)$. If, however, $|F| = p^2$ then $\operatorname{Gal} F/\mathbb{F}_p$ transposes t_1 and t_2 , while $\operatorname{Gal} K/F$ transposes the two epimorphisms corresponding to each t_i , so a unique normal subgroup with quotient $L_2(K) = L_2(p^4)$ is obtained.

Now suppose that -19 is not a square in F, that is, $p \equiv \pm 1 \mod (5)$, so |F| = p, and $p \equiv 2, 3, 8, 10, 12, 13, 14, 15$ or 18 mod (19). In this case, exactly one D_i is a square in F, giving two epimorphisms onto $L = L_2(p)$ with distinct kernels; by adjoining the square root of the other discriminant two epimorphisms onto $L_2(K) = L_2(p^2)$ are obtained, and they are equivalent under Gal K/F.

Similar arguments show that when p = 19 three normal subgroups with quotient $L_2(19)$ are obtained: two of these correspond to epimorphisms with t = u = 4 and

D = 1, and one to an epimorphism with t = u = -5 and D = 0. These three subgroups, and their associated manifolds, will be considered in more detail in § 5.11 of Chapter 5.

3.5 The cases p = 2 and p = 5

This section studies the exceptional primes $p \in \{2, 5\}$ as specified in section § 3.4.

3.5.1 p = 5

L has a single conjugacy class of icosahedral subgroups, since any two such subgroups of $L_2(q)$ are conjugate in $L_2(q^2)$. Take $L_2(5)$ as a representative of these subgroups. Now the normalizer of $L_2(5)$ in \overline{L} is $PGL_2(5) \cong S_5 \cong \operatorname{Aut}(A_5)$, so it follows that in this case there is a single equivalence class of embeddings $\Gamma_0^+ \to \overline{L}$. Representatives for α and β are given by:

$$\alpha \mapsto \frac{1}{2} \begin{pmatrix} t-f & e+1\\ e-1 & t+f \end{pmatrix}, \ \beta \mapsto \frac{1}{2} \begin{pmatrix} e+1 & t+f\\ f-t & 1-e \end{pmatrix}$$
(3.7)

with t = 2, e = 1 and f = 0. Then, arguing as in § 3.4.2 there are two choices for γ , namely $\gamma \mapsto \begin{pmatrix} 0 & 2 \\ 2 & \pm 2 \end{pmatrix}$. Hence there are two normal subgroups with quotient $L_2(5)$, namely N_1 and N_2 , and none with quotient $L_2(5^n)$ for n > 1.

3.5.2 p = 2

If p = 2 then $L_2(2^n)$ has a subgroup $G \cong A_5$ if and only if n is even, in which case all such subgroups are conjugate to $L_2(2^2)$. Thus \overline{L} has a single conjugacy class of such subgroups.

As discussed in § 3.4.1, there are two equivalence classes of embeddings $\Gamma_0^+ \to \overline{L}$, distinguished by the traces of the images of α . These are the elements $t_i \in F_4 \setminus F_2$, that is, the primitive cube roots of 1. Define $\psi_i : \Gamma_0^+ \to L \leq \overline{L}$, for i = 1, 2, by taking $t = t_i$ and sending

$$\begin{split} \alpha \mapsto \left(\begin{array}{cc} t & t+1 \\ t & 0 \end{array} \right), \quad \beta \mapsto \left(\begin{array}{cc} t+1 & 0 \\ t & t \end{array} \right), \quad \text{so} \quad \alpha\beta \mapsto \left(\begin{array}{cc} 0 & 1 \\ 1 & 0 \end{array} \right). \end{split}$$
 If
$$\gamma \mapsto \left(\begin{array}{cc} w & x \\ y & u+w \end{array} \right) \end{split}$$

where $u = t_j$ for some j = 1, 2 then the relations $(\beta\gamma)^2 = (\alpha\beta\gamma)^2 = 1$ give (l+1)w + lx + l(u+w) = y + x = 0, so w = l(x+u) and y = x. Thus $1 = w(u+w) - xy = t(x+u)(u+t(x+u)) + x^2 = (l^2+1)x^2 + tux + tu(u+tu) = tx^2 + tux + u^2$, so $tx^2 + tux + u = 0$, that is, $x^2 + ux + ut^2 = 0$. If u = t this gives $x^2 + tx + 1 = 0$, which is irreducible over F_4 so it has two roots in F_{16} for each value of t; the Galois group Gal F_{16}/F_2 transposes the two possible values t_1, t_2 of t, and for each t the Galois group Gal F_{16}/F_2 and hence one normal subgroup of Γ^+ with quotient $L_2(2^4)$ is obtained. If $u \neq t$ then $x^2 + ux + u^2 = 0$ with roots x = 1 and $x = u^2 = t$, yielding four epimorphisms $\Gamma^+ \to L_2(2^2)$; these four choices of the pair t, x form two orbits under Gal F_4/F_2 , with x = 1 and x = t respectively, so there are two normal subgroups of Γ^+ with quotient $L_2(5)$ and A_5 , this is consistent with the earlier results on these two quotients.

3.6 The main result

The results obtained in § 3.4 and § 3.5 can be summarised in the following theorem:

Theorem 3.1 Γ^+ has only the following normal subgroups with quotient isomorphic to $L_2(q)$ where q is a power of a prime p:

- a) for p = 2 there are two normal subgroups with quotient $L_2(4)$, and one with quotient $L_2(16)$;
- b) for p = 5 there are two normal subgroups with quotient $L_2(5)$, namely those with quotient $L_2(4)$ appearing in (a);
- c) for p = 19 there are three normal subgroups with quotient $L_2(19)$;

- d) for each $p \equiv \pm 1 \mod (5)$ with $p \equiv 2, 3, 8, 10, 12, 13, 14, 15$ or 18 mod (19) there are two normal subgroups with quotient $L_2(p)$ and one with quotient $L_2(p^2)$;
- e) for each p ≡ ±1 mod (5) with p ≡ 1, 4, 5, 6, 7, 9, 11, 16 or 17 mod (19), there are either four normal subgroups with quotient L₂(p) or two with quotient L₂(p²), as (7±5√5)/2 are both squares or both non-squares in F_p;
- f) for each odd $p \equiv \pm 2 \mod (5)$ there are either two normal subgroups with quotient $L_2(p^2)$ or one with quotient $L_2(p^4)$, as $(7 \pm 5\sqrt{5})/2$ are both squares or both non-squares in $F_p(\sqrt{5}) = F_{p^2}$.

3.7 Extending to Γ

Recall that we have

SO

$$\alpha \mapsto \frac{1}{2} \left(\begin{array}{cc} t-f & e+1\\ e-1 & t+f \end{array} \right) \quad ; \quad \beta \mapsto \frac{1}{2} \left(\begin{array}{cc} e+1 & t+f\\ f-t & 1-e \end{array} \right) \quad ; \quad \gamma \mapsto \left(\begin{array}{cc} w & x\\ x & t-w \end{array} \right)$$

and equations $e^2 + f^2 + 3 = t^2 = 1 - t$, 2ew = (e - 1)t - 2fx (since $\operatorname{tr}(\alpha\beta\gamma) = 0$) and $4(e^2 + f^2)x^2 + 4fux + 4e^2 - (e^2 - 1)u^2 = 0$ (since $\det(\gamma) = 1$). Now consider extending a surjection $\phi_+ : \Gamma^+ \to L_2(q)$ to $\phi : \Gamma \to L_2(q)$. Let $g = \begin{pmatrix} x_1 & x_2 \\ x_3 & x_4 \end{pmatrix}$ be the image of a under ϕ . Note first that, since the subgroup $< a, b, d \mid a^2 = b^2 =$ $d^2 = (ab)^5 = (ad)^2 = (bd)^2 = 1 >$ must map faithfully under such a ϕ , we require $20 \mid q(q^2 - 1)$. Since the order of g is 2, we have $x_4 = -x_1$, while $\phi(a\alpha\beta) = \phi(c)$ forces $x_2 = x_3$. Next,

$$\phi(a\alpha) = \phi(b) = \begin{pmatrix} x_1(t-f) + x_2(e-1) & * \\ * & x_2(e+1) - x_1(t+f) \end{pmatrix}$$
$$\operatorname{tr}(a\alpha) = 2ex_2 - 2fx_1 = 0 \text{ and so } g = \begin{pmatrix} x_1 & fx_1/e \\ fx_1/e & -x_1 \end{pmatrix}, \text{ while}$$
$$\phi(a\alpha\beta\gamma) = \phi(d) = \begin{pmatrix} xx_1 - wfx_1/e & * \\ * & fx_1(t-w)/e + xx_1 \end{pmatrix}$$

so $x_1(2xe - 2wf + ft) = \operatorname{tr}(a\alpha\beta\gamma) = 0$. Since we cannot have $x_1 = 0$ (since then $g = \begin{pmatrix} 0 & 0 \\ 0 & 0 \end{pmatrix}$), this forces $x = \frac{f(2w-t)}{2e}$. Finally, $\operatorname{det}(g) = 1$ yields the equation $-x_1^2e^2 - f^2x_1^2 = e^2$ so $x_1^2 = \frac{-e^2}{f^2+e^2}$. To summarise, we have four equations in four unknowns:

(I) $e^2 + f^2 + 3 = 1 - t$

(II)
$$2ew = (e-1)t - 2fx$$

(III)
$$x = \frac{f(2w-t)}{2e}$$

(IV)
$$4(e^2 + f^2)x^2 + 4fux + 4e^2 - (e^2 - 1)u^2 = 0$$

Now (II) and (III) gives $(e^2 + f^2)(t - 2w) - et = 0$. From (I) we get the equation $w = \frac{t+1+et}{4+2t}$. Hence we can write $x = \frac{ft}{2(2+t)}$ (IIIa). Next, (I), (III) and (IV) combine to give $\frac{f^2t^2}{2+t} + 4e^2 - (e^2 - 1)t^2 = 0$, so $(1 - t)f^2 + (7 + 4t)e^2 + (4t - 3) = 0$ (V). Then (V) and (I) gives $e^2 = \frac{4-4t}{6+5t}$, so $f^2 = \frac{-21-7t}{6+5t}$ and $x_1^2 = \frac{4t-4}{11t+17}$.

Theorem 3.2 A map $\phi_+ : \Gamma^+ \to L_2(q)$ can be extended to a map $\phi : \Gamma \to L_2(q)$ with

$$\alpha \mapsto \frac{1}{2} \begin{pmatrix} t-f & e+1\\ e-1 & t+f \end{pmatrix} \quad ; \quad \beta \mapsto \frac{1}{2} \begin{pmatrix} e+1 & t+f\\ f-t & 1-e \end{pmatrix}$$
$$\gamma \mapsto \frac{1}{4+2t} \begin{pmatrix} (1+e)t+1 & ft\\ ft & (1-e)t+1 \end{pmatrix} \quad ; \quad a \mapsto \frac{1}{\sqrt{2+t}} \begin{pmatrix} e & f\\ f & -e \end{pmatrix}$$

provided there exists $f, e \in \mathbb{F}_p$ with $f^2 + e^2 + 3 = t^2$ and $(1 - t)f^2 + (7 + 4t)e^2 + (4t - 3) = 0$. If such a solution does not exist in \mathbb{F}_p , then one exists in \mathbb{F}_{p^2} and our representation is given by

$$\phi(a) = g \quad ; \quad \phi(b) = g\alpha \quad ; \quad \phi(c) = g\alpha\beta \quad ; \quad \phi(d) = g\alpha\beta\gamma$$

Chapter _____

Further maps to linear groups

In chapter 3, all torsion free normal subgroups of the orientation preserving subgroup Γ^+ of the the simplicial Coxeter group $\Gamma = [5,3,5]$ (the Lannér group T_4) whose factor group is of the form $L_2(q)$ were classified. As a result, a family of hyperbolic manifolds with a "large" isometry group was constructed. This aim of this chapter is to extend this classification to the other 8 simplicial co-compact Coxeter groups acting on \mathbb{H}^3 . For ease of exposition, the notation developed by Lannér [La] is used.

The complete list of compact simplicial hyperbolic Coxeter groups in dimension 3 was enumerated by Lannér [La] and is described in detail in § 2.2 of Chapter 2. The nine groups are listed here in Table 4.1.

 $\begin{array}{ll} T_1[2,2,3;3,5,2], & T_4[2,2,5;2,3,5], & T_7[2,3,3;2,5,3], \\ T_2[2,2,3;2,5,3], & T_5[2,3,3;2,4,3], & T_8[2,4,3;2,5,3], \\ T_3[2,2,4;2,3,5], & T_6[2,3,4;2,3,4], & T_9[2,3,5;2,3,5]. \end{array}$

Table 4.1: The nine Lannér groups

Each group is described by a 6-tuple $[\lambda_1, \lambda_2, \lambda_3; \mu_1, \mu_2, \mu_3]$, of numbers, where π/λ_i are the angles between a chosen base face and the other three faces of T, and π/μ_i is the dihedral angle of the edge opposite that labeled with π/λ_i . Recall that these groups can be described as follows: T_1 is the $[5, [3]^2]$ Coxeter group whose Coxeter scheme has one vertex of valence three connected to the three vertices of

valence one, one edge with label 5 and the other two edges unlabeled. T_2 , T_3 and T_4 have linear Coxeter schemes. In this case the labeling for these schemes can be described as a list. The Coxeter scheme for T_2 has edge labels 3, 5, 3. The Coxeter scheme for T_3 has edge labels 4, 3, 5. The Coxeter scheme for T_5 has edge labels 5, 3, 5. The remaining five Lannér diagrams correspond to Coxeter groups whose Coxeter scheme is a square: T_5 is the Coxeter scheme one of whose edges is labeled 4, T_6 is the Coxeter scheme with two opposite edges labeled 4, T_7 is the Coxeter scheme one of whose edges is labeled 5, T_8 has one edge labeled 4 and the opposite edge labeled 5 and T_9 has two opposite edges labeled 5.

4.1 Preliminaries

For each Lannér group T_i , the torsion free normal subgroups of T_i whose factor group has the form $L_2(q)$ will be determined in the same way as done for T_4 in chapter 3. All of the groups T_i contain one of the following two Coxeter schemes as a sub-diagram:

To classify all torsion-free normal subgroups N of $T_i = \Gamma^+$ that satisfy $\Gamma^+/N \cong L_2(q)$, representations of $[5,3]^+$ and $[4,3]^+$ in $L_2(q)$ are constructed. These representations are then extended to full epimorphisms $\Gamma^+ \to L_2(q)$ by determining the conditions under which a suitable image for $\gamma = cd$ can map to $g \in L_2(q)$ satisfying the relations in the presentation for Γ^+ . In all of the Lannér groups, ensuring that all defining relators are mapped faithfully to $L_2(q)$ is sufficient to ensure that the kernel of the map is torsion free.

4.1.1 Coxeter schemes containing [5, 3]

First suppose that the group contains the scheme [5,3]. Then write $\Gamma_0 = [5,3] \cong A_5 \times C_2$, and $\Gamma_0^+ = [5,3]^+ = \langle \alpha, \beta \mid \alpha^5 = \beta^3 = (\alpha\beta)^2 = 1 \rangle \cong A_5$. Working as in

§ 3.4.1, take $\psi_i : \Gamma_0^+ \mapsto A_5$ to be defined by

$$\alpha \mapsto \frac{1}{2} \begin{pmatrix} t-f & e+1 \\ -1+e & t+f \end{pmatrix} \text{ and } \beta \mapsto \frac{1}{2} \begin{pmatrix} e+1 & t+f \\ f-t & 1-e \end{pmatrix}$$

where $e^2 + f^2 + 3 = t^2$, $t = \frac{-1 \pm \sqrt{5}}{2}$, $\alpha = ab$ and $\beta = bc$. Then

$$\alpha\beta\mapsto \left(\begin{array}{cc} 0 & 1\\ -1 & 0 \end{array}\right)$$

Note that if $t_i = \frac{-1+(-1)^i\sqrt{5}}{2}$, then $t_{1+i} = -1 - t_i$, where the subscripts are taken mod 2. Let

$$\gamma \mapsto \left(\begin{array}{cc} w & x \\ y & z \end{array}\right) \in \overline{L},$$

with determinant wz - xy = 1. Then ψ_i can be extended to a homomorphism $\Gamma^+ \to \overline{L}$ if and only if it preserves the relations $\gamma^{v_1} = (\beta \gamma)^{v_2} = (\alpha \beta \gamma)^{v_3} = 1$, where the $v_i \in \{2, 3, 4, 5\}$. Recall that, from quadratic reciprocity, $\sqrt{5} \in F$ if $p \equiv \pm 1 \mod 5$ or $|F| = p^2$.

4.1.2 Coxeter schemes containing [4,3]

Suppose instead that Γ contains the special subgroup $\Gamma_0 = [4,3] \cong S_4 \times C_2$. Let $\overline{\mathbb{F}}_p$ be the algebraic closure of the field \mathbb{F}_p of order p, where p is prime. This is the union of the finite fields \mathbb{F}_q for all powers q of p, with the natural inclusions, so the group $\overline{L} := L_2(\overline{\mathbb{F}}_p)$ is the union of the corresponding groups $L_2(\mathbb{F}_q) = L_2(q)$, with the induced inclusions. It follows that any epimorphism $\Gamma^+ \to L_2(q)$ can be regarded as a homomorphism $\Gamma^+ \to \overline{L}$, by composition with the natural embedding $L_2(q) \to \overline{L}$. Conversely, since Γ^+ is finitely generated, the image of any homomorphism $\Gamma^+ \to \overline{L}$ is contained in a subgroup $L_2(K)$ for some finite subfield K of $\overline{\mathbb{F}}_p$; from Dickson, if $p \equiv \pm 1 \mod 8$, then the group S_4 is in $L_2(p)$. If $p \equiv \pm 3 \mod 8$, then S_4 is in $PGL_2(p)$ and hence lies in $L_2(p^2)$.

If $\theta : \Gamma^+ \to \overline{L}$ is any nontrivial homomorphism, then its restriction ψ to $\Gamma_0^+ = \langle \alpha, \beta \rangle \cong S_4$ must be an isomorphism with a subgroup $G \cong S_4$ of \overline{L} . There is a single conjugacy class of such subgroups G in \overline{L} . This is because, being finite, any pair of such subgroups are both contained in a subgroup $L_2(K)$ for some finite subfield

K of \overline{F}_p ; now $L_2(K)$ has two conjugacy classes of such subgroups, and these are all conjugate in the subgroup $PGL_2(K) \leq L_2(\widetilde{K}) \leq \overline{L}$, where \widetilde{K} is the quadratic extension of K in \overline{F}_p . Let us define two embeddings $\Gamma_0^+ \to \overline{L}$ to be equivalent if they differ by an inner automorphism of \overline{L} . Given any pair of embeddings ψ_i (i = 1, 2), the conjugacy of their images implies that ψ_2 is equivalent to an embedding ψ'_2 with the same image as ψ_1 , so ψ'_2 differs from ψ_1 by an automorphism of S_4 . Since $|\text{Out } S_4| = 1$ it follows that there is one equivalence classes of embeddings.

A representative for this class can be constructed as follows: Elements of $L_2(q)$ of order 4 form a single conjugacy class and have trace $\pm \sqrt{2}$. Let $\Gamma_0^+ = \langle \alpha, \beta | \alpha^4 = \beta^3 = (\alpha \beta)^2 = 1 \rangle \cong S_4$, and then take

$$\alpha \mapsto \frac{1}{2} \left(\begin{array}{cc} t-f & e+1 \\ -1+e & t+f \end{array} \right) \text{ and } \beta \mapsto \frac{1}{2} \left(\begin{array}{cc} e+1 & t+f \\ f-t & 1-e \end{array} \right)$$

where $e^2 + f^2 = -1$, $t = \pm \sqrt{2}$, $\alpha = ab$ and $\beta = bc$. Then, again,

$$\alpha\beta \mapsto \left(\begin{array}{cc} 0 & 1\\ -1 & 0 \end{array}\right)$$

has order 2, as required. Letting

$$\gamma \mapsto \left(\begin{array}{cc} w & x \\ y & z \end{array}\right) \in \overline{L},$$

again with determinant wz - xy = 1, then the map ψ_i can be extended to a homomorphism $\Gamma^+ \to \overline{L}$ if and only if it preserves the relations $\gamma^{v_1} = (\beta \gamma)^{v_2} =$ $(\alpha \beta \gamma)^{v_3} = 1$. Gauss' lemma for quadratic residues gives $\sqrt{2} \in F$ if and only if $p \equiv \pm 1 \mod 8$, or if $|F| = p^2$.

4.1.3 Extending $\Gamma^+ \twoheadrightarrow L_2(q)$ to $\Gamma \twoheadrightarrow L_2(q)$

Suppose now that there exists a normal subgroup N of Γ^+ with $\Gamma^+/N \cong L_2(q)$. A natural question is to ask under what conditions does the map Γ^+/N extend to a map Γ/\widetilde{N} with $\Gamma/\widetilde{N} \cong L_2(q)$ and $N < \widetilde{N}$. Let

$$a \mapsto \left(\begin{array}{cc} x_1 & x_2 \\ x_3 & x_4 \end{array}\right).$$

Then |a| = 2 gives $x_4 = -x_1$ and $c = a\alpha\beta$ with |c| = 2 gives $x_2 = x_3$. Also

$$b = a\alpha \mapsto \frac{1}{2} \begin{pmatrix} x_1 & x_2 \\ x_3 & x_4 \end{pmatrix} \begin{pmatrix} t-f & e+1 \\ -1+e & t+f \end{pmatrix}$$

with trace $\frac{1}{2}(x_1(t-f)+x_2(e-1)+x_3(e+1)+x_4(t+f))=0$. $x_4=-x_1$ and $x_2=x_3$ gives $-x_1f+x_2e=0$. Assuming that $e\neq 0$, then $x_2=\frac{x_1f}{e}$. Now det(a)=1 gives $-x_1^2(e^2+f^2)=e^2$. If the Coxeter scheme contains the subscheme [5,3] then $e^2+f^2=-2-t$ and so $x_1^2=\frac{e^2}{2+t}$. If the Coxeter scheme contains the subscheme [4,3] then $e^2+f^2=-1$ and so $x_1^2=e^2$. Finally,

$$d = a\alpha\beta\gamma = \begin{pmatrix} x_1(y - \frac{fw}{e}) & x_1(z - \frac{fx}{e}) \\ x_1(w + \frac{fy}{e}) & x_1(x + \frac{fz}{e}) \end{pmatrix}$$
(4.1)

and this has trace $x_1(x + y + f\frac{z-w}{e}) = 0$. Assuming that $x_1 = 0$ give a as the zero matrix, a contradiction. Consequently the equation

$$ex + ey + fz - fw = 0 \tag{4.2}$$

must hold. Equation (4.2) can then be combined with the conditions derived for the entries in the matrices α , β and γ to solve for e and f explicitly. If such a solution exists, then the associated normal subgroup $\widetilde{N} < \Gamma$ satisfies the subgroup diagram given in Table 4.2



Table 4.2: Subgroup structure in Γ

4.1.4 A note on the order of elements in $L_2(2^n)$:

Let $L_2(2^n)^{(\infty)}$ be the subgroup of $L_2(2^n)$ fixing ∞ and let $Syl_2^{(\infty)}$ be the Sylow 2-subgroup of $L_2(2^n)^{(\infty)}$, so $Syl_2^{(\infty)} = \left\{ \begin{pmatrix} 1 & \mu \\ 0 & 1 \end{pmatrix} \right\}$ where $\mu \in \mathbb{F}_{2^n}$. Then from chapter

12 of Dickson [Di] the following three results can be obtained:

Lemma 4.1.1 Every nonidentity element of $Syl_2^{(\infty)}$ has order 2. If $L_2(2^n)^{(k)}$ is the subgroup of $L_2(2^n)$ fixing any other element $k \in \mathbb{F}_{2^n} \bigcup \{\infty\}$, then its Sylow 2subgroup $Syl_2^{(k)}$ is conjugate in $L_2(2^n)$ to $Syl_2^{(\infty)}$. These groups have trivial pairwise intersection and together contain all elements of $L_2(2^n)$ of order 2.

Lemma 4.1.2 $L_2(2^n)$ contains $2^{n-1}(2^n + 1)$ cyclic groups of order $2^n - 1$. These groups are all conjugate in $L_2(2^n)$ and together contain $2^{n-1}(2^n + 1)(2^n - 2)$ nontrivial elements of order dividing $2^n - 1$.

Lemma 4.1.3 $L_2(2^n)$ contains $2^{n-1}(2^n - 1)$ cyclic groups of order $2^n + 1$. These groups are all conjugate in $L_2(2^n)$ and together contain $2^{2n-1}(2^n - 1)$ nontrivial elements of order dividing $2^n + 1$.

This forms a complete list of all non-trivial elements of $L_2(2^n)$. As a result, the following results can be stated:

Theorem 4.1 Let n be any positive integer. Then $L_2(2^n)$ contains no element of order 4. Therefore the groups $T_3[2, 2, 4; 2, 3, 5]$, $T_5[2, 3, 3; 2, 4, 3]$, $T_6[2, 3, 4; 2, 3, 4]$ and $T_8[2, 4, 3; 2, 5, 3]$ contain no torsion free normal subgroup whose factor group is isomorphic to $L_2(2^n)$.

4.2 $\Gamma = T_1[2, 2, 3; 3, 5, 2]$

4.2.1 Γ^+ to $L_2(q)$

Let $v_1 = 2$, $v_2 = 3$ and $v_3 = 2$, so $\gamma^2 = (\beta\gamma)^3 = (\alpha\beta\gamma)^2 = 1$. Then $|\gamma| = 2$ gives w + z = 0 (1), while $|\beta\gamma| = 3$ and $|\alpha\beta\gamma| = 2$ give $(e + 1)w + (t + f)y + (f - t)x + (1 - e)z = \pm 1$ (2) and y - x = 0 (3), respectively. Since det $(\gamma) = 1$, then $w^2 + x^2 + 1 = 0$. substituting conditions (1) and (3) back into condition (2) gives 2ew = (u - 2fx), where $u = \pm 1$. Using this and the equation from the determinant gives the quadratic expression:

$$4(e^{2} + f^{2})x^{2} - 4fux + (4e^{2} + 1) = 0.$$
(4.3)

This has discriminant $D = 16f^2 - 16(e^2 + f^2)(1 + 4e^2) = ... = (7 + 4t)(4e)^2$, which is a square in F when (7 + 4t) is a square.

Write $D_i = (7+4t_i)(4e_i)^2$. Then $D_1D_2 = 5(16e_1e_2)^2$ which is a square in F if 5 is a square. 5 is a square mod p if $p \equiv \pm 1 \mod 5$ or if $|F| = p^2$. Because $\operatorname{tr}(\alpha) = \frac{-1\pm\sqrt{5}}{2}$, then either both or neither D_1 and D_2 are squares in F. First suppose that they both are. Then if $p \equiv \pm 1 \mod 5$, for each (t_i, u_j) , where $i, j \in \{1, 2\}$, two epimorphisms onto $L_2(p)$ are obtained, giving a total of 8 epimorphisms. If $p \equiv \pm 2$ mod 5, then $|F| = p^2$, and the Galois automorphism $\phi \in \operatorname{Gal}(F/\mathbb{F}_p)$ interchanges t_1 and t_2 , giving 4 distinct non-conjugate epimorphisms.

Now suppose that neither D_i is a square in F. If $p \equiv \pm 1 \mod 5$, then adjoining $\sqrt{D_1}$ to \mathbb{F}_p 8 epimorphisms are obtained. Now the non-trivial Galois automorphism $\phi \in \operatorname{Gal}(\mathbb{F}_p(\sqrt{D_1})/\mathbb{F}_p)$ swaps $\sqrt{D_1}$ and $-\sqrt{D_1}$, so for any pair (t_i, u_j) , where $i, j \in \{1, 2\}, \phi$ swaps the two solutions for x. Hence 4 distinct epimorphisms $\Gamma \twoheadrightarrow L_2(p^2)$ are obtained. If $p \equiv \pm 2 \mod 5$, then adjoining $\sqrt{D_1}$ to \mathbb{F}_p^2 , 4 epimorphisms are obtained, and again the Galois automorphism swaps $\sqrt{D_1}$ and $-\sqrt{D_1}$, so 2 distinct epimorphisms $\Gamma \twoheadrightarrow L_2(p^4)$ are obtained.

Now the exceptional cases correspond to the discriminant $D = (7 + 4t)4e^2 = 0$. Because it was assumed that $e \neq 0$, then either 4 = 0, so p = 2, or (7 + 4t) = 0, in which case 44 = 0, so $p \in \{2, 11\}$. Now Γ is an index two subgroup of the [5, 3, 4] group studied in § 4.4.1. Because [5, 3, 4] has a unique conjugacy class of subgroups of index 2, Γ must be isomorphic to $[5, 3, 4]^+$ therefore by § 4.4.1 a torsion free epimorphism $\Gamma^+ \rightarrow L_2(2^n)$ cannot be constructed. There remains the case p = 11. In this case, for each set of values t_i, u_j , where $i, j \in \{1, 2\}$, a unique map $\Gamma^+ \rightarrow L_2(11)$ is constructed.

Theorem 4.2 Let $\Gamma = T_1$ and let $D_i = (4e_i)^2(7 + 4t_i)$ be the discriminant of equation 4.3. Then Γ^+ has torsion free normal subgroups N with $\Gamma^+/N \cong L_2(q)$ under the following conditions:

 If p ≡ ±1 mod 5 then either both discriminants D_i are squares in F_p or neither are. If they are both squares then Γ⁺ has 8 distinct torsion-free normal subgroups N with factor group Γ⁺/N ≅ L₂(p). If neither discriminant is a square then Γ^+ has 4 distinct torsion-free normal subgroups N with factor group $\Gamma^+/N \cong L_2(p^2)$.

2) If p ≡ ±2 mod 5 then either both discriminants D_i are squares in F_{p²} or neither are. If they are both squares then Γ⁺ has 4 distinct torsion-free normal subgroups N with factor group Γ⁺/N ≅ L₂(p²). If neither discriminant is a square then Γ⁺ has 2 distinct torsion-free normal subgroups N with factor group Γ⁺/N ≅ L₂(p⁴).

4.2.2 Extending to $\Gamma \twoheadrightarrow L_2(q)$

Let $e^2 + f^2 = -2 - t$, where $t = \frac{-1 \pm \sqrt{5}}{2}$ and $e \neq 0$, y = x, w = -z and $u = \pm 1$ be as in the previous section. Suppose that Γ^+ has a normal subgroup N with $\Gamma^+/N \cong L_2(q)$. From equation 4.2 in § 4.1.3 the condition

$$ex + ey + fz - fw = 0 \tag{4.4}$$

is obtained. From the results obtained in the previous section, the two equations

$$2ew = u - 2fx \tag{4.5}$$

and

$$4(e^{2} + f^{2})x^{2} - 4fux + (4e^{2} + 1) = 0$$
(4.6)

are obtained. Since y = x and w = -z, equation 4.4 becomes $x = \frac{fw}{e}$. Substituting this into equation 4.5 gives $w = \frac{eu}{-2(2+t)}$. Substituting in for x and w in equation 4.6 gives

 $f^2 + (1+4e^2)(2+t) = 0$

Since $e^2 + f^2 = -2 - t$, this can be rewritten to give

$$(7+4t)e^2 = 0$$

Since it is assumed that $e \neq 0$, $7 + 4t = 5 \pm 2\sqrt{5} = 0$. Solving give 5 = 0, so the only possible case with $\Gamma/\widetilde{N} \cong L_2(q)$ is for q = 5.

Theorem 4.3 There exists a normal subgroup $\widetilde{N} \triangleleft \Gamma$ with $\Gamma/\widetilde{N} \cong L_2(q)$ and $\Gamma^+/\Gamma^+ \cap \widetilde{N} \cong L_2(q)$ if and only if p = 5.

4.3 $\Gamma = T_2[2, 2, 3; 2, 5, 3]$

This group has been already extensively studied in the work of [JM]. The classification of all torsion-free normal subgroups N of Γ^+ whose factor group $\Gamma/N \cong L_2(q)$ has been done by Anna Torstensson in her PhD thesis [To], [CMT]. The statement is for completeness sake.

Theorem 4.4 For any prime p there exists a positive integer k such that either $L_2(q)$ or $PGL_2(q)$ is a quotient of Γ^+ by some torsion-free normal subgroup.

4.4
$$\Gamma = T_3[2, 2, 4; 2, 3, 5]$$

4.4.1 Γ^+ to $L_2(q)$

Following the notation of section § 4.1.1, let $v_1 = 4$, $v_2 = 2$ and $v_3 = 2$, so $\gamma^4 = (\beta\gamma)^2 = (\alpha\beta\gamma)^2 = 1$. Then $|\gamma| = 4$ gives $u = \pm\sqrt{2} = w + z$. $|\beta\gamma| = 2$ and $|\alpha\beta\gamma| = 2$ give (e+1)w + (t+f)y + (f-t)x + (1-e)z = y - x = 0, respectively, giving y = x and 2ew + 2fx + (1-e)u = 0, that is, 2ew = (e-1)u - 2fx and so 2ez = 2eu - (e-1)u + 2fx = (e+1)u + 2fx. The determinant wz - xy = 1 then gives

$$4e^{2} = 2ew.2ez - 4e^{2}xy$$

= $((e-1)u - 2fx)((e+1)u + 2fx) - 4e^{2}x^{2}$
= $e^{2}u^{2} - (u+2fx)^{2} - 4e^{2}x^{2}$,

SO

$$2(e^{2} + f^{2})x^{2} + 2fux + (e^{2} + 1) = 0.$$
(4.7)

Equation 4.7 is a quadratic equation for x, and extensions $\Gamma^+ \to L_2(F) < \overline{L}$ of ψ_i are in one-to-one correspondence with its solutions in \overline{F}_p , therefore for each possible choice of the pair t from $\{t_1, t_2\}$ and u from $\{\pm\sqrt{2}\}$ there are at most two such homomorphisms. Such an extension of ψ_i will be denoted by θ_{ij} if $t = t_i$ and $u = u_j$. Now equation (4.7) has discriminant $D = 4f^2u^2 - 8(e^2 + f^2)(e^2 + 1) = 8f^2 - 8(e^2 + f^2)(e^2 + 1) = 4e^2(2t+2)$. Since $e^2 \neq 0$, this is a square if and only if

 $2t + 2 = 1 \pm \sqrt{5}$ is a square. If *D* is a square, then for a choice u_i (provided $D \neq 0$) there are two solutions for *x* in *F*, giving two homomorphisms $\theta_i : \Gamma^+ \rightarrow L_2(F) < L_2(\overline{F})$, so a total of two epimorphisms is obtained. The exceptional case D = 0corresponds to p = 2 which has already been excluded in § 4.1.4.

For $t = t_i$ the discriminants $D_i = 4e_i^2(2t_i + 2)$ satisfy $D_1D_2 = -(8e_2e_2)^2$. If $-1 \in F$, so $p \equiv 1 \mod 4$ or $|F| = p^2$, then either both or neither of D_1 and D_2 are squares in F. First suppose that they are both squares. Then there exists four epimorphisms $\Gamma^+ \to L$. Now $PGL_2(F)$ preserves traces and $B \in PGL_2(F)$ centralises $A_5 \cong \langle \alpha, \beta \mid \ldots \rangle$ iff B = Id. Let $p \equiv 1 \mod 8$. If $p \equiv \pm 1 \mod 5$, then |F| = p and $\operatorname{Aut}(L) = \operatorname{PGL}_2(F)$, so there exists 8 distinct epimorphisms. If, however, $p \equiv \pm 2 \mod 5$ then there exists four epimorphisms $\phi : \Gamma^+ \to L_2(F')$, where F' is a degree two extension of F, forming four conjugacy classes of size two under the action of the Galois automorphism, giving two distinct automorphisms.

Now let $p \equiv -3 \mod 8$. If $p \equiv \pm 1 \mod 5$, then there are four epimorphisms $\phi: \Gamma^+ \twoheadrightarrow L_2(F)$ forming two conjugacy classes of size two under the action of the Galois automorphism, giving two distinct automorphisms. If $p \equiv \pm 2 \mod 5$ then there exists $w \in \mathbb{F}_p$ such that $\sqrt{2} = w\sqrt{5}$. As a result the Galois automorphism simultaneously swaps the traces t_1 , t_2 and the traces u_1 , u_2 , giving two distinct epimorphisms.

Now suppose that neither D_1 nor D_2 are squares in F. Then they are squares in the extension $F(\sqrt{D_i})$. There are several cases to consider:

- a) If $p \equiv \pm 1 \mod 5$ and $p \equiv 1 \mod 8$, so |F| = p, then the Galois group of the extension transposes the two epimorphisms corresponding to each discriminant, so up to conjugacy there exists two epimorphisms onto $L_2(p^2)$.
- b) If $p \equiv \pm 1 \mod 5$ and $p \equiv \pm -3 \mod 8$ then $F = \mathbb{F}(\sqrt{2})$ and the Galois group of the extension $K = F(D_1)$ transposes the two epimorphisms corresponding to each discriminant, while the Galois group of F/\mathbb{F} transposes the traces u_1 and u_2 , so up to conjugacy there exists a unique epimorphism onto $L_2(p^4)$.
- c) If $p \equiv \pm 2 \mod 5$ and $p \equiv \pm 1 \mod 8$, then, arguing as in b) above, there exists, up to conjugacy, two distinct epimorphisms onto $L_2(p^4)$.

d) If $p \equiv \pm 2 \mod 5$ and $p \equiv -3 \mod 8$ write $F = \mathbb{F}_p(\sqrt{5})$. Then $\sqrt{2} = b'\sqrt{5}$ for some $b' \in \mathbb{F}_p$ and up to conjugacy there exists a unique epimorphism onto $L_2(p^4)$.

If $\sqrt{-1} \notin F$, then precisely one of D_1 or D_2 is a square in $L_2(F)$. Now, $\sqrt{-1} \notin F$ if and only if $F = \mathbb{F}_p$ and $p \equiv -1 \mod 4$. Without loss of generality it can be assumed that D_1 is a square. As a result, $p \equiv \pm 1 \mod 5$. Now $p \equiv -1 \mod 4$ so either $p \equiv -1 \mod 8$ or $p \equiv 3 \mod 8$. If $p \equiv 3 \mod 8$ then there are no epimorphisms onto $L_2(p)$, while if $p \equiv -1 \mod 8$ then, up to conjugation by an automorphism of $L_2(q)$, there are 4 distinct epimorphisms. Adjoining the square root of the other discriminant gives four epimorphisms onto $L_2(K)$ (where $K = F(\sqrt{D_2})$), forming two conjugacy classes of epimorphisms. The exceptional case where the discriminant equals zero can only happen if p = 2, which is excluded in § 4.1.4.

The results obtained can be summarised in the following theorem:

Theorem 4.5 Let $\Gamma^+ = T_3^+$ and let D_i be a discriminant for equation 4.7. Then

- If p ≡ ±1 mod 5, p ≡ 1 mod 8, then either Γ⁺ has four distinct normal subgroups N with Γ⁺/N ≅ L₂(p) or it has none. In the second case it has two distinct distinct normal subgroups N with Γ⁺/N ≅ L₂(p²)
- 2) If $p \equiv 1 \mod 4$ and
 - $p \equiv \pm 2 \mod 5$ and $p \equiv 1 \mod 8$, or
 - $p \equiv \pm 1 \mod 5$ and $p \equiv -3 \mod 8$, or
 - $p \equiv \pm 2 \mod 5$ and $p \equiv -3 \mod 8$

then either Γ^+ admits two distinct distinct normal subgroups N with $\Gamma^+/N \cong L_2(p^2)$ or none. If there are no such normal subgroups N, then there is a unique normal subgroup N with $\Gamma^+/N \cong L_2(p^4)$.

3) If p ≡ −1 mod 4, p ≡ ±1 mod 5 and p ≡ −1 mod 8 then there are two distinct normal subgroups N with Γ⁺/N ≅ L₂(p) and there is a uniequ normal subgroup N with Γ⁺/N ≅ L₂(p²)

4.4.2 Extending to $\Gamma \twoheadrightarrow L_2(q)$

Let $e^2 + f^2 = -2 - t$, where $t = \frac{-1 \pm \sqrt{5}}{2}$ and $e \neq 0$, y = x, z = u - w and $u = \pm \sqrt{2}$ be as in the previous section. Suppose that Γ^+ has a normal subgroup N with $\Gamma^+/N \cong L_2(q)$. From equation 4.2 in § 4.1.3 the condition

$$ex + ey + fz - fw = 0 \tag{4.8}$$

is obtained. From the results obtained in the previous section, the two equations

$$2ew = eu - u - 2fx \tag{4.9}$$

and

$$2(e^{2} + f^{2})x^{2} + 2fux + (e^{2} + 1) = 0$$
(4.10)

are obtained. Since y = x and z = u - w, equation 4.8 becomes $x = \frac{2fw-fu}{2e}$. Substituting this into equation 4.9 gives $w = \frac{u(2+t)+eu}{2(2+t)}$. Substituting in for x and w in equation 4.10 gives

$$2f^4 + (2t + 4 + 2e^2)f^2 + 5 + 5e^2 + 3e^2t + 3t = 0$$

Since $e^2 + f^2 = -2 - t$, this can be rewritten to give

 $1 + e^2 = 0$

so $e = \sqrt{-1}$ and hence either $p \equiv 1 \mod 4$ or $q = p^{2n}$. Assume that $p \equiv 1 \mod 4$ or $q = p^{2n}$. Writing $e^2 = -1$ gives $f^2 = -1 - t$, so $f = \pm \sqrt{-1 - t}$. If $t = \frac{-1 \pm \sqrt{5}}{2}$, then $-1 - t = \frac{-1 \pm \sqrt{5}}{2}$. Since $\sqrt{2}$ is already assumed to be in any field with $\Gamma^+ \rightarrow L_2(q)$, f is a square if and only if $\sqrt{-1 \pm \sqrt{5}}$ is.

Theorem 4.6 Suppose there exists a normal subgroup $N \triangleleft \Gamma^+$ with factor group $\Gamma^+/N \cong L_2(q)$ and let t be the trace of the image of α in the quotient group. Then there exists an extension \widetilde{N} of N with $\Gamma/\widetilde{N} \cong L_2(q)$ if and only if $q \cong 1 \mod 4$ and $\sqrt{-1-t}$ lies in \mathbb{F}_q .

4.5 $\Gamma = T_4[2, 2, 5; 2, 3, 5]$

Epimorphic images of this group have been extensively studied in Chapter 3.

4.6 $\Gamma = T_5[2, 3, 3; 2, 4, 3]$

In this case $t = \pm \sqrt{2}$ and $e^2 + f^2 = -1$.

4.6.1 Maps from Γ^+ to $L_2(q)$

Following the notation of section § 4.1.2, let $v_1 = 3$, $v_2 = 2$ and $v_3 = 3$, so $\gamma^3 = (\beta\gamma)^2 = (\alpha\beta\gamma)^3 = 1$. Then $|\gamma| = 3$ gives $\operatorname{tr}(\gamma) = \pm 1 = u$, while $|\beta\gamma| = 2$ and $|\alpha\beta\gamma| = 3$ become the conditions $\operatorname{tr}(\beta\gamma) = 0$ and $\operatorname{tr}(\alpha\beta\gamma) = \pm 1 = k$, respectively. $\operatorname{tr}(\gamma) = u$ forces z = u - w and $\operatorname{tr}(\alpha\beta\gamma) = k$ gives y = x + k. Section § 4.1.4 precludes the exceptional prime p = 2. Write

$$\gamma \mapsto \left(\begin{array}{cc} w & x \\ x+k & u-w \end{array}\right)$$

The condition $\operatorname{tr}(\beta\gamma) = 0$ gives $\frac{1}{2}(2ew+2fx+tk+fk+u-eu) = 0$ while $\operatorname{det}(\gamma) = 1$ gives $wu - w^2 - x^2 - xk = 1$. Rewriting $\operatorname{det}(\gamma) = 1$ as $4e^2\operatorname{det}(\gamma) = 4e^2$, and letting 2ew = eu - u - fk - tk - 2fx, the following quadratic expression is obtained for x:

$$4x^{2} + (-4fkt - 4fu + 4k)x + (-2ft + 2f^{2} - 2fuk - 2utk) = 0 \quad (4.11)$$

Equation 4.11 has discriminant $16(1 + f^2)(1 + 2utk) = -16e^2(1 + 2utk)$. Write $D_i = -16e_i^2(1 + 2t_i)$. Then $D_1D_2 = -7(16e_1e_2)^2$. Now this is a square if $\sqrt{7}$ lies in F, so either $|F| = p^2$ (equivalence of field extensions of \mathbb{F}_p), or, by quadratic reciprocity, $p \equiv 1, 2, 4$ or 4 mod 7. Since $k, u \in \{\pm 1\}$, there are two cases to consider: k = u and k = -u.

Case I: k = u

Equation 4.11 becomes $4x^2 + 4k(1 - f - tf)x + (-2ft + 2f_i^2 - 2f - 2t)$ with discriminant $D = -16e^2(1 + 2t)$. If $p \cong 1, 2$ or 4 mod 7 or if $|F| = p^2$, then either both D_i are squares in F or neither are. Suppose first that they both are. Then it follows that the choice of t will not affect D_i . WLOG choose $t = \sqrt{2}$ and write $D = -16e^2(1 + 2\sqrt{2})$. Then for each value ± 1 of u = k there are two solutions for x and these solutions are equivalent. This can be seen by writing $x = \frac{k}{2}((f + tf - 1) \pm \frac{e}{k}\sqrt{-1 - 2t} \text{ and } 2ew = ek - k - fk - tk - 2fx \text{ becomes}$ $w = \frac{k}{2}((1 + e + et) \pm \frac{-f}{k}\sqrt{-1 - 2t})$. Then clearly the solution for γ does not depend on the value of k. So there are 4 distinct normal subgroups in total.

If neither D_i is a square in F, then they both are in $F(\sqrt{D})$. Now, if $p \cong \pm 1$ mod 8, then $F(\sqrt{D}) \equiv \mathbb{F}_{p^2}$. Let $\overline{\alpha}, \overline{\beta}$ and $\overline{\gamma}$ be the images of α, β and γ under the epimorphism constructed. The maximal subgroups of $L_2(p^2)$ containing $\langle \overline{\alpha}, \overline{\beta} \rangle \cong S_4$ are $L_2(p)$ or $PGL_2(p)$. Since Γ^+ contains no subgroup of index 2, and since the image of γ must lie in $L_2(p^2) \setminus L_2(p)$, it cannot be $PGL_2(p)$ and so, for each value $u = \pm 1$ and each D_i , a unique epimorphism $\Gamma^+ \twoheadrightarrow L_2(p^2)$ is constructed, giving a total of two distinct normal subgroups with factor group $L_2(p^2)$.

If $p \cong \pm 3 \mod 8$, then S_4 is not a subgroup of $L_2(p)$, but $S_4 < PGL_2(p) < L_2(p^2)$. The subgroup $\langle \overline{\alpha}, \overline{\beta} \rangle$ of Γ^+ then embeds into $PGL_2(p) < L_2(p^2)$. From Dickson, [Di, §255], $PGL_2(p)$ is maximal in $L_2(p^2)$, so adjoining $\overline{\gamma}$ to $\langle \overline{\alpha}, \overline{\beta} \rangle$, generates the full group $L_2(p^4)$ provided $\gamma \notin \langle \overline{\alpha}, \overline{\beta} \rangle$. Since neither D_i is a square in \mathbb{F}_{p^2} it follows that $\gamma \in L_2(p^4) \setminus L_2(p^2)$, so $\langle \overline{\alpha}, \overline{\beta}, \overline{\gamma} \rangle \cong L_2(p^4)$. The non-trivial Galois automorphism of \mathbb{F}_{p^2} swaps $\sqrt{2}$ and $-\sqrt{2}$ while $\phi \in \operatorname{Gal}(\mathbb{F}_{p^4}/\mathbb{F}_{p^2})$ swaps D_1 and D_2 . Hence, up to conjugacy, there is a unique epimorphism $\Gamma^+ \to L_2(p^4)$.

If $p \cong 3,5$ or 6 mod 7, then precisely one of D_1 , D_2 is a square in \mathbb{F}_p . By changing labels if necessary, assume that it is D_1 . Then arguing as above, there are two distinct normal subgroups of Γ^+ whose factor group is $L_2(p)$. Adjoining $\sqrt{D_2}$ to F gives a map into $L_2(p^2)$, and again, by the arguments similar to those above, this map is onto.

Case II: k = -u

In this case Equation 4.11 becomes $4x^2 + 4k(1 + f - tf)x + (-2ft + 2f^2 + 2f + 2t)$, an equation with discriminant $D = -16e^2(1 - 2t)$. The analysis of the solutions of this equation is identical to that for the previous case k = u.

Finally, there remains to check when the solutions obtained for k = u and for k = -u coincide. If $p \cong \pm 3 \mod 8$ then, under the action of $\operatorname{Gal}(F/\mathbb{F}_p)$, the

solutions for u = k and u = -k coincide.

Theorem 4.7 Let Γ be the Lannér group $T_5[2,3,3;2,4,3]$ and let Γ^+ be the orientation preserving subgroup.

If $p \equiv \pm 1 \mod 8$ and if $p \equiv 1, 2$ or 4 mod 7, then either both or neither of the discriminants D_i are squares in F. If they both are then, up to automorphisms, there are 4 distinct normal subgroups with factor group $L_2(p)$. If neither are then there are 2 distinct normal subgroups with factor group $L_2(p^2)$.

If $p \equiv \pm 3 \mod 8$ then either both or neither of the discriminants D_i are squares in F. If they both are then there are 2 distinct normal subgroups with factor group $L_2(p^2)$. If neither are then there is a unique normal subgroup with factor group $L_2(p^4)$.

If $p \equiv \pm 1 \mod 8$ and if $p \equiv 3$, 5 or 6 mod 17, then there are 2 distinct normal subgroups with factor group $L_2(p)$ and a unique normal subgroup with factor group $L_2(p^2)$.

4.6.2 The exceptional case p = 7

In the case p = 7, write $D = e^2(1 + 2utk)$. In \mathbb{F}_7 , $\sqrt{2} = \pm 3$. Now if t = 3 then $D = 0 \mod 17$ if u = k, while if t = -3 then D = 0 if u = -k. Choose t = 3. If $D \neq 0 \mod 7$, then u = -k and $D = -16e^2(1 + 2ukt) = e^2 \times 5$. Since 5 is not a square in \mathbb{F}_{17} , there are no epimorphisms $\Gamma \twoheadrightarrow L_2(7)$ for $D \neq 0$. However, there exists an epimorphism $\Gamma^+ \twoheadrightarrow L_2(49)$, corresponding to the quotient Γ^+/K , where K is the unique normal subgroup with this quotient. We now look at the two cases where $D = 0 \mod 7$:

Equation 4.11 becomes $4x^2 = 0$, so x = 0. Since $-1 \equiv 6 \mod 7$ is not a square in \mathbb{F}_7 , choose $e = \sqrt{4}$ and f = 2. Then $e^2 + f^2 = -1$, while 2ew = eu - u - fk - tk - 2fx gives $2ew = -3u \equiv 4u \mod 7$, or $w = 2u/3 \equiv 3u \mod 7$. Then substituting in the values for e, f and t into α , β and γ gives

$$\alpha \mapsto \begin{pmatrix} 4 & 2 \\ 1 & 6 \end{pmatrix} \quad \beta \mapsto \begin{pmatrix} 2 & 3 \\ 6 & 6 \end{pmatrix} \quad \gamma \mapsto k \begin{pmatrix} 4 & 3 \\ 2 & 4 \end{pmatrix}$$

Since $k = \pm 1$ and the image $\overline{\gamma_1}$ of γ lies in $L_2(7)$ both maps are equivalent.

Theorem 4.8 In the exceptional case where p = 7 and $D_i = 0$ a unique epimorphism $\Gamma^+ \twoheadrightarrow L_2(7)$ is recovered.

4.6.3 Extending to $\Gamma \twoheadrightarrow L_2(q)$

Let $e^2 + f^2 = -1$, where $t = \sqrt{2}$ and $e \neq 0$, y = x + k where $k = \pm 1$ and z = u - w where $u = \pm 1$ be as in the previous section. Suppose that Γ^+ has a normal subgroup N with $\Gamma^+/N \cong L_2(q)$. From equation 4.2 in § 4.1.3 the condition

$$ex + ey + fz - fw = 0 \tag{4.12}$$

is obtained. From the results obtained in the previous section, the two equations

$$2ew = eu - u - \int k - tk - 2\int x \tag{4.13}$$

and

$$4x^{2} + (-4fkt - 4fu + 4k)x + (-2ft + 2f^{2} - 2fuk - 2utk) = 0 \quad (4.14)$$

are obtained. Since y = x + k and z = u - w, equation 4.12 becomes $x = \frac{2fw - fu - ek}{2e}$. Substituting this into equation 4.13 gives $w = \frac{u + eu + etk}{2}$. Substituting in for x and w in equation 4.14 gives

$$(1+2tuk)f^2 + 1 + tuk = 0$$

implying that $f^2 = -1$ and consequently $e^2 = 0$, a contradiction, or 1 + tuk = 0. If 1 + tuk = 0, then $1 = 4t^2u^2k^2 = 8$ and so 7 = 0, so p = 7. In this case there is a unique normal subgroup K of Γ with factor group $L_2(7)$. Using GAP, it was shown that K is not torsion free - it contains the two conjugacy classes $\mathcal{C}((abc)^3)$ and $\mathcal{C}((bad)^3)$ of elements of order 2.

Theorem 4.9 Suppose there exists a normal subgroup $N \triangleleft \Gamma^+$ with factor group $\Gamma^+/N \cong L_2(q)$. Then there can exist an extension \widetilde{N} of N with $\Gamma/\widetilde{N} \cong L_2(q)$ if and only if p = 7. The subgroup \widetilde{N} is not torsion free.

4.7 $\Gamma = T_6[2, 3, 4; 2, 3, 4]$

The orientation preserving subgroup of Γ has the following presentation:

$$\Gamma^{+} = \langle \alpha, \beta, \gamma | \alpha^{4} = \beta^{3} = \gamma^{4} = (\alpha \beta)^{2} = (\beta (\gamma)^{2} = (\alpha \beta \gamma)^{3} = 1 \rangle$$

The subgroup Γ^+ is not perfect: If $N = \langle \langle \beta \rangle \rangle$ is the normal closure in Γ^+ of the subgroup $\langle \beta | \beta^3 = 1 \rangle$, then the factor group Γ^+/N is isomorphic to the dihedral group D_3 . Consequently, Γ^+ has an index two subgroup.

4.7.1 Normal subgroups with quotient S_4

Lemma 4.7.1 There are precisely two distinct normal subgroups of Γ^+ whose factor group is isomorphic to S_4 .

Proof: The proof amounts to constructing two non-conjugate epimorphisms $\Gamma^+ \to S_4$. Let $\alpha \mapsto \overline{\alpha} = (1,3,2,4)$ and $\beta \mapsto \overline{\beta} = (1,2,3)$. Then for $\overline{\gamma} \in \{(1,3,2,4), (1,3,4,2)\}$ we get $|\gamma| = 4$, $|\beta\gamma| = 2$ and $|\alpha\beta\gamma| = 3$. This proves existence of the two maps. If $\gamma \mapsto \overline{\gamma} = (1,3,2,4)$ then $\overline{\alpha\gamma} = \overline{\alpha}^2$ has order 2, while if $\gamma \mapsto \overline{\gamma} = (1,3,4,2)$ then $\overline{\alpha\gamma} = (1,4,3)$ has order 3. Hence the maps are not conjugate. Finally it is useful to note that $(1,3,4,2) = (1,3,2,4)^{(2,4)} = \overline{\beta}^{-1} \overline{\alpha\beta}$.

4.7.2 $\Gamma^+ \twoheadrightarrow L_2(q)$

Write α and β as in section 4.1.2. Then $e^2 + f^2 = -1$ and $t = \pm \sqrt{2}$. Let

$$\gamma \mapsto \left(\begin{array}{cc} w & x \\ y & z \end{array}\right)$$

Then $|\gamma| = 4$, $|\beta\gamma| = 2$ and $|\alpha\beta\gamma| = 3$ become the conditions $\operatorname{tr}(\gamma) = \pm\sqrt{2} = u$, $\operatorname{tr}(\beta\gamma) = 0$ and $\operatorname{tr}(\alpha\beta\gamma) = \pm 1 = k$, respectively. $\operatorname{tr}(\gamma) = u$ gives z = u - w while $\operatorname{tr}(\alpha\beta\gamma) = k$ gives y = x + k. Again, by § 4.1.4, the case p = 2 is excluded. So

$$\gamma \mapsto \left(\begin{array}{cc} w & x \\ x+k & u-w \end{array}\right).$$

Now $\operatorname{tr}(\beta\gamma) = 0$ gives $\frac{1}{2}(2ew + 2fx + tk + fk + u - eu) = 0$ while $\operatorname{det}(\gamma) = 1$ gives $wu - w^2 - x^2 - xk = 1$. Rewriting $\operatorname{det}(\gamma) = 1$ as $4e^2 \operatorname{det}(\gamma) = 4e^2$, and letting 2ew = eu - u - fk - tk - 2fx, then x satisfies the quadratic equation

$$4x^{2} + \left(-4ftk - 4fu + 4k\right)x + \left(-2 + f^{2} - 2ft - 2ufk - 2utk\right) = 0 (4.15)$$

This quadratic has determinant $D = -16e^2(3 + 2utk)$, with $t, k \in \{\pm\sqrt{2}\}$. If $\operatorname{sign}(u) = \operatorname{sign}(t)$ and k = -1 or if $\operatorname{sign}(u) = -\operatorname{sign}(t)$ and k = 1 then $D = (4e)^2$. Let $\operatorname{sign}(u) = \operatorname{sign}(t)$ and k = -1. Then x satisfies the quadratic equation

$$4x^2 - 4x + 2 + f^2 = 0$$

and $D = (4e)^2$. Hence $x = \frac{4\pm 4e}{8} = \frac{1}{2}(1\pm e)$. If $x = \frac{1}{2}(1+e)$ then $w = \frac{1}{2}(t-f)$ and γ can be written as

$$\overline{\gamma_1} = \frac{1}{2} \left(\begin{array}{cc} t - f & 1 + e \\ e - 1 & t + f \end{array} \right) = \overline{\alpha}$$

while if $x = \frac{1}{2}(1 - e)$, then $w = \frac{1}{2}(f + t)$ and γ has the form

$$\overline{\gamma_2} = \frac{1}{2} \begin{pmatrix} f+t & 1-e\\ e-1 & t-f \end{pmatrix} = \overline{\beta}^{-1} \overline{\alpha} \overline{\beta}$$

so we recover the two epimorphisms $\Gamma^+ \to S_4$ constructed in the previous section. If $\operatorname{sign}(u) = -\operatorname{sign}(t)$ and k = 1 then the same result is obtained.

Now suppose that sign(u) = sign(t) and k = 1, so x satisfies the quadratic equation

$$4x^2 + (-8tf + 4)x - e^2 - 4tf - 7$$

This has discriminant $D = -7(4e)^2$. Write $D_i = -16e^2(3 + 2ukt_i)$. If $p \equiv 1, 2$ or 4 mod 7, then either both or neither D_i are squares in \mathbb{F}_p . Since one discriminant $D = (4e)^2$ is always a square, it follows that they both are. Hence for $p \equiv \pm 1 \mod 8$ 8 and $p \equiv 1, 2$ or 4 mod 7, there are two distinct normal subgroups whose factor group is $L_2(p)$. If, however, $p \equiv \pm 1 \mod 8$ and $p \equiv 3, 5$ or 6 mod 7, then precisely one discriminant, $D = (4e)^2$, is a square in \mathbb{F}_p . The second one, $D = -7(4e)^2$, is a square in the quadratic extension \mathbb{F}_{p^2} of \mathbb{F}_p , and so the induced map sends Γ^+ onto a subgroup of $L_2(p^2)$. Since Γ^+ has a subgroup of index 2, it is possible that the image is $PGL_2(p)$.

Elements of $PGL_2(p) \setminus L_2(p)$ have non-square determinant. Since it has been assumed that $p \equiv 3, 5$ or 6 mod 7, it follows from quadratic reciprocity that 7 is not a square in the field \mathbb{F}_p . To characterize maps $\Gamma^+ \to PGL_2(p)$, it is sufficient to consider the images $\overline{\gamma}$ of γ whose determinant is 7. The set of equations $\det(\overline{\gamma}) = 7$, $\operatorname{trace}(\alpha\beta\gamma) = \pm 1$, $\operatorname{trace}(\beta\gamma) = 0$ and $\operatorname{trace}(\overline{\gamma}) = \pm\sqrt{2}$ together yield the following quadratic equation for x:

$$4x^{2} + (-4fu - 4ftk + 4k)x + (-25e^{2} - 2ft - 3 - 2utk - 2fuk) = 0$$

This equation has discriminant $D = 16e^2(21 - 2utk)$. Since it is assumed that $\operatorname{sign}(u) = \operatorname{sign}(t)$ and that k = 1 (since otherwise epimorphisms onto S_4 are recovered), then $D = 17(4e)^2$. Then D is a square if and only if $p \equiv 1, 2, 4, 8, 9, 13, 15$ or 16 mod 17, and in this case two distinct normal subgroups with factor group $PGL_2(p)$ are recovered.

Hence, if $p \equiv \pm 1 \mod 8$, $p \equiv 3, 5$ or 6 mod 7 and $p \equiv 1, 2, 4, 8, 9, 13, 15$ or 16 mod 17, then two distinct there are two distinct normal subgroups whose factor group is $PGL_2(p)$. Otherwise there is a unique normal subgroup with factor group $L_2(p^2)$.

Now suppose that $p \equiv \pm 3 \mod 8$. Then $S_4 \not\subset L_2(p)$, but $S_4 \subset PGL_2(p) \subset L_2(p^2)$. Since $PGL_2(p)$ is maximal in $L_2(p^2)$ and S_4 is maximal in $PGL_2(p)$, there are two cases to consider: either Γ^+ has a normal subgroup whose factor group is $PGL_2(p)$, or it has a normal subgroup with factor group $L_2(p^2)$. By construction, the image of γ is a matrix with square determinant and so lies in $L_2(p)$ or in $L_2(p^2) \setminus PGL_2(p)$. If the image of γ lies in $L_2(p)$ then Γ^+ maps to $PGL_2(p)$, while if If the image of γ lies in $L_2(p^2) \setminus PGL_2(p)$.

Theorem 4.10 Let $\Gamma = T_6[2, 3, 4; 2, 3, 4]$ and let D be the discriminant of Equation 4.15. Let t be the trace of α , u be the trace of γ and k be the trace of $\alpha\beta\gamma$. Then

- Suppose that sign(t) = sign(u) and k = −1, or sign(t) = -sign(u) and k = 1. Then two distinct epimorphisms Γ⁺ → S₄ are recovered.
- 2) Suppose that sign(t) = sign(u) and k = 1, or sign(t) = -sign(u) and k = -1. If p ≡ 1, 2 or 4 mod 7 and p ≡ ±1 mod 8 then there are 4 distinct epimorphisms Γ⁺ → L₂(p). If p ≡ ±3 mod 8 or p ≡ 3, 5 or 6 mod 7 then there are 2 distinct epimorphisms Γ⁺ → L₂(p²).

4.7.3 Extending to $\Gamma \twoheadrightarrow L_2(q)$

Let $e^2 + f^2 = -1$, where $t = \sqrt{2}$ and $e \neq 0$, y = x + k where $k = \pm 1$ and z = u - wwhere $u = \pm \sqrt{2}$ be as in the previous section. Suppose that Γ^+ has a normal subgroup N with $\Gamma^+/N \cong L_2(q)$. From equation 4.2 in § 4.1.3 the condition

$$ex + ey + fz - fw = 0 (4.16)$$

is obtained. From the results obtained in the previous section, the two equations

$$2ew = eu - u - fk - tk - 2fx$$
 (4.17)

and

$$4x^{2} + (-4ftk - 4fu + 4k)x + (-2 + f^{2} - 2ft - 2ufk - 2utk) = 0 (4.18)$$

are obtained. Since y = x + k and z = u - w, equation 4.16 becomes $x = \frac{2fw - fu - ek}{2e}$. Substituting this into equation 4.17 gives $w = \frac{u + eu + tek}{2}$. Substituting in for x and w in equation 4.18 gives

$$(3+2tuk)f^2 + 3 + 2tuk = 0 (4.19)$$

implying that $f^2 = -1$ and consequently $e^2 = 0$, a contradiction, or 3 + 2tuk = 0. If 3 + 2tuk = 0, then $9 = 4t^2u^2k^2 = 16$ and so 7 = 0, so p = 7. In this case there is a unique normal subgroup K of Γ with factor group $L_2(7)$. Using GAP, it was shown that K is not torsion free - it contains the four conjugacy classes $\mathcal{C}((dab)^3)$, $\mathcal{C}((abc)^3)$, $\mathcal{C}((bcd)^3)$ and $\mathcal{C}((cda)^3)$ of elements of order 2.

Theorem 4.11 Suppose there exists a normal subgroup $N \triangleleft \Gamma^+$ with factor group $\Gamma^+/N \cong L_2(q)$. Then there can exist an extension \widetilde{N} of N with $\Gamma/\widetilde{N} \cong L_2(q)$ if and only if p = 7. The subgroup \widetilde{N} is not torsion free.

4.8
$$\Gamma = T_7[2, 3, 3; 2, 5, 3]$$

4.8.1 $\Gamma^+ \twoheadrightarrow L_2(q)$

As defined in section 4.1.1, let $v_1 = v_3 = 3$ and $v_2 = 2$, so $\gamma^3 = (\beta \gamma)^2 = (\alpha \beta \gamma)^3 = 1$. Then $|\gamma| = 3$ gives z = u - w, $u = \pm 1$, while $|\alpha \beta \gamma| = 3$ gives y = x + k, with $k = \pm 1$. So

$$\gamma \mapsto \left(\begin{array}{cc} w & x \\ x+k & u-w \end{array}\right) \in \overline{L},$$

and $|\beta\gamma| = 2$ gives 2ew + 2fx + tk + fk + u - eu = 0. det $(\gamma) = 1$ gives $wu - w^2 - x^2 - xk = 1$. Writing this as $4e^2 \det(\gamma) = 4e^2$, the following equality

$$(2ew)2u - (2ew)^2 - (2ex)^2 - (2ex)(2ek) = 4e^2$$
(4.20)

can be obtained. Substituting in for 2ew gives

$$-(8+4t)x^{2} + (4ftk + 4fu - 8k - 4tk)x + (-4 - 4t - 2f^{2} + 2ft + 2fuk + 2utk) = 0$$
(4.21)

a polynomial with discriminant $D = 16e^2(2 + 3t - 2ukt)$. Let $D_i = 16e_i^2(2 + 3t_i - 2kut_i)$, where $t_i = \frac{-1 + (-1)^i \sqrt{5}}{2}$ as usual. Since $k, u \in \{\pm 1\}$, the two cases k = u and k = -u are considered separately.

Case I k = u:

Then $D_i = 16e_i^2(2 + t_i)$ and $D_1D_2 = (16e_1e_2)^2$ and either both or neither D_i are squares in F, giving 8 maps $\Gamma^+ \mapsto L_2(q)$ or none, respectively. If q = p there are 8 distinct maps, while if $q = p^2$ then the 8 images form two conjugacy classes, so 4 distinct maps $\Gamma^+ \mapsto L_2(p^2)$ are recovered. If neither D_i is a root, then adjoining $\sqrt{D_1}$ to F gives maps $\Gamma^+ \mapsto L_2(F(\sqrt{D_1}))$. If $F = \mathbb{F}_p$, $K = \mathbb{F}_p(\sqrt{D_1})$ then the Galois automorphism $\phi \in \text{Gal}(K/F)$ swaps the two solutions $a + b\sqrt{D_1}$, $a - b\sqrt{D_1}$ so 4 distinct maps $\Gamma^+ \mapsto L_2(F(\sqrt{D_1}))$ are constructed. If $F = \mathbb{F}_{p^2}$ so $K = F(\sqrt{D_1})$, 2 epimorphisms $\Gamma^+ \mapsto L_2(F(\sqrt{D_1}))$ are constructed.

Case II k = -u:

Then $D_i = 16e_i^2(2 + 5t_i)$ and $D_1D_2 = -31(16e_1e_2)^2$, so both D_1 and D_2 are squares in F if both 5 and -31 are squares in F. Now -31 is a square in F if either $|F| = p^2$ or if $p \equiv 1, 2, 4, 5, 7, 8, 9, 10, 14, 16, 18, 19, 20, 25$ or 28 mod 31, and 5 is a square in F if $p \equiv \pm 1 \mod 5$ or $|F| = p^2$. First suppose that $p \equiv \pm 1 \mod 5$ and $p \equiv 1, 2, 4, 5, 7, 8, 9, 10, 14, 16, 18, 19, 20, 25$ or 28 mod 31. Then there are either zero or 8 epimorphisms $\Gamma^+ \mapsto L_2(p)$, depending on whether or not D_i are squares or not in F. If they are not, then adjoining, say, $\sqrt{D_1}$ to F gives pairs of conjugate maps $\Gamma^+ \mapsto L_2(F(\sqrt{D_1}))$, giving four distinct epimorphisms $\Gamma^+ \mapsto L_2(p^2)$. If, however, $|F| = p^2$, then if the D_i are squares in F four epimorphisms $\Gamma^+ \mapsto L_2(p^2)$ are constructed. If the D_i are not squares, then adjoining, say, $\sqrt{D_1}$ to F gives maps $\Gamma^+ \mapsto L_2(F(\sqrt{D_1}))$, giving two epimorphisms $\Gamma^+ \mapsto L_2(p^2)$.

Now suppose that -31 is not a square in F, so $p \equiv \pm 1 \mod 5$ and $p \equiv 3, 6, 11, 12, 13, 15, 17, 21, 22, 23, 24, 26, 27, 29, 30 \mod 31$. Then precisely one of D_1 , D_2 is a root in F (without loss of generality say D_1) and 4 distinct maps $\Gamma^+ \mapsto L_2(p)$ are recovered. Adjoining the square root $\sqrt{D_2} \notin F$ gives 4 epimorphisms $\Gamma^+ \mapsto L_2(p^2)$ forming two conjugacy classes of maps under the action of the Galois automorphism Gal(K/F), where $K = F(\sqrt{D_2})$

The following theorems have thus been proved:

Theorem 4.12 Let k = u. Then either both or neither of the discriminants D_i are squares in F. If they both are, then if $p \equiv \pm 1 \mod 5$ gives 8 distinct epimorphisms $\Gamma^+ \twoheadrightarrow L_2(p)$, while if $p \equiv \pm 2 \mod 5$ then 4 distinct epimorphisms $\Gamma^+ \twoheadrightarrow L_2(p^2)$ are obtained. If neither D_i are squares in F, then they both are in $F(\sqrt{D_i})$. If $p \equiv \pm 1$ mod 5 then there are 4 distinct epimorphisms $\Gamma^+ \twoheadrightarrow L_2(p^2)$, while if $p \equiv \pm 2 \mod 5$ there are 2 distinct epimorphisms $\Gamma^+ \twoheadrightarrow L_2(p^4)$.

Theorem 4.13 Let k = -u. Let $\Omega_1 = \{1, 2, 4, 5, 7, 8, 9, 10, 14, 16, 18, 19, 20, 25, 28\}$ be the squares in \mathbb{F}_{31} . Then the following cases hold:

- If p ≡ ±1 mod 5 and p ≡ i mod 31, where i ∈ Ω, then either both discriminants or neither lie in F. If they both do there are 8 epimorphisms Γ⁺ → L₂(p), while if they do not then there are 4 epimorphisms Γ⁺ → L₂(p²).
- If p ≡ ±1 mod 5 then either both discriminants or neither lie in F_{p²}. If they both do then there are 4 epimorphisms Γ⁺ → L₂(p), while if they do not then there are 2 epimorphisms Γ⁺ → L₂(p²).
- 3) If p ≡ ±1 mod 5 and p ≡ i mod 31, where i ∈ F₃₁\Ω. Then precisely one discriminant is a square in F_p and there are 4 epimorphisms Γ⁺ → L₂(p). Adjoining the square root of the other discriminant gives 2 epimorphisms Γ⁺ → L₂(p²).

4.8.2 Case p = 2

As specified in § 3.5 α and β can be written as

$$\alpha \mapsto \begin{pmatrix} t & t+1 \\ t & 0 \end{pmatrix}, \quad \beta \mapsto \begin{pmatrix} t+1 & 0 \\ t & t \end{pmatrix}, \quad \text{so} \quad \alpha\beta \mapsto \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}.$$

where $t \in \mathbb{F}_{p^2}/\mathbb{F}_p$ is a primitive cube root of 1. Writing

$$\gamma \mapsto \left(\begin{array}{cc} w & x \\ y & z \end{array}\right)$$

and noting that $|\gamma| = 3$ and $|\alpha\beta\gamma| = 3$ gives

$$\gamma \mapsto \left(\begin{array}{cc} w & x\\ 1+x & 1+w \end{array}\right)$$

while $|\beta\gamma| = 2$ gives t+w+tx = 0, so w = tx+t. Now det $(\gamma) = (w+1)^2+(x+1)^2 = 1$ and so Equation 4.22 is recovered:

$$tx + t + t^{2}x^{2} + t^{2} + x + x^{2} = 1$$
(4.22)

Since $t^2 = 1 + t$ Equation 4.22 can be rewritten as $tx + tx^2 + x = 0$, so x = 0 or $tx = t + 1 = t^2$ so x = t.

$$x = 0 \implies \gamma \mapsto \begin{pmatrix} t & 0 \\ 1 & 1+t \end{pmatrix} \qquad x = t \implies \gamma \mapsto \begin{pmatrix} 1 & t \\ 1+t & 0 \end{pmatrix} \quad (4.23)$$

Consequently, there is no map $\Gamma^+ \twoheadrightarrow L_2(2^n)$ for n > 2.

Theorem 4.14 There exists no epimorphism $\Gamma^+ \twoheadrightarrow L_2(2^n)$ for n > 2. For n = 2, there are two distinct epimorphisms.

4.8.3 Extending to $\Gamma \twoheadrightarrow L_2(q)$

Let $e^2 + f^2 = -2 - t$, where $t = \frac{-1 \pm \sqrt{5}}{2}$ and $e \neq 0$, y = x + k where $k = \pm 1$ and z = u - w where $u = \pm 1$ be as in the previous section, and p > 2. Suppose that Γ^+ has a normal subgroup N with $\Gamma^+/N \cong L_2(q)$. From equation 4.2 in § 4.1.3 the condition

$$ex + ey + \int z - \int w = 0 \tag{4.24}$$

is obtained. From the results obtained in the previous section, the two equations

$$2ew = eu - u - fk - tk - 2fx (4.25)$$

and

$$-(8+4t)x^{2} + (4ftk + 4fu - 8k - 4tk)x + (-4 - 4t - 2f^{2} + 2ft + 2fuk + 2utk) = 0$$
(4.26)

are obtained. Since y = x + k and z = u - w, equation 4.24 becomes $x = \frac{2fw - fu - ek}{2e}$. Substituting this into equation 4.25 gives $w = \frac{(2+t)u + eu + tek}{2(2+t)}$. Substituting in for x and w in equation 4.26 gives

$$(-2 - 3t + 2utk)f^{2} - 7 - 5t + 2uk + 2utk = 0$$
(4.27)

so $f = \frac{7+5t-2uk-2utk}{-2-3t+2utk}$. It now remains to consider the two cases, sign(u) = sign(k) and -sign(u) = sign(k).

Sign(u) =Sign(k): Write u = k. Then $f^2 = -\frac{5+3t}{2+t}$ and hence $e^2 = -2-t-f^2 = 0$, contradicting the assumption that $e \neq 0$, unless $f^2 = 0$. Now $f^2 = 0$ if and only if $5 + 3t = 5 + 3\frac{-1\pm\sqrt{5}}{2} = 0$. Solving 5 + 3t = 0 gives 4 = 0. So for all primes p > 2, if sign(u) = sign(k) then there is no normal subgroup \widetilde{N} of Γ with factor group $L_2(q)$.

Sign(u) = -Sign(k): Write u = -k. Then $f^2 = -\frac{9+7t}{2+5t}$ and hence $e^2 = 0$, contradicting the assumption that $e \neq 0$, unless $f^2 = 0$. Now $f^2 = 0$ if and only if 9+7t = 0. Solving this gives 124 = 0 so p = 2 (a contradiction) or p = 31. If p = 31 there is a unique normal subgroup K of Γ whose factor group $\Gamma/K \cong L_2(31)$. Since Γ contains conjugacy classes $\mathcal{C}(abc)$, $\mathcal{C}((abc)^3)$, $\mathcal{C}(dab)$ and $\mathcal{C}((dab)^3)$ of elements of order 10 and since $L_2(31)$ has no elements of order 10, K cannot be torsion free.

Theorem 4.15 Suppose there exists a normal subgroup $N \triangleleft \Gamma^+$ with factor group $\Gamma^+/N \cong L_2(q)$. Then there exists an extension \widetilde{N} of N with $\Gamma/\widetilde{N} \cong L_2(q)$ if and only if p = 31. In this case there is a unique normal subgroup \widetilde{N} in Γ with $\Gamma/\widetilde{N} \cong L_2(31)$ and \widetilde{N} is not torsion free.
4.9 $\Gamma = T_8[2, 4, 3; 2, 5, 3]$

4.9.1 $\Gamma^+ \twoheadrightarrow L_2(q)$

Write α and β as they are defined in section § 4.1. Then $e^2 + f^2 + 2 + t = 0$ and $t = \frac{-1 \pm \sqrt{5}}{2}$. Let

$$\gamma \mapsto \left(\begin{array}{cc} w & x \\ y & z \end{array}\right)$$

Then $|\gamma| = 4$, $|\beta\gamma| = 2$ and $|\alpha\beta\gamma| = 3$ become the conditions $\operatorname{tr}(\gamma) = \pm\sqrt{2} = u$, $\operatorname{tr}(\beta\gamma) = 0$ and $\operatorname{tr}(\alpha\beta\gamma) = \pm 1 = k$, respectively. $\operatorname{tr}(\gamma) = u$ forces z = u - w and $\operatorname{tr}(\alpha\beta\gamma) = k$ gives y = x + k. Again we cannot have p = 2. So

$$\gamma \mapsto \left(\begin{array}{cc} w & x \\ x+k & u-w \end{array}\right)$$

Now $\operatorname{tr}(\beta\gamma) = 0$ gives $\frac{1}{2}(2ew + 2fx + tk + fk + u - eu) = 0$ while $\operatorname{det}(\gamma) = 1$ gives $wu - w^2 - x^2 - xk = 1$. Rewriting $\operatorname{det}(\gamma) = 1$ as $4e^2\operatorname{det}(\gamma) = 4e^2$, and letting 2ew = eu - u - fk - tk - 2fx gives the following quadratic expression for x:

$$p(x)_{t_i,u_j,k_s} = (8+4t)x^2 + (-4fu - 4tfk + 4tk + 8k)x + (1+3t+f^2 - 2tku - 2tf - 2fku) = 0.$$
(4.28)

This has discriminant $D = 16[(1 - 2t + 2ukt)f^2 + (-t + 2ku + 2ktu)]$. Write $e^2 = -2 - t - f^2$ in D. Simplifying the resulting expression gives

$$D = 16e^2(2t - 1 - 2ktu).$$

Let $D_j = 16e^2[(2t-1) - 2ktu_j]$. Then $D_1D_2 = -3(16e^2)^2$ and this is a square in F if and only if -3 is a square in F. (note that this number still depends on t, since t is expressed in terms of e and f). By Gauss' lemma for quadratic residues, $\sqrt{-3} \in \mathbb{F}_p$ if and only if $p \equiv 1 \mod 6$.

First suppose that both D_1 and D_2 are squares in F. Then either $p \equiv 1 \mod 6$ or $|F| = p^2$.

• If $p \equiv \pm 1 \mod 5$, $p \equiv \pm 1 \mod 8$ and $p \equiv 1 \mod 6$, then there are 16 distinct epimorphisms $\Gamma^+ \twoheadrightarrow L_2(p)$.

- If $p \equiv \pm 1 \mod 5$ and $p \equiv \pm 3 \mod 8$, then there are 8 distinct epimorphisms $\Gamma^+ \twoheadrightarrow L_2(p^2)$.
- If $p \equiv \pm 2 \mod 5$ and $p \equiv \pm 1 \mod 8$, then there are 8 distinct epimorphisms $\Gamma^+ \twoheadrightarrow L_2(p^2)$.
- If $p \equiv \pm 2 \mod 5$ and $p \equiv \pm 3 \mod 8$, then there are 8 distinct epimorphisms $\Gamma^+ \twoheadrightarrow L_2(p^2)$.

Now suppose that neither D_1 nor D_2 are squares in F. Then they both are in $F(D_1)$ and the nontrivial Galois automorphism of $F(D_i)/F$ swaps $\sqrt{D_j}$ with $-\sqrt{D_j}$, so the Galois automorphism swaps the two solutions to the polynomial $p(x)_{t_i,u_j,k_s}$, for each i, s in $\{0, 1\}$. Then one of the following cases holds:

- If p ≡ ±1 mod 5, p ≡ ±1 mod 8 and p ≡ 1 mod 6, then there are 8 distinct epimorphisms Γ⁺ → L₂(p²)
- If $p \equiv 1 \mod 6$, and suppose one of the following cases holds:

 $p \equiv \pm 1 \mod 5$ and $p \equiv \pm 3 \mod 8$, or $p \equiv \pm 2 \mod 5$ and $p \equiv \pm 1 \mod 8$, or $p \equiv \pm 2 \mod 5$ and $p \equiv \pm 3 \mod 8$

Then there are 4 distinct epimorphisms $\Gamma^+ \rightarrow L_2(p^2)$

Finally, suppose that precisely one of D_1 , D_2 is a square in F. Then $p \equiv \pm 1 \mod 5$, $p \equiv \pm 1 \mod 8$ and $p \not\equiv 1 \mod 6$. Without loss of generality suppose that D_1 is a square in $F = \mathbb{F}_p$. Then there are precisely 8 epimorphisms $\Gamma^+ \to L_2(p)$. Adjoining $\sqrt{D_2}$ to \mathbb{F} to get $F = \mathbb{F}(\sqrt{D_2})$ gives 4 distinct epimorphisms, one for each pair t_i, k_s , for $i, s \in \{0, 1\}$.

Theorem 4.16 Let $p(x)_{i,j,s}$ be as above and let $D_j = 16e^2(2t - 1 - 2ktu_j)$.

Suppose that |F| = p. Then for p ≡ ±1 mod 5, p ≡ ±1 mod 8 and p ≡ 1 mod 6, either both or neither D_j are squares in F. If they both are then for each triple (i, j, s) two solutions are obtained for Equation 4.28 giving a total of 16 epimorphisms Γ⁺ → L₂(p). If neither D_j is a square in F, then they both are in F(√D₁) and there are 8 distinct epimorphisms Γ⁺ → L₂(p²).

- 2) Suppose that |F| = p². Then either both or neither D_j are squares in F. If they both are then for each triple (i, j, s) there is one solution for Equation 4.28 giving a total of 8 epimorphisms Γ⁺ → L₂(p²). If neither D_j is a square in F, then they both are in F(√D₁) and there are 4 distinct epimorphisms Γ⁺ → L₂(p⁴).
- 3) Finally, suppose p ≡ ±1 mod 5, p ≡ ±1 mod 8 and p ≠ 1 mod 6. Then precisely one of D₁, D₂ is a square in F. In this case there are 8 epimorphisms Γ⁺ → L₂(p). Adjoining √D to F, where D is the non-square discriminant gives 4 epimorphisms Γ⁺ → L₂(p²).

4.9.2 Extending to $\Gamma \twoheadrightarrow L_2(q)$

Let $e^2 + f^2 = -2 - l$, where $l = \frac{-1 \pm \sqrt{5}}{2}$ and $e \neq 0$, y = x + k where $k = \pm 1$ and z = u - w where $u = \pm \sqrt{2}$ be as in the previous section. Suppose that Γ^+ has a normal subgroup N with $\Gamma^+/N \cong L_2(q)$. From equation 4.2 in § 4.1.3 the condition

$$ex + ey + fz - fw = 0 (4.29)$$

is obtained. From the results obtained in the previous section, the two equations

$$2ew = eu - u - fk - tk - 2fx (4.30)$$

and

$$(8+4t)x^{2} + (-4fu - 4tfk + 4tk + 8k)x + (1+3t+f^{2} - 2tku - 2tf - 2fku) = 0$$
(4.31)

are obtained. Since y = x + k and z = u - w, equation 4.29 becomes $x = \frac{2fw - fu - ek}{2e}$. Substituting this into equation 4.30 gives $w = \frac{(2+t)u + eu + tek}{2(2+t)}$. Substituting in for x and w in equation 4.31 gives

$$(-1+2t-2tuk)f^{2} - 2uk - 2tuk + t = 0 (4.32)$$

so $f^2 = \frac{2uk+2tuk-t}{-1+2t-2tuk}$. Then $e^2 = -2 - t - f^2 = 0$, contradicting the assumption that $e \neq 0$, unless $f^2 = 0$. Now $f^2 = 0$ if and only if 2uk + 2tuk - t = 0. Write

t = 2uk(1+t). Then $t^2 = 4u^2k^2(1+t)^2$ so 1-t = 16+8t, giving 17 = -9t. Solving this gives 44 = 0, so p = 2 or p = 11. By the previous section $p \neq 2$ and since $11 \equiv 3 \mod 8$, $L_2(11)$ contains no element of order 4. Hence there is no torsion free normal subgroup N of Γ^+ whose factor group is $L_2(11)$.

Theorem 4.17 Suppose there exists a normal subgroup $N \triangleleft \Gamma^+$ with factor group $\Gamma^+/N \cong L_2(q)$. Then there exist no extensions \widetilde{N} of N with $\Gamma/\widetilde{N} \cong L_2(q)$.

4.10 $\Gamma = T_9[2, 3, 5; 2, 3, 5]$

4.10.1 $\Gamma^+ \twoheadrightarrow L_2(q)$

Let α , β and γ be as defined in § 4.1.1. Then $|\gamma| = 5$ and $|\alpha\beta\gamma| = 3$ give y = x + kand z = u - w, where $k = \pm 1$ and $u = \frac{-1 \pm \sqrt{5}}{2}$. $|\beta\gamma| = 2$ gives 2ew + 2fx + tk + fk + u - eu = 0. From $\det(\gamma) = 1$, $wu - w^2 - x^2 - xk = 1$. Multiplying this by $4e^2$ gives

$$(2ew)(2eu) - (2ew)^2 - 4e^2x^2 - 4e^2xk - 4e^2 = 0$$
(4.33)

Substituting for 2ew = eu - u - fk - tk - 2fx in Equation 4.33 and simplifying gives the quadratic equation

$$(8+4t)x^{2} + (-4ftk - 4fu + 8k + 4tk)x + 4+3u + 4t + tu + 2f^{2} + uf^{2} - 2fuk - 2tuk - 2ft = 0$$

$$(4.34)$$

which has the rather unwieldy discriminant $D = 16e^2(2 + 3t + tu - 2tuk + 3u)$. There are two cases: either both α and γ have the same trace, in which case u = t, or they have different traces, in which case u = -1 - t.

Case I u = t:

Then Equation 4.34 becomes

$$\frac{(8+4t)x^2 + (-4ftk - 4ft + 8k + 4tk)x +}{-2ft + f^2t + 5 + 6t + 2f^2 - 2kft - 2k + 2kt = 0}$$
(4.35)

and this has discriminant

$$D = 16e^2 \left((5+2k)t + 3 - 2k \right)$$

Writing $D_i = 16e_i^2((5+2k)t_i + 3 - 2k)$ gives $D_1D_2 = (16e_1e_2)^2(-27 - 28k)$, so

$$D_1 D_2 = \begin{cases} -55(16e_1e_2)^2, \text{ if } k = 1\\ (16e_1e_2)^2, \text{ if } k = -1 \end{cases}$$

Notice that, if k = -1, then Equation 4.36 is recovered

$$(4t+8)x^2 - (4l+8)x + (7+4l+2f^2+f^2t) = 0$$
(4.36)

This can be written as

$$(4t+8)x^{2} - (4t+8)x + \frac{1}{4}(4t+8)(3+t+f^{2}) = 0$$
(4.37)

Substituting in for e^2 in Equation 4.37 and dividing by (4t + 8) gives

$$x^{2} - x + \frac{1}{4}(1 - e^{2}) = 0$$
(4.38)

a quadratic with discriminant $D = e^2$. Hence $x = \frac{1\pm e}{4}$ and so, for each value of $t \in \{t_i\}_{i=1}^2$ there are two solutions, since $e \neq 0$. As a result, if k = -1 there are 4 epimorphisms $\Gamma^+ \twoheadrightarrow L_2(p)$, if $p \equiv \pm 1 \mod 5$, or 2 epimorphisms $\Gamma^+ \twoheadrightarrow L_2(p^2)$, if $p \equiv \pm 2 \mod 5$. If k = 1 then for -55 a square in F, either both or neither D_1 and D_2 are squares in F. If they both are and if $p \equiv \pm 1 \mod 5$ and $p \equiv 1, 3, 4, 5, 9 \mod 11$, then 4 distinct epimorphisms $\Gamma^+ \twoheadrightarrow L_2(p)$ are constructed. If $p \equiv \pm 2 \mod 5$ or $p \equiv 2, 6, 7, 8, 10 \mod 11$, then there are 2 distinct epimorphisms $\Gamma^+ \twoheadrightarrow L_2(p^2)$. If neither discriminant is a square in F, then if $p \equiv \pm 1 \mod 5$ and $p \equiv 1, 3, 4, 5, 9 \mod 11$, then there are 2 distinct epimorphisms $\Gamma^+ \twoheadrightarrow L_2(p^2)$. If neither discriminant is a square in F, then if $p \equiv \pm 1 \mod 5$ and $p \equiv 1, 3, 4, 5, 9 \mod 5$ or $p \equiv 2, 6, 7, 8, 10 \mod 11$ then there is a unique epimorphism $\Gamma^+ \twoheadrightarrow L_2(p^4)$. If $p \equiv \pm 1 \mod 5$ and $p \equiv 2, 6, 7, 8, 10 \mod 11$ then there is a unique epimorphism $\Gamma^+ \twoheadrightarrow L_2(p^4)$. If $p \equiv \pm 1 \mod 5$ and $p \equiv 2, 6, 7, 8, 10 \mod 11$, then there is not. In this case $p \equiv \pm 1 \mod 5$, $p \equiv 2, 6, 7, 8, 10 \mod 11$ and there are 2 distinct epimorphisms $\Gamma^+ \twoheadrightarrow L_2(p)$. Adjoining the square root of the non-square discriminant to F gives a unique epimorphism $\Gamma^+ \twoheadrightarrow L_2(p^2)$.

Theorem 4.18 Let $tr(\alpha) = tr(\gamma)$.

 Suppose that tr(αβγ) = -1. Then if p ≡ ±1 mod 5 there are 4 distinct epimorphisms Γ⁺ → L₂(p). If p ≡ ±2 mod 5 then there are 2 distinct epimorphisms Γ⁺ → L₂(p²). 2) Suppose that $tr(\alpha\beta\gamma) = 1$. Then if -55 is a square in \mathbb{F}_p either either both or neither D_1 , D_2 are squares in F. If they both are squares in F, then

If $p \equiv \pm 1 \mod 5$ and $p \equiv 1, 3, 4, 5, 9 \mod 11$ then there are 4 distinct epimorphisms $\Gamma^+ \rightarrow L_2(p)$.

If $p \equiv \pm 2 \mod 5$ or $p \equiv 2, 6, 7, 8, 10 \mod 11$, then there are 2 distinct epimorphisms $\Gamma^+ \rightarrow L_2(p^2)$.

If neither are squares in F, then

If $p \equiv \pm 1 \mod 5$ and $p \equiv 1, 3, 4, 5, 9 \mod 11$ then there are 2 distinct epimorphisms $\Gamma^+ \rightarrow L_2(p^2)$.

If $p \equiv \pm 2 \mod 5$ or $p \equiv 1, 3, 4, 5, 9 \mod 11$ then there is a unique epimorphism $\Gamma^+ \twoheadrightarrow L_2(p^4)$

3) If p ≡ ±1 mod 5 and p ≡ 2,6,7,8,10 mod 11 then precisely one of D₁, D₂ is a square in F. In this case there are 2 distinct epimorphisms Γ⁺ → L₂(p). Adjoining the square root of the non-square discriminant to F, gives a unique epimorphism Γ⁺ → L₂(p²).

Case II u = -1 - t:

Then 4.34 becomes

$$(8+4t)x^{2} + (-4kft + 4ft + 4f + 8k + 4tk)x + 2k - 2ft + 2fk + t + f^{2} - f^{2}t + 2kft = 0$$
(4.39)

and this has discriminant $D = 32e^2(k-1)$ and so

$$D = \begin{cases} 0, \text{ if } k = 1\\ -(8e)^2, \text{ if } k = -1 \end{cases}$$

So if k = 1 there is a unique epimorphism. Now, letting k = -1, if $p \equiv \pm 1 \mod 5$ then there are 2 distinct epimorphisms onto $L_2(p)$ if $p \equiv 1 \mod 4$. Otherwise there is a unique epimorphism $\Gamma^+ \twoheadrightarrow L_2(p^2)$.

Theorem 4.19 Suppose that $tr(\alpha) \neq tr(\gamma)$. Then

- 1) Suppose that $tr(\alpha\beta\gamma) = 1$. Then if $p \equiv \pm 1 \mod 5$ there are two epimorphisms $\Gamma^+ \twoheadrightarrow L_2(p)$. If $p \equiv \pm 2 \mod 5$ there is a unique epimorphism $\Gamma^+ \twoheadrightarrow L_2(p^2)$.
- 2) Suppose that $tr(\alpha\beta\gamma) = -1$. If $p \equiv \pm 1 \mod 5$ and $p \equiv 1 \mod 4$ then there are 2 epimorphisms $\Gamma^+ \to L_2(p)$. Otherwise there is a unique epimorphism onto $L_2(p^2)$.

4.10.2 Exceptional cases

In the previous section, the discriminant $D = 16e^2((5+2k)t+3-2k)$ if $u_i = t_i$, while $D = 32e^2(k-1)$ if $u_j = t_i$. In the case u = t D = 0 if (5+2k)t+3-2k = 0. If k = 1 then solving this gives $220 \equiv 0 \mod p$, or $p \in \{2, 5, 11\}$. If k = -1 then D = 0 if and only if p = 2. If $u_j = t_i$ then D = 0 if and only if k = 1, giving one unique solution for each value of t_i .

4.10.3 Extending to $\Gamma \twoheadrightarrow L_2(q)$

Let $e^2 + f^2 = -2 - t$, where $t = \frac{-1 \pm \sqrt{5}}{2}$ and $e \neq 0$, y = x + k where $k = \pm 1$ and z = u - w where $u = \frac{-1 \pm \sqrt{5}}{2}$ be as in the previous section. Suppose that Γ^+ has a normal subgroup N with $\Gamma^+/N \cong L_2(q)$. From equation 4.2 in § 4.1.3 the condition

$$ex + ey + fz - fw = 0 \tag{4.40}$$

is obtained. From the results obtained in the previous section, the two equations

$$2ew = eu - u - fk - tk - 2fx \tag{4.41}$$

and

$$(8+4t)x^{2} + (-4ftk - 4fu + 8k + 4tk)x + 4 + 3u + 4t + tu + 2f^{2} + uf^{2} - 2fuk - 2tuk - 2ft = 0$$
(4.42)

are obtained. Since y = x + k and z = u - w, equation 4.40 becomes $x = \frac{2fw - fu - ek}{2e}$. Substituting this into equation 4.41 gives $w = \frac{(2+t)u + eu + tek}{2(2+t)}$. Substituting in for x and w in equation 4.42 gives

$$(2t - 2tuk + u)f^{2} - 2uk - 2tuk + t = 0$$
(4.43)

so $f^2 = \frac{2uk+2tuk-t}{2t-2tuk+u}$ and $e^2 = -2 - t - f^2 = \frac{t+2u+2+ut}{2t-2tuk+u}$.

Case I: $u_i = t_i, i \in \{1, 2\}$

Then $f^2 = \frac{2k-t}{3t+2kt-2k}$ and $e^2 = \frac{-(2t+3)}{3t+2kt-2k}$. If k = 1 then $e^2 = \frac{2t+3}{2-5t}$ and $f^2 = \frac{1+2t}{2t-3}$ while if k = -1 then $e^2 = -1 - t$ and $f^2 = -1$.

Case II: $u_j = l_i, i \neq j$

Then $f^2 = \frac{4k+t+2kt}{t+2k-1}$ and $e^2 = \frac{1+t}{t+2k-1}$. If k = 1 then $e^2 = 1$ and $f^2 = -3 - t$ while if k = -1 then $e^2 = \frac{1+t}{t-3}$ and $f^2 = \frac{t+4}{t-3}$.

Theorem 4.20 Suppose there exists a normal subgroup $N \triangleleft \Gamma^+$ with factor group $\Gamma^+/N \cong L_2(q)$. Then there exists an extension \widetilde{N} of N with $\Gamma/\widetilde{N} \cong L_2(q)$ if and only if the following conditions hold:

	u = t		u = -1 - t	
	k = 1	k = -1	k = 1	k = -1
e^2	$\frac{2t+3}{2-5t}$	-1 - t	1	$\frac{t+1}{t-3}$
f^2	$\frac{1+2t}{2t-3}$	-1	-t - 3	$\frac{t+4}{t-3}$

4.11 Conclusion

In this chapter, all torsion free normal subgroups N of the orientation preserving subgroups Γ^+ of the Lannér groups Γ , whose factor group is a simple group $L_2(q)$, are classified. These subgroups are constructed by classifying the equivalence classes of epimorphisms $\Gamma^+ \to L_2(q)$. For each such subgroup $N \triangleleft \Gamma^+$, necessary and sufficient conditions under which the result extends to torsion free normal subgroups \tilde{N} of Γ are investigated. Chapter 5

Manifolds from the [5, 3, 5] group

5.1 Introduction

In this chapter, various manifolds arising from the action of torsion-free subgroups of the [5,3,5] Coxeter group will be studied. Several invariants (homology, fundamental group, combinatorial structure) of the manifolds will be constructed to distinguish between them.

5.2 The quotient manifolds M_i .

Recall that in Chapter 3 § 3.1 a tetrahedron T in \mathbb{H}^3 with vertices A, B, C, D, and with dihedral angles $\pi/5$, $\pi/3$, $\pi/5$, $\pi/2$, $\pi/2$ and $\pi/2$ along its edges CD, AD, AB, BD, BC, AC was introduced. The natural action of Γ on this tetrahedron equipped \mathbb{H}^3 with a tessellation \mathcal{T} by tetrahedra. Recall further the subgroup $\Gamma_0 = \langle a, b, c \rangle$ of Γ . This group is the isometry group of a dodecahedron, isomorphic to $A_5 \times C_2$. The 120 images of T under Γ_0 form a hyperbolic dodecahedron \mathbb{D} , with dihedral angles $2\pi/5$, and the images of \mathbb{D} under Γ form a dodecahedral tessellation \mathcal{D} of \mathbb{H}^3 .

A flag $\phi = (v, e, f, c)$ of \mathcal{D} , and its associated tetrahedron T in \mathcal{T} , is rightor left-handed if moving away from the vertex v along the edge e while rotating around e from the face f into the interior of the cell c represents a right- or lefthanded screw motion, that is, a clockwise or anticlockwise rotation when viewed from v. Identify a right-handed flag with $1 \in \Gamma$. Then the right- and left-handed flags ϕ_g and tetrahedra T_g correspond to the even and odd elements g of Γ , forming the two cosets Γ^+ and $\Gamma \setminus \Gamma^+$ of Γ^+ in Γ .

In Lemma 3.2.1 in Chapter 3 it was shown that there are two distinct normal subgroups N_1 and N_2 of Γ^+ whose factor group is A_5 .

Theorem 5.1 The quotient manifolds $M_i = \mathbf{H}^3/N_i$ (i = 1, 2) are a chiral pair of Weber-Seifert spaces, formed by identifying opposite faces of a hyperbolic dodecahedron after rotating them through $3\pi/5$ in the positive and negative senses respectively.

Proof: Since Γ_0^+ is mapped isomorphically onto A_5 by each θ_i , we have $\Gamma^+ =$ $N_i\Gamma_0^+ = \Gamma_0^+ N_i$ with $\Gamma_0^+ \cap N_i = 1$, so $\Gamma = N_i\Gamma_0 = \Gamma_0 N_i$ with $\Gamma_0 \cap N_i = 1$. Since the dodecahedron D_1 consists of the 120 tetrahedra in \mathcal{T} forming an orbit of Γ_0 , this shows that D_1 is a fundamental region for the action of each N_i on \mathbb{H}^3 . Each M_i can be formed from D_1 by identifying pairs of boundary points which are equivalent under N_i . Elements of N_i can be regarded as automorphisms, not monodromy permutations. As a result flags ϕ_g and ϕ_h in \mathcal{D} are identified if and only if $N_i g = N_i h$, a condition which implies that g and h are both even or both odd. If g and h are even then they normalise N_i , so $N_i g = g N_i$ and $N_i h = h N_i$, which means that we may equivalently use the condition $gN_i = hN_i$ (which identifies the right-handed flags ϕ_g and ϕ_h under the monodromy action). Using this action has the advantage that if g is a word $g_1 \ldots g_n$ in the generators $g_i \in \{a, b, c, d\}$ of Γ , then successive images $\phi, \phi g_1, \ldots, \phi g_1 \ldots g_n = \phi g$ of a flag ϕ correspond to adjacent tetrahedra in \mathcal{T} ; this makes it easier to determine the action of g as a monodromy permutation than as an automorphism, where successive images could be increasingly far apart. However, if g and h are odd, then $N_i g = g N_i^g =$ $gN_{i'}$, where i' = 3 - i, and similarly for N_ih . Therefore, in constructing M_i the monodromy action of this conjugate subgroup $N_{i'}$, rather than N_i , needs to be used to identify left-handed flags ϕ_g and ϕ_h . Fortunately, in order to determine the identifications of faces which result in M_i , it is sufficient to restrict attention to right-handed flags ϕ_g and ϕ_h , which are identified if and only if g and h have

the same image under θ_i .

The element $(abc)^5$ of Γ represents the antipodal involution in Γ_0 , so as a monodromy permutation it sends the flag ϕ_1 of D_1 to its antipodal flag $-\phi_1$. It follows that the element $g = (abc)^5 d = (\alpha \beta^{-1} \alpha^{-1} \beta)^2 \alpha \gamma$ of Γ^+ sends ϕ_1 to the flag $\phi_g = (-\phi_1)d$ in the adjacent dodecahedron, sharing the same vertex, edge and face as $-\phi_1$. Under θ_1 and θ_2 this element g has the same image in A_5 as $h = \alpha^{-1}$ and α respectively, so the flags ϕ_g and ϕ_h are identified in M_1 and M_2 . Now α^{-1} and α rotate right-handed flags through $2\pi/5$ around their faces in the positive and negative directions (when viewing D_1 from outside). Hence, in M_1 and M_2 , the flag ϕ_g is obtained from ϕ_h by applying a screw motion with a rotation in the positive sense through $\pi - 2\pi/5 = 3\pi/5$ or through $\pi + 2\pi/5 = 7\pi/5 \equiv -3\pi/5 \mod (2\pi)$ respectively. Figures 5.1 shows a dodecahedron with one pair of antipodal faces identified by a $+\frac{3\pi}{5}$ identification. Since Γ_0^+ acts transitively (by automorphisms) on the right-handed flags of D_1 , the same identification applies to every such flag ϕ_h . Consequently, each face of D_1 is identified with its opposite face after a rotation through $3\pi/5$ or $-3\pi/5$. This is the rule for constructing the two Weber-Seifert spaces.

The subgroups N_1 and N_2 are conjugate in Ω , so the manifolds M_1 and M_2 are isometric. However, the only conjugating elements are orientation-reversing elements of $\Omega \setminus \Omega^+$, hence each M_i is the mirror-image of the other.

Note that in the above proof, the monodromy permutations α^{-1} and α rotate left-handed flags in the opposite direction to right-handed flags; however, the earlier remarks imply that in constructing each M_i the roles of θ_1 and θ_2 must be transposed when considering left-handed flags, so all flags are rotated through the same angle.

Each manifold M_i has isometry group Iso $M_i \cong N_{\Omega}(N_i)/N_i = \Omega^+/N_i \cong S_5$, consisting entirely of orientation-preserving elements. The subgroup $\Gamma^+/N_i \cong A_5$ of index 2 preserves the tessellation \mathcal{D}/N_i , acting as the rotation group of the single dodecahedral cell, while r sends \mathcal{D}/N_i to its dual tessellation $\mathcal{D}^*/N_i \cong \mathcal{D}/N_i$. It can be verified, either by hand or by using GAP [GAP], that N_1 and N_2 are the normal closures in Γ^+ of the elements $(abcd)^t = (\alpha \gamma)^t$ for t = 2 and 3. Hence \mathcal{D}/N_1 and \mathcal{D}/N_2 are examples $\{5, 3, 5\}_2$ and $\{5, 3, 5\}_3$ of Coxeter's twisted honeycombs



Figure 5.1: Identification of two faces of a dodecahedron

 $\{p,q,r\}_t$ ([C1], pp. 32–33 and Fig. 23).

The tessellation \mathcal{D}/N_1 consists of one vertex, six edges e_i (all loops), six pentagonal faces, and one dodecahedral cell. Figure 5.2 illustrates the edges in D_1 . Taking the single vertex as base-point, the six loops corresponding to the e_i can be treated as generators of the fundamental group $\pi_1(M_1) \cong N_1$. The abelianisation N_1/N'_1 , where N'_1 is the commutator subgroup of N_1 , is the first integer homology group $H_1(M_1)$. Let f_1 be the face of D_1 incident with ϕ_1 , and let the edges of f_1 , cyclically ordered and directed in the positive orientation, be e_1, \ldots, e_5 . The sixth edge e_6 appears in the 'equatorial' Petrie polygon of D_1 , disjoint from f_1 and its antipodal face $-f_1$: after identifications of faces, this polygon becomes the dotted path $e_1, e_6, e_2, e_6, e_3, e_6, e_4, e_6, e_5, e_6$, as shown in Figure 5.2. (The Petrie polygons of a dodecahedron have length 10, and each edge is identified with four others in the right-left Petrie polygon containing it).

The face f_1 gives rise to the homology relation $e_1 + \cdots + e_5 = 0$. Its neighbour across e_1 is the face adjacent to f_1 containing the edge e_1 . Reading the edges of this second face in the positive orientation gives $-e_1 - e_2 + e_5 + e_6 + e_3 = 0$. The other four neighbours of f_1 across the edges e_2 , e_3 , e_4 and e_5 give relations formed from (1) by cyclically permuting e_1, \ldots, e_5 ; the relations obtained from the remaining six faces of D_1 are simply the negatives of the relations constructed, so



Figure 5.2: $+\frac{3\pi}{5}$ identification with Petrie polygon marked (dashed line)

they can be ignored. Eliminating e_4 (= $e_1 + 2e_2 + 3e_3$), e_5 (= $3e_1 + 2e_2 + e_3$) and e_6 (= $3e_1 - e_2 + 3e_3$) gives $H_1(M_1)$ generated by e_1, e_2 and e_3 with defining relations $5e_1 = 5e_2 = 5e_3 = 0$. Hence $H_1(M_1) \cong \mathbb{Z}_5 \oplus \mathbb{Z}_5 \oplus \mathbb{Z}_5$ (see the comment at the end of [WS]). A presentation for the fundamental group of M_1 has been computed by Lorimer [Lo], and abelianising it confirms this result. The fundamental groups are also computed in Chapter 3 § 3.2 where the presentations are described in detail.

Conjugating by an element of $\Gamma \setminus \Gamma^+$ transposes N_1 and N_2 , so M_2 and its tessellation \mathcal{D}/N_2 are obtained as the mirror images of M_1 and \mathcal{D}/N_1 . Petrie polygons on D_1 can be used to illustrate the chirality of these two tessellations: in each case these closed paths all have length 10, with one of the six edges appearing five times and the other five edges once each (as previously shown in Figure 5.2). Figure 5.2 shows a view of $\mathcal{M}_1 = \mathcal{D}/N_1$ with the outside face in front. In \mathcal{M}_1 the repeated edge of the Petrie path is always followed by a right turn, while in M_2 it is always followed by a left turn. Figures 5.3 and 5.4 illustrate the





Figure 5.3: Right-handed petric path

Figure 5.4: Left-handed petrie path

5.3 The tessellations \mathcal{D}/N'_i by 125 dodecahedra

The group $\Gamma^+/N_1 \cong A_5$ of symmetries of \mathcal{D}/N_1 is generated by a rotation through $2\pi/5$ about the center of the face f_1 , induced by α , and a rotation through $2\pi/3$ about the vertex v_1 , induced by β ; their product is a rotation through π about the mid-point of the edge e_1 . These rotations act on $H_1(M_1)$, regarded as a 3-dimensional vector space over the field GF(5), as the matrices

$$\alpha = \begin{pmatrix} 0 & 1 & 0 \\ 0 & 0 & 1 \\ 1 & 2 & 3 \end{pmatrix}, \quad \beta = \begin{pmatrix} 0 & 0 & 1 \\ -1 & 0 & 0 \\ 0 & -1 & 0 \end{pmatrix} \quad \text{and} \quad \alpha\beta = \begin{pmatrix} -1 & 0 & 0 \\ 0 & -1 & 0 \\ 3 & 2 & 1 \end{pmatrix}$$

with respect to the basis $\{e_1, e_2, e_3\}$. It is easily seen from this that $H_1(M_1)$ is an irreducible module for Γ^+/N_1 , hence also for the full isometry group $\Omega^+/N_1 \cong S_5$ of M_1 . This module is, up to isomorphism, the unique 3-dimensional irreducible module for S_5 over GF(5): it can be identified with the unique nonprincipal irreducible constituent of the natural permutation module for S_5 over this field, or equivalently with the natural module for the special orthogonal group $SO_3(5) \cong S_5$ (this last equivalence can be read off from the Atlas of Finite Groups [Atlas], or equivalently by considering the natural representation of S_5 over the field of five elements. Since 5 divides the order of S_5 , the representation splits as two 1-dimensional representations and an irreducible 3-dimensional representation). As discussed in the

previous section, conjugating by an element of $\Gamma \setminus \Gamma^+$ gives the same conclusion for $H_1(M_2)$.

Since $H_1(M_i)$ is irreducible as an $\operatorname{Isom}(M_i)$ -module, the only nontrivial regular abelian covering of M_i is the manifold $M'_i = \mathbf{H}^3/N'_i$. This has a tessellation \mathcal{D}/N'_i consisting of 125 dodecahedra, in one-to-one correspondence with the elements of $H_1(M_i)$. The symmetry group $G = \Gamma^+/N'_i$ of this tessellation is a semidirect product of an elementary abelian normal subgroup $T = N_i/N'_i \cong H_1(M_i)$ by a complement $H = \Gamma_0^+ N'_i/N'_i \cong A_5$ stabilising a dodecahedron. Similarly, as will be shown later, the full isometry group Ω^+/N'_i of M'_i is a semidirect product of T by S_5 , with the odd permutations in S_5 transposing the tessellation and its dual.

Proposition 5.3.1 The subgroup H of G stabilising a dodecahedron of \mathcal{D}/N'_i has two orbits each of lengths 12, 20 and 30 on the remaining dodecahedra.

Proof: As in the proof of Theorem 5.1, the right-handed flags of \mathcal{D}/N'_i can be identified with the elements of G, and the dodecahedral cells with the cosets gHof its subgroup H. As coset representatives for H in G choose the elements $t \in T$ and write D_t for the dodecahedron corresponding to $t\Pi$. Each element $g \in G$ has a unique factorisation g = th, where $t \in T$ and $h \in H$, and the corresponding flag ϕ_g lies in the dodecahedron D_t . In particular, the flag ϕ_1 corresponding to the identity element of G lies in the dodecahedron D_0 corresponding to the coset tH = H with t = 0 (here it is convenient to use additive notation for the abelian group T, and to write D_0 rather than D_1 for the dodecahedral cell incident with ϕ_1).

Each element $g \in G$ acts as an automorphism of \mathcal{D}/N'_i by left multiplication, sending a coset tH to $g^{-1}tH$, and hence acting on dodecahedra by $D_t \mapsto D_{g^{-1}t}$. The stabiliser of D_0 is H, and its elements h act on cosets by $tH \mapsto h^{-1}tH = t^hH$, so its action on dodecahedra is equivalent to its action by conjugation on T, as a group of linear transformations of this vector space over GF(5). It fixes 0, and by a routine calculation involving eigenspaces and the orbit-stabiliser theorem it is found that H has two orbits each of lengths 12, 20 and 30 on the remaining elements of T (since the eigenspaces of α and β are one-dimensional while that of $\alpha\beta$ is two-dimensional), so the same applies to its action on dodecahedra. If $t \neq 0$ in T then, arguing as in Proposition 5.3.1, it can be shown that t and -t are in the same orbit of H, while the elements $\pm 2t$ are in the other orbit of the same length. The actions of H on its orbits in T of lengths 12, 20 and 30 are equivalent to its actions on the faces, vertices and edges of D_0 , since the stabilisers are the same in each case.

In the action of H on dodecahedra, one orbit of length 12 consists of the dodecahedra D_f meeting D_0 across a common face f (where, from Proposition 5.3.1, fcorresponds as above to some element of T); the other orbit consists of dodecahedra D_{2f} meeting D_f across its face antipodal to f. Continuing like this 'bracelets' D_0 , D_f , D_{2f} , $D_{3f} = D_{-2f}$, $D_{4f} = D_{-f}$ of five dodecahedra are obtained, with -fdenoting the face of D_0 antipodal to f: each dodecahedron meets its two neighbours across a pair of antipodal faces. After five steps any walk through these dodecahedra, passing through the shared faces, returns to $D_{5f} = D_0$ with a twist through π .

The vertex figure at a vertex v of D_0 is an icosahedron $\{3, 5\}$, with each of its 20 faces corresponding to a corner of a dodecahedron incident with v. These corners come in antipodal pairs, so there is a dodecahedron D_v meeting D_0 antipodaly at v. Such dodecahedra D_v form an orbit of H of length 20, and the other orbit of this length consists of dodecahedra D_{2v} meeting D_v antipodaly at its vertex antipodal to v. Continuing in this fashion, 'necklaces' D_0 , D_v , D_{2v} , $D_{3v} = D_{-2v}$, $D_{4v} = D_{-v}$ of five dodecahedra, where -v is the vertex of D_0 antipodal to v, are obtained: each dodecahedron meets its two neighbours antipodal vertices in D_0 will meet four other dodecahedra, and the five dodecahedra join up to form a necklace of five dodecahedra each fixed setwise by an order three rotation fixing the geodesic.

If e is any edge of D_0 , then a geodesic from the center of D_0 to the midpoint of e continues across a face separating two dodecahedra (since there are five dodecahedra about any edge, such a configuration is possible), then passes through the vertex of this face opposite e, and continues across another face, passes through the midpoint of an edge, entering a dodecahedron D_e . Continuing along this geodesic, a 'galaxy' D_0 , D_e , D_{2e} , $D_{3e} = D_{-2e}$, $D_{4e} = D_{-e}$ of five dodecahedra is obtained, where -e is the edge of D_0 antipodal to e: each dodecahedron is separated from

90

its two neighbours by a pair of faces as above, and after five steps the path returns to $D_{5e} = D_0$. The orbits of H of length 30 consist of the dodecahedra D_e and D_{2e} , where e ranges over the edges of D_0 .

This accounts for all 124 dodecahedra $D \neq D_0$ in \mathcal{D}/N'_i , but does not account for all of their incidences with D_0 : there are two dodecahedra meeting D_0 along each edge e, in addition to the dodecahedra D_f and $D_{f'}$ corresponding to the faces f and f' of D_0 incident with e, and there are nine incidences with D_0 at each vertex v, in addition to those already described.

Proposition 5.3.2 II has two orbits each of lengths 20 and 30 on the vertices of \mathcal{D}/N'_i , one orbit of length 10 and one orbit of length 15.

Proof: There are 125 vertices in \mathcal{D}/N'_i . The vertices v of D_0 form an orbit of length 20, as do those joining dodecahedra D_v and D_{2v} in necklaces containing D_0 . The galaxies provide two orbits of length 30, consisting of vertices between D_0 and D_e , and between D_e and D_{2e} . This leaves 25 vertices to be accounted for. Since each axis of 5-fold rotation of D_0 is contained in a bracelet and therefore passes through no vertices, there cannot be any orbits of length dividing 12. Hence the only possible lengths (not exceeding 25) are 5, 10, 15 and 20. The vertex joining D_{2v} and D_{-2v} in a necklace is invariant under the subgroup of order 6 in Hpreserving the pair of vertices $\pm v$, so it lies in an orbit of length dividing 10. This orbit, together with the two of length 20, accounts for all the vertices invariant under a 3-fold rotation of D_0 . As a result there can be no other orbits of length dividing 20. Similarly, the vertex between D_{2e} and D_{-2e} in a galaxy is invariant under the subgroup of order 4 in H preserving the pair of edges $\pm e$, so it lies in an orbit of length dividing 15, and there are no other orbits of length dividing 30. It follows that these two orbits have lengths 10 and 15. 躑

Since \mathcal{D}/N'_i is isomorphic to its dual, the stabiliser in G of a vertex, isomorphic to A_5 , permutes the vertices and dodecahedra in the same way as H permutes the dodecahedra and vertices. In fact the first cohomology group $H^1(A_5, T)$ is a 1-dimensional vector space over GF(5) ([CPS], Theorem 4.2(c)), so there are five conjugacy classes of complements for T in G, two of which consist of the stabilisers of dodecahedra and of vertices. Arguments similar to those in Proposition 5.3.2 show that the complements in the other three classes have orbits of lengths 10, 15, 20, 20, 30 and 30 on the vertices and on the dodecahedra. The half-turn $r \in \Omega^+ \setminus \Gamma^+$ transposes the stabilisers of vertices and of dodecahedra, and also transposes two other classes of complements, leaving the fifth class invariant. Therefore the groups in this last class are contained in complements for T in $\Omega^+/N'_i = \text{Iso } M'_i$, isomorphic to S_5 .

5.4 The tessellation \mathcal{D}/N by 60 dodecahedra

Since N_1 and N_2 are normal subgroups of Ω^+ , so is their intersection $N = N_1 \cap N_2$. Further, as N_1 and N_2 are conjugate by elements of $\Omega \setminus \Omega^+$, N is normal in Ω . Since N_1 and N_2 are distinct maximal normal subgroups of Γ^+ , $N_1N_2 = \Gamma^+$, and therefore $N_i/N \cong \Gamma^+/N_i \cong A_5$ for each i. This shows that the manifold $M = \mathbf{H}^3/N$ is a 60-sheeted regular unbranched covering of each M_i , with covering group A_5 . There is a tessellation \mathcal{D}/N of M by 60 dodecahedra; this has symmetry group Γ/N isomorphic to the wreath product $A_5 \wr C_2$, with the base group $\Gamma^+/N = N_1/N \times N_2/N \cong A_5 \times A_5$ as the orientation-preserving subgroup and $\Gamma_0 N/N \cong A_5 \times C_2$ as the subgroup stabilising a dodecahedron. The isometry r transforms \mathcal{D}/N to its dual tessellation $\mathcal{D}^*/N \cong \mathcal{D}/N$. The normaliser of N in Iso \mathbf{H}^3 is Ω , so M has isometry group Ω/N ; this is isomorphic to the subgroup of index 2 in the wreath product $S_5 \wr C_2$ generated by the complement C_2 and the subgroup of the base group $S_5 \times S_5$ consisting of pairs of permutations with equal parity.

Since N_1 and N_2 are the normal closures in Γ^+ of the elements $(abcd)^t = (\alpha \gamma)^t$ for t = 2 and 3, their intersection N contains the normal closure in Γ^+ of $(abcd)^6$. Indeed, being normal in Γ , N contains the normal closure of $(abcd)^6$ in Γ , and this is also the normal closure in Γ^+ of $(abcd)^6$ and its conjugate $(bcda)^6 = (bacd)^6$. A computer calculation ([C1], p. 45) shows that N contains the normal closure of $(abcd)^6$ in Γ with index 2^9 , so $\langle \langle (abcd)^6 \rangle \rangle_{\Gamma}$ has index 1843200 in Γ . By contrast, it should be noted that the normal closure of $(abcd)^6$ in Γ^+ has index 58,982,400. Thus \mathcal{D}/N has a 2^9 -sheeted covering by Coxeter's tessellation $\{5,3,5\}_{6,6}$ ([C1], p. 45); this is formed from \mathcal{D} by identifying flags which are equivalent under the monodromy permutations $(abcd)^6$ or $(bacd)^6$.

5.5 Subgroups of direct squares

In order to understand the structure of \mathcal{D}/N and its symmetry group, some general facts about certain subgroups of direct squares of groups, such as $A_5 \times A_5$, are required. The following results are all easily verified.

Let S be any group, and let S^2 denote its direct square $S \times S$. For each $\alpha \in \operatorname{Aut} S$ there is a subgroup $S_{\alpha} = \{(s, s\alpha) \mid s \in S\}$ of S^2 which projects isomorphically onto both direct factors; conversely every such subgroup has this form for some automorphism α , so the number of such subgroups is [Aut S]. Two such subgroups S_{α} and S_{β} are conjugate in S^2 if and only if $\alpha^{-1}\beta \in \text{Inn } S$, therefore each conjugacy class contains |Inn S| = |S : Z(S)| subgroups, where Z(S) is the center of S, and there are $|\operatorname{Out} S| = |\operatorname{Aut} S : \operatorname{Inn} S|$ conjugacy classes of them, one for each coset $[\alpha] = \alpha. \text{Inn } S$ of Inn S in Out S. Each conjugacy class of subgroups S_{α} is the set of point stabilisers in a transitive representation of degree |S| of S^2 . The kernel of this representation is $Z_{\alpha} = \{(z, z\alpha) \mid z \in Z(S)\}$, the induced permutation group is S^2/Z_{α} (a central product of two copies of S), and the direct factors of S^2 act as commuting regular normal subgroups, which can be identified with the regular representations of S on itself by left and right multiplication. In particular, if α is taken to be the identity automorphism then the corresponding permutation group is the holomorph Hol S of S, a semidirect product of a regular normal subgroup Sby the diagonal subgroup $S_{\alpha} = \{(s, s) \mid s \in S\}$, acting by conjugation on S as the stabiliser of the identity element.

5.6 The structure of the tessellation \mathcal{D}/N

Consider again the tessellation \mathcal{D}/N . The epimorphism $\theta = (\theta_1, \theta_2) : \Gamma^+ \to A_5 \times A_5$ given by

 $\alpha \mapsto ((13524), (13524)), \quad \beta \mapsto ((123), (123)), \quad \gamma \mapsto ((14352), (13425))$

has kernel N, and this can be used to identify the orientation-preserving symmetry group $G := \Gamma^+/N$ of \mathcal{D}/N with $A_5 \times A_5$.

Proposition 5.6.1 The subgroup H of G stabilising a dodecahedron of \mathcal{D}/N has

two orbits of length 12, and one orbit each of length 15 and 20 on the remaining dodecahedra.

Proof: By taking $S = A_5$ in the previous section, so that Aut $S \cong S_5$ and Z(S) = 1, it can be seen that in G there are $|\operatorname{Out} A_5| = 2$ conjugacy classes each consisting of $|\operatorname{Inn} A_5| = 60$ non-normal subgroups isomorphic to A_5 ; these are the stabilisers in G of dodecahedral cells and of vertices in \mathcal{D}/N , represented by the images under θ of $\Gamma_0^+ = \langle \alpha, \beta \rangle$ and of $\Gamma_1^+ = \langle \beta, \gamma \rangle$, where $\Gamma_1 = \langle b, c, d \rangle$. In particular, since θ_1 and θ_2 agree on α and β , it follows that the stabiliser of the dodecahedron D_1 is the diagonal subgroup $H = \{(g,g) \mid g \in A_5\}$. The direct factors of $G = A_5 \times A_5$ commute, and act regularly on the 60 dodecahedra, which can therefore be labeled with elements of A_5 so that the two factors act as the left and right regular representations of A_5 . It follows that H acts by conjugation, so its orbits on dodecahedra correspond to the conjugacy classes of A_5 . The dodecahedron D_1 corresponds to the class \mathcal{C}_1 , which consists of the identity element, and the other orbits have lengths 15 (corresponding to the class \mathcal{C}_2 of involutions), 20 (corresponding to the class \mathcal{C}_3 of 3-cycles), 12 and 12 (corresponding to the two classes \mathcal{C}_5 and \mathcal{C}'_5 of 5-cycles, containing (12345) and its square).

The above five orbits will be denoted by $\mathcal{O}_1, \mathcal{O}_2, \mathcal{O}_3, \mathcal{O}_5$ and \mathcal{O}'_5 respectively. The subgroups of H stabilising dodecahedra in these orbits are the centralisers of the corresponding elements of A_5 , isomorphic to A_5, V_4, C_3, C_5 and C_5 respectively. This will help to determine how D_1 (and, by regularity, every dodecahedron in \mathcal{D}/N) is related geometrically to the 59 others. By self-duality, similar conclusions will apply to the vertices.

The flags of \mathcal{D} and the tetrahedra of \mathcal{T} are labeled with the elements of Γ , with this group acting regularly by right or left multiplication as a group of monodromy permutations or automorphisms. The flags of \mathcal{D}/N and tetrahedra of \mathcal{T}/N can therefore be labeled with the elements of Γ/N ; here either action of N can be used since this is a normal subgroup of Γ . As a consequence the technical difficulties encountered earlier with the non-normal subgroups N_i can be avoided. In particular, the right-handed flags and tetrahedra can be identified with the elements of G, through its isomorphism with Γ^+/N induced by θ . Those in a given dodecahedral cell form a coset gH of the diagonal subgroup H in G, so the cells of \mathcal{D}/N can be identified with these cosets of H in G. As coset representatives the elements (g, 1) with $g \in A_5$ are chosen, since there are 60 of these, all in different cosets; then write D_g for the corresponding dodecahedral cell of \mathcal{D}/N . A general element $(g_1, g_2) \in G$ then lies in the coset containing $(g_1, g_2)(g_2^{-1}, g_2^{-1}) = (g_1g_2^{-1}, 1)$, represented by (g, 1) where $g = g_1g_2^{-1} \in A_5$, so the flag $\phi = (g_1, g_2)$ lies in the corresponding cell D_g . In particular, the flag $\phi_1 = (1, 1)$ corresponding to the identity element of G lies in the cell D_1 corresponding to the coset gH = H. Let v_1 , e_1 and f_1 denote the vertex, edge and face of D_1 incident with ϕ_1 .

Proposition 5.6.2 Dodecahedra D_g and D_h in \mathcal{D}/N meet across a common face if and only if $g^{-1}h \in \mathcal{C}_5$. In these circumstances, this face is unique, and across its antipodal face D_g meets the dodecahedron $D_{h'}$ satisfying $g^{-1}h' = (g^{-1}h)^{-1} \in \mathcal{C}_5$.

Proof: The element γ , acting as a monodromy permutation, rotates flags of \mathcal{D} around their incident edges, so its image $\gamma \theta = ((14352), (13425)) \in G$ has this effect on the flags of \mathcal{D}/N . Applying the powers of this element to ϕ_1 the flags $((14352)^i, (13425)^i)$ are obtained for $i = 1, \ldots, 4$, and these lie in the dodecahedra D_q for $g = (14352)^i (13425)^{-i} = (13254), (152), (134), (15432)$ respectively. These cells are those that meet D_1 around e_1 . In particular, the last of these cells meets D_1 across f_1 , and the first meets D_1 across the other face of D_1 incident with e_1 . Applying an arbitrary element $(h, h) \in H$ as an automorphism preserving D_1 , it can be seen that the flags $\phi_h = (h^{-1}(14352)^i, h^{-1}(13425)^i)$ are all obtained by rotating the flag (h^{-1}, h^{-1}) of D_1 around its incident edge; these flags ϕ_h lie in the dodecahedra D_g for $g = (13254)^h, (152)^h, (134)^h, (15432)^h$. Thus D_1 meets these dodecahedra D_q around their common edge in the same way as it meets the first four dodecahedra, indexed by $(14352)^{i}(13425)^{-i}$, around e_1 . In particular, it meets a dodecahedron D_g across a face if and only if g is in the conjugacy class \mathcal{C}_5 of A_5 containing (15432), or equivalently its inverse (12345). There are 12 faces of D_1 , and 12 elements $g \in \mathcal{C}_5$, both permuted transitively by H, so each D_q indexed by $g \in \mathcal{C}_5$ meets D_1 across a single face. These 12 dodecahedra D_g form the orbit \mathcal{O}_5 of H on cells. More generally, given any pair of dodecahedra D_g and D_h , by applying an automorphism sending D_g to D_1 , it can be seen that D_g and D_h meet across a common face (which is unique) if and only if $g^{-1}h \in \mathcal{C}_5$.

The monodromy permutation $(abc)^5$, representing the central involution in Γ_0 , sends ϕ_1 to its antipodal flag $-\phi_1$ in D_1 , so $(abc)^5d$ sends ϕ_1 to the adjacent flag $(-\phi_1)d$ in the dodecahedron D_g meeting D_1 across its face $-f_1$ opposite f_1 . Since θ maps $(abc)^5d$ to $((13524)^{-1}, (13524))$, it follows that $g = (13524)^{-2} = (12345)$. This is the inverse of the element (15432) of \mathcal{C}_5 labeling the dodecahedron meeting D_1 across f_1 . It follows that a general dodecahedron D_g meets neighbours D_h and $D_{h'}$ across opposite faces if and only if $g^{-1}h$ and $g^{-1}h'$ are mutually inverse elements of \mathcal{C}_5 .

This result shows that the dual graph of \mathcal{D}/N is isomorphic to the Cayley graph for A_5 with respect to its generating set \mathcal{C}_5 , and by self-duality the same applies to the 1-skeleton of \mathcal{D}/N .



Figure 5.5: Local adjacency about v

Figure 5.6: Partial schematic at v

As seen in the proof of Proposition 5.6.2, D_1 meets $D_{(15432)}$ across f_1 , so this is the dodecahedron D_{f_1} in the notation of § 5.3; it is contained in the orbit of H consisting of the 12 dodecahedra D_f where f is a face of D_1 . Since (15432) is in the conjugacy class C_5 this is the orbit \mathcal{O}_5 . At the face opposite f_1 , D_{f_1} meets $D_{(14253)}$ where $(14253) = (15432)^2$ is a member of the other class \mathcal{C}'_5 of 5-cycles; this lies in a second orbit \mathcal{O}'_5 , consisting of 12 dodecahedra, denoted D_{2f} in the earlier notation. By iterating this argument bracelets of five dodecahedra are obtained, each meeting its two neighbours across a pair of opposite faces; starting with D_1 these dodecahedra lie in the orbits \mathcal{O}_1 , \mathcal{O}_5 , \mathcal{O}'_5 , \mathcal{O}'_5 and \mathcal{O}_5 of H. **Proposition 5.6.3** Dodecahedra D_g and D_h in \mathcal{D}/N meet antipodaly at a common vertex if and only if $g^{-1}h \in \mathcal{C}_3$. In these circumstances, this vertex is unique, and at its antipodal vertex D_g meets $D_{h'}$ antipodaly with $g^{-1}h' = (g^{-1}h)^{-1} \in \mathcal{C}_3$.

Proof: The subgroup $\Gamma_1 = \langle b, c, d \rangle$, in its monodromy action, preserves the set of flags incident with v_1 . Its central involution $(bcd)^5$ sends ϕ_1 to the antipodal flag at v_1 , incident with the dodecahedron D_{v_1} antipodal to D_1 at v_1 , so the element $(bcd)^5 b$ of Γ^+ also sends ϕ_1 to a flag in D_{v_1} . Now θ maps $(bcd)^5 b = \gamma^{-1}(\gamma\beta\gamma^{-1}\beta^{-1})^3$ to ((23)(45), (12)(45)), and $(23)(45)((12)(45))^{-1} = (123) \in \mathcal{C}_3$, so $D_{v_1} = D_{(123)}$. D_1 has 20 vertices, which is the number of elements in \mathcal{C}_3 , so the dodecahedra D_v meeting D_1 antipodaly across a common vertex v form an orbit \mathcal{O}_3 of H, each meeting D_1 at a unique vertex. As in the case of adjacency across faces, it can be seen , by applying automorphisms, that a pair of dodecahedra D_g and D_h meet antipodaly across the antipodal vertex of D_g if and only if $g^{-1}h$ and $g^{-1}h'$ are mutually inverse elements of \mathcal{C}_3 .

By iterating this result, it follows that dodecahedra in \mathcal{D}/N form necklaces of length 3, each dodecahedron meeting its two neighbours antipodaly at an antipodal pair of its vertices. These dodecahedra and incidences can be represented as the vertices and edges of the Cayley graph of A_5 with respect to its generating set \mathcal{C}_3 .

As seen in the proof of Proposition 5.6.2, the dodecahedra $D_{(152)}$ and $D_{(134)}$ meet D_1 along its edge e_1 , and therefore the dodecahedra in \mathcal{O}_3 all meet D_1 across an edge, and more generally D_h meets D_g across an edge if and only if $g^{-1}h \in \mathcal{C}_3$. Now D_g has 30 edges, each corresponding to two such dodecahedra D_h . Consequently, it has 60 such incidences. Since $|\mathcal{C}_3| = 20$, it follows that each of these dodecahedra D_h meets D_g three times along edges, in addition to its antipodal incidence at a vertex v of D_g . Let \overline{v} be the vertex in D_g that is antipodal to v. For any vertex v in a dodecahedron D, the edges opposite v are defined as follows: take a geodesic from v through the center of the face containing v. Then this arc cuts an edge e across from v. Call this edge the edge opposite \overline{v} . Of the two dodecahedra which meet D_g along such an edge, D_h is the nearer to v.

The vertex figure of \mathcal{D}/N at v_1 is an icosahedron $\{3,5\}$, its 20 faces corresponding to the 20 corners of dodecahedra meeting at v_1 . So far 11 of these dodecahedra have been accounted for: there is D_1 itself, three dodecahedra $D_{(15432)}$, $D_{(13254)}$ and $D_{(13542)}$ in \mathcal{O}_5 meeting D_1 across faces incident with v_1 , six dodecahedra $D_{(152)}$, $D_{(134)}$, $D_{(253)}$, $D_{(142)}$, $D_{(135)}$ and $D_{(243)}$ in \mathcal{O}_3 meeting it along edges incident with v_1 , and $D_{(123)} \in \mathcal{O}_3$ meeting D_1 antipodaly at v_1 . Calculations with monodromy permutations show that, of the remaining nine dodecahedra meeting D_1 at v_1 , three of them, $D_{(13452)}$, $D_{(14532)}$ and $D_{(13245)}$ in \mathcal{O}'_5 , meet $D_{(123)}$ across a face, while the remaining six dodecahedra, $D_{(15)(23)}$, $D_{(13)(25)}$, $D_{(12)(35)}$, $D_{(12)(34)}$, $D_{(13)(24)}$ and $D_{(14)(23)}$ in \mathcal{O}_2 , meet $D_{(123)}$ along an edge. Figures 5.7, 5.8 and 5.9 Illustrate the geometrical structure at v. More generally, as described above, two dodecahedra D_g and D_h meet at a common vertex if and only if $g^{-1}h$ is one of the corresponding permutations listed here. Since any meeting between two dodecahedra must be across a common face, along a common edge, or at a common vertex, all meetings between pairs of dodecahedra in \mathcal{D}/N have been accounted for.



Figure 5.7: Vertex structure at v: I

Figure 5.8: Vertex structure at v: II



Figure 5.9: Vertex structure at v: III

5.7 The Poincaré dodecahedral spaces

The chirality exhibited by the Weber-Seifert space also arises in connection with Poincaré's dodecahedral space, where antipodal pairs of faces of a dodecahedron in S^3 , with dihedral angles $2\pi/3$, are identified after a twist through $\pm \pi/5$ to produce a compact 3-manifold, Poincaré's homology sphere. As before, the two possible directions of twisting yield a chiral pair of oriented manifolds P_1 and P_2 , each having only orientation-preserving isometries, and each manifold is the mirror-image of the other. The details of the construction are rather better-known than in the hyperbolic case ([Cox RP, Ch. VIII], [Mon, §3.13], [ST, §62] and [Thu, §1.4.4]), so they will just be outlined here.

Stereographic projection, which allows rotations of S^2 to be represented as Möbius transformations of the Riemann sphere $\mathbf{C} \cup \{\infty\}$, gives an isomorphism $SO(3) \cong PSU(2)$. The natural projection $SU(2) \to PSU(2)$ provides a double covering of SO(3) by

$$SU(2) = \left\{ \begin{pmatrix} u & v \\ -\overline{v} & \overline{u} \end{pmatrix} \mid u, v \in \mathbf{C}, \ u\overline{u} + v\overline{v} = 1 \right\}.$$

By taking real and imaginary parts of u and v as coordinates, SU(2) can be identified with the 3-sphere $S^3 \subset \mathbf{R}^4$, and hence with the multiplicative group Q of unit quaternions. The actions of this group on itself by left and by right multiplication give two subgroups $Q_L, Q_R \cong Q$ in the orientation-preserving isometry group SO(4) of S^3 ; these commute with each other, intersect in their common center of order 2, and generate SO(4), so that SO(4) is the central product Q_LQ_R of Q_L and Q_R , isomorphic to the quotient of $Q \times Q \cong SU(2) \times SU(2)$ obtained by identifying each element of $Q \times Q$ with its negative. The full isometry group O(4) of S^3 is an extension of SO(4) by an orientation-reversing involution induced by conjugation of quaternions, sending (x_1, x_2, x_3, x_4) to $(x_1, -x_2, -x_3, -x_4)$; this inverts each element of Q, so it transposes the two subgroups Q_L and Q_R of SO(4).

There is a single conjugacy class of icosahedral (or dodecahedral) rotation groups $I \cong A_5 \cong L_2(5)$ in SO(3), each I lifting in SU(2) to a binary icosahedral group $\hat{I} \cong \hat{A}_5 \cong SL_2(5)$, an extension of I by a center of order 2. The resulting embedding of \hat{I} in \hat{Q} yields a pair of subgroups $\hat{I}_L, \hat{I}_R \cong \hat{I}$ of Q_L and Q_R , which generate their central product $\hat{I}_L \hat{I}_R$ of order $120^2/2 = 7200$ in SO(4); extending this by conjugation of quaternions, which transposes \hat{I}_L and \hat{I}_R , gives a subgroup Δ of order 14400 in O(4). This is the Coxeter group [5, 3, 3], the symmetry group of the 120-cell {5, 3, 3}, a tessellation \mathcal{E} of S^3 by 120 dodecahedra with dihedral angles $2\pi/3$; the orientation-preserving subgroup $\Delta^+ = [5, 3, 3]^+$ of Δ is $\hat{I}_L \hat{I}_R$.

One can construct \mathcal{E} as the Dirichlet (or Voronoi) tessellation of S^3 corresponding to its discrete subgroup \hat{I} . For each $g \in \hat{I}$ define the Dirichlet region E_g to be the set of points in S^3 which have g as a nearest element of \hat{I} . These 120 sets are permuted regularly by \hat{I}_L and \hat{I}_R , with the diagonal subgroup $H \cong A_5$ of the central product $\Delta^+ = \hat{I}_L \hat{I}_R$ stabilising E_1 and permuting the sets E_g in the same way as it acts by inner automorphisms on the elements $g \in \hat{I}$; the antipodal involution stabilising E_1 acts on these sets by inverting their labels g.

In order to determine the shape of each set E_g , it is convenient to choose the subgroup I of SO(3) to contain the rotation $z \mapsto e^{2\pi i/5}z$ of the Riemann sphere, which lifts to the pair of elements

$$\pm \begin{pmatrix} e^{\pi i/5} & 0\\ 0 & e^{-\pi i/5} \end{pmatrix} \in \hat{I}$$

in SU(2) corresponding to the points $\pm(\cos \pi/5, \sin \pi/5, 0, 0) \in S^3$. The elements of \hat{I} closest to the identity (1, 0, 0, 0) are those maximising their first coordinate,

and with this choice of I these are the 12 members of the conjugacy class of I containing the element $(\cos \pi/5, \sin \pi/5, 0, 0)$. It follows that each E_g is a regular dodecahedron centered at g, meeting E_h across a face if and only if $g^{-1}h$ is conjugate to $(\cos \pi/5, \sin \pi/5, 0, 0)$. These 120 dodecahedra E_g have dihedral angles $2\pi/3$, and are the cells of a tessellation \mathcal{E} of S^3 with Schläfli symbol $\{5, 3, 3\}$ and isometry group Δ . This tessellation has 600 vertices, 1200 edges and 720 faces, and its dual is the 600-cell $\{3, 3, 5\}$. The barycentric subdivision of \mathcal{E} is a tessellation of S^3 by 14400 tetrahedra, each having as its vertices a vertex of \mathcal{E} and the midpoints of an edge, face and cell of \mathcal{E} , all mutually incident. The reflections $\underline{a}, \underline{b}, \underline{c}$ and \underline{d} of S^3 in the sides of a tetrahedron opposite these vertices generate Δ , giving a presentation

$$\Delta = \langle \underline{a}, \underline{b}, \underline{c}, \underline{d} \mid \underline{a}^2 = \underline{b}^2 = \underline{c}^2 = \underline{d}^2 = 1,$$

$$(\underline{ab})^5 = (\underline{bc})^3 = (\underline{cd})^3 = (\underline{ac})^2 = (\underline{ad})^2 = (\underline{bd})^2 = 1\rangle$$

of Δ as a Coxeter group. Setting $\underline{\alpha} = \underline{ab}, \underline{\beta} = \underline{bc}$ and $\underline{\gamma} = \underline{cd}$ as before, the elements $\underline{\alpha}, \underline{\beta}$ and $\underline{\gamma}$ generate the index 2 subgroup Δ^+ of Δ . Let $\hat{I}_R = \langle (abcd)^5 \rangle_{\Delta^+}$ and $\hat{I}_L = \langle (abcd)^3 \rangle_{\Delta^+}$. Then \hat{I}_R and \hat{I}_R generate Δ^+ as a central product and \hat{I}_R and \hat{I}_R have order 120.

The quotients P_1 and P_2 of S^3 by \hat{I}_R and \hat{I}_L are obtained by identifying equivalent pairs of faces of the dodecahedron E_1 , using right multiplication by the conjugates h of $(\cos \pi/5, \sin \pi/5, 0, 0)$ or left multiplication by h^{-1} ; these identify antipodal pairs of faces of E_1 by means of left- or right-handed screw motions with angle $\pi/5$, so each P_i is a Poincaré dodecahedral space. In each case, the isometry group of P_i , induced by the action of the other copy of \hat{I} on S^3 , is isomorphic to I and contains only orientation-preserving elements. The involution which transposes \hat{I}_R and \hat{I}_L induces an orientation-reversing isometry between P_1 and P_2 , so these two spaces form a chiral pair, as noted by Montesinos [Mon, Ch. 3]. Since \hat{I}_R and \hat{I}_L both act freely on the simply connected space S^3 , it follows that P_1 and P_2 have fundamental groups isomorphic to \hat{I} ; this is a perfect group, so their first (and, by duality, second) homology groups are trivial, that is, they are homology spheres.

The quotients of \mathcal{E} by \hat{I}_R and \hat{I}_L are a chiral pair of tessellations of P_1 and P_2 , each consisting of five vertices, ten edges, six faces and one cell. These are Coxeter's twisted honeycombs $\{5, 3, 3\}_t$ for t = 5 and t = 3 respectively, corresponding to the fact that \hat{I}_R and \hat{I}_L are the normal closures in Δ^+ of $(\underline{abcd})^5$ and $(\underline{abcd})^3$ ([Cox TH, Fig. 22], [Mon, Fig 29]). The identification $\{5, 3, 3\}_3$ is illustrated in Figure 5.10



Figure 5.10: Coxeter's $\{5, 3, 3\}_3$

5.8 The structure of the 120-cell

The structure of \mathcal{E} , the 120-cell, can be studied in the same way as were the tessellations \mathcal{D}/N'_i and \mathcal{D}/N , by identifying the flags with the elements of Δ and the dodecahedral cells with the cosets of $\Delta_0 = \langle \underline{a}, \underline{b}, \underline{c} \rangle \cong I \times C_2$ in Δ , or more conveniently of $\Delta_0^+ \cong I$ in Δ^+ . The dodecahedra E_g are labeled by the elements g of $\hat{I}_L \cong \hat{I}$, acting as coset representatives of Δ_0^+ in Δ^+ , so that the orientation-preserving stabiliser Δ_0^+ of D_1 permutes them in the same way as it acts as inner automorphisms on \hat{I} . In particular, the orbits of Δ_0^+ on dodecahedra correspond to the conjugacy classes of \hat{I} . Each conjugacy classes $\mathcal{C} = \mathcal{C}_1, \mathcal{C}_3, \mathcal{C}_5$ or \mathcal{C}'_5 of elements of odd order n = 1, 3 or 5 in I lifts to two classes $\hat{\mathcal{C}}$ and $-\hat{\mathcal{C}}$ in \hat{I} , each inverse-closed and of size $|\mathcal{C}|$, containing elements of order n and 2n respectively. On the other hand, the class \mathcal{C}_2 in I containing the 15 involutions lifts to two mutually inverse classes of 15 elements of order 4 in \hat{I} . The antipodal involution $(\underline{abc})^5$ in $\Delta_0 \setminus \Delta_0^+$ acts by inverting the labels of the dodecahedra, so the orbits of Δ_0^- of length 15 labeled by the elements of order 4 form a single orbit of Δ_0 of length 30.

Proceeding as in § 5.3 and § 5.6, it is shown that bracelets of dodecahedra in \mathcal{E} have length 10, each bracelet through E_1 being labeled with successive powers $g, g^2, \ldots, g^{10} = 1$ of an element $g \in -\hat{\mathcal{C}}'_5$, lying in the classes $-\hat{\mathcal{C}}'_5$, $\hat{\mathcal{C}}_5$, $-\hat{\mathcal{C}}_5$, $\hat{\mathcal{C}}'_5$, $-\hat{\mathcal{C}}_1$, $\hat{\mathcal{C}}'_5$, $-\hat{\mathcal{C}}_5$, $\hat{\mathcal{C}}_5$, $-\hat{\mathcal{C}}_5$, $-\hat{\mathcal{C}}_5$, $-\hat{\mathcal{C}}_5$, $\hat{\mathcal{C}}_5$, $-\hat{\mathcal{C}}_5$,

Every element of \hat{I} whose order divides 10 appears as a label of such a dodecahedron, and in particular the antipodal dodecahedron E_{-1} appears opposite E_1 in each of the six bracelets containing E_1 . Necklaces have length 3, the labels of those through E_1 accounting for the elements of order 3, while their antipodal necklaces each consist of E_{-1} and two dodecahedra labeled by mutually inverse elements of order 6.

The element <u>abcd</u> of Δ has order 30, with the involution $(\underline{abcd})^{15}$ corresponding to $(-1,0,0,0) \in S^3$ and generating the center $\hat{I}_L \cap \hat{I}_R \cong C_2$ of Δ [C1, §11]. The antipodal quotient of \mathcal{E} , by this center, is a tessellation $\{5,3,3\}_{15}$ of SO(3), or equivalently of real projective 3-space $\mathbf{P}^3(\mathbf{R})$, by 60 dodecahedra; it can be constructed as the Dirichlet tessellation of SO(3) corresponding to its discrete subgroup I. In this case each bracelet of dodecahedra has length 5, lifting to a bracelet of length 10 in \mathcal{E} , while each necklace has length 3, lifting to an antipodal pair of necklaces of length 3 in \mathcal{E} .

5.9 The 120-cell and \mathcal{D}/N .

There is a connection between the spherical tessellation $\mathcal{E} = \{5, 3, 3\}$ and the hyperbolic tessellation $\mathcal{D}/N = \{5, 3, 5\}/N$ studied earlier. As shown by Lorimer [Lor], the Coxeter group $\Delta = [5, 3, 3]$ is generated by its elements $\underline{a}, \underline{b}, \underline{c}$ and $\underline{e} = \underline{d}(\underline{abc})^5 \underline{d}(\underline{abc})^5 \underline{d}$; these satisfy the relations

$$\underline{a}^2 = \underline{b}^2 = \underline{c}^2 = \underline{c}^2 = (\underline{a}\underline{b})^5 = (\underline{b}\underline{c})^3 = (\underline{c}\underline{e})^5 = (\underline{a}\underline{c})^2 = (\underline{a}\underline{e})^2 = (\underline{b}\underline{e})^2 = 1,$$

which correspond to the standard defining relations of Γ with $\underline{a}, \underline{b}, \underline{c}$ and \underline{e} replacing a, b, c and d, so there is an epimorphism $\varphi : \Gamma \to \Delta$ given by $a \mapsto \underline{a}, b \mapsto \underline{b}, c \mapsto \underline{c}$ and $d \mapsto \underline{e}$. (Actually, this is the dual of Lorimer's epimorphism $\Gamma \to \Delta^* = [3, 3, 5]$.)

To study all possible rewritings of the presentation of [5, 3, 3], the following geometric description of the 120-cell is considered. In [C1, §9] the 120-cell is de-

composed into two linked tori each consisting of 60 dodecahedra. Cutting one of these tori along a meridinal curve (as illustrated in Figure 5.11) gives a tower of



Figure 5.11: two linked tori with meridian path of one marked

dodecahedra, with a core tower of 10 stacked dodecahedra. Adjacent dodecahedra share a horizontal pentagonal face, and at each edge of this face a further dodecahedron is attached. The outer surface of the tower consists of the 200 free faces of the 50 dodecahedra wrapped around the central tower. Let A and B be planes in S^4 through the horizontal face of D_1 representing the reflecting planes for a and b, and consider how a plane representing d could be constructed. Such a plane must be orthogonal to both A and B in S^3 and so projects (under orthogonal projection) to either a horizontal plane in \mathbb{R}^3 or to a sphere in \mathbb{R}^3 orthogonal to both A and B. By looking in the 120-cell it can be seen that any reflection r whose plane R is orthogonal to A and B is going to be a symmetry of the meridian curve illustrated in Figure 5.12. The stabiliser of this curve is D_{20} (in fact the stabiliser of this curve will act transitively on the dodecahedra of the core of the second torus and so it is the stabiliser of a stalk), so there are 11 possible involutions $\{w_i\}$ whose product with a and b have the required properties. A quick computer check of these elements reveals that the product cw_i has order 2 (three of these), 3 (two of these), 5 (two of these), 6 (two of these) or 10 (two of these). Of the 11 w_i , eight of them (those with cw_i of order 3, 5 or 6 and two of those with cw_i of order 2) form a generating set $[a, b, c, w_i]$ for Δ . This also proves the existence of an epimorphism $\theta : [5, 3, 6] \rightarrow [5, 3, 3]$.

To find a more useful description of φ , its restriction to Δ^+ is first constructed. This group is the central product $\hat{I}_L \hat{I}_R$ of two copies of $\hat{I} \cong SL_2(5)$, that is, the



Figure 5.12: Two hundred faces of $\{5,3,3\}$ forming the boundary torus of a dodecahedral tower. Meridian path bounded by dotted lines

quotient of $SL_2(5) \times SL_2(5)$ obtained by identifying each pair of matrices with its negative. The center of this group is generated by the involution $\pm(I_2, -I_2)$, where I_2 is the identity matrix in $SL_2(5)$, and the central quotient, obtained by identifying each matrix with its negative, is $L_2(5) \times L_2(5) \cong A_5 \times A_5$. Define $\varphi^+: \Gamma^+ \to \Delta^+$ by

$$\alpha \mapsto \pm(A, A), \quad \beta \mapsto \pm(B, B), \quad \gamma \mapsto \pm(C_1, C_2)$$

where

$$A = \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}, \quad B = \begin{pmatrix} -1 & -1 \\ 1 & 0 \end{pmatrix}, \quad C_1 = \begin{pmatrix} 0 & -2 \\ -2 & 2 \end{pmatrix} \text{ and } C_2 = \begin{pmatrix} 0 & 2 \\ 2 & 2 \end{pmatrix}$$

are elements of $SL_2(5)$. It is routine to check that these images of α , β and γ generate Δ^+ and satisfy the defining relations of Γ^+ , so φ^+ is an epimorphism. Moreover, the epimorphism $SL_2(5) \to A_5$ given by $A \mapsto (13524)$ and $B \mapsto (123)$ sends $C_1 = [B^{-1}AB, A]^2$ to (14352) and $C_2 = BC_1B^{-1}$ to (13425), so φ^+ is a lifting of the epimorphism $\theta : \Gamma^+ \to A_5 \times A_5$ considered in § 5.7. Now Δ is an extension of Δ^+ by the involution $\pm (M_1, M_2) \mapsto \pm (M_2, M_1)$ which interchanges the two copies of $SL_2(5)$. Therefore $\varphi^+ : \Gamma^+ \to \Delta^+$ can be extended to an epimorphism $\varphi : \Gamma \to \Delta$ by sending $(abc)^5$, an involution which commutes with α and β and conjugates γ to $[\alpha^{-1},\beta]^2 \alpha \gamma^{-1} \beta^{-1} \alpha^{-1} \beta [\alpha^{-1},\beta]^2$, to this involution, since a calculation shows that this has the corresponding effect on the images of α , β and γ in Δ^+ .

Theorem 5.2 The kernel $K = \ker \varphi$ is the unique normal subgroup of Γ with $\Gamma/K \cong \Delta$. It is a subgroup of index 2 in N.

Proof: Since $\varphi : \Gamma \to \Delta$ is an epimorphism, the first isomorphism theorem gives $\Gamma/K \cong \Delta$. Since Γ^+ is perfect and Δ is not, any normal subgroup L with $\Gamma/L \cong \Delta$ must be contained in Γ^+ , and hence in N since it follows from Lemma 3.2.1 that N is the only normal subgroup of Γ^+ with quotient $A_5 \times A_5$. This implies that the epimorphism $\theta : \Gamma^+ \to \Gamma^+/N \cong A_5 \times A_5$ must lift to an epimorphism $\hat{\theta} : \Gamma^+ \to \Delta^+$ with kernel L. Now any element g of odd order n in $A_5 \cong L_2(5) \cong I$ lifts to two elements \hat{g} and $-\hat{g}$ of orders n and 2n in $\hat{A}_5 \cong SL_2(5) \cong \hat{I}$, so any element (g, h) of odd order n in $A_5 \times A_5$ lifts to two elements $\pm(\hat{g}, \hat{h})$ and $\pm(\hat{g}, -\hat{h})$ of orders n and 2n in Δ^+ . Since Γ^+ is generated by its elements α, β and γ of odd orders n = 5, 3 and 5, any lifting $\hat{\theta}$ of θ must send these generators to the unique elements of the same orders n covering their images in $A_5 \times A_5$. This shows that θ has only one lift $\hat{\theta}$, namely φ , so L = K, a subgroup of index 2 in N.

Using GAP, a presentation was obtained for N. It has a presentation on 53 generators and 57 relations. The abelianisation $N/N' \cong \mathbb{Z}^{41} \oplus \mathbb{Z}_2^{12}$ shows that N cannot be presented with fewer generators.

In the action of Γ as a monodromy group on the flags of any quotient of \mathcal{D} , one can regard the first three generators a, b and c as describing how flags fit together to form dodecahedral cells, and d as describing how these cells meet across faces; there is a similar interpretation for the generators $\underline{a}, \ldots, \underline{d}$ of Δ , acting on quotients of \mathcal{E} . Now the epimorphism $\varphi : \Gamma \to \Delta$ sends the first three generators of Γ to those of Δ , so that under the induced isomorphism $\Gamma/K \to \Delta$, dodecahedra of \mathcal{D}/K are matched up with dodecahedra of \mathcal{E} ; however, φ sends d to \underline{e} rather than \underline{d} , so that adjacency of dodecahedra in \mathcal{D}/K corresponds to the effect of \underline{e} in identifying faces of dodecahedra in \mathcal{E} .

Now consider the action of the element \underline{e} of Δ as a monodromy permutation on a flag $\phi = (v, e, f, E)$ incident with a face f on a dodecahedron E of the 120-cell \mathcal{E} . Reading $\underline{e} = \underline{d}(\underline{abc})^5 \underline{d}(\underline{abc})^5 \underline{d}$ from left to right, it can be seen that \underline{d} sends ϕ to the adjacent flag (v, e, f, E_f) in the dodecahedron E_f which meets E across their common face f; then $(\underline{abc})^5$ sends this to the antipodal flag of E_f , on the face f' of E_f antipodal to f; applying \underline{d} again, the adjacent flag in the dodecahedron E_{2f} which meets E_f across f' is obtained; then applying $(\underline{abc})^5$ the antipodal flag of E_{2f} , on the face f'' of E_{2f} antipodal to f' is obtained; finally applying \underline{d} the adjacent flag in the dodecahedron E_{3f} which meets E_{2f} across f'' is recovered. This means that, if \underline{d} is replaced with \underline{e} as a monodromy permutation and leaving $\underline{a}, \underline{b}$ and \underline{c} unchanged, then the 120 dodecahedra E of \mathcal{E} can be reassembled with this new rule for adjacency: each face f of E is identified, not with the corresponding face f of E_f , but with the face f'' of E_{3f} , three steps rather than one around a bracelet $E, E_f, E_{2f}, E_{3f}, \ldots, E_{-f}$. Combinatorially, this is possible, but since \underline{ce} has order 5 these new identifications would require each edge to be surrounded by five dodecahedra, rather than three. Therefore spherical dodecahedra E in \mathcal{E} need to be replaced with hyperbolic dodecahedra D, so that they can have dihedral angles $2\pi/5$. Doing this, and making the corresponding identifications of each face f of D with the face f'' of D_{3f} , the tessellation \mathcal{D}/K of \mathbf{H}^3/K by 120 dodecahedra

to be replaced with hyperbolic dodecahedra D, so that they can have dihedral angles $2\pi/5$. Doing this, and making the corresponding identifications of each face f of D with the face f'' of D_{3f} , the tessellation \mathcal{D}/K of \mathbf{H}^3/K by 120 dodecahedra corresponding to the normal subgroup K of Γ is obtained. This has symmetry group $\Gamma/K \cong \Delta$, and its quotient by the center $N/K \cong C_2$ is the tessellation \mathcal{D}/N studied earlier. Alternatively, \mathcal{D}/N can be obtained directly by applying the above reassembling process to the 60 dodecahedra in the antipodal quotient of \mathcal{E} discussed in the preceding section. (This reassembling process can be seen as a 3-dimensional analogue of the operations on maps considered in dimension 2 by Wilson [Wil] and Jones and Thornton [JT]).

5.10 The tessellation \mathcal{D}/K

The tessellation \mathcal{D}/K is an unbranched double covering of \mathcal{D}/N , each dodecahedral cell in the latter lifting to two in the former. The local properties of these two tessellations are similar, but globally they differ. As in the case of \mathcal{D}/N , but with $SL_2(5)$ and Δ^+ replacing A_5 and $A_5 \times A_5$, the right-handed flags and the dodecahedra of \mathcal{D}/K can be labeled with the elements of Δ^+ and $SL_2(5)$ respectively, so that a flag $\pm(M_1, M_2)$ is in the dodecahedron labeled with the (well-defined) element $M_1M_2^{-1}$ of $SL_2(5)$. As in the case of \mathcal{E} , the diagonal subgroup $H \cong A_5$ of the central product Δ^+ stabilises a dodecahedron D_1 , and its action on the 120 dodecahedra corresponds to the action of Inn $SL_2(5) \cong L_2(5) \cong A_5$ on the elements of $SL_2(5)$. In particular, the orbits of H on the dodecahedra correspond to the conjugacy classes of $SL_2(5)$, as described in § 5.9. It follows that each orbit \mathcal{O} of H on dodecahedra in \mathcal{D}/N lifts to two orbits of size $|\mathcal{O}|$ on dodecahedra in \mathcal{D}/K , and that the labeling of these dodecahedra is compatible with the lifting of elements of A_5 to $SL_2(5)$.

By considering monodromy permutations of flags, as in the case of \mathcal{D}/N , the possible adjacencies between pairs of dodecahedra D_P and D_Q labeled with matrices $P, Q \in SL_2(5)$ can be determined: starting with the neighbours of the dodecahedron D_I labeled with the identity matrix $I = I_2$, apply automorphisms to consider general dodecahedra. For instance, the element γ^{-1} of Γ^+ has image $\pm (\binom{2}{2} \binom{2}{3}, \binom{2}{3} \binom{3}{0})$ in Δ^+ , so D_I meets D_M across a common face where

$$M = \begin{pmatrix} 2 & 2 \\ 2 & 0 \end{pmatrix} \begin{pmatrix} 2 & 3 \\ 3 & 0 \end{pmatrix}^{-1} = -A^2 \in SL_2(5),$$

and it follows from this that D_P meets D_Q across a common face if and only if $P^{-1}Q$ is in the conjugacy class $-\hat{\mathcal{C}}_5$ of $SL_2(5)$ containing $-A^2$. In fact, whereas the element $(abc)^5 d$ of Γ^+ , which sends flags one step around a bracelet of dodecahedra, has an image of order 5 in $A_5 \times A_5$, its image $\pm (-A^{-1}, A)$ in Δ^+ has order 10: its fifth power is mapped to the central involution $\pm(-I_2, I_2)$, which is not in the image of Γ_0 , so each bracelet of five dodecahedra in \mathcal{D}/N lifts to a bracelet of ten dodecahedra in \mathcal{D}/K . (This is obtained, as in the preceding section, from a bracelet of ten dodecahedra in \mathcal{E} , with dodecahedra three steps apart in \mathcal{E} giving rise to adjacent dodecahedra in \mathcal{D}/N .) On the other hand, the element $(bcd)^5(abc)^5$ of Γ^+ , which sends flags one step around a necklace of dodecahedra, has an image $\pm (B^{-1}, B)$ of order 3 in Δ^+ ; it follows from this that D_P and D_Q meet antipodaly across a vertex if and only if $P^{-1}Q \in \hat{\mathcal{C}}_3$, and that each necklace of length 3 in \mathcal{D}/N lifts to a pair of necklaces of length 3 in \mathcal{D}/K . The dodecahedron $D_{(123)}$ in \mathcal{D}/N meets D_1 antipodaly at v_1 and also along three edges; this lifts to two dodecahedra in \mathcal{D}/K , namely D_B meeting D_I antipodaly at a vertex, and D_{-B} meeting it along three edges. Thus D_P and D_Q meet along an edge if and only if $P^{-1}Q \in -\hat{\mathcal{C}}_3$, so it is no longer possible in \mathcal{D}/K for a pair of dodecahedra to meet in both ways, as happened in \mathcal{D}/N .

5.11 Quotients of Γ^+ isomorphic to $L_2(19)$

After $L_2(4) \cong L_2(5) \cong A_5$, the next smallest epimorphic image of Γ^+ is $L_2(19)$ (in Theorem 3.1 of § 3.6 it was shown that no epimorphisms onto $L_2(7)$, $L_2(8)$, $L_2(9)$, $L_2(11)$, $L_2(13)$ or $L_2(17)$). As is shown in section § 3.4, there are three normal subgroups with quotient $L_2(19)$. In this section, these three subgroups K_i will be considered in more detail, together with their associated tessellations \mathbb{H}^3/K_i .

5.11.1 The three normal quotients and their extensions

Recall in § 3.4 of Chapter 3 that maps $\Gamma^+ \to L_2(q)$ were characterised by matrices with entries in terms of e, f, t, u, w and x, where $e^2 + f^2 = -2 - t, u, t = \frac{-1 \pm \sqrt{5}}{2}$. When p = 19 take u = t, e = 4 and f = -t, giving epimorphisms $\Gamma^+ \to L = L_2(19)$ defined by

$$\alpha \mapsto \begin{pmatrix} t & 12 \\ 11 & 0 \end{pmatrix}, \quad \beta \mapsto \begin{pmatrix} 12 & 0 \\ -t & 8 \end{pmatrix}, \quad \gamma \mapsto \begin{pmatrix} w & x \\ y & z \end{pmatrix},$$

where $t = t_i = 4, -5$ for i = 1, 2. Here y = x, w = ((e-1)u - 2fx)/2e = (5x-2)t, z = u - w = (3 - 5x)t and x is a root of $(e^2 + f^2)x^2 + ftx + e^2 - \frac{1}{4}(e^2 - 1)t^2 = 0,$ that is, $(t+2)x^2 - (t-1)x + (t+2) = 0$ or equivalently $x^2 + (2 - 3t)x + 1 = 0.$ When t = -5 this quadratic equation becomes $x^2 - 2x + 1 = 0$, with a single root x = 1, so there is an epimorphism $\theta_0 : \Gamma^+ \to L$ defined by

$$\gamma \mapsto \left(\begin{array}{cc} 4 & 1 \\ 1 & 10 \end{array}\right).$$

When t = 4 we have $x^2 + 9x + 1 = 0$, which has two roots x = -4, -5, so there are epimorphisms $\theta_1, \theta_2 : \Gamma^+ \to L$ defined by

$$\gamma \mapsto \left(\begin{array}{cc} 7 & -4 \\ -4 & -3 \end{array}\right) \quad \text{or} \quad \left(\begin{array}{cc} 6 & -5 \\ -5 & -2 \end{array}\right)$$

respectively. Therefore, there are three normal subgroups $K_i = \ker \theta_i$ of Γ^+ with $\Gamma^+/K_i \cong L_2(19)$ for i = 0, 1, 2.

The epimorphism $\theta_0: \Gamma^+ \to L$ can be extended to an epimorphism $\theta_0: \Gamma \to L$

by sending

$$\begin{array}{c} a \mapsto \left(\begin{array}{cc} 1 & 6 \\ 6 & -1 \end{array} \right), \quad b \mapsto \left(\begin{array}{cc} 4 & -7 \\ -3 & -4 \end{array} \right), \quad c \mapsto \left(\begin{array}{cc} -6 & 1 \\ 1 & 6 \end{array} \right), \quad d \mapsto \left(\begin{array}{cc} -4 & 4 \\ -9 & 4 \end{array} \right),$$

so the kernel is a normal subgroup N_0 of Γ with $\Gamma/N_0 \cong L$. This shows that $K_0 = N_0^+ = N_0 \cap \Gamma^+$ is normal in Γ , with $\Gamma/K_0 \cong L \times C_2$. This is because N_0 contains the conjugacy classes with representatives $(abc)^5$ and $(bcd)^5$. These representatives are central in $\langle a, b, c \rangle$ and $\langle b, c, d \rangle$, respectively. The two epimorphisms θ_1 and θ_2 , on the other hand, cannot be extended to Γ : when t = 4 the only possible image of b in $PGL_2(19)$, inverting the images of α and β , is $\binom{2}{7} \binom{2}{17}$, but then $d = b\beta\gamma$ does not map to an involution; thus K_1 and K_2 are not normal in Γ . Letting d act on the generators α , β and γ_1 by conjugation,

If N is any normal subgroup of Γ with quotient L, then N^+ is a normal subgroup of Γ with $\Gamma/N^+ \cong L \times C_2$ and $\Gamma^+/N^+ \cong L$. This last isomorphism implies that $N^+ = K_i$ for i = 0, 1 or 2, and since K_1 and K_2 are not normal in Γ , then $N^+ = K_0$. Thus N contains K_0 with index 2, so it maps onto a normal subgroup of order 2 in $\Gamma/K_0 \cong L \times C_2$; the only such subgroup is the direct factor $N_0/K_0 \cong C_2$, so N is its inverse image N_0 in Γ . Thus N_0 is the only normal subgroup of Γ with quotient L. Using GAP, a presentation for N_0 and K_0 was recovered. The abelianisations $N_0/N'_0 \cong \mathbb{Z}_2^{57}$ and $K_0/K'_0 \cong \mathbb{Z}^{53}$ were also computed. These give the first homology of the associated manifolds \mathbb{H}^3/N_0 and \mathbb{H}^3/K_0 , respectively.

The involution $\binom{9}{9} \binom{8}{10}$ in $PGL_2(19) \setminus L$ induces an automorphism of L transposing the images of a and d, and also those of b and c, so K_0 is normal in Ω , with $\Omega/K_0 \cong PGL_2(19) \times C_2$. This is the isometry group of the manifold \mathbb{H}^3/K_0 , with orientation-preserving subgroup $\Omega^+/K_0 \cong PGL_2(19)$; the tessellation $\mathcal{K}_0 = \mathcal{D}/K_0$ which it carries is reflexible and self-dual, with symmetry group $\Gamma/K_0 \cong L \times C_2$ and orientation-preserving subgroup $\Gamma^+/K_0 \cong L$. As already noted, the direct factor C_2 is generated by the common image of the involutions $(abc)^5$ and $(bcd)^5$ in $\Gamma \setminus \Gamma^+$, which applies the antipodal symmetry to each dodecahedral cell; factoring this out, it can be seen that the orbifold \mathbb{H}^3/N_0 is tessellated by the antipodal quotients of these dodecahedra.

The effect of the half-turn $r \in \Omega^+$ on Γ^+ is to transpose α and γ^{-1} , and to invert β . Conjugation by the element $\binom{10 \ 10}{7 \ 9} \in L$ has the same effect on the images
of these elements under θ_1 , so K_1 is normalised by r and is thus normal in Ω^+ with $\Omega^+/K_1 \cong L \times C_2$. This is the isometry group of the manifold \mathbb{H}^3/K_1 , consisting entirely of orientation-preserving transformations. The subgroup $\Gamma^+/K_1 \cong L$ is the symmetry group of the tessellation $\mathcal{K}_1 = \mathcal{D}/K_1$, which is chiral and self-dual. The same conclusions apply to K_2 and its quotient manifold and tessellation, with the element $\binom{7}{1} \binom{7}{12} \in L$ imitating r in this case. Since K_1 and K_2 are conjugate under elements of $\Gamma \setminus \Gamma^+$, their corresponding quotient manifolds and tessellations are mirror images of each other.

5.11.2 Factorisation in $L_2(19)$

The following method allows the structure of each of the three tessellations \mathcal{K}_i associated with $L = L_2(19)$ to be determined, by identifying which dodecahedron contains the flag labeled by any given element of L. The 57 dodecahedra may be identified with the cosets gG in L of the subgroup $G \cong A_5$, which is the stabiliser of one dodecahedron D_0 in the monodromy representation of L on flags. Let Bdenote a Borel subgroup of L, that is, the normaliser $C_{19} : C_9$ of a Sylow 19subgroup of L. Then $|B \cap G|$ divides gcd(171, 60) = 3, so B has an orbit of length divisible by |B|/3 = 57 on the dodecahedra, and therefore acts transitively on them. Thus L = BG and $C := B \cap G \cong C_3$, so each element $h \in L$ has the form h = bg where $b \in B$ and $g \in G$. A second factorisation h = b'g', for $b' \in B$ and $g' \in G$, exists if and only if b' = bc and $g' = c^{-1}g$ where $c \in C$. Therefore the cosets bC of C in B can be used to represent the cosets bG of G in L, and hence to represent the dodecahedra in \mathcal{K}_i . (Unfortunately, there is no subgroup of L which acts regularly on the dodecahedra and whose elements could therefore provide coset representatives for G in L.)

In order to locate flags within specific dodecahedra, the coset bG of G which contains a given element $h \in L$ needs to be determined. That is, h needs to be factorised as a product h = bg where $b \in B$ and $g \in G$. B can be assumed to be the stabiliser of ∞ in the natural representation of L as a group of Möbius transformations of the projective line $P^1(19) = F_{19} \cup \{\infty\}$, so that g maps ∞ to the same point $p \in P^1(19)$ as h does. A list of 20 elements $g_p \in G$, each sending ∞ to a different point p, is constructed (these will be coset representatives for C in G, so there are three arbitrary choices for each p). Given any $h \in L$, write it as the image under $\theta_i : \Gamma^+ \to L$ of a word w in the generators α, β and γ of Γ^+ , and compute the image p of ∞ under h by composing the Möbius transformations corresponding to the generators appearing in w. Then take g to be the corresponding element $g_p \in G$ from the list, and write $b = hg^{-1} \in B$, so that h = bg as required, and the flag corresponding to h lies in the dodecahedron corresponding to the coset bC.

5.11.3 A note on the above method

Factorisation in $L_2(19)$ has no nice extension to a more general case $L_2(q)$. For q = 19 the subgroup $B = N_{L_2(19)}(C_{19})$, where C_{19} is the abelian subgroup stabilising the point ∞ in the action of $L_2(19)$ on the projective line $\mathbb{F}_{19} \cup \infty$, was used. This subgroup has index q+1 = 20 in $L_2(19)$. For all but finitely many q, q+1 > |G| = 60, so the decomposition L = BG cannot arise for large q.

5.11.4 Structure of the quotient tessellations for $L_2(19)$

In each of the cases i = 0, 1 or 2, the group $L = L_2(19)$ permutes the $|L|/|A_5| = 57$ dodecahedra in $\mathcal{K}_i = \mathcal{D}/\mathcal{K}_i$ transitively, with each dodecahedron stabilised by a subgroup $G \cong A_5$. In order to understand how these dodecahedra are joined together, more information is needed on how the stabiliser of one of the dodecahedra permutes the others. Equivalently, the suborbits for the action of L on the cosets of a subgroup $G \cong A_5$ are needed. To do this, a result due to Jones and Zvonkin [JZ] is used:

Lemma 5.11.1 (Jones-Zvonkin) If \mathcal{G} and \mathcal{H} are finite conjugacy classes of subgroups of a group S, then the number ν of groups $G \in \mathcal{G}$ containing a particular $H \in \mathcal{H}$ is given by

$$\nu = |\tilde{H}: H| \sum_{i=1}^{t} \frac{1}{m_i}$$

with $\tilde{H} = N_S(H)$, and the subgroups $H \in \mathcal{H}$ contained in G form t conjugacy classes under the action of $\tilde{G} = N_S(G)$, with the groups H in the *i*-th class satisfying $|N_{\tilde{G}}(H) : H| = m_i$. The proof of the Lemma in [JZ] shows that the ν groups $G \in \mathcal{G}$ containing H form t orbits of lengths $|\tilde{H} : H|/m_i$ under the action of \tilde{H} . In particular, if all subgroups $H \in \mathcal{H}$ contained in G are conjugate in \tilde{G} , then t = 1 and $\nu = |\tilde{H} : H|/m$ where $m = m_1 = |N_{\tilde{G}}(H) : H|$.

This can be applied to find the suborbit-lengths for the action of S = L on the cosets of a subgroup $G \cong A_5$. There are two conjugacy classes of such subgroups, which fuse in Aut $G = PGL_2(19)$, so \mathcal{G} can be taken to be either of these classes. They are maximal subgroups, so $\tilde{G} = G$ and the action of S can be identified with its action by conjugation on \mathcal{G} . This action, which is primitive, has degree $|S:G| = \frac{1}{2}q(q^2 - 1)/60 = 3420/60 = 57$.

Let H to be an arbitrary subgroup of G. Then the number of points fixed by H is the number of conjugates of G in S that contain H. By comparing this with the corresponding numbers for supergroups of H in G, the number of suborbits of G with H as a point stabiliser can be determined.

- If *H* ≅ *A*₄ then *H* = *H*, *t* = 1 and *m*₁ = 1, so the Lemma gives *ν* = 1, that is, *H* is contained in no other conjugate of *G*. Thus there is no suborbit with *A*₄ stabilisers. This argument applies for all *q* ≡ ±3 mod (8); however, if *q* ≡ ±1 mod (8) then *H* ≅ *S*₄ with *t* = 1 and *m*₁ = 1 so *ν* = 2 giving a suborbit of length |*G* : *H*| = 5.
- 2. If $H \cong D_5$ then $\tilde{H} \cong D_{10}$, t = 1 and $m_1 = 1$, so the Lemma gives $\nu = 2$, and hence there is one suborbit of length 6 with D_5 stabilisers, G acting as $L_2(5)$, or equivalently on antipodal pairs of faces of a dodecahedron. This applies for all $q \equiv \pm 1 \mod (20)$, since $\tilde{H} = D_{10}$; if $q \equiv \pm 11 \mod (20)$ then $\tilde{H} = H$ so there are no such suborbits.
- 3. If $H \cong D_3$ then $\tilde{H} = H$, t = 1 and $m_1 = 1$, so the Lemma gives $\nu = 1$, that is, there is no suborbit with stabiliser D_3 . This applies for all $q \equiv \pm 5 \mod (12)$, since $\tilde{H} = H$; if $q \equiv \pm 1 \mod (12)$ then $\tilde{H} \cong D_6$ giving $\nu = 2$, so there is a suborbit of length 10 with D_3 stabilisers.
- 4. If $H \cong C_5$ then $\tilde{H} = D_{10}$, t = 1 and $m_1 = 2$ since $N_{\tilde{G}}(H) = N_G(H) \cong D_5$, so $\nu = 2$. The two conjugates of G containing H also contain its supergroup D_5 , as shown in (2), so there is no suborbit with C_5 stabilisers.

- 5. If $H \cong V_4$ then $\tilde{H} \cong A_4$, t = 1 and $m_1 = 3$ since $N_{\tilde{G}}(H) = N_G(H) \cong A_4$, so $\nu = 1$ and there is no suborbit with V_4 stabilisers. More generally, if $q \equiv \pm 1$ mod 8 then $N_S(H) = S_4$ so $\nu = 2$, so there is get 1 suborbit of length 15. If $q \equiv \pm 3 \mod 8$ then $N_S(H) = A_4$ and $\nu = 1$, there is no suborbit.
- 6. If H ≅ C₃ then H̃ ≅ D₉, t = 1 and m₁ = 2 since N_{G̃}(H) = N_G(H) ≅ D₃, so ν = 3. The two conjugates other than G itself are not fixed by any supergroup of H in G, so there is one suborbit of length 20 with C₃ stabilisers; G acts as on the vertices of a dodecahedron, each stabiliser fixing two points. More generally, H̃ = D_{q∓1} depending on whether q ≡ ±1 mod 3. Also, N_{G̃}(H) = N_G(H) = D₃, t = 1 so m₁ = 2. Then ν = q∓1/6. If q ≡ ±5 mod 12 then there is one suborbit of length 2 with C₃ stabilizers. If q ≡ ±1 mod 12 then there are two suborbits of length 10 with D₃ stabilizers.
- 7. If $H \cong C_2$ then $\tilde{H} \cong D_{10}$, t = 1 and $m_1 = 2$ since $N_{\tilde{G}}(H) = N_G(H) \cong V_4$, so $\nu = 5$, giving four conjugates other than G fixed by H. Now there are 15 subgroups $H \cong C_2$ in G, each fixing four points other than G, giving $15 \times 4 = 60$ fixed points. There are six subgroups D_5 in G, each containing five subgroups C_2 , so there are $6 \times 5 = 30$ pairs $D_5 > C_2$; since the 15 subgroups C_2 are all conjugate in G, each is contained in 30/15 = 2 subgroups D_5 of G. It follows that H fixes two points in the orbit of length 6 given in (2), so it has 4 - 2 = 2 points other than G outside that orbit. There are no other supergroups of H in G which can arise as stabilisers for these points, so Ghas an orbit of length |G : H| = 30 with H acting as the stabiliser of two points; this is equivalent to the action of G on edges of a dodecahedron.
- 8. There cannot be II = 1 since then |G:II| = 60 > 57, so G could not have a suborbit of length 60.

5.11.5 The three tessellations

To summarise, it was shown that G splits \mathcal{G} into orbits of lengths 1, 6, 20 and 30 with stabilisers G, D_5, C_3 and C_2 . Since 1+6+20+30=57 is the number of cells in the tessellations, this forms a complete list. This will now be interpreted in terms of the tessellations $\mathcal{K} = \mathcal{K}_i = \mathcal{D}/K_i$ for i = 0, 1, 2. General features common to

all three tessellations will be described. More detailed features which distinguish between the three tessellations require a more precise examination of how L is generated as a quotient of Γ^+ in each case.

If D_0 is a dodecahedral cell in \mathcal{K} , then there are at most 12 dodecahedra which meet D_0 across common faces; this set of dodecahedra must be invariant under the stabiliser $G \cong A_5$ of D_0 in L, so it is a union of orbits of G. None of these dodecahedra can be equal to D_0 , otherwise regularity and connectedness would imply that all dodecahedra in \mathcal{K} would be equal to each other. Since there is only one orbit of G of length between 1 and 20, it follows that these dodecahedra form the orbit of length 6, each meeting D_0 twice; since this pair of common faces must be invariant under the stabiliser D_5 of the neighbouring dodecahedron, they must be an antipodal pair $\pm f$. Denoting this neighbour by $D_{\pm f}$, and applying the same argument to it, it can be seen that $D_{\pm f}$ meets D_0 across its faces $\pm f$, so these two dodecahedra form a bracelet of length 2. Note that this gives a $3\pi/5$ "twist" to the tessellation.

The tessellation from θ_0

For a dodecahedron D_0 , let e and -e be a pair of antipodal edges stabilized setwise by an element of $\langle \alpha, \beta \rangle$ of order two. Let ϕ be a flag in D_0 sharing an edge with e, and let v be the vertex common to them both. Under the epimorphism θ_0 the antipodal flag at the vertex v, $\phi \gamma^2 \beta^2 \gamma^2 \beta^2 \gamma \beta^2$, is also the flag $\phi \alpha^3 \beta \alpha^4 \beta \alpha^2$ at the antipodal vertex -v of D_0 . So the dodecahedron antipodal to D_0 across v is D_0 and v coincides with -v under the identification, as illustrated in Figure 5.13.

If D_i , D_j meet D_0 only along an edge of D_0 , then they also meet D_0 across the edge -e as shown in Figure 5.14. This contributes 30 dodecahedra to the tessellation, forming a single orbit of G with C_2 stabilizers. So far 37 dodecahedra have been counted, leaving 20 as yet unaccounted for. These form a single orbit of dodecahedra disjoint from D_0 . These 20 dodecahedra come in ten pairs $\{D_v, D_{-v}\}$, each pair fixed setwise by a rotation of order 3 fixing two antipodal vertices of D_0 . Let v, -v be such a pair of vertices in D_0 , and let D_f meet D_0 along a face containing v. Then D_f has two additional faces f_1 and f_2 which also contain v. These faces are not shared with D_0 . Let e_{f_i} be the edge in f_i opposite v. Then D_v





Figure 5.13: Vertex identification for \mathcal{M}_0

Figure 5.14: Edge pairings for \mathcal{M}_0

meets one of these edges and D_{-v} meets the other. Of the 6 dodecahedra meeting D_v across a face, three of them are dodecahedra sharing an edge with D_0 and three are vertex dodecahedra disjoint from D_0 . Since the automorphisms β preserves the properties of the tessellation, this forces the three vertex dodecahedra to share a common vertex with D_v .

The tessellations from θ_1 and θ_2

For the epimorphisms θ_1 and θ_2 , D_0 meets another dodecahedron D_v at v so that D_0 and D_v correspond to antipodal faces of the icosahedral vertex figure $I = \{3, 5\}$ at v. Since D_0 has 20 vertices v these dodecahedra D_v form an orbit of G of length 20, with D_v and D_{-v} having the same stabiliser C_3 where -v denotes the vertex of D_0 antipodal to v. Since there are no further orbits with this stabiliser, D_v and D_{-v} meet at a common vertex, antipodal in each to v and -v respectively, so that D_0 , D_v and D_{-v} form a necklace of length 3.

This accounts for 1 + 6 + 20 = 27 of the 57 dodecahedra, so 30 dodecahedra remain, forming a single orbit of G with C_2 stabilisers. Now D_0 has 30 edges e, each surrounded by five dodecahedra; one of these is D_0 , and two are of the form D_f for the two faces f of D_0 incident with e, so two others meet D_0 along e. This gives $30 \times 2 = 60$ such incidences, two for each edge, so each of the 30 remaining dodecahedra meets D_0 along a pair of edges, which must be invariant under the stabiliser $H \cong C_2$. These incidences can be used to distinguish between the two tessellations: if p is a Petrie path on D_0 containing both e and -e, and let D_{e_1} and D_{e_2} be the two dodecahedra incident with e. Then the D_{e_i} meet D_0 along the edges adjacent to -e in p. In the case of θ_1 , p is a right-left-right Petrie path; in the case of θ_2 , p is a left-right-left Petrie path. Figure 5.15 shows the edge identification for the manifold \mathcal{M}_2 associated with the kernel of the map $\theta_2 : \Gamma^+ \to L_2(19)$.

This accounts for all the dodecahedra, although not all incidences with D_0 : at each of the 20 vertices v, the vertex figure is an icosahedron whose 20 faces represent the corners of dodecahedra meeting D_0 at v: one of these dodecahedra is D_0 , three further dodecahedra have the form D_f for faces f incident with v, while six dodecahedra meet D_0 along the three edges incident with v. A further dodecahedron is D_v . There are nine remaining dodecahedra unaccounted for. Of these nine, six are antipodal dodecahedra, three of which are antipodal to the vertices bounding the edges containing v. Figure 5.16 illustrates this. The three dodecahedra meeting D_v across a face and which are not antipodal to D_0 meet D_0 across three of the six edges left after deleting all edges bounding a face containing v or -v. The other three edges meet dodecahedra which are face adjacent to D_{-v} . The three edges are also permuted regularly by the automorphism action of β .

Edge pairing for M₂



Figure 5.15: Edge pairings for \mathcal{M}_2



Figure 5.16: Vertex identification for \mathcal{M}_1

Chapter 6

Manifolds from the other groups

In this chapter minimal index torsion free normal subgroups and minimal index torsion free subgroups of the eight Lannér groups T_1 , T_2 , T_3 , T_5 , T_6 , T_7 , T_8 and T_9 , introduced in Table 2.3 of § 2.2, will be constructed. Manifolds arising from the group T_2 have been extensively studied by Jones and Mednykh [JM]. Their main results are included here for completeness only. The group T_4 is studied extensively in Chapter 5.

Let $\mathcal{L}(G)$ be the lowest common multiple of the orders of all finite subgroups of an arbitrary group G and let $\mathcal{M}(G)$ be the minimum index of a torsion-free subgroup of G. $\mathcal{L}(G)$ will divide $\mathcal{M}(G)$ since, for any torsion free subgroup H of G, the action of a representative g of a conjugacy class $\mathcal{C}(g)$ of torsion elements on the cosets of H in G fails to fix any coset. Hence |g| divides the index of H in G. It is known that, for Fuchsian groups G, $\mathcal{M}(G)/\mathcal{L}(G)$ is either 1 or 2 ([EEK]). Jones and Reid [JR] have demonstrated that, for Kleinian groups G there is no global bound on the ratio $\mathcal{M}(G)/\mathcal{L}(G)$ by constructing a sequence of groups Γ_k with $\mathcal{M}(\Gamma_k)/\mathcal{L}(\Gamma_k) > k$, for any $k \in \mathbb{N}$. It is interesting to note that, by contrast, several of the orientation preserving subgroups Γ^+ of the Lannér groups Γ have a minimal index torsion free normal subgroup with index equal to $\mathcal{L}(\Gamma^+)$.

6.1 $\Gamma = T_1[2, 2, 3; 3, 5, 2]$

Recall that $T_1[2, 2, 3; 3, 5, 2]$ is the Coxeter group with Coxeter diagram



where a, b, c and d are Coxeter generators of Γ . Let $\alpha = ab$, $\beta = bc$ and $\gamma = cd$. Then the orientation preserving subgroup Γ^+ has the following presentation:

$$\Gamma^{+} = \langle \alpha, \beta, \gamma \mid \alpha^{5} = \beta^{3} = \gamma^{2} = (\alpha\beta)^{2} = (\beta\gamma)^{3} = (\alpha\beta\gamma)^{2} = 1 \rangle$$

6.1.1 Torsion in Γ and in Γ^+

Conjugacy classes of torsion elements in Γ are listed in Table 6.1. The restriction of Table 6.1 to representatives of conjugacy classes lying in Γ^+ is given in Table 6.2

Order	Representative	
Order 2	$a, ac, ad, cd, acd, (abc)^5, (abd)^5$	
Order 3	bc	
Order 4	cbd	
Order 5	$ab, (ab)^2$	
Order 6	$c(ab)^2, d(ab)^2$	
Order 10	$abc, (abc)^3, abd, (abd)^3$	

Table 6.1: Conjugacy class representatives for elements of finite order in Γ

Order	Representative
Order 2	ac, ad, cd
Order 3	bc
Order 5	$ab, (ab)^2$

Table 6.2: Conjugacy class representatives for elements of finite order in Γ^+

6.1.2 Minimal index torsion free subgroups

Lemma 6.1.1 Γ^+ has a unique normal subgroup K with $\Gamma^+/K \cong A_5$. The kernel of the map contains the conjugacy class $\mathcal{C}(\gamma)$ of torsion elements of Γ^+ .

Proof: Adjoining the relation $\gamma = 1$ to the presentation for Γ^+ gives a group with presentation $\langle \alpha, \beta, \gamma | \alpha^5 = \beta^3 = (\alpha\beta)^2 = 1 \rangle$, so $\Gamma^+ / \langle \langle \gamma \rangle \rangle_{\Gamma^+} \cong A_5$. Let $\alpha \mapsto (1, 3, 5, 2, 4) = \overline{\alpha}$ and $\beta \mapsto (1, 2, 3) = \overline{\beta}$. Since A_5 contains no elements g of order 2 which satisfy the conditions $(\overline{\beta}g)^3 = 1$ and $(\overline{\alpha}\overline{\beta}g)^2 = 1$, this completes the proof.

Corollary 6.1.1 There exists a minimal index torsion free subgroup of Γ^+ of index 120.

Proof: This result was obtained computationally. The unique normal subgroup K of index 60 in Γ^+ has 2047 conjugacy classes of subgroups of index 2. 1024 of the class representatives avoid the conjugacy class $C(\gamma)$.

The representatives of subgroups avoiding $C(\gamma)$ can be further partitioned by their abelian invariants - this partitions the set of representatives into 10 classes, four of which have positive first Betti number. The normalisers of the subgroups of K have indices in $\{6, 10, 12, 15, 20, 30, 60\}$. Of the 1024 conjugacy class representatives avoiding $C(\gamma)$, 8 have normalizer of index 6 in Γ^+ .

Lemma 6.1.2 There are no normal subgroups N of Γ^+ with $\Gamma^+/N \cong A_6$

Proof: Let $\langle \alpha, \beta \rangle \hookrightarrow A_5 \subset A_6$ be an embedding of A_5 in A_6 . Up to automorphisms of A_6 , this embedding is unique. Let $\alpha \mapsto (1,3,5,2,4)$, $\beta \mapsto (1,2,3)$ and $\gamma \mapsto g$, so that α and β generate $A_5 < A_6$. A_6 has 45 elements g of order 2, 30 of which lie in $A_6 \setminus A_5$. Of these 30 elements, only (1,2)(3,6), (1,3)(2,6) and (1,6)(2,3)satisfy |(1,2,3)g| = 3. However, $|\alpha\beta(1,2)(3,6)| = 3$, $|\alpha\beta((1,3)(2,6)| = 3$ and $|\alpha\beta(1,6)(2,3)| = 4$. So $|\alpha\beta g| \neq 2$, and the proof is complete.

Lemma 6.1.3

- 1) There are precisely two torsion free normal subgroups K_1 , K_2 whose factor group $\Gamma^+/K_i \equiv L_2(11)$.
- 2) These are the smallest index torsion free normal subgroups.
- 3) Their intersection has index 660^2 in Γ^+ and is normal in $\Omega = \langle \Gamma, \tau \rangle$, the order 2 extension of Γ in $Isom(\mathbb{H}^3)$.

Proof:

1) In chapter 4 sections § 4.1.1 and § 4.2 it was shown that homomorphisms $\Gamma^+ \rightarrow L_2(11)$ can be characterised by maps of the form

$$\alpha \mapsto \frac{1}{2} \begin{pmatrix} t-f & e+1\\ -1+e & t+f \end{pmatrix}, \beta \mapsto \frac{1}{2} \begin{pmatrix} e+1 & t+f\\ f-t & 1-e \end{pmatrix} \text{ and } \gamma \mapsto \begin{pmatrix} w & x\\ x & -w \end{pmatrix}$$

where $t = \frac{-1\pm\sqrt{5}}{2}$, $e^2 + f^2 = -2 - t$ and ew = u - fx, where $u = \pm 1$ and $\sqrt{5} = 4$. If $t = \frac{-1+\sqrt{5}}{2}$ then $t = 7 \mod 11$, so $-2 - t = -9 \equiv 2 \mod 11$. Setting e = f = 1 gives $e^2 + f^2 = 2$, as required. Then x satisfies the quadratic equation

$$2x^2 - 2ux + 2 = 0 (6.1)$$

which has discriminant $D = 4e^2(1+t) = 4(8) = (4\sqrt{2})^2$. Now $\sqrt{2} \notin \mathbb{F}_{11}$, so equation 6.1 has no solution over \mathbb{F}_{11} . Letting $t = \frac{-1-\sqrt{5}}{2}$, gives t = 3, so $-2 - t - = -5 \equiv 6 \mod 11$. Writing e = 4, f = 1, so $e^2 = 5$, $f^2 = 1$ and $e^2 + f^2 = 6$, x satisfies the polynomial

$$6x^2 - 2ux + 6 = 0 \tag{6.2}$$

which has discriminant $D = 4.5(1+3) = 5^2 \mod 11$. Solving equation 6.2 gives $x = 2u \pm 5$, So for each value of u two solutions for x are obtained, giving four solutions in total.

Case I: u = 1: Then $x = 2 \pm 5 = 7$ or $-3 \equiv 8 \mod 11$ and 4w = 1 - x

• If x = 7 then 4w = 1 - 7 = -6 so $w \equiv 4 \mod 11$.

• If x = 8 then 4w = 1 - 8 = -7 so $w \equiv 1 \mod 11$.

Therefore

$$\gamma \mapsto \tilde{\gamma} \in \left\{ \left(\begin{array}{cc} 4 & 7 \\ 7 & -4 \end{array} \right), \left(\begin{array}{cc} 1 & 8 \\ 8 & -1 \end{array} \right) \right\}$$

Case II: u = -1: Then $x = -2 \pm 5 = 3$ or $-7 \equiv 4 \mod 11$ and 4w = -1 - x

- If x = 3 then 4w = -1 3 = -4 so we get $w \equiv -1 \mod 11$.
- If x = 4 then 4w = -1 4 = -5 so we get $w \equiv 7 \mod 11$.

Therefore

$$\gamma \mapsto \tilde{\gamma} \in \left\{ \left(\begin{array}{cc} -1 & 3 \\ 3 & 1 \end{array} \right), \left(\begin{array}{cc} 7 & 4 \\ 4 & -7 \end{array} \right) \right\}$$

Observe that the solutions in the case u = -1 are simply those from u = 1 multiplied by -1 and so are equivalent in $L_2(11)$. Now $\operatorname{Aut}(L_2(11)) = PGL_2(11)$. The action of the outer automorphism on $L_2(11)$ conjugates the two conjugacy classes of A_5 in $L_2(11)$. Since A_5 is maximal in $L_2(q)$, there is no automorphism of $L_2(11)$ which fixes A_5 and conjugates the two solutions of $\overline{\gamma}$. Thus there are two distinct kernels K_1 and K_2 , corresponding to the distinct epimorphisms $\theta_1, \theta_2: \Gamma^+ \to L_2(11)$, as required.

- 2) Since Γ^+ is perfect, it suffices to consider only non-abelian simple quotients. The factor group of any torsion-free normal subgroup must contain faithful images of the special subgroups $A_5 \cong \langle \alpha, \beta \rangle$ and $S_4 \cong \langle \beta, \gamma \rangle$. From lemma 6.1.1 it has already been shown that there are no torsion free normal subgroups whose factor group is A_5 . Neither is there a torsion free normal subgroup whose factor group is $L_2(7)$ or $L_2(8)$, since neither $L_2(7)$ nor $L_2(8)$ contain elements of order 5. Lemma 6.1.2 proves that there is no map to A_6 , and the result follows.
- 3) Take the product map $\theta = (\theta_1, \theta_2) : \Gamma^+ \twoheadrightarrow L_2(11) \times L_2(11)$. Then letting $\ker(\theta) = K, \Gamma^+/K \cong L_2(11) \times L_2(11)$ and $K_1 \cap K_2 \leq K$. By composition with the projections onto each factor, it is clear that $K < K_i$. Thus $K \leq K_1 \cap K_2$, so $K = K_1 \cap K_2$. The K_i are not normal in Γ : If they were then there would exist some element $g \in L_2(11)$ of order 2 such that $g\alpha, g\alpha\beta$ and $g\alpha\beta\gamma$ all have order 2. Referring back to section § 4.2.2 of Chapter 4, this becomes the condition $e^2 = \frac{-1}{4}$ and $f^2 = \frac{-7}{4} t$, or $e^2 = 8$ and $f^2 = 9$. Since 8 is not

a square in \mathbb{F}_{11} , the K_i are not normal in Γ . So for any element $g \in \Gamma \setminus \Gamma^+$ $K_i^g \neq K_i$. Since K_1 and K_2 are the unique normal subgroups of Γ^+ with factor group $L_2(11)$, $K_i^g = K_j$, for $\{i, j\} \in \{1, 2\}$ and $g \in \Gamma \setminus \Gamma^+$. Then $K_1 \cap K_2$ is normal in Γ .

Write $\Omega = \langle \Gamma, \tau \rangle$, where τ is the graph automorphism of the Coxeter diagram for Γ that fixes the generators a and b and transposes the generators c and d. Since Γ^+ is normal in Ω , and K is the unique normal subgroup of Γ^+ with factor group $L_2(11) \times L_2(11)$, K is characteristic in Γ^+ and hence normal in Ω .

6.2 $\Gamma = T_2[2, 2, 3; 2, 5, 3]$

The small manifolds arising from this group are studied in detail by Jones and Mednykh in [JM]. Their conclusions are stated in this section for completeness only. Let Γ^+ be the orientation preserving subgroup of Γ and let Ω be the normalizer of Γ in Isom (\mathbb{H}^3). Then Ω is a split extension of Γ by a cyclic group of order 2. Ω has an orientation preserving subgroup Ω^+ . Both results rely on the conjecture that the minimal volume hyperbolic 3-orbifold is \mathbb{H}^3/Ω , where $\Omega = \langle \Gamma, \tau \rangle$ is an index two extension of T_2 . Work done by Gehring and Martin [GM1], [GM2] suggests that this group is the Kleinian group with smallest co-volume.

Theorem 6.1 There is a unique torsion-free normal subgroup K_0 of Ω of least index (= 2640) in Ω . This is a subgroup of Γ^+ , with $\Gamma^+/K_0 \cong L_2(11)$ and $\Omega/K_0 \cong PGL_2(11) \times C_2$. The corresponding manifold $\mathcal{M}_0 = \mathcal{H}^3/K_0$ is the smallest hyperbolic 3-manifold with a large isometry group. It is orientable, with Iso $\mathcal{M}_0 \cong PGL_2(11) \times C_2$, and is tessellated by 11 hyperbolic icosahedra, each meeting the 10 others across two antipodal faces. The isometries preserving this tessellation form a subgroup $\Gamma/K_0 \cong L_2(11) \times C_2$ of index 2 in Iso \mathcal{M}_0 , while the remaining isometries transform the tessellation to its dual. The first integer homology group $H_1(\mathcal{M}_0)$ of \mathcal{M}_0 is isomorphic to \mathbf{Z}^{10} .

Theorem 6.2 There are two torsion-free normal subgroups K_1 and K_2 of Ω^+ of

least index (= 720) in Ω^+ . They are subgroups of Γ^+ which are conjugate in Ω , with $\Gamma^+/K_i \cong L_2(9) \cong A_6$ and $\Omega^+/K_i \cong PGL_2(9)$ for i = 1, 2. The manifolds $\mathcal{M}_i = \mathcal{H}^3/K_i$ are the smallest orientable 3-manifolds with large orientation-preserving isometry groups. They form a chiral pair, with Iso $\mathcal{M}_i = \text{Iso}^+ \mathcal{M}_i \cong PGL_2(9)$, and they are tessellated by six hyperbolic icosahedra, each meeting the other five across a set of four faces with tetrahedral symmetry. The isometries preserving this tessellation form a subgroup $\Gamma^+/K_i \cong L_2(9)$ of index 2 in Iso \mathcal{M}_i , while the remaining isometries transform the tessellation to its dual. The first integer homology group $H_1(\mathcal{M}_i)$ of \mathcal{M}_i is isomorphic to \mathbf{Z}_3^6 .

6.3 $\Gamma = T_3[2, 2, 4; 2, 3, 5]$

In this section minimal index torsion free normal subgroups and minimal index torsion free subgroups of Γ^+ will be constructed. Some details on the construction of the associated manifolds will be included. The famous tessellation of \mathbb{H}^3 by rightangled dodecahedra is a consequence of this group. Figure 6.1, from the archives of the Geometry Center [Geom], illustrates the structure of this tessellation. This figure clearly demonstrates the octahedral vertex figure as well as the dodecahedral cells. It will be shown that the smallest manifold arising from the action of Γ^+ on \mathbb{H}^3 is tessellated by 2 such dodecahedral cells, while the smallest manifold with maximal symmetry is tessellated by 22 dodecahedra.



Figure 6.1: Tessellation of \mathbb{H}^3 by right-angled dodecahedra

Recall that $T_3[2, 2, 4; 2, 3, 5]$ is the Coxeter group with Coxeter diagram

where a, b, c and d are Coxeter generators of Γ . Let $\alpha = ab$, $\beta = bc$ and $\gamma = cd$. Then the orientation preserving subgroup Γ^+ has the following presentation:

$$\Gamma^+ = \langle \alpha, \beta, \gamma \mid \alpha^5 = \beta^3 = \gamma^4 = (\alpha \beta)^2 = (\beta \gamma)^2 = (\alpha \beta \gamma)^2 = 1 \rangle$$

6.3.1 Torsion in Γ and in Γ^+

Representatives of the conjugacy classes of torsion elements in Γ are given in Table 6.3 The restriction of Table 6.3 to representatives of classes lying in Γ^+ is given in Table 6.4.

Order	Representative
Order 2	$a, d, ac, ad, (cd)^2, a(cd)^2, (bcd)^3, (abc)^5$
Order 3	bc
Order 4	$cd, b(cd)^2$
Order 5	$ab, (ab)^2$
Order 6	$c(ab)^2, bcd$
Order 10	$abc, (abc)^3, abd, (ab)^2d$

Table 6.3: Conjugacy class representatives for elements of finite order in Γ

Order	Representative
Order 2	$ac, ad, (cd)^2$
Order 3	bc
Order 4	cd
Order 5	$ab, (ab)^2$

Table 6.4: Conjugacy class representatives for elements of finite order in Γ^+

 Γ^+ has 647 conjugacy classes of subgroups of index less than or equal to 60. A representative for each class was determined using the program Lowx [Lx]. For each representative, the action of Γ^+ on its cosets was calculated and the results are listed in Appendix A.3. Among the 647 induced permutation groups, 20 of the induced actions have kernels which fail to avoid torsion:

- The induced action on the representative with index 2.
- The induced action on the representative with index 5 which yields an epimorphism onto A_5 .
- The induced action on the representative with index 6 which yields an epimorphism onto A_5 .
- Both actions induced by the representatives with index 10 which yield an epimorphism onto A_5 .
- Three of the actions induced by the nine conjugacy classes of subgroups of index 12. These correspond to the induced permutation representations A₅ and A₅ × C₂ as listed in Appendix A.3
- The induced action on the representative with index 15 which yields an epimorphism onto A_5 .
- Three of the actions induced by the thirteen conjugacy classes of subgroups of index 20. These correspond to the induced permutation representations A_5 and $A_5 \times C_2$ as listed in Appendix A.3.
- One of the actions induced by the twenty four conjugacy classes of subgroups of index 24. This corresponds to the induced permutation representation $A_5 \times C_2$ listed in Appendix A.3.
- Three of the actions induced by the nineteen conjugacy class of subgroups of index 30. These correspond to the induced permutation representations A₅ and A₅ × C₂ as listed in Appendix A.3.
- One of the actions induced by the forty eight conjugacy class of subgroups of index 40, giving $A_5 \times C_2$
- Three of the actions induced by the three hundred and twenty three conjugacy class of subgroups of index 30.

6.3.2 Minimal index torsion free subgroups

Lemma 6.3.1 Γ^+ has no torsion free normal subgroup N whose factor group Γ^+/N is isomorphic to $L_2(7)$ or $L_2(8)$.

Proof: The proof is immediate since neither $L_2(7)$ nor $L_2(8)$ contain elements of order 5.

Lemma 6.3.2 Γ^+ has no torsion free normal subgroup N whose factor group $\Gamma'N$ is isomorphic to A_5 or S_5 .

Proof: The case for A_5 is trivial, as A_5 contains no elements of order 4. Suppose $\Gamma^+ \twoheadrightarrow S_5$. Let $\alpha = ab$, $\beta = bc$ and $\gamma = cd$. Since $\langle \alpha, \beta \rangle \cong A_5$, we can let $\alpha \mapsto (1,3,5,2,4)$ and $\beta \mapsto (1,2,3)$ and suppose that $\gamma \mapsto g$. Now $|\{g \in S_5| |g| = 4\}| = 30$. Of these elements, six of them satisfy $|\beta\gamma| = 2$: These are (1,3,4,2), (1,3,5,2), (1,3,2,4), (1,5,3,2), (1,3,2,5) and (1,4,3,2). Then for $g \in \{(1,3,4,2), (1,3,5,2), (1,3,2,4), (1,5,3,2)\}$, $|\alpha\beta g| = 4$ while if $g \in \{(1,3,2,5), (1,4,3,2)\}$ then $|\alpha\beta g| = 6$.

Lemma 6.3.3 Let Γ^+ be the orientation preserving subgroup of $\Gamma = T_3[2, 2, 4; 2, 3, 5]$. Then

- Γ⁺ has a unique normal subgroup K with factor group Γ⁺/K ≅ A₅ which has 2 classes of torsion free subgroups H_i of index 120 in Γ. These are the minimal index torsion free subgroups of Γ⁺ and they have abelianisation H_i/[H_i, H_i] ≅ Z ⊕ Z₂⁵.
- 2) For each minimal index torsion free subgroup H_i of Γ^+ , the associated manifold $\mathcal{M}_i(K, 1) = \mathbb{H}^3/H_i$ is a 2-fold branched cover of \mathbb{H}^3/K . Further, there exist manifolds $\mathcal{M}_i(K, n)$ tessellated by 2n right-angled dodecahedra for any $n \in \mathbb{N}$.

Proof: 1): Write $\alpha \mapsto (1,3,5,2,4)$ and $\beta \mapsto (1,2,3)$. Let $\gamma \mapsto g$ where g is some element of A_5 such that the order of g divides 4. Since A_5 contains no elements of order 4, then $|g| \in \{1,2\}$. If $\gamma \mapsto ()$, the identity element of A_5 , then $|\beta\gamma| = 2$ forces $\beta = ()$, a contradiction. So |g| = 2. There is a unique element in $(2,4)(3,5) \in A_5$ satisfying $|\beta g| = 2$ and $|\alpha\beta g| = 2$. Let $\theta : \Gamma^+ \to A_5$ be the induced map constructed and write $K = \text{Ker}(\theta)$. Then by construction K contains the conjugacy class $\mathcal{C}(\gamma^2)$ of 2-torsion in Γ^+ . GAP was used to investigate index 2 subgroups of K. There are 7 classes of subgroups in K, $\{H_i|1 \leq i \leq 7\}$ of index 2, 5 of which avoid the conjugacy class $\mathcal{C}(\gamma^2)$ of torsion. Taking the abelianisation of each of these 5 subgroups yields 2 subgroups, H_1 and H_2 , with positive first Betti number. Their abelianization, $H_i/[H_i, H_i]$, was calculated using GAP and in each case this was found to be $\mathbb{Z} \oplus \mathbb{Z}_2^5$. The fact that Γ^+ is not perfect (it maps onto C_2) means that it remains to consider the possibility that Γ^+ maps onto some small solvable group with torsion free kernel. Now, in any homomorphism from Γ or Γ^+ to a solvable group, α and β must go to the identity since they generate a perfect A_5 . Setting $\alpha = \beta = 1$ in the presentation of Γ^+ gives C_2 . Hence the derived series terminates with a subgroup of index 2 in Γ^+ . So, any quotient of Γ^+ with torsion free kernel has the form $H : C_2$ or H, where H is perfect, Hence, in looking for small index normal subgroups is it sufficient to consider simple groups H or groups $H : C_2$, Since it has already been established that Γ^+ has no torsion free kernel with quotient A_5 , the result follows.

2): Let $\mathcal{M}_i(K, 1) = \mathbb{H}^3/H_i$ be the manifold associated to each H_i . Then, by construction, each $\mathcal{M}_i(K, 1)$ is a 2-fold branched cover of the orbifold \mathbb{H}^3/K . Since \mathbb{H}^3 carries a tessellation by right angled dodecahedra, it follows that $\mathcal{M}_i(K, 1)$ is tessellated by 2 right angled dodecahedra. By composing with the map $\mathbb{Z} \to \mathbb{Z}/n\mathbb{Z}$, characteristic subgroups K_n of index n in each H_i can be obtained. The associated manifold $\mathcal{M}_i(K, n)$ is a regular n-fold cover of the manifold $\mathcal{M}_i = \mathbb{H}^3/H_i$. Hence the manifolds $\mathcal{M}_i(K, n)$ are tesselated by 2n right angled dodecahedra.

6.3.3 Minimal index torsion free normal subgroups

Lemma 6.3.4 Γ^+ has no torsion free normal subgroups N with $\Gamma^+/N \cong A_6$ or S_6 .

Proof: Suppose there exists a torsion free normal subgroup. $A_6 \cong L_2(9)$ has 2 conjugacy classes of subgroups isomorphic to A_5 . Representatives for each class are $\langle (1,3,5,2,4), (1,2,3) \rangle$ and $\langle (1,2,3,4,6), (1,5,6)(2,4,3) \rangle$. In the first case (when $\langle \alpha, \beta \rangle \rightarrow \langle (1,3,5,2,4), (1,2,3) \rangle$) there are nine elements g of order 4 in A_6 satisfying $|\beta g| = 2$, listed below:

 $\{(1,3,4,2)(5,6), (1,3,5,2)(4,6), (1,3,6,2)(4,5), (1,3,2,4)(5,6), (1,3,2,5)(4,6), (1,3,2,6)(4,5), (1,4,3,2)(5,6), (1,5,3,2)(4,6), (1,6,3,2)(4,5)\}$

For each g in the above list, the word $\alpha\beta g$ has order 5, 5, 3, 5, 4, 4, 4, 5, 4, respectively. In the second case, when $\langle \alpha, \beta \rangle \rightarrow \langle (1, 2, 3, 4, 6), (1, 5, 6)(2, 4, 3) \rangle$, again there are nine elements g of order 4 in A_6 satisfying $|\beta g| = 2$. They are:

$$g \in \{(1,2)(3,4,6,5), (1,2,3,5)(4,6), (1,3)(2,6,5,4), (1,3,4,5)(2,6), (1,4)(2,3,6,5), (1,4,2,5)(3,6), (1,6,4,2)(3,5), (1,6,2,3)(4,5), (1,6,3,4)(2,5)\}$$

For each g in the above list, the word $\alpha\beta g$ has order 5, 4, 5, 4, 3, 5, 4, 4, 5, respectively. Therefore there are no torsion free normal subgroups N whose factor group $\Gamma^+/N \cong A_6$. The case for $\Gamma^+ \not\rightarrow S_6$ is similar.

Lemma 6.3.5 Γ^+ has precisely two distinct torsion free normal subgroups N_1 and N_2 with $\Gamma^+/N_i \cong PGL_2(11)$.

Proof: From Chapter 4, § 4.4.1, $t = \frac{-1 \pm \sqrt{5}}{2}$ and $\sqrt{5} = 4$ in \mathbb{F}_{11} . If $t = \frac{-1 + \sqrt{5}}{2}$ then t = 7, and if $t = \frac{-1 - \sqrt{5}}{2}$ then t = 3.

Case I: t = 7: Then $e^2 + f^2 = -2 - 7 = -9 \equiv 2 \mod 11$, giving

$$\alpha \mapsto \begin{pmatrix} 3 & 1 \\ 0 & 4 \end{pmatrix}$$
 and $\beta \mapsto \begin{pmatrix} 1 & 4 \\ -3 & 0 \end{pmatrix}$. Write $\gamma \mapsto \begin{pmatrix} a & b \\ c & d \end{pmatrix}$.

Next, $|\alpha\beta\gamma| = 2$ gives the identity

$$(\alpha\beta\gamma)^2 \mapsto \begin{pmatrix} c^2 - ad & cd - bd \\ ab - ac & b^2 - ad \end{pmatrix} = Id \in PGL_2(11)$$

and this forces b = c. Then

$$(\beta\gamma)^2 \mapsto \left(\begin{array}{cc} (a+4b)^2 + 8a(b+4d) & (b+4d)(a+b) \\ 8a(a+b) & 9b^2 + 8a(b+5d) \end{array}\right)$$

gives the equations 8a(a + b) = 0 and (b + 4d)(a + b) = 0. If a = 0 then either b = 0, in which case γ is the zero matrix, or b = -4d, giving

$$(\beta\gamma)^2 \mapsto d^2 \left(\begin{array}{cc} 3 & 0\\ 0 & 1 \end{array}\right)$$

as the identity matrix in $PGL_2(11)$, a contradiction. If $a \neq 0$, then b = -a and so

$$\gamma \mapsto \left(\begin{array}{cc} -b & b \\ b & d \end{array}\right)$$

Then

$$\gamma^{4} \mapsto \begin{pmatrix} 8b^{4} + 4b^{2} + 1 & -8b^{4} - 8b^{2} - 4 - \frac{1}{b^{2}} \\ -8b^{4} - 8b^{2} - 4 - \frac{1}{b^{2}} & 8b^{4} + 12b^{2} + 9 + \frac{1}{b^{4}} + \frac{4}{b^{2}} \end{pmatrix}$$

Writing $8b^4 = -8b^2 - 4 - \frac{1}{b^2}$ gives

$$\gamma^4 \mapsto \begin{pmatrix} -4b^2 - 3 - \frac{1}{b^2} & 0\\ 0 & 4b^2 + 5 + \frac{3}{b^2} + \frac{1}{b^4} \end{pmatrix}$$

Since these must be equal for this to be an element of $PGL_2(11)$, b must satisfy the equation

$$8b^6 + 8b^4 + 4b^2 + 1 = 0$$

Solving for $8x^3 + 8x^2 + 4x + 1 = 0$ give x a cube root of unity, so b is a sixth root of unity. Since 6 does not divide 10 = 11 - 1, there are no maps $\Gamma^+ \rightarrow PGL_2(11)$ with trace $(\alpha) = 7$.

Case II: t = 3: Then $e^2 + f^2 = -5 \equiv 6 \mod 11$, and set

$$\alpha \mapsto \begin{pmatrix} -2 & 1 \\ 0 & 5 \end{pmatrix}$$
 and $\beta \mapsto \begin{pmatrix} 1 & 5 \\ 2 & 0 \end{pmatrix}$. Write $\gamma \mapsto \begin{pmatrix} a & b \\ c & d \end{pmatrix}$.

Then

$$(\alpha\beta\gamma)^2 \mapsto \left(\begin{array}{cc} c^2 - ad & cd - bc\\ ab - ac & b^2 - ad \end{array}\right) = Id \in PGL_2(11)$$

forces b = c. Then

$$(\beta\gamma)^2 \mapsto \left(\begin{array}{cc} (a+5b)^2 + 2a(b+5d) & (b+5d)(a+7b) \\ 2a(a+7b) & 4b^2 + 2a(b+5d) \end{array}\right)$$

This gives 2a(a + 7b) = 0 and (b + 5d)(a + 7b) = 0. If a = 0 then either b = 0, forcing γ to be the zero matrix, or b = -5d, and this gives

$$(\beta\gamma)^2 \mapsto b^2 \left(\begin{array}{cc} 4 & 0\\ 0 & 3 \end{array}\right)$$

which cannot be the identity matrix in $PGL_2(11)$. So $a \neq 0$. Thus a = -7b, giving

$$(\beta\gamma)^2 \mapsto 4b(3b-d) \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix}$$

Thus

$$\gamma \mapsto \left(\begin{array}{cc} -7b & b \\ b & d \end{array}\right)$$

subject to $\gamma^4 = Id \in PGL_2(11)$. Now

$$\gamma^4 \mapsto g = \left(\begin{array}{ccc} 8b^4 + 8db^3 + d^2b^2 & 6b^4 + 7db^3 + 4d^2b^2 + d^3b \\ 6b^4 + 7db^3 + 4d^2b^2 + d^3b & 6b^4 + 8db^3 + 3d^2b^2 + d^4 \end{array}\right)$$

so $b(6b^3 + 7db^2 + 4d^2b + d^3) = 0$. Since *b* cannot be 0, then $6b^3 + 7db^2 + 4d^2b + d^3 = 0$. This cubic equation has three solutions, $b \in \{5d, 6d, 8d\}$. If b = 8d then $det(g) = d^2$, so writing d = 1 gives $g \in PSL_2(11)$. For b = 5d, $det(g) = 6d^2$ and for b = 6d, $det(g) = 10d^2$. Because neither 6 nor 10 are squares in \mathbb{F}_{11} , neither matrix lies in $SL_2(11)$. They must then lie in $PGL_2(11)$. The images of α and β generate an A_5 subgroup of $PGL_2(11)$. Now A_5 is not maximal in $PGL_2(11)$ - it is contained in a maximal $L_2(11)$ subgroup. However, A_5 is maximal in $L_2(11)$, and the images of γ are, by construction, not contained in $L_2(11)$. The group elements in $PGL_2(11) \setminus L_2(11)$ acts by interchanging the two conjugacy classes of A_5 in $L_2(11)$, so they do not fix any given A_5 . Since $PGL_2(11)$ has no outer automorphisms, the two maps $\Gamma^+ \twoheadrightarrow PGL_2(11)$ constructed are inequivalent, and their kernels are two distinct normal subgroups K_1 and K_2 in Γ^+ .

Since Γ^+ is not perfect, it is still possible that Γ^+ maps onto a small solvable group. However, in any map from Γ^+ to a solvable group, α and β must map to the identity, since they generate a perfect group. Then $\Gamma^+/\{\alpha, \beta\} \cong C_2$, so the derived series terminates with a subgroup of index 2 in Γ^+ . Hence any quotient of Γ^+ has the form $G \triangleleft C_2$, where G is a perfect group. Hence for small index subgroups, it suffices to consider either G simple or extensions of G by a cyclic group of order 2. If the extension has the form $G \times C_2$, then it follows that Γ^+ also maps onto G. Hence the following result is obtained:

Theorem 6.3 The smallest index torsion free normal subgroups of Γ^+ arise from epimorphisms $\Gamma^+ \rightarrow PGL_2(11)$. There are two such maps $\theta_1, \theta_2 : \Gamma^+ \rightarrow PGL_2(11)$ corresponding to two distinct normal subgroups K_1 and K_2 in Γ^+ .

Using Lowx and GAP, generating sets for K_1 and K_2 were found. Both subgroups arise from the action of Γ^+ on conjugacy classes of subgroups of index 12 and they are conjugate in Γ . The abelianisation of each K_i , $K_i/[K_i, K_i] \cong \mathbb{Z}_2^{10} \oplus \mathbb{Z}_{11}^3$ and their intersection is a normal subgroup of Γ^+ of index 660*1320. The elements generating K_1 and K_2 as subgroups of Γ^+ are listed in Table 6.5

Elements of the generating	Elements of the generating
set for K_1	set for K_2
bcbdcbabcdcbdabcdcba	bcdcbabcdcdbabcdbcbu
cbabcdcbabdcdcbabcdb	cbdcbabcdcdbabcdcbab
cdcbabcdcbdabcdbcbab	dcbabcdcbadbcdcbabcb
acbdcbabcdcdbabcdcbaba	acbabcdcbabdcdcbabcdba
adcbabcdcbadbcdcbabcba	acdcbabcdcbdabcdbcbaba
babacbabadcbdabcdababcda	babacbdcbabcdcdbabcdcaba
babacdcbabcdcdbabcdbcaba	babadcbabadcdbabcdababca
babdcbabadcbdabcdababcba	babcbabadcbadbcdababcdba
bacbabadcbabdcdababcdaba	bacbabdcbabadcdbabcdabca
bacbabcdcbabdcdcbabcdbab	badcbabadcbadbcdababcaba
bacbadcbabadcdbabcdbabca	badcbabcdcbadbcdcbabcbab
bacdcbabcdcbdabcdbcababa	bcbabadcbabadcdbabcdbaba
bdcbabadcbabdcdababcbaba	

Table 6.5: The generating sets for K_1 and K_2 in Theorem 6.3

6.3.4 Manifold structure

From the previous section, α , β and γ have torsion-free representations in $PGL_2(11)$ given by $\alpha \mapsto \tilde{\alpha} = \begin{pmatrix} -2 & 1 \\ 0 & 5 \end{pmatrix} \beta \mapsto \tilde{\beta} = \begin{pmatrix} 1 & 5 \\ 2 & 0 \end{pmatrix}$ and $\gamma \mapsto \tilde{\gamma} \in \begin{cases} \begin{pmatrix} -2 & 5 \\ 5 & 1 \end{pmatrix}, \begin{pmatrix} 2 & 6 \\ 6 & 1 \end{pmatrix} \end{cases}$. Corresponding to the two subgroups K_1 and K_2 are two hyperbolic manifolds $\mathcal{M}_i \cong \mathbb{H}^3/K_i$. The structure of these manifolds can be determined using the group structure of $PGL_2(11)$. In $PGL_2(11)$, $\langle \tilde{\alpha}, \tilde{\beta} \rangle \cong A_5$, and there exists a complementary dihedral subgroup D_{11} of order 22 which can be used to index the cosets of A_5 in $PGL_2(11)$. Write

$$D_{11} \cong \left\langle \left(\begin{array}{cc} 1 & 2 \\ 0 & 1 \end{array} \right), \left(\begin{array}{cc} -1 & 5 \\ 0 & 1 \end{array} \right) \right\rangle$$

The following method can be used to determine the structure of each of the two tessellations \mathcal{K}_i associated with $L = PGL_2(11)$, by identifying which dodecahedron contains the flag labeled by any given element of L. The 22 dodecahedra may be identified with the cosets gG in L of the subgroup $G \cong A_5$, which is the stabiliser of one dodecahedron D_0 in the monodromy representation of L on flags. Let B be the Borel subgroup of L, that is, a subgroup of the normaliser $C_{11}: C_{10}$ of a Sylow 11-subgroup of L. Let $\tilde{B} \subset B$ denote the unique subgroup D_{11} of B. Then $|\tilde{B} \cap G| = 1$ so \tilde{B} has an orbit of length 22 on the dodecahedra and therefore acts transitively on them. As a result $L = \tilde{B}G$ and each element $h \in L$ has a unique form h = bg, where $b \in \tilde{B}$ and $g \in G$. The flag corresponding to h lies in the dodecahedron corresponding to the coset bG, and position of the flag in the dodecahedron is determined by g.

6.3.5 Other manifolds and minimal index torsion free subgroups

The next smallest index torsion free normal subgroup arises from the action of Γ^+ on a subgroup of index 20. This group was found by studying the induced actions of Γ^+ on low index subgroups. Table A.3 of Appendix A summarises the induced actions. The induced permutation representation is $G = A_5 \ltimes H$, where H is elementary abelian of order 32. The resulting manifold is tessellated by 32 dodecahedra and the fundamental group in this case has abelianisation $\mathbb{Z}^{15} \oplus \mathbb{Z}_2^2$. As a consequence a tower of manifolds with maximal isometry group is recovered. Each manifold in this tower consists of $32n^6$ dodecahedra and is a regular n^6 -fold cover of the manifold \mathbb{H}^3/G . Additionally, GAP was used to verify that G has subgroups H of index 120 in G whose preimages correspond to subgroups of index 120 in Γ^+ that avoid all conjugacy classes of torsion.

6.4 $\Gamma = T_4[2, 2, 5; 2, 3, 5]$

The manifolds arising from torsion free subgroups and torsion free normal subgroups of the group T_5^+ have already been studied in detail in chapter 5.

6.5 $\Gamma = T_5[2, 3, 3; 2, 4, 3]$

Recall that $T_5[2, 3, 3; 2, 4, 3]$ is the Coxeter group with Coxeter diagram



where a, b, c and d are Coxeter generators of Γ . Let $\alpha = ab$, $\beta = bc$ and $\gamma = cd$. Then the orientation preserving subgroup Γ^+ has the following presentation:

$$\Gamma^+ = \langle \alpha, \beta, \gamma \mid \alpha^4 = \beta^3 = \gamma^3 = (\alpha \beta)^2 = (\beta \gamma)^2 = (\alpha \beta \gamma)^3 = 1 \rangle$$

6.5.1 Torsion in Γ and in Γ^+

Representations of the conjugacy classes of torsion elements in Γ are given in Table 6.6 The restriction of this list to representatives of classes lying in Γ^+ is given in Table 6.7

Order	Representative	
Order 2	$a, ac, bd, (ab)^2, (abc)^3, (bad)^3$	
Order 3	bc	
Order 4	$ab, c(ab)^2, d(ab)^2, adc, bcd$	
Order 6	abc	

Table 6.6: Conjugacy class representatives for elements of finite order in Γ

Order	Representative
Order 2	$ac, bd, (ab)^2$
Order 3	bc
Order 4	ab

Table 6.7: Conjugacy class representatives for elements of finite order in Γ^+

Lemma 6.5.1 There are no torsion free normal subgroups of Γ^+ whose factor group is S_4 or S_5 .

Proof: Any torsion free normal subgroup of Γ^+ must contain $\langle \alpha, \beta \rangle \equiv S_4$. First, suppose that there is such a subgroup N with $\Gamma^+/N \simeq S_4$. Then it can be assumed that $\alpha \mapsto (1, 2, 3, 4)$. Therefore $\beta \in \{(1, 3, 2), (1, 4, 2), (1, 4, 3), (2, 4, 3)\}$ and $\alpha\beta \in \{(3, 4), (2, 3), (1, 2), (1, 4)\}$. Now $|\gamma| = 3$ and $|\alpha\beta\gamma| = 3$ can therefore both be written as a product of an even number of transpositions. But $\alpha\beta$ has an odd number of transpositions, a contradiction! Since S_5 has a unique conjugacy class of elements of order 4, the result extends to S_5 .

In particular, since A_5 has no element of order 4, the following result is immediate:

Proposition 6.5.1 Γ^+ has no torsion free normal subgroup whose factor group is isomorphic to A_5 .

In Theorem 4.8 in chapter 4 it was shown that there is a unique normal subgroup K in Γ^+ whose factor group is $L_2(7)$. Since it is unique, it is characteristic. Γ^+ is normal in the normalizer Ω of Γ . Consequently K is normal in Ω

Note: In Table A.5 of Appendix A it can be seen that there are two conjugacy classes of subgroups of index 8 in Γ^+ . Calculating their cores gives the same normal subgroup K of Γ^+ in each case, verifying Theorem 4.8.

The group Γ^+ is perfect: Abelianizing the presentation for Γ^+ , the identities $\alpha^4 = \beta^3 = \gamma^3 = 1$, $(\alpha\beta)^2 = \alpha^2\beta^2 = 1$, $(\beta\gamma)^2 = \beta^2\gamma^2 = 1$ and $(\alpha\beta\gamma)^3 = \alpha^3 = 1$ are obtained. Hence $\alpha = 1$, so $\beta^2 = \beta^3 = 1$. Therefore $\beta = 1$, so $\gamma^2 = \gamma^3 = 1$, and so Γ^+ is a perfect group. Combining the above results, the following theorem can be stated:

Theorem 6.4 Let K be a smallest index torsion free normal subgroup of Γ^+ . Then K is unique, K has index 168 in Γ^+ and $\Gamma^+/K \cong L_2(7)$. Furthermore, K is normal in Ω , the normalizer of Γ in $Isom(\mathbb{H}^3)$.

Using GAP, a presentation for K was obtained. K is the fundamental group of the associated hyperbolic manifold $\mathcal{M} = \mathbb{H}^3/K$. A presentation for $\pi_1(\mathcal{M})$ is given here:

$$\begin{aligned} \pi_1(M) &= \langle F_1, F_2, F_3, F_4, F_5, F_6 \mid \\ &F_6F_1^{-1}F_6^{-1}F_3F_5^{-1}F_2F_5F_2^{-1}F_3^{-1}F_1 = 1, \\ &F_4^{-1}F_1^{-1}F_5^{-1}F_1^{-1}F_3F_2F_1F_5F_3^{-1}F_2^{-1}F_4F_1 = 1, \\ &F_3F_6^{-1}F_2^{-1}F_4F_3F_2F_3^{-1}F_4^{-1}F_1^{-1}F_2^{-1}F_3^{-1}F_1F_6F_2 = 1, \\ &F_1F_6^{-1}F_1^{-1}F_4^{-1}F_2F_6F_5^{-1}F_1^{-1}F_2^{-1}F_6^{-1}F_5F_1F_4F_6 = 1, \\ &F_4^{-1}F_1^{-1}F_6^{-1}F_3F_5^{-1}F_4F_3F_2F_3^{-1}F_4^{-1}F_5F_3^{-1}F_2^{-1}F_4F_1F_6 = 1, \\ &F_1^{-1}F_3F_2F_1F_4F_3F_5^{-1}F_2^{-1}F_3^{-1}F_4^{-1}F_5F_3^{-1}F_6F_1F_4F_1^{-1}F_6^{-1} = 1 \end{aligned}$$

The manifold constructed differs from the manifolds constructed using the groups T_1 , T_2 , T_3 and T_4 . Those manifolds had as fundamental regions hyperbolic dodecahedra, and the manifolds had a straightforward decomposition into dodecahedra identified across faces. The manifolds arising from the groups T_5 to T_9 have a more complicated structure. The associated fundamental regions are pleated (or creased or pinched) across their faces. Figures 6.2 and 6.3 illustrate the difference between a standard cube and a pleated cube:



Figure 6.2: Standard hyperbolic cube

Figure 6.3: Pinched hyperbolic cube

The added creases across faces of a cube give a "pinched point" at the midpoint of each edge. Visualization of the resulting manifolds is correspondingly harder. For example, the group stabilizing the vertex C in the center of a face is, in the case of T_5 , an octahedral group! The abelianisation of $K = \pi_1(\mathcal{M})$ is free abelian on six generators: $\pi_1(\mathcal{M})/[\pi_{\infty}(\mathcal{M}), \pi_{\infty}(\mathcal{M})] = \mathbb{Z}^6$, and as a result an infinite tower of covers of \mathcal{M} by manifolds \mathcal{M}_n consisting of 7n cubical cells can be constructed.

Note: The group $\pi_1(\mathcal{M})$ is not the torsion free subgroup of smallest index in Γ^+ . There are 5 conjugacy classes of such groups of index 24 in Γ^+ . There are four distinct normal subgroups whose factor group is $L_2(23)$. In each case the preimage under the induced epimorphism $\Gamma^+ \twoheadrightarrow L_2(11)$ of $C_{23} \ltimes C_{11} < L_2(11)$ is a torsion free subgroup of index 24. Since $C_{23} \ltimes C_{11}$ is maximal in $L_2(23)$, each preimage has trivial normalizer in Γ^+ , so the associated hyperbolic manifolds have very little symmetry. The Sylow 7-subgroup C_7 of $L_2(7)$ has index 24 and its preimage is also a torsion free subgroup. The normalizer of C_7 in $L_2(7)$ has index 8 in $L_2(7)$ and pulls back to an index 8 normalizer of H in Γ^+ . Since there are no futher conjugacy classes of subgroups of index 24 that are torsion free, this completes the list.

Let H be a Sylow 7-subgroup of $L_2(7)$. The elements of H can be used to index the 7 cosets of S_4 in $L_2(7)$, where S_4 stabilizes a cubical cell in the tessellation of \mathcal{M} . Letting the cosets correspond to cubical cells in the decomposition of \mathcal{M} , the manifold can then be reconstructed combinatorially as in the previous sections.

6.5.2 The action of $L_2(7)$ on $H_1(\mathcal{M})$

The subgroup $\Gamma^+/N_1 \cong L_2(7)$ of isometries of the manifold $\mathcal{M} = \mathbb{H}^3/K$ is generated by a rotation through $\pi/2$ about the center of a square face f_1 , induced by α , a rotation through $2\pi/3$ about a vertex v_1 adjacent to f_i , induced by β and a rotation through $2\pi/3$ about an edge adjacent to v_1 . These rotations act on $H_1(M_1)$, regarded as a 6-dimensional module over \mathbb{Z} , as the matrices

$$\alpha \mapsto \begin{pmatrix} 1 & 0 & 0 & 1 & 0 & 0 \\ 0 & -1 & 0 & -1 & -1 & -1 \\ 1 & 0 & 1 & 1 & 0 & 0 \\ -2 & 0 & 0 & -1 & 0 & 0 \\ 1 & 1 & 0 & 1 & 1 & 0 \\ 1 & 1 & 0 & 1 & 0 & 1 \end{pmatrix} \qquad \beta \mapsto \begin{pmatrix} 0 & 0 & 0 & 0 & -1 & 1 \\ 0 & 0 & 1 & 0 & 0 & -1 \\ 0 & 1 & 0 & 0 & 0 & 1 \\ -1 & 0 & 0 & 0 & 1 & -1 \\ 1 & 0 & -1 & 0 & -1 & 1 \\ 0 & 0 & -1 & -1 & -1 & 1 \end{pmatrix}$$

$$\gamma \mapsto \begin{pmatrix} 0 & 0 & 0 & -1 & 0 \\ -1 & 0 & 0 & 0 & 0 & 0 \\ -1 & -1 & -1 & -2 & -2 & 0 \\ 1 & 0 & 0 & 1 & 1 & 1 \\ 0 & 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & -1 & -1 & -1 & 0 \end{pmatrix}$$

According to the Atlas of Finite Groups [Atlas], there are 6 irreducible characters for $L_2(7)$, with degrees, 1, 3, 3, 6, 7, and 8. The values of the characters of an element of order 2 are 1, -1, -1, 2, -1 and 0, respectively. Since the element $\alpha\beta$ is an element of order 2 and has trace -2, the representation cannot be primitive, so the character splits as a sum of two irreducible characters: either as $\chi_2 + \chi_5$ or as $2\chi_2$. Since the trace of α is 2 and the trace of β is 0, the action splits as $2\chi_2$. So there exists a covering manifold \mathcal{M}_n of \mathcal{M} whose covering transformations (deck transformations) form an abelian group, and over which the isometry group extends to the group $L_2(7) \rtimes \mathbb{Z}_n^6$.

6.5.3 Computational results

 Γ^+ has 62 conjugacy classes of subgroups of index less than or equal to twenty. Letting Γ^+ act on the cosets of a representative of each conjugacy class induces a representation $\Gamma^+ \to G$, where G is some finite permutation group. Unless $G = A_5$, the induced map faithfully carries all conjugacy classes of torsion elements of Γ^+ . The smallest torsion-free normal subgroup is, as previously observed, the kernel K of a map $K = \ker\theta : \Gamma^+ \to L_2(7)$. It is interesting to note that, after A_5 , the next two smallest alternating groups that arise as the quotient of Γ^+ by some torsion free normal subgroup are A_{11} and A_{19} .

6.6 $\Gamma = T_6[2, 3, 4; 2, 3, 4]$

Recall that $T_6[2, 3, 4; 2, 4, 3]$ is the Coxeter group with Coxeter diagram



where a, b, c and d are Coxeter generators of Γ . Let $\alpha = ab$, $\beta = bc$ and $\gamma = cd$. Then the orientation preserving subgroup Γ^+ has the following presentation:

$$\Gamma^{+} = \langle \alpha, \beta, \gamma \mid \alpha^{4} = \beta^{3} = \gamma^{4} = (\alpha \beta)^{2} = (\beta \gamma)^{2} = (\alpha \beta \gamma)^{3} = 1 \rangle$$

6.6.1 Torsion in Γ and in Γ^+

Representations of the conjugacy classes of torsion elements in Γ are given in Table 6.8 The restriction of Table 6.8 to representatives of classes lying in Γ^+ is given in Table 6.9

Order	Representative
Order 2	$a, b, ac, bd, (ab)^2, (cd)^2, (abc)^3, (dab)^3, (cda)^3, (bcd)^3.$
Order 3	bc, ad.
Order 4	$ab, cd, c(ab)^2, d(ab)^2, a(cd)^2, b(cd)^2.$
Order 6	abc, dab, acd, bcd.

Table 6.8: Conjugacy class representatives for elements of finite order in Γ

ſ	Order	Representative	
	Order 2	$ac, bd, (ab)^2, (cd)^2.$	
	Order 3	bc, ad.	
	Order 4	ab, cd.	

Table 6.9: Conjugacy class representatives for elements of finite order in Γ^+

6.6.2 Maps to S_4

Lemma 6.6.1 There are precisely two distinct torsion free normal subgroups of Γ^+ whose factor group is isomorphic to S_4 .

Proof: The proof amounts to constructing two non-conjugate epimorphisms $\Gamma^+ \twoheadrightarrow S_4$. Let $\alpha \mapsto (1,3,2,4)$ and $\beta \mapsto (1,2,3)$. Then for $\gamma \in \{(1,3,2,4), (1,3,4,2)\}$ we get $|\gamma| = 4$, $|\beta\gamma| = 2$ and $|\alpha\beta\gamma| = 3$. This proves existence of the two maps. If $\gamma \mapsto (1,3,2,4)$ then the common image of $\alpha\gamma$ and of α^2 has order 2, while if $\gamma \mapsto (1,3,4,2)$ then $\alpha\gamma \mapsto (1,4,3)$ has order 3. Hence the maps are not conjugate.

As can be seen in Table A.6 in Appendix A, there are six conjugacy classes of subgroups of index 4 in Γ^+ . Letting Γ^+ act on the cosets of a representative in each case gives the following results:

- 2 kernels K_1, K_2 such that $\Gamma^+/K_i \cong S_4$ with $K_i/[K_i, K_i] \cong \mathbb{Z}_2 \oplus \mathbb{Z}_3 \oplus \mathbb{Z}_8^2$
- 4 kernels K_3, K_4, K_5, K_6 with $\Gamma^+/K_i \cong S_4$ and $K_i/[K_i, K_i] \cong \mathbb{Z}_2^5 \oplus \mathbb{Z}_3$.

The four kernels with abelianisation $\mathbb{Z}_2^5 \oplus \mathbb{Z}_3$ contain either the conjugacy class $\mathcal{C}(ab)^2$ or the conjugacy class $\mathcal{C}(cd)^2$, while the two kernels with abelianisation $\mathbb{Z}_2 \oplus \mathbb{Z}_3 \oplus \mathbb{Z}_8^2$ avoid the conjugacy classes of torsion elements and are distinct. Therefore they are the kernels of the maps constructed in Lemma 6.6.1. Their intersection is a normal subgroup of Γ^+ of index 288. Both of these kernels, K_1 and K_2 , have three conjugacy classes of subgroups of index 4 whose abelianisation has positive first Betti number. Consequently an infinite family of manifolds $M(K_i, 4n)$ can be constructed as 4n-fold covers of $M(K_i) = \mathbb{H}^3/K_i$. These manifolds are tessellated by 4n cubical cells. If the associated fundamental groups, $\pi_1(M(K_i, 4n)) < \Gamma^+$ are normal in Γ^+ , then the manifolds $M(K_i, 4n)$ will exhibit a high degree of symmetry. Presentations for the two kernels are given in Table 6.10 and Table 6.11:

$$\begin{split} K_{1} &:= \langle F_{1}, F_{2}, F_{3} | \\ F_{1}^{-1}F_{2}^{-1}F_{1}^{-1}F_{3}F_{1}^{-2}F_{2}^{-1}F_{1}^{-1}F_{2}^{-2}F_{1}F_{3}F_{2}^{-2} = 1, \\ F_{3}F_{1}^{-1}F_{3}F_{1}^{-2}F_{2}^{-1}F_{1}^{-1}F_{3}F_{1}^{-1}F_{3}^{2}F_{2}^{-1}F_{1}F_{3} = 1, \\ F_{1}^{-1}F_{3}F_{1}^{-2}F_{2}^{-1}F_{1}^{-1}F_{3}^{-1}F_{1}^{-1}F_{2}F_{3}^{-2}F_{1}^{-1}F_{2}F_{3}^{-1}F_{1}^{-1}F_{2}^{2}F_{3}^{-1}F_{1}^{-1}F_{2} = 1 \rangle \end{split}$$

Table 6.10: Presentation for K_1

As a subgroup of Γ^+ , K_1 has generators {bdca, dcba, bcdbab}. In the presentation given in Table 6.10, $F_1 = bdca$, $F_2 = dcba$ and $F_3 = bcdbab$.

$$\begin{split} K_{2} &:= \langle F_{1}, F_{2}, F_{3} | \\ F_{2}F_{1}^{-1}F_{2}F_{1}^{-1}F_{3}F_{1}^{-1}F_{2}F_{1}^{-1}F_{2}^{2}F_{1}F_{3}F_{1}F_{2} = 1, \\ F_{1}F_{2}^{-1}F_{1}F_{3}^{-2}F_{1}^{-1}F_{3}^{-1}F_{1}^{-1}F_{2}^{-1}F_{1}^{-1}F_{3}^{-1}F_{1}^{-1}F_{3}^{-2} = 1, \\ F_{2}^{-1}F_{3}^{2}F_{1}^{-1}F_{2}F_{1}^{-1}F_{3}F_{1}^{-1}F_{2}^{-1}F_{1}^{-1}F_{3}^{-1}F_{1}^{-2}F_{2}F_{1}^{-1}F_{3}F_{1}^{-1}F_{2}^{-1}F_{1}^{-1}F_{3}^{-1}F_{1}^{-1}F_{2}^{-1} = 1 \rangle \\ \text{Table 6.11: Presentation for } K_{2} \end{split}$$

As a subgroup of Γ^+ , K_2 has generators {*cbda*. *dacb*, *abcbdaba*}. In the presentation given in Table 6.11, $F_1 = cbda$, $F_2 = dacb$ and $F_3 = abcbdaba$.

6.6.3 Geometrical description of the two manifolds

From Lemma 6.6.1 there are two distinct epimorphisms $\Gamma^+ \to S_4$ with torsion-free kernel. Letting S_4 act on the right as a monodromy action on flags, manifolds $\mathcal{M} = \mathbb{H}^3/K_i$ can be constructed by identifications on faces of a pinched cube. Letting $\alpha, \gamma \mapsto (1, 3, 2, 4)$ gives identifications as shown in Figure 6.4. Letting $\gamma \mapsto (1, 3, 4, 2)$ (the image of $\beta^{-1}\alpha\beta$ under the maps constructed in Lemma 6.6.1) gives identifications as shown in Figure 6.5. Two of the identifications, red square to red square and blue square to blue square, are illustrated in these diagrams. The vertex shared by two blue squares and a red square in Figure 6.4 is the "pinched point" introduced in § 6.5.1. Similarly for the vertex shared by two red squares and a blue square in Figure 6.4.



B

Figure 6.4: Identifications for $\gamma = (1, 3, 2, 4)$

Figure 6.5: Identifications for $\gamma = (1, 3, 4, 2)$

6.6.4 Maps to $L_2(7)$

Further small index torsion free normal subgroups can be constructed by pulling back images of normal subgroups in $\Gamma^+/K_1 \cap K_2 \cong S_4 \times S_4$. It is also interesting to study the small non-abelian simple groups which arise as quotients of Γ^+ by some torsion-free normal subgroup. The smallest such group is $L_2(7)$. There are two conjugacy classes of subgroups of index 7 in Γ^+ . Their cores have index 168 in Γ^+ and they are contained in each other, so there is a unique normal subgroup K_3 satisfying $\Gamma^+/K_3 \cong L_2(7)$. The resulting manifold has $\frac{168}{24} = 7$ pinched cubical cells. Since K_3 is the unique normal subgroup of Γ^+ with quotient $L_2(7)$, it is characteristic in Γ^+ and hence it is normal in Ω , the normalizer of Γ in Isom(\mathbb{H}^3). A presentation for K_3 is given in Table 6.12. The generators of K_3 , $K_3/[K_3, K_3]$, is \mathbb{Z}^{13} .

$$\begin{split} &K = \langle F_1, F_2, F_3, F_4, F_5, F_6, F_7, F_8, F_9, F_{10}, F_{11}, F_{12}, F_{13}, w | \\ &F_4F_{12}^{-1}F_{13}F_5F_{13}^{-1}F_4^{-1}F_{12}F_5^{-1} = 1, \ F_{12}^{-1}F_{10}F_{11}^{-1}F_{10}^{-1}F_2F_{12}F_{11}F_2^{-1} = 1 \\ &F_{10}^{-1}F_{13}^{-1}F_6^{-1}F_7^{-1}F_6F_{10}F_{13}F_7 = 1, \ F_{12}F_1F_9^{-1}F_{12}^{-1}F_6^{-1}F_1^{-1}F_6F_9 = 1 \\ &F_4F_{10}^{-1}F_9F_8F_9^{-1}F_4^{-1}F_{10}F_8^{-1} = 1, \ F_{13}^{-1}F_2^{-1}F_9F_3^{-1}F_2F_9^{-1}F_{13}F_3 = 1, \\ &F_6F_{13}F_5F_3F_1^{-1}F_5^{-1}F_3^{-1}F_{13}^{-1}F_6^{-1}F_1 = 1, \ w = F_{11}F_{10}^{-1}F_4F_{13}F_3, \\ &F_3^{-1}F_{10}F_7^{-1}F_{13}^{-1}F_{10}^{-1}F_9F_8F_9^{-1}F_{13}F_3F_7F_8^{-1} = 1, \\ &F_1^{-1}F_{11}F_8^{-1}F_{12}F_1F_9^{-1}F_{12}^{-1}F_{10}F_{11}^{-1}F_{10}^{-1}F_9F_8 = 1, \\ &F_6F_{13}F_5F_2F_8^{-1}F_9^{-1}F_6^{-1}F_5^{-1}F_{13}^{-1}F_2^{-1}F_9F_8 = 1, \\ &F_6F_{13}F_5F_2F_8^{-1}F_9^{-1}F_6^{-1}F_5^{-1}F_{13}^{-1}F_2^{-1}F_9F_8 = 1, \\ &F_7F_{11}F_{10}^{-1}F_{12}F_5^{-1}F_{10}F_7^{-1}F_{13}^{-1}F_{12}^{-1}F_{13}F_5 = 1, \\ &F_10F_{11}^{-1}F_6F_{13}F_3F_4F_{11}F_{10}^{-1}F_6^{-1}F_3^{-1}F_{13}^{-1}F_{13}^{-1}F_4^{-1} = 1, \\ &F_1F_9^{-1}F_2F_{13}F_7F_{10}^{-1}F_4F_9F_1^{-1}F_{12}^{-1}F_{13}F_7^{-1}F_{13}^{-1}F_4^{-1}F_{12} = 1, \\ &F_7F_8^{-1}F_9^{-1}F_2F_{13}F_5F_1(F_7^{-1}F_5^{-1}F_{13}^{-1}F_2^{-1}F_9F_8F_1^{-1})^w = 1 \rangle \end{split}$$

Tabl	.e 6	5.12:	Presentation	for	K_{2}
Tabl	e f	512:	Presentation	for	-K
1			1 10001100101011	101	· · · ,

$F_1 = badcbabcda$	$F_2 = bcdabadcba$	$F_3 = bcdcbadcda$		
$F_4 = cdabadcbab$	$F_5 = cdcbadcdab$	$F_6 = dabadcbabc$		
$F_7 = dcbabcdaba$	$F_8 = acdcbadcdaba$	$F_9 = bacbadcdbcda$		
$F_{10} = cbadcdbcdaba$ $F_{11} = babcdacdcadcbdcdbcdaba$ $F_{12} = acbadcdbcdaba$				
$F_{13} = abcdabadcacbadcdbdabcabacbadcd$				

Table 6.13: Generators for K_3 in Γ^+

Thinking of $L_2(7)$ as the factor group $SL_2(7)/\{\pm I\}$, the induced map $\Gamma^+ \twoheadrightarrow L_2(7)$ obtained by the composition $\Gamma^+ \twoheadrightarrow \Gamma^+/K \cong L_2(7)$ can be expressed as an epimorphism $\theta: \Gamma^+ \to L_2(7)$ defined by the following identifications:

$$\alpha \mapsto \left[\left(\begin{array}{cc} 0 & -3 \\ -2 & -3 \end{array} \right) \right], \ \beta \mapsto \left[\left(\begin{array}{cc} 3 & 0 \\ 2 & -2 \end{array} \right) \right] \text{ and } \gamma \mapsto \left[\left(\begin{array}{cc} 2 & -1 \\ -3 & 2 \end{array} \right) \right]$$

6.6.5 Maps to $A_6 \cong L_2(9)$

There are 18 conjugacy classes of subgroups of index 6 in Γ^+ . Letting Γ^+ act on them induces epimorphisms onto D_6 , S_4 and A_6 . There are two conjugacy classes of subgroups of index 6 whose cores have index 360 in Γ^+ , and these cores coincide. This unique core K_4 is torsion free. The map $\Gamma^+ \to \Gamma^+/K_4 \cong A_6$ can be characterized by $\alpha \mapsto (1,3,4,2)(5,6), \beta \mapsto (1,4,3)$ and $\gamma \mapsto (1,6,3,4)(2,5)$. The group A_6 can be written as the quotient group $SL_2(9)/\{\pm I\}$. As equivalence classes of matrices, the map $\Gamma^+ \to L_2(9)$ can be expressed as

$$\alpha \mapsto \left[\begin{pmatrix} 0 & 1 \\ 2 & \sqrt{2} \end{pmatrix} \right] \quad , \quad \beta \mapsto \left[\begin{pmatrix} 0 & 2 + 2\sqrt{2} \\ 2 + 2\sqrt{2} & 1 \end{pmatrix} \right]$$

and $\gamma \mapsto \left[\begin{pmatrix} 2 + 2\sqrt{2} & 1 \\ 0 & 1 + 2\sqrt{2} \end{pmatrix} \right]$

Since K_4 is the unique normal subgroup of Γ^+ with quotient A_6 , it is characteristic in Γ^+ and hence it is normal in Ω , the normalizer of Γ in $\text{Isom}(\mathbb{H}^3)$. The following lemma summarizes these results:

Lemma 6.6.2 There is a unique torsion free normal subgroup K_4 of Γ^+ whose factor group is $A_6 \cong L_2(9)$. Furthermore, K is normal in Ω .

6.7 $\Gamma = T_7[2, 3, 3; 2, 5, 3]$

Recall that $T_7[2,3,3;2,5,3]$ is the Coxeter group with Coxeter diagram



where a, b, c and d are Coxeter generators of Γ . Let $\alpha = ab$, $\beta = bc$ and $\gamma = cd$. Then the orientation preserving subgroup Γ^+ has the following presentation:

$$\Gamma^+ = \langle \alpha, \beta, \gamma \mid \alpha^5 = \beta^3 = \gamma^3 = (\alpha \beta)^2 = (\beta \gamma)^2 = (\alpha \beta \gamma)^3 = 1 \rangle$$

6.7.1 Torsion in Γ and in Γ^+

Representatives of the conjugacy classes of torsion elements in Γ are given in Table 6.14. The restriction of this list to representatives of classes lying in Γ^+ is given in Table 6.15.

Order	Representative	
Order 2	$a, ac, bd, (abc)^5$ and $(dab)^5$	
Order 3	bc	
Order 4	adc and bcd	
Order 5	ab and $(ab)^5$	
Order 6	$c(ab)^2$ and $d(ab)^2$	
Order 10	$abc, (abc)^3, dab \text{ and } (dab)^3$	

Table 6.14: Conjugacy class representatives for elements of finite order in Γ

Order	Representative
Order 2	ac and bd
Order 3	bc
Order 5	ab and $(ab)^5$

Table 6.15: Conjugacy class representatives for elements of finite order in Γ^+

Lemma 6.7.1 Proper normal subgroups of Γ^+ are torsion free

Proof: The proof is essentially the same as Corollary 3.1.1 for the group $T_4 = [5, 3, 5]$ in chapter 3.

6.7.2 Maps to A_5

Lemma 6.7.2

- i) Γ^+ has two distinct normals subgroups K_1 , K_2 whose factor groups are A_5 .
- ii) K_1 and K_2 are conjugate in Γ .
- iii) The intersection $K_1 \cap K_2$ is normal in Ω , the full isometry group of Γ , and $\Gamma^+/K_1 \cap K_2 \cong A_5 \times A_5$.

Proof:

- i) Let $\alpha \mapsto (1,3,5,2,4), \beta \mapsto (1,2,3)$ be defined as in Chapter 3, Lemma 3.2.1. Then for $\gamma \in \{(2,3,4), (2,3,5)\}$, the remaining conditions $|\gamma| = 3, |\beta\gamma| = 2$ and $|\alpha\beta\gamma| = 3$ are satisfied. Since (1,3,5,2,4)(2,3,4) = (1,4)(3,5) and (1,3,5,2,4)(2,3,5) = (1,5,3,2,4), it is clear that the two normal subgroups K_1 and K_2 induce inequivalent epimorphisms and hence are distinct.
- ii) Let $d \in \Gamma \setminus \Gamma^+$ act by conjugation on $\{\alpha, \beta, \gamma\}$. Then $\alpha^d = (\alpha^{-1})^{\beta\gamma}$, $\beta^d = (\beta^{-1})^{\gamma}$ and $\gamma^d = \gamma^{-1}$. Therefore, conjugating by d induce the outer automorphism $\{(1, 3, 5, 2, 4), (1, 2, 3), (2, 3, 4)\} \mapsto \{(1, 3, 2, 4, 5), (1, 4, 3), (2, 4, 3)\}$. Conjugating $\{(1, 3, 2, 4, 5), (1, 4, 3), (2, 4, 3)\}$ by $g = (2, 4, 5) = \beta^2 \alpha^2 \beta^2$ gives $\{(1, 3, 5, 2, 4), (1, 2, 3), (2, 3, 5)\}$, hence the K_i are conjugate in Γ .
- iii) The intersection of K_1 and K_2 is clearly normal in Γ^+ : Since the effect of $d \in \Gamma \setminus \Gamma^+$ is to transpose K_1 with K_2 then it fixes the intersection, and as a result $K_1 \cap K_2$ is normal in Γ . Finally, the effect of the outer automorphism $\tau \in \Omega \setminus \Gamma$ is to transpose the generators a and d with b and c. The equivalent action on the generators of Γ^+ is to send α to α^{-1} , β to $\alpha\beta\gamma$ and γ to γ^{-1} . For $\alpha = (1, 3, 5, 2, 4), \beta = (1, 2, 3)$ and $\gamma = (2, 3, 4)$ the element g = (1, 2, 3) and $\gamma = (2, 3, 5)$ the element $g = (1, 4)(2, 3) \in A_5$ has the required action.
Consequently, the index 2 orientation preserving subgroup $\Omega^+ = \langle \Gamma^+, \tau \rangle$ of Ω has two normal subgroups \widetilde{K}_1 and \widetilde{K}_2 of index 60, each of whose factor group is A_5 . The groups K_i arise as index two subgroups of the \widetilde{K}_i , and the factor groups $\Omega^+/K_i \cong A_5 \times C_2$. Since both K_i are normal in $\Omega^+, K_1 \cap K_2$ is also normal in Ω^+ . Hence $K_1 \cap K_2$ is normal in Ω . It remains to show that $\Gamma^+/K_1 \cap K_2 \cong A_5 \times A_5$. Let $\phi : \Gamma^+ \twoheadrightarrow A_5 \times A_5$ be defined by the maps

$$\begin{array}{rcl} \alpha & \mapsto & (& (1,3,5,2,4), & (1,3,5,2,4) &) \\ \beta & \mapsto & (& (1,2,3), & (1,2,3) &) \\ \gamma & \mapsto & (& (2,3,4), & (2,3,5) &) \end{array}$$

Then, by composition with the restriction to each direct product, $\operatorname{Ker}(\phi)$ is a normal subgroup of both K_1 and K_2 , and $\operatorname{Ker}(\phi) \subseteq K_1 \cap K_2$. Since $\operatorname{Ker}(\phi)$ has index 60 in both K_1 and K_2 , $\operatorname{Ker}(\phi) = K_1 \cap K_2$.

6.7.3 Presentations for K_i

A generating set for K_1 as a subgroup of Γ^+ is $\{babdabca, (bcda)^2, d(abc)^2a, (dabc)^2, a(cdab)^2a\}$. More formally, K_1 can be given the abstract presentation listed in Table 6.16.

Table 6.16: Presentation for the group K_1 of Lemma 6.7.2

Table 6.17 describes each generator of K_1 as given in Table 6.16 in terms of the generators of Γ^+ .

Generator	Representative
of K_1	in Γ^+
F_1	babdabca
F_2	b da b ca b a
F_3	$(cba)^2 dbca$
F_4	$abadbc(babc)^3$
F_5	$ad(acb)^2ab$

Table 6.17: Representating the generators of K_1 as elements of Γ^+

of Γ^+ А generating for K_2 subgroup set as а is ${bacbadba, cbabdaba, (cdab)^2, dabacbab, (dcba)^2, a(dabc)^2a}.$ More formally, K_2 can be given the abstract presentation in Table 6.18.

$$\begin{split} K_1 &= \langle F_1, F_2, F_3, F_4, F_5 | \\ F_2 F_5 F_2^{-1} F_3 F_4 F_1 F_3 F_1 F_2^{-1} F_1 F_4 F_5^{-1} F_2 F_1^{-1} = 1, \\ F_2 F_5^{-1} F_2 F_1^{-1} F_3^{-1} F_5 F_4^{-1} F_1^{-1} F_2 F_5 F_3^{-1} F_2 F_5^{-1} F_4^{-1} = 1, \\ F_2^{-1} F_3 F_1 F_2^{-1} F_5^2 F_2^{-1} F_3 F_4 F_1 F_5 F_3^{-1} F_4 F_5 F_2^{-1} F_3 = 1, \\ F_1^{-1} F_4^{-1} F_3^{-1} F_2 F_5^{-1} F_2^{-1} F_1 F_4 F_1 F_5 F_3^{-1} F_2 F_5^{-1} F_4^{-1} F_1 F_2^{-1} F_5 F_4^{-1} = 1, \\ F_1 F_4 F_3^{-1} F_4 F_5 F_2^{-1} F_3 F_4 F_1 F_5 F_3^{-1} F_2 F_5^{-1} F_4^{-1} F_1 F_2^{-1} F_5 F_4^{-1} = 1, \\ F_5 F_4^{-1} F_1^{-1} F_2 F_5 F_2^{-1} F_3 F_1 F_2^{-1} F_5 F_4^{-1} F_3 F_1 F_5 F_3^{-1} F_2 F_5^{-1} F_2^{-1} F_2^{-1} F_2^{-1} F_1 F_4 F_1 F_5 F_3^{-1} F_2 F_5^{-1} F_4^{-1} F_2^{-1} F_5 F_4^{-1} = 1, \\ F_5 F_4^{-1} F_1^{-1} F_2 F_5 F_2^{-1} F_3 F_1 F_2^{-1} F_5 F_4^{-1} F_3 F_1 F_5 F_3^{-1} F_2 F_5^{-1} F_4^{-1} F_2 F_5 F_2^{-1} F_3 F_4 F_1 = 1 \rangle \end{split}$$

Table 6.18: Presentation for the group K_2 of Lemma 6.7.2

Table 6.19 describes each generator of K_2 as given in Table 6.18 in terms of the generators of Γ^+

Generator	Representative
of K_1	in Γ^+
F_1	bacbadba
F_2	cbabdaba
F_3	$(cdab)^2$
F_4	dabacbab
F_5	acbacdabab

Table 6.19: Representing the generators of K_2 as elements of Γ^+

Both K_1 and K_2 have abelianization $\mathbb{Z}_2 \oplus \mathbb{Z}_4^4$. Both groups have 15 conjugacy classes of subgroups of index 2 whose abelianization has positive first Betti number, none of which is normal in Γ^+ . Their intersection $K_1 \cap K_2$ has abelianisation $\mathbb{Z}^{59} \oplus \mathbb{Z}_2^{30}$.

6.8 $\Gamma = T_8[2, 4, 3; 2, 5, 3]$

Recall that $T_8[2, 4, 3; 2, 5, 3]$ is the Coxeter group with Coxeter diagram



where a, b, c and d are Coxeter generators of Γ . Let $\alpha = ab$, $\beta = bc$ and $\gamma = cd$. Then the orientation preserving subgroup Γ^+ has the following presentation:

$$\Gamma^{+} = \langle \alpha, \beta, \gamma \mid \alpha^{5} = \beta^{3} = \gamma^{4} = (\alpha\beta)^{2} = (\beta\gamma)^{2} = (\alpha\beta\gamma)^{3} = 1 \rangle$$

6.8.1 Torsion in Γ and in Γ^+

Representatives of the conjugacy classes of torsion elements in Γ are given in Table 6.20. The restriction of this list to representatives of classes lying in Γ^+ is given in Table 6.21.

Order	Representative
Order 2	$a, ac, bd, (cd)^2, (adc)^3, (bcd)^3, (abc)^5, (dab)^5$
Order 3	bc, ad
Order 4	$cd, a(cd)^2, b(cd)^2$
Order 5	$ab, (ab)^2$
Order 6	$adc, bcd, c(ab)^2, d(ab)^2$
Order 10	$abc, (abc)^3$

Table 6.20: Conjugacy class representatives for elements of finite order in Γ

Order	Representative
Order 2	$ac, bd, (cd)^2$
Order 3	bc, ad
Order 4	cd
Order 5	$ab, (ab)^2$

Table 6.21: Conjugacy class representatives for elements of finite order in Γ^+

6.8.2 Maps to finite groups

Lemma 6.8.1 Γ^+ contains no normal subgroups whose factor group is A_5 .

Proof: Suppose that there exists some $K \triangleleft \Gamma^+$ whose factor group $\Gamma^+/K \cong A_5$. Let $\theta : \Gamma^+ \twoheadrightarrow A_5$ be the induced permutation representation. Adding $\gamma = 1$ to the presentation for Γ^+ gives the trivial group, so $\gamma \nleftrightarrow 1$. Write $\alpha \mapsto (1, 3, 5, 2, 4)$ and $\beta \mapsto (1, 2, 3)$. Since A_5 contains no elements of order 4, let γ map to some element of order 2. A quick check reveals that no element of order 2 in A_5 satisfies the required conditions.

Lemma 6.8.2 Γ^+ contains no normal subgroups whose factor group is $L_2(7)$.

Proof: Suppose that there exists some $K \triangleleft \Gamma^+$ whose factor group $\Gamma^+/K \cong L_2(7)$. Let $\theta : \Gamma^+ \twoheadrightarrow L_2(7)$ be the induced permutation representation. Since $7 \equiv \pm 2 \mod 5$, $L_2(7)$ contains no element of order 5, so the conjugacy class $\mathcal{C}(\alpha)$ lies in K. Adding the relation $\alpha = 1$ to the presentation for Γ^+ gives the trivial group. So K cannot exist. **Lemma 6.8.3** There exists a unique torsion free normal subgroup K of Γ^+ whose factor group $\Gamma^+/K \cong A_6 \cong L_2(9)$

Proof: Write

$$\alpha \mapsto \frac{1}{2} \begin{pmatrix} t-f & e+1\\ -1+e & t+f \end{pmatrix} \text{ and } \beta \mapsto \frac{1}{2} \begin{pmatrix} e+1 & t+f\\ f-t & 1-e \end{pmatrix}$$

where $e^2 + f^2 = -2 - t$, $t = \frac{-1 \pm \sqrt{5}}{2}$. Then

$$\gamma \mapsto \left(\begin{array}{cc} w & x \\ x+k & u-w \end{array}\right)$$

where $u = \pm \sqrt{2}$, $k = \pm 1$, and x satisfies the quadratic equation

$$(8+4t)x^2 + (-4fu - 4ftk + 4tk + 8k)x + (1+3t + f^2 - 2utk - 2ft - 2fuk) = 0.$$
(6.3)

Taking everything mod 3, $t = 2(2 \pm \sqrt{2})$, since $\sqrt{5} \equiv \sqrt{2} \mod 3$.

Case I

Suppose first that $t = -(2 + \sqrt{2}) = 1 + 2\sqrt{2}$. Then letting $e = 1 + 2\sqrt{2}$ and f = 0, $e^2 + f^2 = \sqrt{2} = -2 - t$ and equation 6.3 becomes

$$2\sqrt{2}x^2 + 2\sqrt{2}kx + (1 + uk + 2\sqrt{2}uk) = 0$$
(6.4)

which has discriminant $D = \sqrt{2} + 2 + uk + \sqrt{2}uk$. Now $u = \pm\sqrt{2}$ and $k = \pm 1$. Then $D \in \mathbb{F}_3$ if and only if $\sqrt{2} + \sqrt{2}uk = 0$ if and only if $\operatorname{sign}(u) = \operatorname{sign}(k)$ and this forces D = 0. Otherwise $D = \sqrt{2} + 2 + 2 + \sqrt{2} = 1 + 2\sqrt{2}$. For this to be a square in $\mathbb{F}_3(\sqrt{2})$ there must exist $a, b \in \mathbb{F}_3$ with $(a + b\sqrt{2})^2 = a^2 + 2b^2 + 2ab\sqrt{2} = 1 + 2\sqrt{2}$, so $a^2 + 2b^2 = 1$ (1) and 2ab = 2 (2). (2) forces $a, b \in \mathbb{F}_3^* = \{1, 2\}$. Then $a^2 = b^2 = 1$, and hence $a^2 + 2b^2 \equiv 0 \mod 3$. Thus D is not a square.

1) Letting $u = \sqrt{2}$ and k = -1 gives x as a solution to the quadratic $2x^2 + x + 2 = 0$. By the previous paragraph D = 0 so there exists a unique solution, x = 2. Since 2ew = eu - u - tk (chapter 4 § 4.9), substituting in for e, u, t and k gives $2(1+2\sqrt{2})w = 2+2\sqrt{2}$. Multiplying both sides by $(2+\sqrt{2})^{-1} = 1+\sqrt{2}$ gives $w = \sqrt{2}$. Consequently

$$\gamma \mapsto \left(\begin{array}{cc} \sqrt{2} & 2\\ 1 & 0 \end{array}\right)$$

2) Letting $u = -\sqrt{2}$ and k = 1 gives x as a solution to the quadratic $x^2 + x + 1 = 0$. Again it is required that D = 0 so there is a unique solution for x, x = 1. In this case 2ew = eu - u - tk becomes $2(1 + 2\sqrt{2})w = 1 + \sqrt{2}$. Multiplying both sides by $(2 + \sqrt{2})^{-1} = 1 + \sqrt{2}$ gives $w = 2\sqrt{2}$. Consequently

$$\gamma \mapsto \left(\begin{array}{cc} 2\sqrt{2} & 1\\ 2 & 0 \end{array}\right)$$

Since the value of γ in both solutions differs only by ± 1 , both solutions are equivalent.

Case II

Now let $t = -(2-\sqrt{2}) = 1+\sqrt{2}$. Then letting $e = 1+\sqrt{2}$ and f = 0, $e^2+f^2+3 \equiv t^2 \mod 3$ and equation 6.3 becomes

$$\sqrt{2}x^2 + \sqrt{2}kx + (1 + uk + \sqrt{2}uk) = 0 \tag{6.5}$$

which has discriminant $D = 2 + 2\sqrt{2} + uk + 2\sqrt{2}uk$. Then $D \in \mathbb{F}_3$ if and only if $2\sqrt{2} + \sqrt{2}uk = 0$, forcing sign(u) = sign(k). If sign(u) \neq sign(k) then $D = 1 + \sqrt{2}$. If there is $a + b\sqrt{2} = \omega \in \mathbb{F}_3(\sqrt{2})$ with $a, b \in \mathbb{F}_3$ such that $\omega^2 = D$ then $a^2 + 2b^2 = 1$ and 2ab = 1. 2ab = 1 forces $a, b \neq 0$ so $a^2 = b^2 = 1$ and 0 = 1, a contradiction. Hence sign(u) = sign(k) and D = 0.

1) Letting $u = \sqrt{2}$ and k = 1 gives x as a solution to the quadratic $x^2 + 2x + = 0$. By the previous paragraph D = 0 so there exists a unique solution, x = 2. Since 2ew = eu - u - tk (chapter 4 § 4.9), substituting in for e, u, t and k gives $2(1+\sqrt{2})w = 1+2\sqrt{2}$. Multiplying both sides by $(2+2\sqrt{2})^{-1} = 1+2\sqrt{2}$ gives $w = \sqrt{2}$. Consequently

$$\gamma \mapsto \left(\begin{array}{cc} \sqrt{2} & 1\\ 2 & 0 \end{array}\right)$$

2) Letting $u = -\sqrt{2}$ and k = -1 gives x as a solution to the quadratic $x^2 + x + = 0$. D = 0 gives a unique solution x = 1. Again, using 2ew = eu - u - tk, $2(1 + \sqrt{2})w = 2 + \sqrt{2}$. Multiplying both sides by $(2 + 2\sqrt{2})^{-1} = 1 + 2\sqrt{2}$ gives $w = 2\sqrt{2}$ and as a result

$$\gamma \mapsto \left(\begin{array}{cc} \sqrt{2} & 2\\ 1 & 0 \end{array}\right)$$

Since the value of γ in both solutions differs only by ± 1 , both solutions are equivalent. Finally, there is an automorphism of A_6 that transposes the two epimorphisms, so there is a unique such kernel.

Note: The above proof is included as a demonstration of theory provided in Chapter 4. A more elementary proof would go as follows: Since both conjugacy classes of A_5 in A_6 are conjugate under the outer automorphism of S_6 , it suffices to consider only $A_5 \cong \langle (1,3,5,2,4), (1,2,3) \rangle$. Then there exists a unique element g of order 5 in A_6 satisfying |(1,2,3)g| = 2 and |(2,4,3,5)g| = 3.

The group Γ^+ is perfect: Abelianizing the presentation for Γ^+ , the identities $\alpha^4 = \beta^3 = \gamma^4 = 1$, $(\alpha\beta)^2 = \alpha^2\beta^2 = 1$, $(\beta\gamma)^2 = \beta^2\gamma^2 = 1$ and $(\alpha\beta\gamma)^3 = \alpha^3\gamma^3 = 1$ are obtained. Squaring $\beta^2\gamma^2$ gives $\beta = 1$. Hence $\alpha = 1$, so $\gamma^4 = \gamma^3 = 1$, and so Γ^+ is a perfect group. Combining the above results, the following theorem can be stated:

Theorem 6.5 There exists a unique torsion free normal subgroup K of Γ^+ with factor group A_6 . Furthermore, this subgroup K is the smallest index torsion free normal subgroup of Γ^+ .

Theorem 6.5 describes the smallest index torsion free normal subgroup, but this is not the smallest index torsion free subgroup of Γ^+ . There is a unique normal subgroup N whose factor group is $L_2(11)$. The conjugacy class $\mathcal{C}(\gamma^2)$ of elements of order 2 lies in K. Let θ be the induced map $\Gamma^+ \to L_2(11)$. Then the preimage $K.C_{11}$ of the subgroup C_{11} of $L_2(11)$ of order 11 under the map θ is an index 60 subgroup which avoids all conjugacy classes of torsion, excepting the class $\mathcal{C}(\gamma^2)$ of elements of order 2. Using GAP, torsion free subgroups of $K.C_{11}$ of of index 2 were found.

6.9 $\Gamma = T_9[2, 3, 5; 2, 3, 5]$

Recall that $T_9[2, 5, 3; 2, 5, 3]$ is the Coxeter group with Coxeter diagram

$$a 5 b$$

 $5 c d$

where a, b, c and d are Coxeter generators of Γ . Let $\alpha = ab$, $\beta = bc$ and $\gamma = cd$. Then the orientation preserving subgroup Γ^+ has the following presentation:

$$\Gamma^+ = \langle \alpha, \beta, \gamma \mid \alpha^5 = \beta^3 = \gamma^5 = (\alpha\beta)^2 = (\beta\gamma)^2 = (\alpha\beta\gamma)^3 = 1 \rangle$$

6.9.1 Torsion in Γ and in Γ^+

Representatives of the conjugacy classes of torsion elements in Γ are given in Table 6.22. The restriction of this list to representatives of classes lying in Γ^+ is given in Table 6.23.

Order	Representative
Order 2	a, ac, bd
Order 3	bc, ad
Order 5	$ab, (ab)^2, cd, (cd)^2$
Order 6	$c(ab)^2, d(ab)^2, a(dc)^2, b(cd)^2$
Order 10	$abc, bad, cda, bcd, (abc)^3, (bad)^3, (cda)^3, (bcd)^3$

Table 6.22: Conjugacy class representatives for elements of finite order in Γ

Order	Representative					
Order 2	ac, bd					
Order 3	bc, ad					
Order 5	$ab, cd, (ab)^2, (cd)^2$					

Table 6.23: Conjugacy class representatives for elements of finite order in Γ^+

6.9.2 Maps to $L_2(5) \cong A_5$

Lemma 6.9.1 There are three distinct torsion free normal subgroups K_1 , K_2 and K_3 of Γ^+ with factor group A_5 .

Proof: Let $\alpha \mapsto \overline{\alpha} = (1,3,5,2,4)$ and $\beta \mapsto \overline{\beta} = (1,2,3)$. Then, for $\overline{\gamma} \in \{(1,3,5,2,4), (1,5,3,4,2), (1,5,3,2,4)\}$, the conditions $|\overline{\gamma}| = 5$, $|\overline{\beta}\overline{\gamma}| = 2$ and $|\overline{\alpha}\overline{\beta}\overline{\gamma}| = 3$ are satisfied. Hence there are three torsion-free normal subgroups whose factor group is A_5 . The elements (1,3,5,2,4) and (1,5,3,4,2) lie in the same conjugacy class, while (1,5,3,4,2) lies in the other conjugacy class of elements of order 5 in A_5 . To prove that these groups are distinct, it suffices to consider the image of the word $\alpha\gamma$ under the three maps.

Setting $\overline{\gamma} = (1, 5, 3, 4, 2)$ gives $|\overline{\alpha\gamma}| = 5$, letting $\overline{\gamma} = (1, 5, 3, 4, 2)$ gives $|\overline{\alpha\gamma}| = 3$ and letting $\overline{\gamma} = (1, 5, 3, 2, 4)$ gives $|\overline{\alpha\gamma}| = 2$. Consequently, the epimorphisms are inequivalent and so there are precisely three distinct normal subgroups, as desired.

The above lemma gives three values for γ satisfying the presentation of Γ^+ . Some facts are known about the manifolds in each case:

Case I: $\alpha \mapsto (1,3,5,2,4), \beta \mapsto (1,2,3)$ so $\alpha\beta = (2,4)(3,5)$. Then let $\gamma \mapsto (1,3,5,2,4)$ (ie $\gamma = \alpha$), so $\alpha\gamma \mapsto (1,5,2,3,4)$. The kernel of this map, K_1 , contains the subgroup $\langle \langle (\alpha\gamma)^5 \rangle \rangle_{\Gamma^+}$ (this subgroup has infinite index in Γ^+) and K_1 has abelianisation $\mathbb{Z}_2 \oplus \mathbb{Z}_3^4 \oplus \mathbb{Z}_4^4 \oplus \mathbb{Z}_5^3$.

Case II: $\alpha \mapsto (1,3,5,2,4), \beta \mapsto (1,2,3)$ so $\alpha\beta = (2,4)(3,5)$. If $\gamma \mapsto (1,5,3,4,2)$ is conjugate to α and $\alpha\gamma \mapsto (1,4,5)$. The kernel of this map, K_2 , contains the subgroup $\langle \langle (\alpha\gamma)^3 \rangle \rangle_{\Gamma^+}$ with finite index (this subgroup has index 622,080,000 in Γ^+) and K_2 has abelianisation $\mathbb{Z}_2 \oplus \mathbb{Z}_3^4 \oplus \mathbb{Z}_4^4 \oplus \mathbb{Z}_5^3$.

Case III: $\alpha \mapsto (1,3,5,2,4), \beta \mapsto (1,2,3)$ so $\alpha\beta = (2,4)(3,5)$. Then let $\gamma \mapsto (1,5,3,2,4)$, so $\gamma \not\sim \alpha$ in A_5 , but is in S_5 , and $\alpha\gamma \mapsto (1,2)(4,5)$. The kernel of this map, K_3 , is the subgroup $K = \langle \langle (\alpha\gamma)^2 \rangle \rangle_{\Gamma^+}$ and K_3 has abelianisation \mathbb{Z}^{11} . Consequently there exists a k-celled manifold arising from this Coxeter group for every $k \in \mathbb{N}$. The associated manifold $\mathcal{M}_z = \mathbb{H}^3/K_3$ is Zimmermann's manifold [Z].

The group K_3 is generated by the words shown in Table 6.24

 $(bcda)^2, (cdab)^2, (dabc)^2, (dcba)^2, acdabcdaba, (dabc)^2)^a, babacdcdca, babcbabada, babcdcdcba, badababcba, bcbabadaba, cbdacdcadb$

Table 6.24: A generating set for K_3

As an abstract group, K_3 has a presentation as given in Table 6.25. The relationship between the generators of K_3 as a subgroup of Γ^+ and the generators of K_3 as an abstract group is summarised in Table 6.26

$$\begin{split} K_{3} &= \langle F_{1}, F_{2}, F_{3}, F_{4}, F_{5}, F_{6}, F_{7}, F_{8}, F_{9}, F_{10}, F_{11} \mid \\ F_{7}^{-1}F_{9}F_{4}^{-1}F_{2}F_{7}F_{2}^{-1}F_{9}^{-1}F_{4} &= 1, F_{10}^{-1}F_{3}^{-1}F_{5}^{-1}F_{4}F_{10}F_{1}F_{4}^{-1}F_{3}F_{5}F_{1}^{-1} = 1, \\ F_{6}F_{8}F_{2}^{-1}F_{6}^{-1}F_{4}F_{11}F_{8}^{-1}F_{4}^{-1}F_{2}F_{11}^{-1} &= 1, \\ F_{11}F_{2}^{-1}F_{5}F_{3}F_{10}F_{11}^{-1}F_{5}^{-1}F_{10}F_{1}^{-1}F_{3}^{-1}F_{2} = 1, \\ F_{10}^{-1}F_{3}^{-1}F_{9}^{-1}F_{4}F_{7}^{-1}F_{9}F_{1}^{-1}F_{7}F_{10}F_{1}F_{4}^{-1}F_{3} &= 1, \\ F_{9}F_{6}^{-1}F_{3}F_{7}^{-1}F_{1}F_{9}^{-1}F_{7}F_{4}^{-1}F_{6}F_{3}^{-1}F_{4}F_{1}^{-1} &= 1, \\ F_{9}F_{6}^{-1}F_{3}F_{7}^{-1}F_{1}F_{9}^{-1}F_{7}F_{4}^{-1}F_{6}F_{2}F_{8}^{-1}F_{6}^{-1} &= 1, \\ F_{5}^{-1}F_{10}^{-1}F_{8}F_{2}^{-1}F_{5}F_{3}F_{10}F_{3}^{-1}F_{6}F_{2}F_{8}^{-1}F_{6}^{-1} &= 1, \\ F_{6}F_{10}F_{5}F_{11}F_{7}F_{8}^{-1}F_{6}^{-1}F_{5}^{-1}F_{10}^{-1}F_{8}F_{11}^{-1}F_{7}^{-1} &= 1, \\ F_{6}F_{10}F_{5}F_{11}F_{7}F_{8}^{-1}F_{6}F_{2}F_{8}^{-1}F_{6}^{-1}F_{2}^{-1}F_{4}F_{1}^{-1}F_{9} &= 1, \\ F_{10}F_{8}F_{11}^{-1}F_{7}^{-1}F_{9}F_{5}^{-1}F_{2}F_{8}^{-1}F_{10}F_{5}F_{11}F_{7}F_{2}^{-1}F_{9}^{-1} &= 1, \\ F_{10}F_{8}F_{11}^{-1}F_{7}^{-1}F_{9}F_{5}^{-1}F_{2}F_{8}^{-1}F_{10}F_{5}F_{11}F_{7}F_{2}^{-1}F_{9}^{-1} &= 1, \\ F_{10}F_{8}F_{11}^{-1}F_{7}^{-1}F_{9}F_{5}^{-1}F_{2}F_{8}^{-1}F_{10}F_{5}F_{11}F_{7}F_{9}^{-1}F_{1} &= 1, \\ F_{10}F_{8}F_{11}^{-1}F_{7}^{-1}F_{9}F_{5}^{-1}F_{2}F_{8}^{-1}F_{10}F_{5}F_{11}F_{7}F_{9}^{-1}F_{1} &= 1, \\ F_{5}^{-1}F_{2}F_{8}^{-1}F_{6}^{-1}F_{2}^{-1}F_{4}F_{1}^{-1}F_{7}F_{4}^{-1}F_{5}F_{6}F_{8}F_{7}^{-1}F_{1} &= 1, \\ F_{5}^{-1}F_{2}F_{8}^{-1}F_{6}^{-1}F_{2}^{-1}F_{4}F_{1}^{-1}F_{7}F_{4}^{-1}F_{5}F_{6}F_{8}F_{7}^{-1}F_{1} &= 1, \\ F_{6}F_{3}^{-1}F_{2}F_{11}F_{2}^{-1}F_{4}F_{10}F_{9}F_{6}^{-1}F_{3}F_{11}^{-1}F_{4}^{-1}F_{3}F_{10}^{-1}F_{3}^{-1}F_{9}^{-1} &= 1 \rangle \end{split}$$

Table 6.25: A presentation for K_3

Generator in Γ	Image in K_3
$(dcba)^2$	F_1
babacdedea	F_2
babc babada	F_3
babcdcdcba	F_4
badababcba	F_5
bcbabadaba	F_6
$(cdab)^2$	$F_{10}^{-1}F3^{-1}F_5^{-1}F_2F_8^{-1}F_6^{-1}$
acdabcdaba	$F_7 F_2^{-1} F_9^{-1} F_4 F_7^{-1} F_9 F_1^{-1} F_9$
$(bcda)^2$	$F_4 F_7^{-1} F_9 F_1^{-1} F_7$
$(cdab)^2$	$F_7 F_2^{-1} F_9^{-1} F_4 F_7^{-1} F_9 F_1^{-1} F_7 F_4^{-1}$
$((dabc)^2)^a$	$F_{10}^{-1}F_3^{-1}F_5^{-1}F_2F_8^{-1}F_{10}F_5$
cbdacdcadb	$F_7 F_{11} F_2^{-1} F_5 F_3 F_{10} F_3^{-1} F_6 F_9^{-1}$

Table 6.26: The images of the generators of K_3 as a subgroup of Γ^+ in H

As has been previously observed, $H_1/[H_1, H_1] \cong \mathbb{Z}^{11}$. The induced action of $A_5 \cong \Gamma^+/H_1$ on $H_1/[H_1, H_1] \cong \mathbb{Z}^{11}$ equips \mathbb{Z}^{11} with a module structure $\theta : A_5 \hookrightarrow GL_{11}(\mathbb{Z})$. The injection θ is defined by the two maps:

	$\begin{pmatrix} 1 \end{pmatrix}$	0	0	0	0	0	0	0	0	0	0)
	1	0	0	1	1	0	0	0	0	0	-1
	0	0	1	0	0	1	0	-1	0	-1	0
	0	0	0	0	0	0	1	0	1	0	0
	0	0	0	0	0	0	0	0	0	-1	1
$\alpha \mapsto$	0	0	0	0	-1	0	0	1	0	0	1
	-1	0	0	-1	0	0	-1	0	0	0	1
	0	0	0	0	-1	0	0	0	0	0	0
	0	-1	0	0	0	0	0	-1	0	0	-1
	0	0	1	0	0	0	0	0	0	-1	1.
	0	0	0	0	0	0	0	0	0	0	1)

ar	10	1
ar	10	1

	$\begin{pmatrix} 0 \end{pmatrix}$	0	0	0	-1	0	0	0	0	1	0
	-1	0	1	0	-1	1	0	0	0	0	1
	1	0	0	0	0	0	0	0	0	0	0
	0	0	0	0	1	0	0	0	0	0	-1
	1	-1	-1	0	0	0	0	-1	0	0	-1
$\beta \mapsto$	-1	-1	-1	-1	-1	-1	-1	0	-1	1	0
	1	0	0	0	0	0	1	0	1	-1	0
	-1	0	-1	-1	0	-1	-1	0	-1	1	0
	0	0	0	0	1	0	0	0	0	0	0
	1	-1	0	0	0	0	0	-1	0	0	-1
	1	0	0	1	0	0	1	0	0	0	-1

According to the Atlas of Finite Groups [Atlas], there are 5 irreducible characters for A_5 , with degrees, 1, 3, 3, 4, and 5, so this representation cannot be primitive. Hence the character splits as a sum of irreducible characters. The traces of the conjugacy class representatives of A_5 in this representation are given as follows: $\operatorname{tr}(\alpha) = 1$, $\operatorname{tr}(\alpha^2) = 1$, $\operatorname{tr}(\beta) = -1$ and $\operatorname{tr}(\alpha\beta) = -1$. A calculation involving characters of representatives of the conjugacy classes of elements of A_5 gives this module as a sum of the 3-, 3- and 5-dimensional irreducible modules of A_5 . So, for any positive integer n, there exists a covering manifold $\mathcal{M}_z(n)$ of \mathcal{M}_z whose covering transformations (deck transformations) form an abelian group, and over which the isometry group extends to the group $A_5 \rtimes \mathbb{Z}_n^{11}$.

6.9.3 Further interesting subgroups

Lemma 6.9.2 There are two distinct normal subgroups K_4 and K_5 of Γ^+ whose factor group is $A_6 \cong L_2(9)$.

Proof: There are two conjugacy classes of A_5 in A_6 . Letting $\alpha \mapsto \alpha_1 = (13524)$ and $\beta \mapsto \beta_1 = (123)$, we can have $\gamma_1 \in \{(14326), (16325)\}$ satisfying $|\gamma_1| = 5$, $|\beta_1\gamma_1| = 2$ and $|\alpha_1\beta_1\gamma_1| = 3$. Note that α_1^2 is conjugate to γ_1 in each case. Letting $\alpha \mapsto \alpha_2 = (13542)$ and $\beta \mapsto \beta_2 = (143)$ we get $\gamma_2 \in \{(14326), (16325)\}$. Notice that conjugating by $(24) \in S_6 \setminus A_6$ will transpose α_1, β_1 and γ_1 with α_2, β_2 and γ_2

Lemma 6.9.3 There is a unique normal subgroup K_6 of Γ^+ whose factor group is $L_2(11)$.

Proof: Let $\phi : \Gamma^+ \to L_2(11)$ be a map from Γ^+ to $L_2(11)$ as constructed in § 4.10, and $K = \text{Ker}(\theta)$ be the kernel of the map. There are two cases to consider: the trace($\overline{\alpha}$) =trace($\overline{\gamma}$) and trace($\overline{\alpha}$) \neq trace($\overline{\gamma}$).

Case 1 - trace($\overline{\alpha}$) = trace($\overline{\gamma}$):

Suppose first that $\operatorname{trace}(\overline{\alpha}) = \operatorname{trace}(\overline{\gamma}) = t$. Then $t = (-1 \pm \sqrt{5})/2 = 7$ or 3. If $\operatorname{trace}(\overline{\alpha}) = 7$ then

$$\alpha \mapsto \overline{\alpha} = \begin{pmatrix} 3 & 1 \\ 0 & 4 \end{pmatrix} \text{ and } \beta \mapsto \overline{\beta} = \begin{pmatrix} 1 & 4 \\ 8 & 0 \end{pmatrix}$$

Suppose first that $\operatorname{trace}(\overline{\alpha}\overline{\beta}\overline{\gamma}) = 1$: then there is no solution for $\gamma \mapsto g \in L_2(11)$. If $\operatorname{trace}(\overline{\alpha}\overline{\beta}\overline{\gamma}) = -1$ then two distinct solutions for γ :

$$\gamma \mapsto g_1 = \begin{pmatrix} 4 & 0 \\ -1 & 3 \end{pmatrix} = (\overline{\alpha})^{\overline{\beta}} \text{ and } \gamma \mapsto g_2 = \begin{pmatrix} 3 & 1 \\ 0 & 4 \end{pmatrix} = \overline{\alpha}$$

Hence two epimorphisms θ_1 , $\theta_2 : \Gamma^+ \to A_5$ are constructed. The kernels K_{θ_1} , K_{θ_2} of these maps are distinct torsion free normal subgroups of Γ^+ whose factor group Γ^+/K_{θ_i} is an A_5 subgroup of $L_2(11)$. If trace($\overline{\alpha}$) = 3 then

$$\alpha \mapsto \overline{\alpha} = \begin{pmatrix} 10 & 3 \\ 2 & 4 \end{pmatrix} \text{ and } \beta \mapsto \overline{\beta} = \begin{pmatrix} 3 & 4 \\ 1 & 9 \end{pmatrix}$$

Assume first that $\operatorname{trace}(\overline{\alpha}\overline{\beta}\overline{\gamma}) = 1$. Then, solving the equations in § 4.10, a unique solution

$$\gamma \mapsto g_3 = \left(\begin{array}{cc} 10 & 8\\ 9 & 4 \end{array}\right)$$

is recovered. Since the order of $\overline{\alpha}g_3$ is six, $\overline{\alpha}g_3 \notin A_5 < L_2(11)$, so $g_3 \notin A_5$. As A_5 is maximal in $L_2(11)$, it follows that the images $\overline{\alpha}, \overline{\beta}$ and g_3 generate $L_2(11)$. Hence the kernel of the map, K_{θ_3} , is a distinct torsion free normal subgroup of Γ^+ whose factor group Γ^+/K_{θ_3} isomorphic to $L_2(11)$. Next, assuming that trace $(\overline{\alpha}\overline{\beta}\overline{\gamma}) = -1$ then, solving the equations in § 4.10, two distinct solutions

$$\gamma \mapsto g_4 = \begin{pmatrix} 10 & 3\\ 2 & 4 \end{pmatrix} = \overline{\alpha} \text{ and } \gamma \mapsto g_5 = \begin{pmatrix} 4 & 9\\ 8 & 10 \end{pmatrix} = (\overline{\alpha})^{\overline{\beta}}$$

are recovered. The kernels K_{θ_4} , K_{θ_5} of these maps are distinct torsion free normal subgroups of Γ^+ whose factor group Γ^+/K_{θ_i} is an A_5 subgroup of $L_2(11)$.

Case 2 - trace(α) \neq trace(γ):

Now suppose that $\operatorname{trace}(\overline{\alpha}) \neq \operatorname{trace}(\overline{\gamma})$. As in the previous case, $t = (-1 \pm \sqrt{5})/2 = 7$ or 3. If $\operatorname{trace}(\overline{\alpha}) = 7$ then

$$\alpha \mapsto \overline{\alpha} = \begin{pmatrix} 3 & 1 \\ 0 & 4 \end{pmatrix} \text{ and } \beta \mapsto \overline{\beta} = \begin{pmatrix} 1 & 4 \\ 8 & 0 \end{pmatrix}$$

Suppose first that $\operatorname{trace}(\overline{\alpha}\overline{\beta}\overline{\gamma}) = 1$: then a unique solution

$$\gamma \mapsto g_6 = \begin{pmatrix} 10 & 8 \\ 9 & 4 \end{pmatrix} = (\overline{\alpha}^2)^{\overline{\beta}\overline{\alpha}^{-1}}$$

is recovered, and the image of α , β and γ generate an A_5 subgroup of $L_2(11)$. If $\operatorname{trace}(\overline{\alpha}\overline{\beta}\overline{\gamma}) = 1$: then there is no solution for $\gamma \mapsto g \in L_2(11)$. If $\operatorname{trace}(\overline{\alpha}) = 3$ then

$$\alpha \mapsto \overline{\alpha} = \begin{pmatrix} 10 & 3 \\ 2 & 4 \end{pmatrix} \text{ and } \beta \mapsto \overline{\beta} = \begin{pmatrix} 3 & 4 \\ 1 & 9 \end{pmatrix}$$

Assume first that $\operatorname{trace}(\overline{\alpha}\overline{\beta}\overline{\gamma}) = 1$. Then, solving the equations in § 4.10, a unique solution

$$\gamma \mapsto g_7 = \begin{pmatrix} 3 & 10 \\ 0 & 4 \end{pmatrix} = (\overline{\alpha}^2)^{\overline{\beta}\overline{\alpha}^{-1}}$$

for γ is recovered, and the image of α , β and γ generate an A_5 subgroup of $L_2(11)$. If trace $(\overline{\alpha\beta\gamma}) = 1$: then there is no solution for $\gamma \mapsto g \in L_2(11)$.

Hence there is a unique map $\theta_3 : \Gamma^+ \to L_2(11)$. The kernel of the map K_{θ_3} , is a unique torsion free subgroup of Γ^+ whose factor group is $L_2(11)$. The epimorphism is characterised by the matrices

$$\alpha \mapsto \begin{pmatrix} 10 & 3 \\ 2 & 4 \end{pmatrix} \qquad \beta \mapsto \begin{pmatrix} 3 & 4 \\ 1 & 9 \end{pmatrix} \qquad \gamma \mapsto \begin{pmatrix} 10 & 8 \\ 9 & 4 \end{pmatrix}$$

Computer calculations with GAP yielded $K/K' \cong \mathbb{Z}^{41}$.

Lemma 6.9.4 There are two distinct normal subgroups K_7 and K_8 of Γ^+ whose factor group is $L_2(19)$.

Proof: Proceeding as in the previous lemma, two non-conjugate maps θ_1, θ_2 : $\Gamma^+ \twoheadrightarrow L_2(19)$ were constructed, one for each case $t = t_1$, $u = t_2$ and $t = t_2$, $u = u_1$:

$$\alpha \mapsto \begin{pmatrix} 0 & 8 \\ 7 & 14 \end{pmatrix} \qquad \beta \mapsto \begin{pmatrix} 8 & 14 \\ 0 & 12 \end{pmatrix} \qquad \gamma \mapsto \begin{pmatrix} 17 & 7 \\ 6 & 7 \end{pmatrix}$$
$$\alpha \mapsto \begin{pmatrix} 0 & 13 \\ 16 & 5 \end{pmatrix} \qquad \beta \mapsto \begin{pmatrix} 8 & 1 \\ 0 & 12 \end{pmatrix} \qquad \gamma \mapsto \begin{pmatrix} 5 & 11 \\ 4 & 9 \end{pmatrix}$$

The kernels of the maps are distinct non-conjugate torsion free normal subgroups of Γ^+ . In both cases the abelianization has the form $\mathbb{Z}_2 \oplus \mathbb{Z}_3^{37} \oplus \mathbb{Z}_5^{18} \oplus \mathbb{Z}_8^{18} \oplus \mathbb{Z}_{11}^{18} \oplus \mathbb{Z}_{17}^{10} \oplus \mathbb{Z}_{19}^{3} \oplus \mathbb{Z}_{19}^{18} \oplus \mathbb{$

6.9.4 Maps to M_{12}

 Γ^+ has a natural description as the quotient of a free product with amalgamation. Write

$$A_5 = \langle \alpha, \beta \mid \alpha^5 = \beta^3 = (\alpha\beta)^2 = 1 \rangle$$

$$A'_5 = \langle \beta', \gamma \mid \gamma^5 = \beta'^3 = (\beta'\gamma)^2 = 1 \rangle$$

and consider the free product with amalgamation, $A_5 *_{\{\beta = \beta'\}} A'_5$. This is a group with presentation

$$\langle \alpha, \beta, \gamma \mid \alpha^5 = \beta^3 = \gamma^5 = (\alpha \beta)^2 = (\beta \gamma)^2 = 1 \rangle$$

Then Γ^+ can be written as the quotient group $A_5 *_{\{\beta=\beta'\}} A'_5 / \{(\alpha\beta\gamma)^3\}$.

If Γ^+ has a subgroup H of index n, then, using the action of Γ^+ on the cosets of H, a permutation group G with permutation representation on n points can be constructed. The kernel of the induced map $\Gamma^+ \to G$ is the core of H in Γ^+ . If Ghas a minimal value n_0 of n for which it has a faithful permutation representation on n points, then no action of Γ^+ on subgroups of index less than n_0 can generate an induced map $\Gamma^+ \to G$. The group A_5 has transitive permutation representations on 5, 6, 10 and 12 points, corresponding to right coset actions on subgroups A_4 , D_{10} , D_6 and C_5 . In each case the action is unique. Choose generators x, y of A_5 so that $x^2 = y^3 = (xy^2)^5 = 1$. Diagrams representing the actions of A_5 on n points (*Higman* diagrams), for n = 5, 6, 10, 12, can be constructed as shown in Figure 6.6 and Figure 6.7. In these diagrams, heavy dots represent points fixed by the action of y; triangles represent vertices permuted in an anti-clockwise direction by y and any two vertices interchanged by x are joined by a dotted line. As well as these transitive actions, A_5 has a trivial action on one vertex.



Figure 6.6: Higman diagrams for A_5 acting on 5, 6 and 10 points.



Figure 6.7: Higman diagram for A_5 acting on 12 points.

Recall that Γ^+ has presentation

$$\Gamma^{+} = \langle \alpha, \beta, \gamma \mid \alpha^{5} = \beta^{3} = \gamma^{5} = (\alpha\beta)^{2} = (\beta\gamma)^{2} = 1 \rangle$$

By writing $x = \alpha \beta$, $y = \beta$ and $z = \beta \gamma$, the presentation for Γ^+ can be rewritten as

$$\begin{split} \Gamma^+ &= \langle x, y, z \mid x^2 = y^3 = z^2 = 1, (xy^2)^5 = (y^2 z)^5 = (xy^2 z)^3 = 1 \rangle \\ &= \langle x, y, z \mid x^2 = y^3 = z^2 = 1, (xy^2)^5 = (y^2 z)^5 = 1 \rangle / \{ (xy^2 z)^3 \} \\ &= A_5 *_{y'=y} A'_5 / \{ (xy^2 z)^3 \} \end{split}$$

where $A_5 = \langle x, y | x^2 = y^3 = (xy^2)^5 = 1 \rangle$ and $A'_5 = \langle y', z | y'^3 = z^2 = (y'^2 z)^5 = 1 \rangle$ Permutation representations for Γ^+ on *n* points can then be constructed by amalgamating Higman diagrams for A_5 on *n* points and adding the additional condition that $(xy^2z)^3 = 1$. The process goes as follows: suppose that a permutation representation of Γ^+ on *n* points, for some integer *n*, is required. Take two Higman diagrams D_1 and D_2 , each describing the action of A_5 on *n* points. Let A_5 act on D_1 and A'_5 act on D_2 . These two diagrams are now amalgamated by identifying the triangles in D_1 with those in D_2 . The resulting amalgamated diagram satisfies the relations of the free product with amalgamation, $A_5 *_{\{y'=y\}} A'_5$. The remaining vertices of D_1 not already identified with those of D_2 are then paired so that the final diagram satisfies the additional condition $(xy^2z)^3 = 1$.

It is instructive at this point to consider an example. Suppose that all possible permutation representations of Γ^+ on 5 points are desired. Take two copies, D_1 and D_2 , of the diagram representing the action of A_5 on 5 points. Let D_1 describe the action of A_5 and D_2 describe the action of A'_5 . Since Γ^+ can be expressed as the quotient of the free product $A_5 *_{\{y'=y\}} A'_5$, the two diagrams D_1 and D_2 can be amalgamated by identifying triangles. There are three possible ways of doing this, giving diagrams \tilde{D}_1 , \tilde{D}_2 and \tilde{D}_3 . For each diagram \tilde{D}_i , the condition $(xy^2z)^3 = 1$ in the presentation of Γ^+ is then used to figure out all admissible actions of z on the 5 points. Figure 6.8 illustrates all these admissible diagrams, which represent three distinct non-conjugate epimorphisms from Γ^+ to A_5 , corresponding to three distinct normal subgroups of Γ^+ whose factor groups are all isomorphic to A_5 . This verifies the results of Lemma 6.9.1, where it was shown that there are three distinct normal subgroups with factor group A_5 .

The Mathieu group M_{12} is the automorphism group of the S(5, 6, 12) Steiner system. The smallest index subgroup of M_{12} is the Mathieu group M_{11} , the automorphism group of the S(4, 5, 11) Steiner system. There are two conjugacy classes of maximal subgroups M_{11} in M_{12} : one is the natural subgroup M_{11} acting on 11 points. A representative of the second class has a transitive representation on 12 points. The action by right multiplication of M_{12} on the 12 cosets of an M_{11} subgroup induces a permutation representation for M_{12} on 12 points. This is the smallest permutation representation for M_{12} . The rest of this section is devoted to proving that Γ^+ has two distinct normal subgroups N_1 and N_2 whose factor



Figure 6.8: Representation of the maps $\Gamma^+ \to A_5$ constructed in Lemma 6.9.1. Black dotted lines correspond to the action of x on points, and red dotted lines correspond to the action of z on points.

group is M_{12} . Higman diagrams for the action of A_5 on 12 points will be combined as described above and GAP will be used to classify the resulting permutation groups.

Theorem 6.6 There are two distinct normal subgroups K_{10} and K_{11} of Γ^+ whose factor group is M_{12} .

Proof: Assume that $\Gamma^+ = A_5 *_{y'=y} A'_5 / \{(xy^2z)^3\}$, where $A_5 = \langle x, y | x^2 = y^3 = (xy^2)^5 = 1 \rangle$ and $A'_5 = \langle y', z | y'^3 = z^2 = (y'^2z)^5 = 1 \rangle$. Permutation representations of A_5 (and A'_5) on 12 points are in 1-1 correspondence with partitions of 12 into orbits of A_5 of size 1, 5, 6, 10 and 12. Admissible partitions are

 $\{1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1\}, \{5, 1, 1, 1, 1, 1, 1, 1\}, \{6, 1, 1, 1, 1, 1\}, \{5, 5, 1, 1\}, \{5, 6, 1\}, \{6, 6\}, \{10, 1, 1\}, \{12\}$

The proof now proceeds by listing all combinations of pairs of these partitions which yield transitive representations of $A_5 *_{y=y'} A'_5$ on 12 points and imposing the condition that $(xy^2z)^3 = 1$. The resulting permutation representation will describe a transitive permutation representation of Γ^+ on 12 points. In general, if a partition involves n_i orbits of length i, for i = 1, 5, 6, 10, 12, then y has $n_1 + 2n_5 + n_{10}$ fixed points, since it fixes 1, 2, 0, 1 or 0 points in each orbit of length 1, 5, 6, 10 or 12 points, respectively. Then the partition can only be combined with one where y'has the same number of fixed points. There are a total of 49 possible pairs of partitions. The proof is divided into seven cases: in each case it is assumed that the partition induced by the action of A_5 is fixed. The only restriction on the second partition (the partition induced by A'_5) is that the action of y' fixes the same number of points as y does.

- 1) One of the partitions is $\{5, 1, 1, 1, 1, 1, 1, 1\}$: In this case there are 9 points fixed by the action of y. Hence the second partition must also be of type $\{5, 1, 1, 1, 1, 1, 1\}$ and there is no way of amalgamating these two diagrams to obtain a transitive representation on 12 points. So no transitive permutation diagram of an action of Γ^+ on 12 points can be constructed by amalgamating two A_5 actions with one of the actions being the $\{5, 1, 1, 1, 1, 1, 1, 1\}$ permutation representation of A_5 on 12 points.
- 2) One of the partitions is $\{6, 1, 1, 1, 1, 1, 1\}$: Since this contains 6 points stabilized by y, this partition can only be combined with partitions of type $\{5, 5, 1, 1\}$ or type $\{6, 1, 1, 1, 1, 1, 1\}$. Clearly any amalgamation of two copies of $\{6, 1, 1, 1, 1, 1, 1\}$ will result in an intransitive permutation representation with 6 fixed points, while any amalgamation of $\{6, 1, 1, 1, 1, 1, 1\}$ with $\{5, 5, 1, 1\}$ will have 2 fixed points. So no transitive permutation diagram of an action of Γ^+ on 12 points can be constructed by amalgamating two A_5 actions with one of the actions being the $\{6, 1, 1, 1, 1, 1\}$ permutation representation of A_5 on 12 points.
- 3) One of the partitions is $\{5, 5, 1, 1\}$: Since this contains 6 points stabilized by y, then by the result of 2) this action can only be combined with $\{5, 5, 1, 1\}$. Let s_1 and s_2 be the two points fixed by the action of A_5 . Transitivity forces the points s_1 and s_2 to be transposed by the action of $z \in A'_5$ with points not fixed by y. Similarly, x must transpose the points v_1 and v_2 fixed by y' and z. Then the only to construct a transitive diagram must be to construct a diagram similar to that of Figure 6.9



Figure 6.9: A possible transitive action of Γ^+ on 12 points

It only remains to find all such diagrams: Starting with a permutation representation of A_5 on 5 points, let S be a point permuted by the action of xand action of y. Let y' and z fix a vertex s_f . Since the final diagram must be transitive on 12 points, this vertex must be transposed by the action of x. Diagrams a) and c) in Figure 6.10 illustrate the two possible positions of s_f . The choice of S is dependent on the choice of s_f . For a fixed choice of the position of s_f (and consequently a fixed choice of the position of S), the image of S under the action of $(xy^2z)^2$ is investigated. Write $\mathcal{V} = (xy^2z)^2(S)$. There are 4 possible configurations and they are illustrated in Figure 6.10:



Figure 6.10: Images of S under the action of $(xy^2z)^2$

Connectivity excludes the cases a) and c), since in those cases it is impossible to adjoin an action of xy^2 such that $xy^2z(\mathcal{V}) = S$. In cases b) and d), the action of z fixes \mathcal{V} , so $xy^2z(\mathcal{V}) = S$ if and only if the action of z transposes two points permuted by y, which is impossible. So no transitive permutation diagram of an action of Γ^+ on 12 points can be constructed by amalgamating two A_5 actions with one of the actions being the $\{5, 5, 1, 1\}$ permutation representation of A_5 on 12 points. 4) One of the partitions is $\{5, 6, 1\}$: Since this contains 3 points stabilized by y, this partition can only be combined with partitions of type $\{5, 6, 1\}$ or $\{10, 1, 1\}$. Suppose first that two diagrams of the form $\{5, 6, 1\}$ are used. Then transitivity forces a diagram containing the subscheme illustrated in Figure 6.11, with triangle 4 amalgamated either with triangle 2 or with triangle 3. However, no amalgamation of triangles 2 and 3 or triangles 2 and



Figure 6.11: Partial subscheme containing the $\{5, 6, 1\}$ Higman diagram

4 satisfies the condition $(xy^2z)^3(s) = s$. Consequently all resulting permutation diagrams will fail to satisfy $(xy^2z)^3 = 1$. Suppose now that diagrams of the form $\{5, 6, 1\}$ and $\{10, 1, 1\}$ are used. Then the two vertices representing trivial actions of y' and z must be swapped by an action of x, while the vertex representing the trivial action of x and y must be swapped by an action of z. So a diagram as illustrated in Figure 6.12 is recovered, where triangle 4 is superimposed on one of triangles 1, 2 or 3.



Figure 6.12: Partial subscheme containing the $\{5, 6, 1\}$ Higman diagram

Now, the action of $(xy^2z)^2$ sends s to $z(v_1)$. If z fixes v_1 , then $(xy^2z)^3(s) = z(v_2) \neq s$, while if z swaps v_1 , then a transitive permutation representation on 12 points must contain a subdiagram as shown in Figure 6.13, which is impossible.



Figure 6.13: Diagram 3

So no transitive permutation diagram of an action of Γ^+ on 12 points can be constructed by amalgamating two A_5 actions with one of the actions being the $\{5, 6, 1\}$ permutation representation of A_5 on 12 points.

5) One of the partitions is $\{6, 6\}$: Since this contains no points stabilized by y, this diagram can only be combined with a diagram of the form $\{6, 6\}$ or a diagram of the form $\{12\}$. Suppose first that two diagrams of the form $\{6, 6\}$ are used. Let Figure 6.14 represent the action of A_5 on 12 points. Then the action of A'_5 on 12 points is similar. To find all amalgamated diagrams \widetilde{D} satisfying the relations $A_5 *_{\{y=y'\}} A'_5$ and the condition $(xy^2z)^3 = 1$, it suffices to consider all ways of adding the action of z to the diagram \mathcal{D} in Figure 6.14 so that, for any vertex \mathcal{V} in \mathcal{D} , $(xy^2z)^3(\mathcal{V}) = \mathcal{V}$.



Figure 6.14: Diagram \mathcal{D} : A_5 action of type $\{6, 6\}$ on 12 points

Suppose first that the action of z is to swap s_1 with some vertex v_i . If it assumed that z also swaps s_2 and s_3 , then $(xy^2z)^2$ sends s_4 to $z(s_6)$. If z fixes s_6 , then $(xy^2z)^3$ sends s_4 to $z(s_5)$. Hence $(xy^2z)^3(s_4) = z(s_5)$ and this is s_4 if and only if z swaps s_4 with s_5 . Since the diagram for the action of y and z is of type $\{6, 6\}$, this forces z to swap s_6 with some v_j , contradicting the assumption that it fixes s_6 . So z swaps s_6 with some v_j . But then $(xy^2z)(s_4) =$ $z(s_6) = v_j$, for some j and it is impossible for $xy^2z(v_j) = s_4$. So z cannot swap s_2 and s_3 .

Let z fix s_2 and s_3 . Then $(xy^2z)^2(s_4) = s_3$, so $(xy^2z)^3(s_4) = s_4$ if and only if $xy^2z(s_3) = z(s_6) = s_4$. Hence z swaps s_4 and s_6 . This forces $z(s_5) = v_i$, for

some i. But then $(zy^2z)^2(s_5) = v_i$ and $xy^2z(v_i) = s_5$ forces $z(s_5) = v_1$. Then $(xy^2z)^3(v_2) = v_2$ if and only if z swaps v_4 with v_6 . Consequently z swaps s_1 and v_5 . The resulting group action is illustrated in Figure 6.15 and describes the group $A_5 \rtimes (C_2 \times C_2 \times C_2 \times C_2)$



Figure 6.15: Action of Γ^+ on 12 points

By symmetry arguments, there is also only one solution if it is assumed that z swaps s_2 with some v_i . The last case is if z swaps s_3 with some v_i . Suppose then that z swaps s_1 and s_2 . Then $(xy^2z)^2(s_1) = z(s_3)$, a contradiction, since there is no way of satisfying $xy^2z(v_i) = s_2$, for any i. So z fixes s_1 and s_2 . Then $xy^2z(s_2) = z(s_3)$, and since $xy^2z(s_4) = s_3$, z must swap s_4 with some v_j . Now suppose that z swaps s_5 and s_6 . Then $(xy^2z)^2(s_3) = z(s_4)$, or $xy^2z(z(s_4)) = s_3$. Similarly, $(xy^2z)^2(s_4) = z(s_3)$, or $xy^2z(z(s_3)) = s_4$. Combining these results, we get $s_4zxy^2z = s_3$ and $s_3zxy^2z = s_4$, so $s_4zxy^2z^2xy^2z = s_4$. Since $x = \alpha\beta = ac$, $y = \beta = bc$ and $z = \beta\gamma = bd$, the identity $zxy^2z^2xy^2z = 1$ becomes $zxy^2xy^2z = bdacbcbcacbcbcbd = bdababbd = (\alpha^2)^{\beta\gamma} = 1$, a contradiction. So z does not swap s_5 and s_6 . Then $(xy^2z)^2(s_6) = z(s_4)$, again a contradiction.

Instead let the second partition be of type $\{12\}$. Let Figure 6.16 represent the action of A_5 on 12 points and Figure 6.14 represent the action of A'_5 . Suppose first that z fixes s_1 and s_2 , as shown in Figure 6.16:



Figure 6.16: Diagram 5

If z fixes v_1 , then $(xy^2z)(v_1) = s_2$, so z must transpose s_4 and s_5 . Consequently, $(xy^2z)^2(s_3) = z(s_3)$. If z fixes s_3 then $(xy^2z)^3(s_4) = v_1 \neq s_2$, so z

cannot fix s_3 . Since $(xy^2z)(z(s_3)) = s_3$, z must transpose v_2 and s_3 . But then $(xy^2z)^3(v_3) = v_4 \neq v_3$. So v_1 cannot be fixed by z. Then there are only 3 possible partial diagrams with v_1 transposed by z, shown in Figure 6.17:



Figure 6.17: Possible diagrams

In I), $(xy^2z)^2(v_2) = z(v_3)$. If $z(v_3) = v_3$, then $(xy^2z)^3(v_2) = xy^2z(v_3) = s_3 \neq v_2$, so $z(v_3) = v_2$. But then $(xy^2z)^3(v_2) = v_4 \neq v_2$. In a similar manner, diagram II) fails (since $(xy^2z)^2(v_3) = v_3$) and diagram III) fails (there is no way to complete the diagram so that $(xy^2z)^3(s_6) = s_6$). Hence z cannot fix s_1 and s_2 .

Now suppose that z transposes s_1 and s_2 . Then there are two diagrams, illustrated in Figure 6.18, in which all conditions are met:



Figure 6.18: Admissible diagrams

Diagram IV) describes an action of $A_5 \ltimes (C_2 \times C_2 \times C_2 \times C_2 \times C_2)$ on 12 points, and diagram V) describes an action of A_{12} on 12 points.

The remaining case is when z fixes one of s_1 or s_2 . By symmetry arguments, it suffices to consider when z fixes s_2 and transposes s_1 with some other point. Calculations similar to those above show that there are two admissible diagrams, both yielding an action of Γ^+ on 12 points. In each case the kernel in Γ^+ of this action is a normal subgroup whose factor group is $A_5 \times ((C_2 \times C_2 \times C_2 \times C_2 \times C_2))$.

- 6) One of the partitions is $\{10, 1, 1\}$: Since this contains 3 points stabilized by y, this partition can only be combined with partitions of type $\{5, 6, 2\}$ or of type $\{10, 1, 1\}$. By 4) above, it suffices to consider only the case $\{10, 1, 1\}$. Since there are two points with trivial action and only one point fixed by the action of y but permuted by x or z, it is clear that there is no way of combining these diagrams to get a transitive representation on 12 points.
- 7) One of the partitions is {12}: The only remaining case is to combine this with {12}. Calculations similar to those done in the previous sections give 6 possible diagrams:



Figure 6.19: Transitive action of A_5 on 12 points with generators x, y and z



Figure 6.20: Transitive action of A_5 on 12 points with generators x, y and z



Figure 6.21: Transitive action of A_5 on 12 points with generators x, y and z

•



Figure 6.22: Transitive action of $L_2(11)$ on 12 points with generators x, y and z



Figure 6.23: Transitive action of M_{12} on 12 points with generators x, y and z



Figure 6.24: Transitive action of M_{12} on 12 points with generators x, y and z

The first three diagrams, Figures 6.19, 6.20 and 6.21, correspond to transitive actions of A_5 on 12 points and arise from the three maps $\Gamma^+ \to A_5$ constructed in Lemma 6.9.1. The fourth diagram, Figure 6.22, corresponds to a transitive action of $L_2(11)$ on 12 points. The associated epimorphism, $\Gamma^+ \to L_2(11)$, was constructed in Lemma 6.9.3. The final two diagrams. Figures 6.23 and 6.24, correspond to transitive actions of M_{12} on 12 points. the last two diagrams describe different epimorphisms, since the element $\alpha^2\gamma = (xy^2)^2y^2z$ has order 4 for the representation described in Figure 6.23, and order 6 in the representation described in Figure 6.24.

As a corollary, from the representation described in diagram V), the following result is immediate:

Corollary 6.9.1 The group Γ^+ has a unique normal subgroup with factor group A_{12}

-

6.10 Concluding remarks

In this chapter, examples of manifolds arising from torsion-free subgroups of the 9 Lannér groups were constructed. Using the isometry groups of these manifolds, combinatorial descriptions of several of the manifolds were given, taking advantage of the size of the isometry groups. In some interesting cases, the induced action of these isometries on the first homology was computed. This additional information was then used to investigate the construction of arbitrarily large manifolds whose isometry group is also large.

For each group, a minimal index torsion free subgroup was determined. The orientation preserving subgroups of the groups T_1 , T_2 , T_3 , T_5 and T_8 have non-normal minimal index torsion-free subgroups, while orientation preserving subgroups of the groups T_4 , T_6 , T_7 and T_9 all have minimal index torsion-free normal subgroups.

Chapter 7

The 4-dimensional compact simplicial Coxeter groups

7.1 Motivation

In 1985, Mike Davis constructed his celebrated "Davis manifold" [Da], providing the first explicit construction of a four-dimensional hyperbolic manifold. This manifold arose from the existence of a torsion free subgroup of index 14400 in the group $\Gamma_4 = [5, 3, 3, 5]$, shown in Table 7.1. The manifold has Euler characteristic 26. By the Gauss-Bonnet theorem [KZ], the volume of a complete finite volume hyperbolic 4-manifold \mathcal{M} is given by $\operatorname{Vol}(\mathcal{M}) = \frac{4\pi^2}{3}\chi(\mathcal{M})$: thus the volume of the Davis Manifold is $26 \times \frac{4\pi^2}{3}$.

Since then, an outstanding question has been whether or not Γ_1 has a minimal index torsion free subgroup K. Since the least common multiple of the orders of conjugacy classes of maximal finite subgroups of Γ_1 is 14400, K must have index 14400. If such a subgroup does exist, then the associated manifold will have Euler characteristic 1 and therefore it would be an example of a smallest volume compact hyperbolic 4-manifold. This chapter will study the subgroup structure of Γ_1 and seek to outline a possible approach to the construction of a smallest volume compact hyperbolic 4-manifold. The five co-compact four-dimensional hyperbolic simplicial Coxeter groups are listed in Table 7.1.



Table 7.1: The 5 co-compact simplicial Coxeter groups with a hyperbolic 4-simplex as a fundamental region

The groups Γ_1 and Γ_2 deserve particular attention. In his paper, Davis [Da] constructed an explicit epimorphism from Γ_4 onto the [5, 3, 3] Coxeter group. This group is the full symmetry group of the 120-cell, first introduced in § 5.8. The Davis manifold has Euler characteristic 26. In his concluding remarks, Davis notes that Γ_1 and Γ_2 have rational Euler characteristic $\frac{1}{14400}$ and $\frac{17}{28800}$, respectively, and speculates as to the existence of torsion free subgroups of indices 14400 and 28000 in these groups. Such subgroups would yield compact hyperbolic manifolds of Euler characteristic 1 and 17, respectively. The analogous problem for noncompact manifolds has been solved by John Ratcliffe and Steven Tschantz in [RT1]. The manifolds they constructed arise from torsion-free subgroups of index 16 × 120 of the Coxeter group shown in Table 7.2. More recently, using computational



Table 7.2: Simplicial Coxeter group with Non-co-compact simplex Δ^4 in \mathbb{H}^4

methods, Marston Conder and Colin Maclachlan [CM] constructed a torsion free subgroup of index 8×14400 in Γ_1 . This corresponds to the smallest volume co-compact hyperbolic 4-manifold currently known.

In this chapter a different approach is used to construct subgroups of Γ_1 of index 8×14400 . The subgroups constructed are different to those of Conder and

Maclachlan. Additionally, some obstructions to finding a torsion free subgroup of index 14400 are presented. The method applied combines computational techniques with classical group theory, and in particular makes use of the classification of the finite simple groups. Furthermore, it provides a method to answer the question whether Γ_1 has a torsion free subgroup of index 14400.

7.2 Conjugacy class representatives

Any attempt to construct a manifold from the action of a subgroup of Γ_i on \mathbb{H}^4 requires information on the conjugacy classes of torsion elements: namely what they are and how they are avoided. Representatives of the conjugacy classes of torsion elements in Γ_1 are characterised using Theorem 2.2 in Chapter 2

This identifies the torsion elements, up to conjugacy. Therefore, for Γ_1 , any conjugacy class of elements of finite order lies in one of the following five special subgroups listed in Table 7.3

Table 7.3: Maximal finite subgroups of Γ_1

It suffices to consider only the conjugacy classes of elements of prime order. The conjugacy classes for each maximal subgroup G_1, G_2, G_3, G_4 and G_5 are summarised in Table 7.4, Table 7.6, Table 7.5, Table 7.7 and Table 7.8, respectively.

Order of representative	Class representative
2	a
2	ac
2	$(abc)^5$
2	$(abcd)^{15}$
3	bc
3	$(abcd)^{10}$
5	ab
5	$(ab)^2$
5	$(abcd)^6$
5	$(abcd)^{12}$
5	$(abcd)^{10}(abc)^{-2}$

Table 7.4: Conjugacy classes of elements of prime order in $G_1 = [5, 3, 3]$

Order of representative	Class representative
2	a
2	d
2	ad
3	de
5	ab
. 5	$(ab)^{2}$

Table 7.5: Conjugacy classes of elements of prime order in $G_3\cong D_{10}\times D_6$

Order of representative	Class representative
2	е
2	a
2	ae
2	ac
2	ace
2	$(abc)^5$
2	$e(abc)^5$
3	bc
5	ab
5	$(ab)^{2}$

Table 7.6: Conjugacy classes of elements of prime order in $G_2 = [5,3] \times C_2 \cong A_5 \times C_2 \times C_2$

Order of representative	Class representative
2	a
2	c
2	ce
2	ac
2	ace
3	cd

Table 7.7: Conjugacy classes of elements of prime order in $G_4\cong C_2\times S_4$

Order of representative	Class representative
2	b
2	bd
3	bc
5	bcde

Table 7.8: Conjugacy classes of elements of prime order in $G_5\cong S_5$

The elements listed in the previous tables account for all conjugacy classes of torsion of prime order in Γ . However, in many cases the same class has been counted more than once. For example, a and e represent different conjugacy classes of elements of order 2 in G_3 . However, a is conjugate to b in G_1 while b is conjugate to e in G_5 , so a and e represent the same conjugacy class in Γ . After removing multiple references for the same conjugacy class, representatives of the conjugacy classes of torsion of prime order in Γ are given by the following elements:

Order 2: a, ac, ace, (abc)⁵, (abc)⁵e and (abcd)¹⁵.
Order 3: bc and (abcd)¹⁰.
Order 5: ab, (ab)², bcde, (abcd)⁶, (abcd)¹² and (abcd)¹⁰(abc)⁻².

7.3 Subgroups of the [5, 3, 3, 3] Coxeter group

As previously observed, torsion free subgroups of the [5, 3, 3, 3] Coxeter group Γ correspond to hyperbolic 4-manifolds. If Γ were to contain a torsion free subgroup of index 14400, then the associated hyperbolic 4-manifold would have Euler characteristic 1, and would thus be an example of a smallest volume compact hyperbolic 4-manifold. In this section, examples of the smallest currently known manifolds are constructed. It will be shown that these examples differ from those constructed in [CM].

7.3.1 Low index subgroups of Γ

Using Lowx [Lx], a complete list of conjugacy class representatives of subgroups of [5, 3, 3, 3] of index at most 720 was obtained. The number of such subgroups and their index in each case is summarised in Table 7.9

Index	# classes	Index	# classes	Index	# classes
2	1	385	1	595	6
85	2	420	1	600	17
120	2	425	2	624	4
136	2	436	1	625	1
156	2	445	1	640	2
170	4	480	18	641	1
240	10	505	1	650	7
255	3	510	21	660	1
272	6	521	1	675	2
300	1	540	1	676	1
312	4	544	8	680	15
325	1	555	2	685.	4
340	4	556	1	691	2
360	2	565	1	720	13

Table 7.9: Number of classes of subgroups of index \leq 720 in Γ

The data in Table 7.9 will be used to prove that if K is a torsion free normal subgroup of Γ , then Γ/K contains no sporadic simple group, nor any extension of one, among its epimorphic images. This first result proves that the construction used in Davis paper [Da] cannot be used for this group.

Proposition 7.3.1 There exists no epimorphism $\varphi : \Gamma \mapsto \Gamma_0 := [5,3,3]$ with torsion free kernel.

Proof: Suppose there exists such a map. Let a, b, c, d, e be the standard Coxeter generators for $\Gamma = [5, 3, 3, 3]$, and define $\varphi(a) := \underline{a}, \varphi(b) := \underline{b}, \varphi(c) := \underline{c}, \varphi(d) := \underline{d}$. Then $\underline{a}, \underline{b}, \underline{c}, \underline{d}$ form a generating set for Γ_0 which satisfies the Coxeter relations for Γ_0 , and so form a set of Coxeter generators for Γ_0 . It remains to find an image for e under φ such that $\underline{a}\varphi(e), \underline{b}\varphi(e)$ and $\underline{c}\varphi(e)$ all have order 2 and that $\underline{d}\varphi(e)$ has order 3. Recalling the decomposition in § 5.9 of the 120-cell into two linked tori consisting of 60 dodecahedra, one of these tori can be cut along a meridian curve to recover a tower of dodecahedra, which contains a core tower of 10 stacked dodecahedra. Adjacent dodecahedra share a horizontal pentagonal face, and at each edge a further dodecahedron is attached. The outer face of the tower consists of the 200 free faces of the 50 dodecahedra wrapped around the central tower. The core tower is stabilised by the dihedral group D_{10} generated by <u>a</u> and <u>b</u>. Hence e must map to some element of order 2 with $|\underline{ag}| = |\underline{bg}| = 2$. In § 5.9 it has been proved that any element g of order 2 in Γ_0 which satisfies $|\underline{ag}| = |\underline{bg}| = 2$ lies in a dihedral group D_{20} which stabilizes the meridian curve as illustrated in Figure 5.12. There are 11 possible involutions w_i lying in this subgroup. Three of them, $(\underline{ababc})^3$, d^w , where $w = (\underline{cbabcd})^2(\underline{abc})^3\underline{abdc}$ and $(\underline{abcd})^{15}$, satisfy $|\underline{cg}| = 2$. However, their product with <u>d</u> has orders 10, 5 and 2, respectively. Since it was required that <u>de</u> have order 3, this proves that no such map exists.

Proposition 7.3.2 Let w = abcde in Γ . Then for i < 25, $i \neq 17$, the following results hold:

- 1) $\langle \langle w^i \rangle \rangle_{\Gamma} = \Gamma$ if *i* is odd.
- 2) $\langle \langle w^i \rangle \rangle_{\Gamma} = \Gamma^+$ if *i* is even.
- For i = 17, Γ/⟨⟨wⁱ⟩⟩_Γ ≅ S₄(4), the 4-dimensional simple symplectic group over the field of 4 elements. ⟨⟨wⁱ⟩⟩_Γ is the unique such subgroup with this quotient. The kernel of the map Γ → Γ/⟨⟨wⁱ⟩⟩_Γ contains the conjugacy class C((abcd)¹⁵) of torsion elements.

Proof: The first two results have been verified computationally. The third result was obtained as follows: Γ has two conjugacy classes of subgroups of index 85. These were obtained by using the program Lowx. The induced action of Γ on the cosets of a representative of one of these classes gives a permutation representation $\theta: \Gamma \to S_{85}$. The resulting group was found, using GAP, to be the simple group $S_4(4)$. The Atlas of simple groups [Atlas] lists $S_4(4)$ as having two conjugacy classes of maximal subgroups of index 85. These lift, under the inverse of θ , to representatives of the two conjugacy classes of subgroups of index 85 in Γ . Hence the kernel of θ is unique.

7.3.2 Maps to $S_4(4)$

Let Γ act on the cosets of a representative of either of the two conjugacy classes of subgroups of index 85. This action induces a permutation representation θ : $\Gamma \rightarrow S_4(4)$. From the Atlas of Finite Groups [Atlas], it can be seen that $S_4(4)$ has two conjugacy classes of subgroups of index 85, so these subgroups lift to two conjugacy classes of subgroups of index 85 in Γ . Since Γ has only two conjugacy classes of subgroups of index 85, they must coincide with the lifted subgroups. Since $S_4(4)$ is simple, the core must be the same in each case, giving the epimorphism $\theta: \Gamma \twoheadrightarrow S_4(4)$. Write $K = \ker(\theta)$. GAP [GAP] was used to study the actions of each conjugacy class of torsion elements of Γ on the cosets of a subgroup of index 85 in Γ . It was discovered that the conjugacy class $\mathcal{C}((abcd)^{15})$ of elements of order 2 in Γ acts with fixed points on the 85 cosets, and hence lies in K. The group $S_4(4)$ contains conjugacy class of Sylow-17 groups Syl_{17} which have index 4×14400 in $S_4(4)$. The normalizer $N_{S_4(4)}(Syl_{17}) = N(Syl_{17})$ has index 14400 in $S_4(4)$ and is an extension of Syl_{17} by a cyclic group of order 4. Any element of order 2 lying in $N(S_{17})$ acts faithfully by conjugation, since otherwise it would commute with Syl_{17} and hence $S_4(4)$ would have an element of order 34, a contradiction. GAP was used to study the images of representatives of conjugacy classes of torsion elements of Γ . It was found that elements of the conjugacy class represented by ac act by inversions on Syl_{17} , therefore the preimage $N(S_{17})K$ contains elements of the conjugacy class $\mathcal{C}(ac)$.

The preimage of the group Syl_{17} under θ clearly cannot contain any of the torsion elements and is an extension of K by Syl_{17} . This group was constructed using GAP and the conjugacy classes of index 2 subgroups were determined. There are fifteen such conjugacy classes of subgroups, 8 of which avoid the conjugacy class $C((abcd)^{15})$. Their abelianisation is one of the three following types:

$$\mathbb{Z}_4 \oplus \mathbb{Z}_4 \oplus \mathbb{Z}_4 \oplus \mathbb{Z}_{17} \tag{7.1}$$

$$\mathbb{Z}_2 \oplus \mathbb{Z}_2 \oplus \mathbb{Z}_2 \oplus \mathbb{Z}_4 \oplus \mathbb{Z}_4 \oplus \mathbb{Z}_{17} \tag{7.2}$$

$$\mathbb{Z}_2 \oplus \mathbb{Z}_2 \oplus \mathbb{Z}_2 \oplus \mathbb{Z}_4 \oplus \mathbb{Z}_8 \oplus \mathbb{Z}_{17}$$
(7.3)

These were found by constructing an explicit presentation for H, and hence for
the index 2 subgroups.

(a) A subgroup with abelianisation $\mathbb{Z}_4^3 \oplus \mathbb{Z}_{17}$

Let $H_1 < \Gamma$ be the subgroup generated by the following elements (these generators were found using GAP):

cdcbabacbdedcbabcdabcababcdebabc, abacbabadedcbabacbabcdecdbcababcdbc, bacbedcbabacbadcbabedcbabcdecababcdbca, abacbabacbabcdcbaedcbabcdabcababcdedbca, ababadcbabacbdcedcbabcdeabcdabcababcdebcd, babacdcbabacbadedcbabacdebcababcdcbabcababc

Then H_1 has index 8×14400 in Γ . Letting Γ act by right multiplication on the cosets of H_1 gives an induced permutation representation on 8×14400 points. Now H_1 contains an element of a conjugacy class C(g) of Γ if and only if the induced permutation action of g fixes some points. Using GAP, the induced actions of representatives of each of the conjugacy classes of Γ were computed. Since no representative of a conjugacy class in Γ fixed a point, H_1 avoids all conjugacy classes of torsion elements. The resultant manifold has Euler characteristic 8 and therefore has volume $\frac{32\pi^2}{3}$. The abelianisation of H, $H_1/[H_1, H_1]$, is a finite group of type $\mathbb{Z}_4^3 \oplus \mathbb{Z}_{17}$. Since the second generator has odd length, so H_1 is not a subgroup of Γ^+ .

(b) A subgroup with abelianisation $\mathbb{Z}_2^3 \oplus \mathbb{Z}_4^2 \oplus \mathbb{Z}_{17}$

Let $H_2 < \Gamma$ be the subgroup generated by the following elements (these generators were found using GAP):

ababdcbabecdcabab, bcbabacbdcbabedcbabcababcdecdbabc, adcbaedcbabacbabdcbabacdedbcababcdabcababcba, bcbabacdcbabacbabcdcbabecdcbabcdbabcdbca, acbabcdcbabacbadcedcbabacbdcababcdebabcdbabcababcde Then H_2 has index 8×14400 in Γ . Letting Γ act by right multiplication on the cosets of H_2 gives an induced permutation representation on 8×14400 points. Now H_2 contains an element of a conjugacy class C(g) of Γ if and only if the induced permutation action of g fixes some points. Using GAP, the induced actions of representatives of each of the conjugacy classes of Γ were computed. Since no representative of a conjugacy class in Γ fixed a point, H_2 avoids all conjugacy classes of torsion elements. The resultant manifold has Euler characteristic 8 and therefore has volume $\frac{32\pi^2}{3}$. The abelianisation of H_2 , $H_2/[H_2, H_2]$, is a finite group of type $\mathbb{Z}_2^3 \oplus \mathbb{Z}_4^2 \oplus \mathbb{Z}_{17}$. Since the first generator has odd length, so H_2 is not a subgroup of Γ^+ .

(c) A subgroup with abelianisation $\mathbb{Z}_2^3 \oplus \mathbb{Z}_4 \oplus \mathbb{Z}_8 \oplus \mathbb{Z}_{17}$

Let $H_3 < \Gamma$ be the subgroup generated by the following elements (these generators were found using GAP):

Then H_3 has index 8×14400 in Γ . Letting Γ act by right multiplication on the cosets of H_3 gives an induced permutation representation on 8×14400 points. Now H_3 contains an element of a conjugacy class $\mathcal{C}(g)$ of Γ if and only if the induced permutation action of g fixes some points. Using GAP, the induced actions of representatives of each of the conjugacy classes of Γ were computed. Since no representative of a conjugacy class in Γ fixed a point, H_3 avoids all conjugacy classes of torsion elements. The resultant manifold has Euler characteristic 8 and therefore has volume $\frac{32\pi^2}{3}$. The abelianisation of H_3 , $H_3/[H_3, H_3]$, is a finite group of type $\mathbb{Z}_2^3 \oplus \mathbb{Z}_4 \oplus \mathbb{Z}_8 \oplus \mathbb{Z}_{17}$. Since the first generator has odd length, so H_3 is not a subgroup of Γ^+ .

7.3.3 Maps to $S_4(5)$

There are two conjugacy classes of subgroups of index 156 in Γ . Letting Γ act on the cosets of a representative of either class produces a permutation representation $\Gamma \to H_4 < S_{156}$ into the symmetric group on 156 points. GAP was used to construct the map, and the group H_4 was discovered to be the group $S_4(5).2$, a group containing $S_4(5)$ as a normal subgroup of index 2. $S_4(5)$ is a simple symplectic group of order $2^{6}.3^{2}.5^{4}.13$. The group $S_4(5).2$ is its automorphism group and is isomorphic to $SO_5(5)$

Proposition 7.3.3 $S_4(5)$ contains no subgroup of index 7200.

Proof: A subgroup S of index 7200 would have order $650 = 2.5^2.13$, so Sylow's theorems imply that S has a unique and hence normal Sylow 13-subgroup C_{13} . Since C_{13} has no automorphism of order 5, elements of order 5 and 13 in S must commute, giving elements of order 65. But $S_4(5)$ has no elements of this order.

7.4 Characterising maps to simple groups

Theorem 7.1 Let $H < \Gamma$ have index 14400 and suppose that H is torsion free. Let $K \triangleleft \Gamma$ be the core of H in Γ . Then Γ/K does not admit a map onto any of the sporadic simple groups.

Proof: Suppose it did. Let \overline{H} be the image of H in $\overline{\Gamma} = \Gamma/K$ and suppose that $\varphi: \Gamma/K \to G$ is a map from the factor group $\overline{\Gamma}$ to some sporadic simple group G. Then the image of \overline{H} under φ is a subgroup of index dividing 14400 in a sporadic group and avoids the images of all the conjugacy classes of torsion. Using the list of all conjugacy classes of subgroups of Γ with index ≤ 720 (Table 7.9), first look at all sporadic simple groups S with maximal subgroups of index less than 720. These are $M_i, i \in \{11, 12, 22, 23, 24\}, J_2, J_3, HS, McL$ and Co_3 . GAP was used to compute the actions of Γ on representatives of the conjugacy classes in each case. None of the above sporadic groups appeared in the resulting list of permutation groups. Among the remaining groups, $Co_1, Fi_{23}, Fi'_{24}, HN, Th, B, M O'N, Ly$

and J_4 contain no subgroup of index ≤ 14400 (see chapter 5 of [W]), and are therefore eliminated. The remaining groups are Suz, Co_2 , He, Fi_{22} , J_3 and Ru.

Suz contains one class of maximal subgroups of index < 14400. A representative of this class has index 1782 in Suz. Therefore Suz contains no subgroup of index 14400 since 1782 does not divide 14400. Co_2 contains one class of maximal subgroups of index ≤ 14400 . A representative of this class has index 2300 in Co_2 , and as a result it also cannot contain a subgroup of index 14400. He contains two classs of maximal subgroups of index \leq 14400: one class has index 2058 and has representative $S_4(4)$: 2, the other has index 8330 with representative $2^2 L_3(4) S_3$. Neither 2058 nor 8330 divide 14400. Therefore the respective subgroups cannot contain A subgroup of index 14400. Fi_{22} contains three classes of maximal subgroups of index ≤ 14400 : two classes of type $O_7(3)$ with index 14080 and one class of type $2.U_6(2)$ with index 3510. Because 14080 and 3510 do not divide 14400, Fi_{22} contains no subgroups of index 14400. Similarly, J_3 has one class of maximal subgroups of index less than 14400, and because this class has index 6156 in J_3 , J_3 has no subgroup of index 14400. Finally, Ru also has one class of maximal subgroups of index less than 14400. This class has index 4060 in Ru, therefore Rualso contains no subgroup of index 14400.

Chapter 8

Conclusion and further work

8.1 Conclusion

In this work all torsion free normal subgroups of the index two orientation preserving subgroups of the Lannér groups whose factor groups have the form $PSL_2(q)$, where $q = p^n$ and p is a prime, are classified. In the case of each group, some examples of manifolds were constructed and their homology computed. Minimal index torsion free subgroups of each Lannér group are also constructed. The computational techniques developed in this work were applied to the 4-dimensional Coxeter group [5,3,3,3] to study minimal volume compact hyperbolic manifolds.

In chapter 3, subgroups of the orientation preserving subgroup Γ^+ of the Lannér group $\Gamma = T_4 = [5,3,5]$ are studied. All such subgroups whose factor group is $L_2(q)$ are classified. The conditions under which these subgroups are normal in any extension of Γ^+ in Isom(\mathbb{H}^3) is also investigated. The special case p = 5 yields two normal subgroups N_1 and N_2 , whose associated manifolds $\mathcal{M}_{\gamma} = \mathbb{H}^3/N_i$ are a chiral pair of Seifert-Weber manifolds.

Chapter 4 generalises the results of chapter 3 to the other 8 Lannér groups. The classification of such subgroups, in the case of $\Gamma = T_2$, was previously described by Anna Torstensson in her PhD thesis [To]. A statement of her results was included for completeness. A classification of all torsion-free normal subgroups $N \triangleleft \Gamma^+ \triangleleft \Gamma =$

 T_i of the remaining 7 Lannér groups T_i , whose factor groups are of the form $L_2(q)$, is given. Some conditions are also provided under which such subgroups N remain normal in an extension of Γ^+ in Isom(\mathbb{H}^3).

As well as the theoretical results of chapters 3 and 4, explicit examples of some of the manifolds were given. In chapter 5, several of the manifolds arising from torsion-free subgroups $N \triangleleft \Gamma^+ = [5, 3, 5]^+$ were constructed. A combinatorial description of these manifolds, using the group theoretic structure of their isometry groups, was given. Interesting chiral pairs of manifolds were discovered, and their chirality exhibited using the subgroup structure of Γ .

In chapter 6, manifolds exhibiting a large degree of symmetry which arise from the other 8 Lannér groups were investigated. The smallest manifolds with a large degree of symmetry which were associated to the Lannér group $\Gamma = T_2$ were described in a paper of Jones and Mednykh [JM]. A statement of their main results was included for completeness. For each of the remaining 7 Lannér groups, a brief description of the associated manifolds \mathcal{M} is given. In the case of both the Lannér groups T_5 and T_9 , the smallest index torsion-free normal subgroup $N \cong \pi_1(\mathcal{M})$ has a free abelianisation $N/[N, N] \cong H_1(\mathcal{M})$. The action of the isometry group Isom(\mathcal{M}) on this free Z-ring is computed. As a result a construction for an arbitrarily large hyperbolic 3-manifold with a large isometry group is given.

Computational techniques were developed to construct complete lists of conjugacy classes of subgroups of low index in these groups. A summary of these results is provided in the Appendices. These lists were then used to test the theoretical results proved in this thesis and also to search for specific normal subgroups whose factor groups were of interest.

The computational techniques developed in this work were applied to the 4dimensional Coxeter group [5, 3, 3, 3] and a study was done of the low index subgroups of this group. The existence of torsion free subgroups of index 115200 was established and a possible approach towards determining the minimal index torsion-free subgroup of this group is outlined.

8.2 Further work

Based on the work of this thesis, there are several interesting avenues of research that can be pursued in the future. Chapter 6 includes only a selection of the results obtained from the Lannér groups. It would be interesting to study more of these manifolds, providing researchers in hyperbolic geometry and abstract polytopes with a library of "ready-made" manifolds with which to test hypotheses. Using the tables in Appendix A and the methods described in chapters 5 and 6, many more interesting manifolds can be identified and constructed combinatorically.

The methods used in chapters 3 and 4 can also be applied to the 23 non-compact simplicial Coxeter groups acting on \mathbb{H}^3 . They are often referred to as "Quasi-Lannér Groups". Vinberg has shown in [V1] that 6 of these are non-arithmetic, so torsionfree subgroups of these groups can be used to construct non-arithmetic manifolds. If, further, the subgroups are normal, then the manifolds will exhibit a high degree of symmetry.

Generalising to higher dimensions, the combination of group theory and computational techniques used in chapter 7 can be extended to characterise examples of cusped hyperbolic manifolds in dimensions 6 and 8. By combining computational techniques and classical group theory, it becomes possible to construct explicit subgroups avoiding torsion.

Additionally, it would be interesting to determine the full isometry group in $\operatorname{Isom}(\mathbb{H}^n)$ of each of these groups, and to determine subgroups of Γ that are torsion free whose associated manifolds \mathcal{M} have maximal symmetry. Note that, in \mathbb{H}^3 at least, the isometry group of \mathcal{M} is still finite.



Tables of low index subgroups for the simplicial groups

A.1 $\Gamma = T_1[2, 2, 3; 3, 5, 2]$

 Γ^+ has 3359 conjugacy classes of low index subgroups of index ≤ 60 . Their index, the induced permutation group and, where possible, the abelianisation of the kernel of the map are listed here.

Index	# classes	Induced action	Abelianization
5	1	A_5	\mathbb{Z}_2^{11}
6	1	A_5	\mathbb{Z}_2^{11}
10	1	A_5	\mathbb{Z}_2^{11}
11	4	$PSL_{2}(11)$	$\mathbb{Z}_2^{10}\oplus\mathbb{Z}_{11}^3$
12	9	$\begin{cases} 1 \Rightarrow A_5 \\ 2 \Rightarrow L_2(11) \\ 2 \Rightarrow A_5 \ltimes H_1 \\ 2 \Rightarrow A_5 \ltimes H_1 \\ 2 \Rightarrow A_5 \ltimes H_1 \end{cases}$	$egin{array}{c} \mathbb{Z}_2^{11} \ \mathbb{Z}_2^{10} \oplus \mathbb{Z}_{11}^3 \ \mathbb{Z}_2^{15} \oplus \mathbb{Z}_2^{14} \ \mathbb{Z}_2^{27} \ \mathbb{Z}^{31} \end{array}$
15	1	A_5	\mathbb{Z}_2^{11}
16	4	$A_5 \ltimes H_2$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^2$

Index	# classes	Induced action	Abelianization
18	3	$A_5 \ltimes H_3$	not computed
		$\left(1 \Rightarrow A_5 \right)$	\mathbb{Z}_2^{11}
		$4 \Rightarrow A_5 \ltimes H_2$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^2$
		$2 \Rightarrow A_5 \ltimes H_1$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^{14}$
20	29	$\begin{cases} 2 \Rightarrow A_5 \ltimes H_1 \end{cases}$	\mathbb{Z}^{27}
		$2 \Rightarrow A_5 \ltimes H_1$	\mathbb{Z}^{31}
		$2 \Rightarrow PSL_2(19)$	$\mathbb{Z}_2^{18}\oplus\mathbb{Z}_{19}^3$
		$\left(16 \Rightarrow A_5 \ltimes H_4 \right)$	not computed
22	1	A_{22}	not computed
		$(1 \Rightarrow A_5 \ltimes H_1)$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^{14}$
		$1 \Rightarrow A_5 \ltimes H_1$	\mathbb{Z}^{27}
		$1 \Rightarrow A_5 \ltimes II_1$	\mathbb{Z}^{31}
24	22	$\left\{ 4 \Rightarrow A_5 \ltimes H_5 \right.$	\mathbb{Z}^{43}
		$8 \Rightarrow A_5 \ltimes H_6$	not computed
1		$4 \Rightarrow A_5 \ltimes H_7$	not computed
r		$\begin{array}{c} 3 \Rightarrow A_5 \ltimes H_{24} \end{array}$	not computed
26	1	$L_2(25)$	\mathbb{Z}^{78}
28	1	$S_{6}(2)$	not computed
		$\int 1 \Rightarrow A_5$	711
		$4 \Rightarrow A_5 \ltimes H_2$	$\frac{2}{7/157/2}$
20	41	$4 \Rightarrow L_2(29)$	\square \square_2
- 50	41	$16 \Rightarrow A_5 \ltimes H_4$	not computed
		$3 \Rightarrow A_5 \ltimes H_8$	not computed
		$1 \Rightarrow A_5 \ltimes H_9$	
		$4 \Rightarrow A_5 \ltimes H_1$	$\mathbb{Z}^{15} \oplus \mathbb{Z}_2 14$
		$4 \Rightarrow A_5 \ltimes H_1$	\mathbb{Z}^{27}
32	22	$\left\{ 4 \Rightarrow A_5 \ltimes H_1 \right.$	\mathbb{Z}^{31}
		$2 \Rightarrow L_2(31)$	not computed
		$ 8 \Rightarrow A_5 \ltimes H_6 $	not computed

•

Index	# classes	Induced action	Abelianization
34	3	A ₃₄	not computed
		$\int 9 \Rightarrow A_5 \ltimes H_{21}$	not computed
		$1 \Rightarrow S_6(2)$	not computed
		$12 \Rightarrow A_5 \ltimes H_{20}$	not computed
		$1 \Rightarrow A_5 \ltimes H_{19}$	not computed
		$2 \Rightarrow A_5 \ltimes H_{18}$	not computed
		$12 \Rightarrow A_5 \ltimes H_{24}$	not computed
36	73	$\begin{cases} 12 \Rightarrow A_5 \ltimes H_{25} \end{cases}$	not computed
		$2 \Rightarrow \Lambda_5 \ltimes H_{17}$	not computed
		$4 \Rightarrow A_5 \ltimes H_{26}$	not computed
		$5 \Rightarrow A_5 \ltimes (\times_{i=1}^6 A_6)$	not computed
		$7 \Rightarrow A_5 \ltimes (H_1 \ltimes (\times_{i=1}^6 A_5))$	not computed
		$2 \Rightarrow A_5 \ltimes (\times_{i=1}^6 A_6)$	not computed
		$\left(4 \Rightarrow A_5 \ltimes (II_1 \ltimes (\times_{i=1}^6 A_6)) \right)$	not computed
38	2	A ₃₈	not computed
		$\begin{array}{c} 3 \Rightarrow A_5 \ltimes II_2 \end{array}$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^2$
		$2 \Rightarrow A_5 \ltimes H_1$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^{14}$
	64	$3 \Rightarrow A_5 \ltimes H_1$	\mathbb{Z}^{27}
		$3 \Rightarrow A_5 \ltimes H_1$	\mathbb{Z}^{31}
		$4 \Rightarrow A_5 \ltimes H_5$	\mathbb{Z}^{43}
40		$\begin{cases} 12 \Rightarrow A_5 \ltimes H_4 \end{cases}$	not computed
		$12 \Rightarrow A_5 \ltimes H_7$	not computed
		$12 \Rightarrow A_5 \ltimes H_{22}$	not computed
		$6 \Rightarrow A_5 \ltimes II_{23}$	not computed
		$2 \Rightarrow A_5 \ltimes (\times_{i=1}^5 A_8)$	not computed
		$\left(\begin{array}{c}4\Rightarrow A_{40}\end{array}\right)$	not computed
		$\left(3 \Rightarrow A_5 \ltimes H_{16} \right)$	not computed
49	70	$4 \Rightarrow A_5 \ltimes (\times_{i=1}^5 A_8)$	not computed
44	10	$0 \Rightarrow A_5 \ltimes (H_1 \ltimes (\times_{i=1}^6 A_7)))$	not computed
		$(3 \Rightarrow A_{42})$	not computed

Index	# classes	Induced action	Abelianization	
	0	$\int 7 \Rightarrow A_{22} \ltimes H_{14}$	not computed	
44	9	$\begin{array}{c} 2 \Rightarrow A_{44} \end{array}$	not computed	
45	1	$A_5 \ltimes H_{10}$	not computed	
46	18	$18 \Rightarrow \Lambda_{46}$	not computed	
		$\int 1 \Rightarrow 3840$	not computed	
		$18 \Rightarrow 122880$	not computed	
		$6 \Rightarrow 1399680$	not computed	
		$8 \Rightarrow 3932160$	not computed	
		$18 \Rightarrow 7864320$	not computed	
		$2 \Rightarrow 113374080$	not computed	
		$20 \Rightarrow 251658240$	not computed	
		$9 \Rightarrow 5733089280$	not computed	
		$12 \Rightarrow 11466178560$	not computed	
		$24 \Rightarrow 16106127360$	not computed	
48		$4 \Rightarrow 82649704320$	not computed	
	220	$4 \Rightarrow 5289581076480$	not computed	
	<i>397</i>	$2 \Rightarrow 23482733690880$	not computed	
		$4 \Rightarrow 46965467381760$	not computed	
		$3 \Rightarrow 1348984441405440$	not computed	
			$8 \Rightarrow 1502894956216320$	not computed
		$16 \Rightarrow 3005789912432640$	not computed	
		$1 \Rightarrow 17118912860651520$	not computed	
		$2 \Rightarrow 34237825721303040$	not computed	
		$25 \Rightarrow 43167502124974080$	not computed	
		$18 \Rightarrow 1095610423081697280$	not computed	
		$8 \Rightarrow A_5 \ltimes (\times_{i=1}^6 A_8)$	not computed	
		$118 \Rightarrow A_5 \ltimes (H_1 \ltimes (\times_{i=1}^6 A_8))$	not computed	
		$1 \Rightarrow A_{48}$	not computed	

Index	# classes	Induced action	A belianization
		$(1 \Rightarrow A_5 \ltimes H_{12})$	not computed
		$3 \Rightarrow A_5 \ltimes (\times_{i=1}^{10} A_5)$	not computed
50	27	$\begin{cases} 1 \Rightarrow A_5 \ltimes H_{14} \end{cases}$	not computed
		$8 \Rightarrow A_5 \ltimes H_{15}$	not computed
		$14 \Rightarrow A_{50}$	not computed
		$1 \Rightarrow L_2(25)$	\mathbb{Z}^{78}
52	24	$\begin{cases} 6 \Rightarrow L_2(25) \ltimes H_{11} \end{cases}$	not computed
		$17 \Rightarrow A_{52}$	not computed
53	1	A ₅₃	not computed
54	403	{	
		$4 \Rightarrow L_2(11)$	$\mathbb{Z}_2^{10}\oplus\mathbb{Z}_{11}^3$
	14	$4 \Rightarrow A_5 \ltimes L_2(11)$	not computed
55	14	$4 \Rightarrow L_2(11) \ltimes (\times_{i=1}^{11} A_5)$	not computed
		$\left(\begin{array}{c}2\Rightarrow A_{55}\end{array}\right)$	not computed
		$\int 1 \Rightarrow S_6(2)$	not computed
56	31	$\begin{cases} 14 \Rightarrow S_6(2) \ltimes H_{10} \end{cases}$	not computed
		$16 \Rightarrow A_{56}$	not computed
57	4	$4 \Rightarrow L_2(19)$	$\mathbb{Z}_2^{18}\oplus\mathbb{Z}_{19}^3$
58	78	$78 \Rightarrow A_{58}$	not computed
59	1	$1 \Rightarrow A_{59}$	not computed

There are also 2063 conjugacy classes of subgroups of index 60 in Γ^+ . No information on the induced actions was computed.

- 1) H_1 is elementary abelian of order 32.
- 2) H_2 is elementary abelian of order 16.
- 3) $H_3/[H_3, H_3]$ is elementary abelian of order 32, and $[H_3, H_3]$ is elementary abelian of order 729.
- 4) H_4 is elementary abelian of order 1024.
- 5) H_5 is elementary abelian of order 64.

- 6) $H_6/[H_6, H_6]$ is elementary abelian of order 64, and $[H_6, H_6]$ is elementary abelian of order 32.
- 7) H_7 is elementary abelian of order 2048.
- 8) $H_8/[H_8, H_8]$ is elementary abelian of order 32, and $[H_8, H_8]$ is elementary abelian of order 5⁶.
- 9) $H_9/[H_9, H_9]$ is elementary abelian of order 32, and $[H_9, H_9]$ is elementary abelian of order 3^{10} .
- 10) H_{10} is elementary abelian of order 2^{21} .
- 11) H_{11} is elementary abelian of order 2^{13} .
- 12) $H_{12}/[H_{12}, H_{12}]$ is elementary abelian of order 32, and $[H_{12}, H_{12}]$ is elementary abelian of order 5¹⁰.
- 13) $H_{13}/[H_{13}, H_{13}]$ is elementary abelian of order 16, and $[H_{13}, H_{13}]$ is elementary abelian of order 5¹⁵.
- 14) $H_{14}/[H_{14}, H_{14}]$ is elementary abelian of order 32, and $[H_{14}, H_{14}]$ is $\times_{i=1}^{10} A_5$.

15)
$$H_{15} = \times_{i=1}^{5} A_{10}$$
.

- 16) $H_{16}/[H_{16}, H_{16}]$ is elementary abelian of order 32, and $[H_{16}, H_{16}]$ is elementary abelian of order 7⁶.
- 17) $H_{17}/[H_{17}, H_{17}]$ is elementary abelian of order 2¹¹, and $[H_{17}, H_{17}]$ is elementary abelian of order 3¹².
- 18) $H_{18}/[H_{18}, H_{18}]$ is elementary abelian of order 2⁶, and $[H_{18}, H_{18}]$ is elementary abelian of order 3¹².
- 19) $H_{19}/[H_{19}, H_{19}]$ is elementary abelian of order 2⁵, and $[H_{19}, H_{19}]$ is elementary abelian of order 3¹².
- 20) $H_{20}/[H_{20}, H_{20}]$ is elementary abelian of order 2⁶, and $[H_{20}, H_{20}]$ is elementary abelian of order 3⁶.
- 21) $H_{21}/[H_{21}, H_{21}]$ is elementary abelian of order 2⁵, and $[H_{21}, H_{21}]$ is elementary abelian of order 3⁶.

- 22) $H_{22}/[H_{22}, H_{22}]$ is elementary abelian of order 2^{10} , and $[H_{22}, H_{22}]$ is elementary abelian of order 16.
- 23) $H_{23}/[H_{23}, H_{23}]$ is elementary abelian of order 2¹¹, and $[H_{23}, H_{23}]$ is elementary abelian of order 16.
- 24) $H_{24}/[H_{24}, H_{24}]$ is elementary abelian of order 2⁵. Write $H'_{24} = [H_{24}, H_{24}]$. Then $H'_{24}/[H'_{24}, H'_{24}]$ is elementary abelian of order 3⁶ and $[H_{24}, H_{24}]$ is elementary abelian of order 2¹².
- 25) $H_{25}/[H_{25}, H_{25}]$ is elementary abelian of order 2⁶. Write $H'_{25} = [H_{25}, H_{25}]$. Then $H'_{25}/[H'_{25}, H'_{25}]$ is elementary abelian of order 3⁶ and $[H_{25}, H_{25}]$ is elementary abelian of order 2¹².
- 26) $H_{26}/[H_{26}, H_{26}]$ is elementary abelian of order 2⁶. Write $H'_{26} = [H_{26}, H_{26}]$. Then $H'_{26}/[H'_{26}, H'_{26}]$ is elementary abelian of order 3⁶ and $[H_{26}, H_{26}]$ is elementary abelian of order 2¹².

A.2 $\Gamma = T_2[2, 2, 3; 2, 5, 3]$

 Γ^+ has 143 conjugacy classes of low index subgroups of index ≤ 60 . Their index, the induced permutation group and, where possible, the abelianisation of the kernel of the map are listed here.

Index	# classes	Induced action	Abelianization
6	4	A_6	\mathbb{Z}_3^6
10	2	A_6	\mathbb{Z}_3^6
11	2	$L_2(11)$	\mathbb{Z}^{10}
12	1	$L_2(11)$	\mathbb{Z}^{10}
15	4	A_6	\mathbb{Z}_3^6
17	1	$L_2(16)$	$\mathbb{Z}^{12}\oplus\mathbb{Z}_2^6$
20	4	$ \left\{\begin{array}{l} 2 \Rightarrow A_6 \\ 2 \Rightarrow L_2(19) \end{array}\right. $	$\mathbb{Z}_3^6 \ \mathbb{Z}_3^{19} \oplus \mathbb{Z}_{19}^3$
22	2	$L_2(11) \ltimes H_1$	not computed
26	1	$L_2(25)$	\mathbb{Z}^{51}
30	10	$\begin{cases} 4 \Rightarrow A_6 \\ 2 \Rightarrow L_2(29) \\ 4 \Rightarrow A_6 \ltimes H_2 \end{cases}$	$\mathbb{Z}_3^6 \ \mathbb{Z}_3^{28} \oplus \mathbb{Z}_5^{15} \oplus \mathbb{Z}_{29}^3 \ ext{not computed}$
33	2	$2 \Rightarrow L_2(11) \ltimes H_3$	not computed
36	10	$\begin{cases} 2 \Rightarrow A_6 \\ 4 \Rightarrow A_6 \times A_6 \\ 4 \Rightarrow A_6 \ltimes (\times_{i=1}^6 A_5) \end{cases}$	\mathbb{Z}_3^6 not computed not computed
40	6	$ \left\{\begin{array}{l} 2 \Rightarrow A_6\\ 4 \Rightarrow A_{40} \end{array}\right. $	not computed not computed
42	2	$2 \Rightarrow L_2(41)$	$\mathbb{Z}^{40}\oplus\mathbb{Z}^{18}_{41}$
45	2	$2 \Rightarrow A_6$	\mathbb{Z}_3^6
50	4	$\Rightarrow A_6 \rtimes H_4$	not computed

Index	# classes	Induced action	Abelianization
51	4	$\int 1 \Rightarrow L_2(16)$	$\mathbb{Z}^{12}\oplus\mathbb{Z}_2^6$
01		$\left(3 \Rightarrow L_2(16) \rtimes H_5 \right)$	not computed
52	3	$\begin{cases} 1 \Rightarrow L_2(25) \end{cases}$	\mathbb{Z}^{51}
01		$\left(2 \Rightarrow L_{(25)} \rtimes H_6 \right)$	not computed
53	2	$2 \Rightarrow A_{53}$	not computed
55	2	$2 \Rightarrow L_2(11)$	\mathbb{Z}^{10}
57	4	$4 \Rightarrow L_2(19)$	$\mathbb{Z}_3^{19}\oplus\mathbb{Z}_{19}^3$
		$4 \Rightarrow A_6$	\mathbb{Z}_3^6
	71	$1 \Rightarrow L_2(11)$	\mathbb{Z}^{10}
		$2 \Rightarrow L_2(19)$	$\mathbb{Z}_3^{19}\oplus\mathbb{Z}_{19}^3$
		$2 \Rightarrow L_2(29)$	$\mathbb{Z}_3^{28}\oplus\mathbb{Z}_5^{15}\oplus\mathbb{Z}_{29}^3$
		$2 \Rightarrow L_2(59)$	$\mathbb{Z}_3^{116}\oplus\mathbb{Z}_5^{58}\oplus$
			$\mathbb{Z}^{58}_{11}\oplus\mathbb{Z}^3_{59}$
60		$2 \Rightarrow L_2(59)$	$\mathbb{Z}_3^{58}\oplus\mathbb{Z}_5^{58}\oplus$
00			$\mathbb{Z}_7^{58}\oplus\mathbb{Z}_{19}^{58}\oplus\mathbb{Z}_{59}^3$
		$4 \Rightarrow A_6 \times A_6$	not computed
		$36 \Rightarrow A_6 \ltimes II_7$	not computed
		$6 \Rightarrow L_2(19) \ltimes H_8$	not computed
		$4 \Rightarrow A_6 \ltimes (\times_{i=1}^6 A_5)$	not computed
		$4 \Rightarrow A_6 \ltimes (\times_{i=1}^{10} A_5)$	not computed
		$4 \Rightarrow A_6 \ltimes H_9$	not computed

The groups H_i

- 1) H_1 is an elementary abelian group of order 1024.
- 2) H_2 is $\times_{i=1}^6 A_5$.
- 3) H_3 has commutator subgroup H' elementary abelian of order 3^{11} . The factor group H/H' is also elementary abelian and has order 2^{10} .
- 4) $H_4 = \times_{i=1}^{10} A_5.$
- 5) H_5 is elementary abelian of order 3^{16} .
- 6) H_6 is elementary abelian of order 2^{13} .

- 7) H_7 is elementary abelian of order 3^6 .
- 8) H_8 is elementary abelian of order 3¹9.
- 9) H_9 is a group of order 782757789696000000. Its structure is not known but is possibly the semidirect product of 6 copies of A_5 with an elementary abelian group of order $2^{24} = 16777216$.

A.3 $\Gamma = T_3[2, 2, 4; 2, 3, 5]$

There are 647 conjugacy classes of subgroups of index ≤ 60 in Γ^+ . Their index, the induced permutation group and, where possible, the abelianisation of the kernel of the map are listed here. Note that the subgroups arising from the representatives of subgroups of index ≤ 10 all contain the torsion representative γ^2 . The kernel of the map $\Gamma^+ \rightarrow PGL_2(11)$ is torsion free.

Index	$\# \ classes$	Induced action	Abelianization
2	1	\mathbb{Z}_2	0
5	1	A_5	$\mathbb{Z}_2^5 \oplus \mathbb{Z}_4$
6	1	A_5	$\mathbb{Z}_2^5\oplus\mathbb{Z}_4$
10	2	$\begin{cases} 1 \Rightarrow A_5 \\ 1 \Rightarrow A_5 \times C_2 \end{cases}$	$\mathbb{Z}_2^5 \oplus \mathbb{Z}_4$ \mathbb{Z}_2^{11}
12	9	$\begin{cases} 1 \Rightarrow A_5 \\ 2 \Rightarrow A_5 \times C_2 \\ 2 \Rightarrow PGL_2(11) \\ 4 \Rightarrow H_1 \times (A_5 \ltimes H_2) \end{cases}$	$egin{array}{c} \mathbb{Z}_2^5 \oplus \mathbb{Z}_4 \ \mathbb{Z}_2^{11} \ \mathbb{Z}_2^{10} \oplus \mathbb{Z}_{11}^3 \ \mathbb{Z}^{31} \end{array}$
15	1	A_5	$\mathbb{Z}_2^5\oplus\mathbb{Z}_4$
18	1	$(A_5 \times C_2) \ltimes H_3$	not computed
20	13	$\begin{cases} 1 \Rightarrow A_5 \\ 2 \Rightarrow A_5 \times C_2 \\ 4 \Rightarrow A_5 \ltimes H_2 \end{cases}$	$egin{array}{c} \mathbb{Z}_2^5 \oplus \mathbb{Z}_4 \ \mathbb{Z}_2^{11} \ \mathbb{Z}^{15} \oplus \mathbb{Z}_2^2 \end{array}$
20	10	$ \begin{array}{c} 4 \Rightarrow A_5 \times H_2 \\ 4 \Rightarrow A_5 \ltimes H_1 \\ 2 \Rightarrow PGL_2(19) \end{array} $	\mathbb{Z}^{31} $\mathbb{Z}^{18}_2 \oplus \mathbb{Z}^3_{19}$
22	3	$\begin{cases} 2 \Rightarrow PGL_2(11) \\ 1 \Rightarrow S_{22} \end{cases}$	$\mathbb{Z}_2^{10} \oplus \mathbb{Z}_{11}^3$ not computed

Index	# classes	Induced action	Abelianization
		$(1 \Rightarrow A_5 \times C_2)$	\mathbb{Z}_2^{11}
		$2 \Rightarrow PGL_2(11)$	$\mathbb{Z}_2^{10}\oplus\mathbb{Z}_{11}^3$
		$2 \Rightarrow A_5 \ltimes H_4$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^{14}$
24	22	$\left\{ \begin{array}{c} 2 \Rightarrow A_5 \ltimes H_4 \end{array} \right.$	\mathbb{Z}^{27}
		$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	\mathbb{Z}^{31}
		$8 \Rightarrow (A_5 \times C_2) \ltimes H_2$	not computed
		$1 \Rightarrow (A_5 \times C_2) \ltimes H_5$	not computed
26	1	$L_2(25)$	$\mathbb{Z}^{26}\oplus\mathbb{Z}_2^{27}$
28	1	$S_6(2)$	not computed
		$\int 1 \Rightarrow A_5$	$\mathbb{Z}_2^5\oplus\mathbb{Z}_4$
		$2 \Rightarrow A_5 \times C_2$	\mathbb{Z}_2^{11}
		$8 \Rightarrow A_5 \ltimes H_2$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^2$
00	10	$4 \Rightarrow PGL_2(29)$	not computed
30	19	$1 \Rightarrow (A_5 \times C_2) \ltimes H_6$	not computed
		$1 \Rightarrow (A_5 \times C_2) \ltimes H_7$	not computed
	1	$1 \Rightarrow (A_5 \times C_2) \ltimes H_8$	not computed
		$1 \Rightarrow (A_5 \times C_2) \ltimes H_9$	not computed
20	4	$\int 2 \Rightarrow (A_5 \times C_2) \ltimes H_2$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^2$
32	4	$\begin{cases} 2 \Rightarrow L_2(31) \end{cases}$	not computed
34	1	A ₃₄	not computed
		$(1 \Rightarrow S_6(2))$	not computed
		$7 \Rightarrow (A_5 \times C_2) \ltimes H_3$	not computed
		$2 \Rightarrow (A_5 \times C_2) \ltimes H_{10}$	not computed
36	22	$\left \begin{array}{c} 4 \end{array} \right 8 \Rightarrow (A_5 \times C_2) \ltimes H_{11}$	not computed
		$2 \Rightarrow (A_5 \times C_2) \ltimes H_{12}$	not computed
		$1 \Rightarrow (A_5 \times C_2) \ltimes H_8$	not computed
		$1 \Rightarrow (A_5 \times C_2) \ltimes H_9$	not computed

Index	# classes	Induced action	Abelianization
	<u> </u>	$ (2 \Rightarrow A_5 \times C_2) $	$\mathbb{Z}_2^{\overline{11}}$
		$7 \Rightarrow A_5 \ltimes H_2$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^2$
		$2 \Rightarrow A_5 \ltimes H_1$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^{14}$
		$2 \Rightarrow A_5 \ltimes H_1$	\mathbb{Z}^{27}
		$10 \Rightarrow A_5 \ltimes H_1$	\mathbb{Z}^{32}
40	48	$\begin{cases} 2 \Rightarrow PGL_2(19) \end{cases}$	$\mathbb{Z}_2^{18}\oplus\mathbb{Z}_{19}^3$
		$8 \Rightarrow A_5 \ltimes H_{13}$	not computed
		$8 \Rightarrow A_5 \ltimes H_{14}$	not computed
		$4 \Rightarrow A_5 \ltimes H_{15}$	not computed
		$2 \Rightarrow (A_5 \times C_2) \ltimes H_{16}$	not computed
		$2 \Rightarrow A_{40}$	not computed
	8	$(1 \Rightarrow A_5 \ltimes H_{17})$	not computed
49		$2 \Rightarrow A_5 \ltimes H_{18}$	not computed
42		$4 \Rightarrow (A_5 \times C_2) \ltimes H_{19}$	not computed
		$\left(1 \Rightarrow A_{42} \right)$	not computed
4.4	7	$\int 1 \Rightarrow S_{22}$	not computed
44	1	$\begin{cases} 6 \Rightarrow S_{22} \ltimes H_{20} \end{cases}$	not computed
45	1	$A_5 \ltimes H_{21}$	not computed
46	1	$\int 2 \Rightarrow A_{46}$	not computed
40	4	$2 \Rightarrow S_{46}$	not computed

Index	# classes	Induced action	A belianization
		$(3 \Rightarrow (A_5 \times C_2) \ltimes H_2)$	not computed
		$4 \Rightarrow (A_5 \times C_2) \ltimes H_1$	not computed
		$20 \Rightarrow (A_5 \times C_2) \ltimes H_{24}$	not computed
		$8 \Rightarrow (A_5 \times C_2) \ltimes H_{22}$	not computed
		$10 \Rightarrow (A_5 \times C_2) \ltimes H_{23}$	not computed
		$16 \Rightarrow (A_5 \times C_2) \ltimes H_{33}$	not computed
		$7 \Rightarrow (A_5 \times C_2) \ltimes H_{34}$	not computed
48	89	$\begin{array}{c} 2 \Rightarrow (A_5 \times C_2) \ltimes H_{37} \end{array}$	not computed
		$1 \Rightarrow \Lambda_5 \ltimes \left(\times_{i=1}^6 L_2(7) \right)$	not computed
		$2 \Rightarrow (A_5 \times C_2) \ltimes H_{36}$	not computed
		$3 \Rightarrow (A_5 \times C_2) \ltimes H_{35}$	not computed
		$2 \Rightarrow (A_5 \times C_2) \ltimes H_{31}$	not computed
		$2 \Rightarrow A_5 \ltimes (\times_{i=1}^6 A_8)$	not computed
		$8 \Rightarrow (A_5 \times C_2) \ltimes (H_2 \ltimes (\times_{i=1}^6 A_8))$	not computed
		$\left(1 \Rightarrow A_{48} \right)$	not computed
		$\int 1 \Rightarrow A_5 \ltimes H_{30}$	not computed
		$1 \Rightarrow A_5 \ltimes (\times_{i=1}^{10} A_5)$	not computed
50	5	$\left\{ 1 \Rightarrow A_5 \ltimes (H_2 \ltimes (\times_{i=1}^{10} A_5)) \right\}$	not computed
		$1 \Rightarrow A_{50}$	not computed
		$\left(1 \Rightarrow S_{50} \right)$	not computed
		$\int 1 \Rightarrow L_2(25)$	not computed
		$2 \Rightarrow L_2(25) \ltimes C_2$	not computed
52	10	$\left \begin{array}{c} 4 \Rightarrow L_2(25) \ltimes H_{25} \end{array} \right $	not computed
		$2 \Rightarrow A_{52}$	not computed
		$\left(1 \Rightarrow S_{52} \right)$	not computed
53	1	$1 \Rightarrow A_{53}$	not computed

,

Index	# classes	Induced action	Abelianization
		$(1 \Rightarrow (A_5 \times C_2) \ltimes H_{27})$	not computed
		$1 \Rightarrow (A_5 \times C_2) \ltimes H_{28}$	not computed
		$1 \Rightarrow (A_5 \times C_2) \ltimes H_{29}$	not computed
		$4 \Rightarrow (A_5 \times C_2) \ltimes H_{32}$	not computed
54	27	$\left\{ 7 \Rightarrow A_5 \ltimes \left(\times_{i=1}^6 A_9 \right) \right\}$	not computed
		$4 \Rightarrow (A_5 \times C_2) \ltimes (\times_{i=1}^6 A_9)$	not computed
		$7 \Rightarrow (A_5 \times C_2) \ltimes (H_2 \ltimes (\times_{i=1}^6 A_9))$	not computed
		$1 \Rightarrow A_{54}$	not computed
		$\left(1 \Rightarrow S_{54} \right)$	not computed
55	4	$4 \Rightarrow PGL_2(11)$	$\mathbb{Z}_2^{10}\oplus\mathbb{Z}_{11}^3$
		$\int 1 \Rightarrow S_6(2)$	not computed
56	0	$2 \Rightarrow S_6(2) \ltimes C_2$	not computed
50	9	$4 \Rightarrow S_6(2) \ltimes H_{26}$	not computed
		$\left(2 \Rightarrow S_{56} \right)$	not computed
58	8	$\int 6 \Rightarrow A_{58}$	not computed
58	0	$\begin{array}{c} 2 \Rightarrow S_{58} \end{array}$	not computed
59	1	$1 \Rightarrow S_{59}$	not computed

There are also 323 conjugacy classes of subgroups of index 60 in Γ^+ . No information on the induced actions was computed.

- 1) H_1 is elementary abelian of order 64.
- 2) H_2 is elementary abelian of order 32.
- 3) H'_3 is elementary abelian of order 729, and H_3/H'_3 is elementary abelian of order 32.
- 4) $H_4 = \mathbb{Z}_2 \oplus \mathbb{Z}_2 \oplus \mathbb{Z}_2 \oplus \mathbb{Z}_2 \oplus \mathbb{Z}_4.$
- 5) H_5/H'_5 is elementary abelian of order 32. Now $K = H'_5$ has an elementary derived subgroup of order 2^{12} and the factor group K/K' is elementary abelian of order 3^5 .

- 6) H'_6 is elementary abelian of order 5⁶ and H_6/H'_6 is elementary abelian of order 32.
- 7) H'_7 is elementary abelian of order 3^{10} and H_7/H'_7 is elementary abelian of order 32.
- 8) H_8 is the direct product of six copies of A_5 .
- 9) H'_9 is the direct product of six copies of A_5 and H_9/H'_9 is elementary abelian of order 32.
- 10) H'_{10} is elementary abelian of order 3^{12} and H_{10}/H'_{10} is elementary abelian of order 32.
- 11) $H_{11} = H_2 \ltimes (H_2 \ltimes (H/H' \ltimes H'))$ where H' is elementary abelian of order 2^{12} and H/H' is elementary abelian of order 729.
- 12) H'_{12} is elementary abelian of order 3^{12} and H_{12}/H'_{12} is elementary abelian of order 2^{11} .
- 13) H'_{13} is elementary abelian of order 16 and $H_{13}/H_{13}' = \mathbb{Z}_2^5 \oplus \mathbb{Z}_4$
- 14) H'_{14} is elementary abelian of order 32 and $H_{14}/H_{14}' = \mathbb{Z}_2^5 \oplus \mathbb{Z}_4$
- 15) H'_{15} is elementary abelian of order 512 and $H_{14}/H_{14}' = \mathbb{Z}_2^5 \oplus \mathbb{Z}_4$
- 16) $H_{16} = A_8 \times A_8 \times A_8 \times A_8 \times A_8$
- 17) H'_{17} is elementary abelian of order 7⁶ and $H_{17}/H17'$ is elementary abelian of order 64.
- 18) $H_{18} = A_7 \times A_7 \times A_7 \times A_7 \times A_7 \times A_7$
- 19) H_{19}
- 20) H_{20} is elementary abelian of order $2^{2}1$
- 21) H_{21} has an elementary abelian derived subgroup of order 3^{15} . The factor group H_{21}/H'_{21} is elementary abelian of order 2^5 .
- 22) $H_{22}/[H_{22}, H_{22}]$ is elementary abelian of order 2⁵. $[H_{22}, H_{22}]$ is elementary abelian of order 2¹¹.

- 23) $H_{23}/[H_{23}, H_{23}]$ is elementary abelian of order 2⁶. $[H_{23}, H_{23}]$ is elementary abelian of order 2¹¹.
- 24) H_{24} is elementary abelian of order 2^{11} .
- 25) H_{25} is elementary abelian of order 2^{13} .
- 26) H_{26} is elementary abelian of order 2^{21} .
- 27) $H_{27}/[H_{27}, H_{27}]$ is elementary abelian of order 2⁵. $[H_{27}, H_{27}] \cong \times_{i=1}^{6} C_{9}$.
- 28) $H_{28}/[H_{28}, H_{28}]$ is elementary abelian of order 2⁵. $[H_{28}, H_{28}]$ is elementary abelian of order 3¹².
- 29) $H_{29}/[H_{29}, H_{29}]$ is elementary abelian of order 2^{11} . $[H_{28}, H_{28}]$ is elementary abelian of order 3^{12} .
- 29) $H_{29}/[H_{29}, H_{29}]$ is elementary abelian of order 2^{11} . $[H_{28}, H_{28}]$ is elementary abelian of order 3^{12} .
- 30) $H_{30}/[H_{30}, H_{30}]$ is elementary abelian of order 2⁵. $[H_{30}, H_{30}]$ is elementary abelian of order 5¹⁰.
- 31) $H_{31}/[H_{31}, H_{31}]$ is elementary abelian of order 2¹⁰. Write $H'_{31} = [H_{31}, H_{31}]$. Then $H'_{31}/[H'_{31}, H'_{31}]$ is elementary abelian of order 3¹² and $[H_{31}, H_{31}]$ is elementary abelian of order 2²⁴
- 32) Let $H_{32}^{(i)} = [H_{32}^{(i-1)}, H_{32}^{(i-1)}]$ be the *i*th derived subgroup of H_{32} , and $K^{(i)} = H_{32}^{(i-1)}/H_{32}^{(i)}$. Then the $K^{(i)}$ are elementary abelian, $|H^{(1)}| = 2^5$, $|H^{(2)}| = 3^6$, $|H^{(3)}| = 2^{12}$ and $H_{32}^{(3)}$ is also elementary abelian of order 3^{18} .
- 33) $H_{33}/[H_{33}, H_{33}]$ is elementary abelian of order 2¹¹. $[H_{33}, H_{33}]$ is elementary abelian of order 2¹¹.
- 34) $H_{34}/[H_{34}, H_{34}]$ is elementary abelian of order 2⁵. Write $H'_{34} = [H_{34}, H_{34}]$. Then $H'_{34}/[H'_{34}, H'_{34}]$ is elementary abelian of order 3⁶ and $[H_{34}, H_{34}]$ is elementary abelian of order 2¹².
- 35) $H_{35}/[H_{35}, H_{35}]$ is elementary abelian of order 2⁵ and $[H_{35}, H_{35}] \cong \times_{i=1}^{6} L_2(7)$.

- 36) $H_{36}/[H_{36}, H_{36}]$ is elementary abelian of order 2⁵. Write $H'_{36} = [H_{36}, H_{36}]$. Then $H'_{36}/[H'_{36}, H'_{36}]$ is elementary abelian of order 3¹² and $[H_{36}, H_{36}]$ is elementary abelian of order 2²⁴.
- 37) $H_{37}/[H_{37}, H_{37}]$ is elementary abelian of order 2⁵. Write $H'_{37} = [H_{37}, H_{37}]$. Then $H'_{37}/[H'_{37}, H'_{37}]$ is elementary abelian of order 3⁶ and $[H_{37}, H_{37}]$ is elementary abelian of order 2²⁴.

A.4 $\Gamma = T_4[2, 2, 5; 2, 3, 5]$

There are 147 conjugacy classes of subgroups of index ≤ 60 in Γ^+ . Their index, the induced permutation group and, where possible, the abelianisation of the kernel of the map are listed here.

Index	# classes	Induced action	Abelianization
5	2	$2 \Rightarrow A_5$	\mathbb{Z}_5^3
6	2	$2 \Rightarrow A_5$	\mathbb{Z}_5^3
10	2	$2 \Rightarrow A_5$	\mathbb{Z}_5^3
12	2	$2 \Rightarrow A_5$	\mathbb{Z}_5^3
15		$2 \Rightarrow A_5$	\mathbb{Z}_5^3
17	1	$L_2(16)$	$\mathbb{Z}^{33}\oplus\mathbb{Z}_2^{12}$
		$(2 \Rightarrow A_5)$	\mathbb{Z}_5^3
20	5	$\begin{cases} 1 \Rightarrow L_2(19) \end{cases}$	\mathbb{Z}^{56}
		$2 \Rightarrow L_2(19)$	$\mathbb{Z}_5^{18}\oplus\mathbb{Z}_{19}^3$
~-	-	$\int 1 \Rightarrow A_5 \times A_5$	$\mathbb{Z}^{41}\oplus\mathbb{Z}_2^{12}$
25	5	$\begin{cases} 4 \Rightarrow A_{25} \end{cases}$	not computed
		i $2 \Rightarrow A_5$	\mathbb{Z}_5^3
		$2 \Rightarrow A_5 \times A_5$	$\mathbb{Z}^{41}\oplus\mathbb{Z}_2^{12}$
	~~	$10 \Rightarrow A_5 \ltimes H_1$	$\mathbb{Z}_{5}^{15} \oplus \mathbb{Z}_{25}^{16} \oplus \mathbb{Z}_{125}^{11}$
30	25	$2 \Rightarrow L_2(29)$	not computed
		$2 \Rightarrow A_5 \ltimes H_2$	not computed
		$7 \Rightarrow A_{30}$	not computed
32	2	$2 \Rightarrow L_2(31)$	not computed
34	1	$L_2(16) \ltimes H_3$	not computed
2.0		$\int 1 \Rightarrow A_5 \times A_5$	$\mathbb{Z}^{41}\oplus\mathbb{Z}^{12}_2$
36	3	$\begin{cases} 2 \Rightarrow A_5 \ltimes H_2 \end{cases}$	not computed
42	2	$2 \Rightarrow L_2(41)$	not computed
46	2	$2 \Rightarrow A_{46}$	not computed

Index	# classes	Induced action	Abelianization	
-0		$(2 \Rightarrow A_5 \times A_5)$	$\mathbb{Z}^{41}\oplus\mathbb{Z}_2^{12}$	
		$2 \Rightarrow A_5 \ltimes H_1$	$\mathbb{Z}_{5}^{15} \oplus \mathbb{Z}_{25}^{16} \oplus \mathbb{Z}_{125}^{11}$	
		$2 \Rightarrow L_2(49)$	not computed	
90		$1 \Rightarrow (A_5 \times A_5) \ltimes H_3$	not computed	
		$2 \Rightarrow A_{25} \ltimes H_4$	not computed	
		$2 \Rightarrow A_5 \ltimes H_5$	not computed	
		$1 \Rightarrow L_2(16)$	$\mathbb{Z}^{33}\oplus\mathbb{Z}_2^{12}$	
51	5	$\begin{cases} 3 \Rightarrow L_2(16) \ltimes H_6 \end{cases}$	not computed	
		$1 \Rightarrow L_2(16) \ltimes H_7$	not computed	
r 7	4	$\int 2 \Rightarrow A_5 \ltimes H_8$	not computed	
99		$\left\{\begin{array}{c}2\Rightarrow A_{55}\end{array}\right.$	not computed	
57	6	$\int 2 \Rightarrow L_2(19)$	\mathbb{Z}^{56}	
97	0	$\left(\begin{array}{c} 4 \Rightarrow L_2(19) \end{array}\right)$	$\mathbb{Z}_5^{18}\oplus\mathbb{Z}_{19}^3$	
59	1	A_{59}	not computed	
		$(2 \Rightarrow A_5)$	\mathbb{Z}_5^3	
	64	$2 \Rightarrow L_2(19)$	$\mathbb{Z}_5^{18}\oplus\mathbb{Z}_{19}^3$	
		$1 \Rightarrow L_2(19)$	\mathbb{Z}^{56}	
		$6 \Rightarrow A_5 \times A_5$	$\mathbb{Z}^{41}\oplus\mathbb{Z}_2^{12}$	
		$10 \Rightarrow A_5 \ltimes H_1$	$\mathbb{Z}_{5}^{15} \oplus \mathbb{Z}_{25}^{16} \oplus \mathbb{Z}_{125}^{11}$	
			$2 \Rightarrow L_2(29)$	not computed
60		$2 \Rightarrow L_2(59)$	not computed	
60		$4 \Rightarrow A_5 \ltimes (\times_{i=1}^6 A_5)$	not computed	
		$3 \Rightarrow L_2(19) \ltimes (\times_{i=1}^{19} C_3)$	not computed	
		$4 \Rightarrow A_5 \ltimes (\times_{i=1}^6 A_6)$	not computed	
		$2 \Rightarrow A_5 \ltimes (\times_{i=1}^5 M_{11})$	not computed	
		$8 \Rightarrow A_5 \ltimes \left(\times_{i=1}^5 M_{12} \right)$	not computed	
		· •	I	
		$6 \Rightarrow A_{30} \ltimes H_9$	not computed	

- 1) H_1 is elementary abelian of order 125.
- 2) H_2 is the direct product of six copies of A_5 .
- 3) H_3 is elementary abelian of size 2^{16} .

- 4) H_4 is elementary abelian of size 2^{24} .
- 5) H_5 is the direct product of six copies of A_{10} .
- 6) H_6 is elementary abelian of size 3^{16} .
- 7) H'_7 is elementary abelian of size 3^{17} and H/H' is elementary abelian of size 2^{16} .
- 8) H_8 is the direct product of six copies of M_11 .
- 9) H_9 is elementary abelian of order 2^{29} .

A.5 $\Gamma = T_5[2, 3, 3; 2, 4, 3]$

There are 62 conjugacy classes of subgroups of index ≤ 20 in Γ^+ . Their index, the induced permutation group and, where possible, the abelianisation of the kernel of the map are listed here.

Index	# classes	Induced action	Abelianization
5	1	A_5	\mathbb{Z}_2^{11}
6	1	A_5	\mathbb{Z}_2^{11}
7	2	$L_2(7)$	\mathbb{Z}^6
8	5	$\begin{cases} 1 \Rightarrow L_2(7) \\ 4 \Rightarrow A_5 \ltimes H_1 \end{cases}$	\mathbb{Z}^6 \mathbb{Z}^{27}
10	1	A_5	\mathbb{Z}_2^{11}
11	1	A_{11}	not computed
		$\left(1 \Rightarrow A_5 \right)$	\mathbb{Z}_2^{11}
12	7	$\begin{cases} 4 \Rightarrow A_5 \ltimes H_2 \end{cases}$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^{10}\oplus\mathbb{Z}_4^3$
		$2 \Rightarrow A_5 \ltimes H_2$	$\mathbb{Z}^{25}\oplus\mathbb{Z}_2^8$
1.4	C	$\int 2 \Rightarrow L_2(7)$	\mathbb{Z}^{6}
14	0	$\begin{cases} 4 \Rightarrow L_2(7) \ltimes H_3 \end{cases}$	\mathbb{Z}^{27}
15	1	A_5	\mathbb{Z}_2^{11}
1.0	C	$\int 4 \Rightarrow A_5 \ltimes H_1$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^2$
10	0	$\begin{array}{c} 2 \Rightarrow L_2(7) \ltimes H_4 \end{array}$	\mathbb{Z}^{193}
18	2	$L_2(17)$	$\mathbb{Z}_2^2\oplus\mathbb{Z}_4^{16}\oplus\mathbb{Z}_{17}^3$
19	2	A_{19}	not computed
		$\int 1 \Rightarrow A_5$	$\oplus_{i=1}^{11}\mathbb{Z}_2$
		$4 \Rightarrow A_5 \ltimes H_1$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^2$
20	27	$\left \begin{array}{c} 4 \Rightarrow A_5 \ltimes H_2 \end{array} \right $	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^{10}\oplus\mathbb{Z}_4^3$
		$2 \Rightarrow A_5 \ltimes H_2$	$\mathbb{Z}^{25}\oplus\mathbb{Z}_2^8$
		$16 \Rightarrow A_5 \ltimes H_5$	$\mathbb{Z}^{325}\oplus\mathbb{Z}_2^{50}\oplus\mathbb{Z}_4^{24}\oplus\mathbb{Z}_8^{52}$

- 1) H_1 is elementary abelian of order 16.
- 2) H_2 is elementary abelian of order 32.
- 3) H_3 is elementary abelian of order 8.
- 4) H_4 where |H/H'| = 64 is elementary abelian and H' is cyclic of order 2.
- 5) H_3 is elementary abelian of order 1024.

A.6 $\Gamma = T_6[2, 3, 4; 2, 3, 4]$

There are 388 conjugacy classes of subgroups of index ≤ 20 in Γ^+ . Their index, the induced permutation group and, where possible, the abelianisation of the kernel of the map are listed here.

Index	# classes	Induced action	Abelianization
2	1	C_2	$\mathbb{Z}_3\oplus\mathbb{Z}_3$
3	4	$ \left\{\begin{array}{l} 2 \Rightarrow D_6 \\ 2 \Rightarrow D_6 \end{array}\right. $	$\mathbb{Z}_3 \ \mathbb{Z}_2^4 \oplus \mathbb{Z}_3$
4	6	$\begin{cases} 4 \Rightarrow S_4 \\ 2 \Rightarrow S_4 \end{cases}$	$\mathbb{Z}_2^5 \oplus \mathbb{Z}_3 \ \mathbb{Z}_2 \oplus \mathbb{Z}_3 \oplus \mathbb{Z}_8^2$
0	10	$ \begin{pmatrix} 2 \Rightarrow D_6 \\ 2 \Rightarrow D_6 \\ 4 \Rightarrow C \end{cases} $	\mathbb{Z}_3 $\mathbb{Z}_2^4\oplus\mathbb{Z}_3$ $\mathbb{Z}_5^5\oplus\mathbb{Z}_3$
6	18	$\begin{cases} 4 \Rightarrow S_4 \\ 2 \Rightarrow S_4 \\ 2 \Rightarrow A_6 \end{cases}$	$\mathbb{Z}_2^\circ \oplus \mathbb{Z}_3 \ \mathbb{Z}_2 \oplus \mathbb{Z}_3 \oplus \mathbb{Z}_8^2 \ \mathbb{Z}^9 \oplus \mathbb{Z}_2^2 \oplus \mathbb{Z}_3^6$
7	2	$2 \Rightarrow L_2(7)$	\mathbb{Z}^{13}
8	25	$ \left\{\begin{array}{l} 4 \Rightarrow S_4 \\ 2 \Rightarrow S_4 \\ 2 \Rightarrow C_2 \ltimes (H_1) \\ 1 \Rightarrow L_2(7) \\ 12 \Rightarrow C_2 \ltimes H_2 \\ 4 \Rightarrow L_2(7) \ltimes H_3 \end{array}\right. $	$\mathbb{Z}_2^5 \oplus \mathbb{Z}_3$ $\mathbb{Z}_2 \oplus \mathbb{Z}_3 \oplus \mathbb{Z}_8^2$ $\mathbb{Z}_3 \oplus \mathbb{Z}_8^3$ \mathbb{Z}^{13} $\mathbb{Z}_3 \oplus \mathbb{Z}_4 \oplus \mathbb{Z}_{16}^2$ \mathbb{Z}^{55}
9	1	\hat{D}_{18}	\mathbb{Z}_2^8
10	1	A_6	$\mathbb{Z}^9 \oplus \mathbb{Z}_2^2 \oplus \mathbb{Z}_3^6$
11	4	S ₁₁	not computed

Indon	H alacsas	Induced action	Abelianization	
Thuex	# clusses		$\mathbb{Z}^{11} \oplus \mathbb{Z}^2$	
			$12 \Rightarrow S_4$ $12 \Rightarrow C_2 \ltimes (H_4)$	$\mathbb{Z}_2 \oplus \mathbb{Z}_4$ $\mathbb{Z}_2^5 \oplus \mathbb{Z}_8^2$
		$6 \Rightarrow C_2 \ltimes H_4$	$\mathbb{Z}_3 \oplus \mathbb{Z}_8^3$	
		$16 \Rightarrow C_2 \ltimes H_1$	$\mathbb{Z}_2^{48}\oplus\mathbb{Z}_4$	
		$4 \Rightarrow C_2 \ltimes H_5$	$\mathbb{Z}_2^{11} \oplus \mathbb{Z}_4^2$	
10	08	$12 \Rightarrow C_2 \ltimes H_5$	$\mathbb{Z}^9\oplus\mathbb{Z}_2^6$	
12	90	$2 \Rightarrow C_2 \ltimes H_5$	$\mathbb{Z}^9 \oplus \mathbb{Z}_2^8$	
		$2 \Rightarrow C_2 \times A_6$	$\mathbb{Z}^{19}\oplus\mathbb{Z}_3^6$	
		$24 \Rightarrow C_2 \times H_6$	$\mathbb{Z}_3^{25} \oplus \mathbb{Z}_{16}^3$	
		$2 \Rightarrow PGL_2(11)$	$\mathbb{Z}^{11}\oplus\mathbb{Z}^{11}_3\oplus\mathbb{Z}^{10}_4\oplus\mathbb{Z}_5\oplus\mathbb{Z}^3_{11}$	
		$4 \Rightarrow A_6 \ltimes H_7$	$\mathbb{Z}^{273}\oplus\mathbb{Z}_2^{32}\oplus\mathbb{Z}_3^6$	
		$\begin{array}{c} 2 \Rightarrow M_{12} \end{array}$	not computed	
13	2	$2 \Rightarrow S_{13}$	not computed	
		$(2 \Rightarrow L_2(7))$	\mathbb{Z}^{13}	
	34 8	$2 \Rightarrow C_2 \times L_2(7)$	$\mathbb{Z}^{13}\oplus\mathbb{Z}_3^2$	
			$4 \Rightarrow L_2(7) \ltimes H_3$	not computed
14		$\begin{cases} 4 \Rightarrow C_2 \times (L_2(7) \ltimes H_3) \end{cases}$	$\mathbb{Z}^{55}\oplus\mathbb{Z}_2^{28}\oplus\mathbb{Z}_3^2$	
		$8 \Rightarrow L_2(7) \ltimes H_8$	$\mathbb{Z}^{251}\oplus\mathbb{Z}_2^8$	
		$8 \Rightarrow C_2 \times (L_2(7) \ltimes H_8)$	not computed	
		$\left \begin{array}{c} 4 \Rightarrow A_{14} \end{array} \right $	not computed	
15		$\left(\begin{array}{c}2\Rightarrow A_6\end{array}\right)$	$\mathbb{Z}^9\oplus\mathbb{Z}_2^2\oplus\mathbb{Z}_3^6$	
		$\left \begin{array}{c} 2 \Rightarrow C_2 \ltimes H_9 \end{array} \right $	not computed	
		$4 \Rightarrow C_2 \ltimes H_{10}$	not computed	

Index	# classes	Induced action	Abelianization
		$(2 \Rightarrow C_2 \ltimes H_1)$	as above
		$6 \Rightarrow C_2 \ltimes (C_3 \ltimes H_{12})$	$\mathbb{Z}_3\oplus\mathbb{Z}_4\oplus\mathbb{Z}_{16}^2$
		$2 \Rightarrow C_2 \ltimes H_5$	$\mathbb{Z}_2^{48}\oplus\mathbb{Z}_4$
		$7 \Rightarrow C_2 \ltimes H_5$	$\mathbb{Z}^9 \oplus \mathbb{Z}_2^8$
10	46	$1 \Rightarrow L_2(7) \times C_2$	$\mathbb{Z}^{13}\oplus\mathbb{Z}_3^2$
16		$\begin{cases} 8 \Rightarrow C_2 \ltimes (C_3 \ltimes H_{13}) \end{cases}$	$\mathbb{Z}_3^{13}\oplus\mathbb{Z}_8^2\oplus\mathbb{Z}_{32}$
		$8 \Rightarrow C_2 \ltimes (C_3 \ltimes (C_3 \ltimes H_{13}))$	$\mathbb{Z}^{33}\oplus\mathbb{Z}_2^6$
		$4 \Rightarrow C_2 \ltimes (L_2(7) \ltimes H_3)$	$\mathbb{Z}^{55}\oplus\mathbb{Z}_2^{28}\oplus\mathbb{Z}_3^2$
		$4 \Rightarrow A_6 \ltimes H_{11}$	$\mathbb{Z}^{129}\oplus\mathbb{Z}_2^{32}\oplus\mathbb{Z}_3^6$
		$4 \Rightarrow L_2(7) \ltimes (C_2 \ltimes H_8)$	not computed
17	4	$4 \Rightarrow A_{17}$	not computed
10	0.9	$\int 28 \Rightarrow A_{18}$	not computed
18	93	$\begin{cases} 12 \Rightarrow S_{18} \end{cases}$	not computed
19	6	$6 \Rightarrow A_{19}$	not computed
		$(1 \Rightarrow A_6)$	$\mathbb{Z}^9\oplus\mathbb{Z}_2^2\oplus\mathbb{Z}_3^6$
20	35	$2 \Rightarrow A_6 \times C_2$	$\mathbb{Z}^{19}\oplus\mathbb{Z}_3^6$
		$8 \Rightarrow A_6 \ltimes H_7$	not computed
		$\begin{cases} 4 \Rightarrow A_6 \ltimes H_{14} \end{cases}$	not computed
		$4 \Rightarrow S_4 \wr A_5$	not computed
		$4 \Rightarrow H_{15} \wr A_5$	not computed
		$12 \Rightarrow S_{20}$	not computed

- 1) H'_1 is elementary abelian of order 16 and $H/H' = C_3$ is cyclic.
- 2) $H_2 = C_3 \ltimes (C2 \times C2 \times C4 \times C4).$
- 3) H_3 is elementary abelian of order 8.
- 4) H'_4 is elementary abelian of order 4 and H/H' is elementary abelian of order 9.
- 5) H'_5 is elementary abelian of order 16 and H/H' is elementary abelian of order 9.
- 6) $H_6 = C_2 \ltimes (C_3 \ltimes H)$, where H' is elementary abelian of order 8 and H/H' is elementary abelian of order 16.
- 7) H_7 is elementary abelian of order 32.
- 8) II_8 is elementary abelian of order 64.
- 9) $H_9 = C_3 \ltimes (A_5 \ltimes (A_5 \times A_5)).$
- 10) $H_{10} = C_3 \ltimes (C_2 \ltimes (C_2 \ltimes (A_5 \ltimes (A_5 \times A_5)))).$
- 11) H_{11} is elementary abelian of order 16.
- 12) $H'_{12} = C_2$ and H_{12}/H'_{12} is elementary abelian of order 16.
- 13) H'_{13} is elementary abelian of order 4 and H_{13}/H'_{13} is elementary abelian of order 16.
- 14) H_{14} is elementary abelian of order 512.
- 15) $H_{15} = S_4 \ltimes H_3$.

A.7 $\Gamma = T_7[2, 3, 3; 2, 5, 3]$

Index	# classes	Induced action	A belianization
5	2	$2 \Rightarrow A_5$	$\mathbb{Z}_2\oplus\mathbb{Z}_4^4$
6	2	$2 \Rightarrow A_5$	$\mathbb{Z}_2 \oplus \mathbb{Z}_4^4$
7	1	A_7	\mathbb{Z}^{69}
10	4	$\begin{cases} 2 \Rightarrow A_5 \\ 2 \Rightarrow A_{10} \end{cases}$	$\mathbb{Z}_2 \oplus \mathbb{Z}_4^4$ not computed
11	4	$4 \Rightarrow L_2(11)$	$\mathbb{Z}_2\oplus\mathbb{Z}_8^{10}\oplus\mathbb{Z}_{11}^3$
12	8	$\begin{cases} 2 \Rightarrow A_5 \\ 2 \Rightarrow L_2(11) \\ 4 \Rightarrow A_5 \ltimes H_1 \end{cases}$	$\mathbb{Z}_2 \oplus \mathbb{Z}_4^4$ $\mathbb{Z}_2 \oplus \mathbb{Z}_8^{10} \oplus \mathbb{Z}_{11}^3$ $\mathbb{Z}^{15} \oplus \mathbb{Z}_2^8 \oplus \mathbb{Z}_4 \oplus \mathbb{Z}_8^{10} \oplus \mathbb{Z}_{32}$
14	3	$3 \Rightarrow A_7 \ltimes H_2$	not computed
15	4	$\left \begin{cases} 2 \Rightarrow A_5 \\ 2 \Rightarrow A_7 \end{cases} \right $	$\mathbb{Z}_2\oplus\mathbb{Z}_4^4 \ \mathbb{Z}^{69}$
16	8	$8 \Rightarrow A_5 \ltimes H_3$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2\oplus\mathbb{Z}_4^{10}\oplus\mathbb{Z}_{16}$
20	20	$\begin{cases} 2 \Rightarrow A_5 \\ 8 \Rightarrow A_5 \ltimes H_3 \\ 4 \Rightarrow A_5 \ltimes H_1 \\ 6 \Rightarrow A_{10} \ltimes H_4 \end{cases}$	$\mathbb{Z}_{2} \oplus \mathbb{Z}_{4}^{4}$ $\mathbb{Z}^{15} \oplus \mathbb{Z}_{2} \oplus \mathbb{Z}_{4}^{10} \oplus \mathbb{Z}_{16}$ $\mathbb{Z}^{15} \oplus \mathbb{Z}_{2}^{8} \oplus \mathbb{Z}_{4} \oplus \mathbb{Z}_{8}^{10} \oplus \mathbb{Z}_{32}$ not computed

- 1) H_1 is elementary abelian of order 32.
- 2) H_2 is elementary abelian of order 64.
- 3) H_3 is elementary abelian of order 16.
- 3) H_3 is elementary abelian of order 512.

A.8 $\Gamma = T_8[2, 4, 3; 2, 5, 3]$

There are 103 conjugacy classes of subgroups of index ≤ 20 in Γ^+ . Their index, the induced permutation group and, where possible, the abelianisation of the kernel of the map are listed here.

Index	# classes	Induced action	A belianization	
G	6	$\int 4 \Rightarrow A_6$	$\mathbb{Z}_2^{61} \oplus \mathbb{Z}_3^6$	
0	0	$\begin{array}{c} 2 \Rightarrow A_6 \end{array}$	$\mathbb{Z}^{20}\oplus\mathbb{Z}_3^2$	
10	7	$\int 3 \Rightarrow A_6$	as above	
10	1	$\left\{ 4 \Rightarrow \Lambda_{10} \right\}$	not computed	
11	2	$L_2(11)$	$\mathbb{Z}_{10}\oplus\mathbb{Z}_2^{101}$	
		$\int 1 \Rightarrow L_2(11)$	as above	
10	11	$2 \Rightarrow M_{12}$	not computed	
12		$6 \Rightarrow A_6 \ltimes H_1$	not computed	
		$\left(2 \Rightarrow A_{12} \right)$	not computed	
15	19	$\int 6 \Rightarrow A_6$	as above	
15	12	$\begin{cases} 6 \Rightarrow A_{12} \end{cases}$	not computed	
		$4 \Rightarrow A_6 \ltimes H_2$	$\mathbb{Z}^{215}\oplus\mathbb{Z}^3_2$	
16	17	$) 8 \Rightarrow A_6 \ltimes H_2$	$\mathbb{Z}_2^{42} \oplus \mathbb{Z}_3^{21} \oplus \mathbb{Z}_8^{15}$	
10	11		$\oplus \mathbb{Z}^8_{16} \oplus \mathbb{Z}^{52}_{32}$	
		$\left(5 \Rightarrow A_{16} \right)$	not computed	
17	10	$\int 1 \Rightarrow L_2(16)$	$\mathbb{Z}^{17}\oplus\mathbb{Z}_2^{652}\oplus\mathbb{Z}_4^{12}$	
11	10	$\begin{array}{c} 9 \Rightarrow A_{17} \end{array}$	not computed	
	e	$\left(2 \Rightarrow A_6 \ltimes C_3 \right)$	\mathbb{Z}^{62}	
18	8	$\left\{ 2 \Rightarrow A_6 \ltimes H_3 \right.$	not computed	
		$4 \Rightarrow A_{18}$	not computed	
Index	# classes	Induced action		Abelianization
-------	-----------	----------------	------------------------------------	--
			$1 \Rightarrow A_6$	$\mathbb{Z}^{20}\oplus\mathbb{Z}_3^2$
			$2 \Rightarrow A_6$	$\mathbb{Z}_2^{61}\oplus\mathbb{Z}_3^6$
			$2 \Rightarrow L_2(19)$	$\mathbb{Z}_2^{571}\oplus\mathbb{Z}_3^{19}\oplus\mathbb{Z}_{19}^3$
20	29	<	$12 \Rightarrow A_6 \ltimes H_1$	not computed
			$6 \Rightarrow A_6 \ltimes H_4$	not computed
			$4 \Rightarrow A_{10} \ltimes H_4$	not computed
			$2 \Rightarrow A_{20}$	not computed

- 1) H_1 is elementary abelian of order 32.
- 2) H_2 is elementary abelian of order 16.
- 3) H'_3 is elementary abelian of order 729 and H_3/H'_3 .
- 4) H_4 is elementary abelian of order 512. is elementary abelian of order 32.

A.9 $\Gamma = T_9[2, 3, 5; 2, 3, 5]$

 Γ^+ has 457 conjugacy classes of low index subgroups of index ≤ 24 . Their index, the induced permutation group and, where possible, the abelianisation of the kernel of the map are listed here.

Index	# classes	Induced action	Abelianization
5		$\int 2 \Rightarrow A_5$	$\mathbb{Z}_2\oplus\mathbb{Z}_3^4\oplus\mathbb{Z}_4^4\oplus\mathbb{Z}_5^3$
	J	$ 1 \Rightarrow A_5 $	\mathbb{Z}^{11}
6	7	$\left(2 \Rightarrow A_5 \right)$	$\mathbb{Z}_2\oplus\mathbb{Z}_3^4\oplus\mathbb{Z}_4^4\oplus\mathbb{Z}_5^3$
		$\left\{ 1 \Rightarrow A_5 \right.$	\mathbb{Z}^{11}
		$4 \Rightarrow A_6$	$\mathbb{Z}^9\oplus\mathbb{Z}_2^{16}\oplus\mathbb{Z}_3^6$
10	7	$i 2 \Rightarrow A_5$	$\mathbb{Z}_2\oplus\mathbb{Z}_3^4\oplus\mathbb{Z}_4^4\oplus\mathbb{Z}_5^3$
		$1 \Rightarrow A_5$	\mathbb{Z}^{11}
		$2 \Rightarrow A_6$	$\mathbb{Z}^9\oplus\mathbb{Z}_2^{16}\oplus\mathbb{Z}_3^6$
		$\begin{array}{c} 2 \Rightarrow A_{10} \end{array}$	not computed
11	10	$\int 2 \Rightarrow L_2(11)$	\mathbb{Z}^{41}
		$\begin{cases} 8 \Rightarrow A_{10} \end{cases}$	not computed
12	20	$2 \Rightarrow A_5$	$\mathbb{Z}_2\oplus\mathbb{Z}_3^4\oplus\mathbb{Z}_4^4\oplus\mathbb{Z}_5^3$
		$1 \Rightarrow A_5$	\mathbb{Z}^{11}
		$1 \Rightarrow L_2(11)$	\mathbb{Z}^{41}
		$4 \Rightarrow A_5 \ltimes H_1$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^8\oplus\mathbb{Z}_3^4\oplus\mathbb{Z}_4\oplus$
			$\mathbb{Z}_5^3 \oplus \mathbb{Z}_8^{10} \oplus \mathbb{Z}_{19}^{12} \oplus \mathbb{Z}_{32}$
		$6 \Rightarrow A_5 \ltimes H_1$	$\mathbb{Z}^{103}\oplus\mathbb{Z}_2^2$
		$4 \Rightarrow M_{12}$	not computed
		$2 \Rightarrow A_{12}$	not computed

Index	# classes	Induced action	Abelianization
15	26	$3 \Rightarrow A_5$	computed above
		$4 \Rightarrow A_6$	computed above
		$6 \Rightarrow A_5 \ltimes H_2$	$\mathbb{Z}^{40}\oplus\mathbb{Z}_2^{81}\oplus\mathbb{Z}_3^{32}\oplus$
			$\mathbb{Z}_4^4 \oplus \mathbb{Z}_5^{33} \oplus \mathbb{Z}_7^{20} \oplus \mathbb{Z}_9^{12}$
		$3 \Rightarrow A_5 \ltimes H_2$	$\mathbb{Z}^{191}\oplus\mathbb{Z}_2^{40}\oplus\mathbb{Z}_3^2$
		$10 \Rightarrow A_{15}$	not computed
16	40	$4 \Rightarrow A_5 \ltimes H_3$	\mathbb{Z}^{71}
		$8 \Rightarrow A_5 \ltimes H_3$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2\oplus\mathbb{Z}_3^4\oplus$
			$\mathbb{Z}_4^{10}\oplus\mathbb{Z}_5^3\oplus\mathbb{Z}_{16}$
		$8 \Rightarrow A_6 \ltimes H_3$	$\mathbb{Z}^{129}\oplus\mathbb{Z}_2^{56}\oplus\mathbb{Z}_3^6\oplus\mathbb{Z}_4^{16}$
		$\begin{array}{c} 20 \Rightarrow A_{16} \end{array}$	not computed
18	17	$\int 1 \Rightarrow A_5 \ltimes H_4$	not computed
		$4 \Rightarrow A_5 \ltimes H_5$	not computed
		$8 \Rightarrow A_5 \ltimes H_6$	not computed
		$\left(4 \Rightarrow A_{18} \right)$	not computed
		$\int 3 \Rightarrow A_5$	computed above
	108	$2 \Rightarrow A_6$	computed above
		$12 \Rightarrow A_5 \ltimes H_3$	computed above
		$10 \Rightarrow A_5 \ltimes H_1$	computed above
20		$\begin{array}{c} 2 \Rightarrow L_2(19) \end{array}$	$\mathbb{Z}_2\oplus\mathbb{Z}_3^{37}\oplus\mathbb{Z}_5^{18}\oplus$
20			$\mathbb{Z}_{8}^{18} \oplus \mathbb{Z}_{11}^{18} \oplus \mathbb{Z}_{17}^{20} \oplus \mathbb{Z}_{19}^{3}$
		$16 \Rightarrow A_5 \ltimes H_7$	not computed
		$4 \Rightarrow A_6 \ltimes H_8$	not computed
		$15 \Rightarrow A_5 \ltimes H_9$	not computed
21	34	$\left \begin{array}{c}38\Rightarrow A_{20}\end{array}\right $	not computed
		$\int 10 \Rightarrow L_3(4)$	not computed
		$24 \Rightarrow A_{21}$	not computed

Index	# classes	Induced action	Abelianization
22	30	$6 \Rightarrow L_3(4) \ltimes H_{10}$	not computed
		$\begin{cases} 8 \Rightarrow A_{11} \ltimes H_{10} \end{cases}$	not computed
		$16 \Rightarrow A_{22}$	not computed
		$(2 \Rightarrow A_5 \ltimes H_1)$	$\mathbb{Z}^{15}\oplus\mathbb{Z}_2^8\oplus\mathbb{Z}_3^4\oplus\mathbb{Z}_4\oplus$
24	155		$\mathbb{Z}_5^3 \oplus \mathbb{Z}_8^{10} \oplus \mathbb{Z}_{19}^{12} \oplus \mathbb{Z}_{32}$
		$3 \Rightarrow A_5 \ltimes H_1$	$\mathbb{Z}^{103}\oplus\mathbb{Z}_2^2$
		$4 \Rightarrow A_5 \ltimes H_{11}$	$\mathbb{Z}^{167}\oplus\mathbb{Z}_2^{32}$
		$4 \Rightarrow A_5 \ltimes H_{12}$	not computed
		$4 \Rightarrow A_5 \ltimes H_{13}$	not computed
		$36 \Rightarrow A_5 \ltimes H_{14}$	not computed
		$1 \Rightarrow L_2(11) \ltimes H_{13}$	not computed
		$5 \Rightarrow A_5 \ltimes H_{15}$	not computed
		$12 \Rightarrow A_5 \ltimes H_{16}$	not computed
		$4 \Rightarrow M_{12} \ltimes H_{13}$	not computed
		$40 \Rightarrow A_5 \ltimes (H_1 \ltimes H_{16})$	not computed
		$2 \Rightarrow A_{12} \ltimes H_7$	not computed
		$\begin{array}{c} 38 \Rightarrow A_{24} \end{array}$	not computed

- 1) H_1 is an elementary abelian group of order 32.
- 2) H_2 is an elementary abelian group of order 81.
- 3) H_3 is an elementary abelian group of order 16.
- 4) H_4 is an elementary abelian group of order 3^5 .
- 5) H_5 is an elementary abelian group of order 3^6 .
- 6) H_6 is the semidirect product of two elementary abelian groups. H'_6 is elementary abelian of order 3^6 and the factor group is elementary abelian order 32.
- 7) H_7 is an elementary abelian group of order 1024.
- 8) H_8 is an elementary abelian group of order 512.

- 9) H_9 is the semidirect product of two elementary abelian groups. H'_9 is elementary abelian of order 2^{10} and the factor group is elementary abelian order 3^4 .
- 10) H_{10} is an elementary abelian group of order 2^{10} .
- 11) H_{11} is an elementary abelian group of order 64.
- 12) $H_{12} = C_4 \times C_4 \times C_4 \times C_4 \times C_4.$
- 13) H_{13} is an elementary abelian group of order $2^{11} = 2048$.
- 14) H_{14} has order 2048 and has elementary abelian derived subgroup of order 32. The factor group is also elementary abelian
- 15) H_{15} is the semidirect product of two elementary abelian groups. H'_{15} is elementary abelian of order 2^{12} and the factor group is elementary abelian order 3^5 .
- 16) H_{16} is the semidirect product of two elementary abelian groups. H'_{16} is elementary abelian of order 2^{12} and the factor group is elementary abelian order 3^{6} .

Bibliography

- [Al] D. Allcock, Infinitely many hyperbolic Coxeter groups through dimension 19, http://www.ma.utexas.edu/ allcock/research/doubling7.pdf
- [An1] E. M. Andreev, On convex polyhedra in Lobachevskij space, Mat. Sb. Nov. Ser. 81 (1970), pp. 445–478 (Russian); English translation Math. USSR Sb. 10 (1970), pp. 413–440.
- [An2] M. T. Anderson, Scalar Curvature and Geometrization Conjectures for 3-Manifolds, Comp. Geom. 30, (1997), pp 49-81.
- [Atlas] J. H. Conway, R. T. Curtis, S. P. Norton, R. A. Parker, R. A. Wilson, Atlas of Finite Groups, Clarendon Press, Oxford, (1985).
- [B] N. Bourbaki, Groupes et algebres de Lie, Chapters 4-6, Hermann, Paris, (1968).
- [BH] B. Brink and R. Howlett, A finiteness property and an automatic structure for Coxeter Groups, Math. Ann. 296 (1993), pp 179-190.
- [CFJR] T. Chinburg, E. Friedman, K. N. Jones, A. W. Reid, *The smallest volume arithmetic hyperbolic 3-manifold*, Annali della Scuola Normale Superiore di Pisa **30** (2001), pp. 1-40.
- [CM] M. Conder and C. Maclachlan, Compact hyperbolic 4-manifolds of small volume, Proc. AMS, 133 (2005), pp. 2469-2476.

- [CMT] M. Conder, G. Martin and A. Torstensson, Maximal symmetry groups of hyperbolic 3-manifolds, New Zealand Journal of Mathematics 35 (2006), pp. 37-62.
- [CPS] E. Cline, B. Parshall and L. Scott, Cohomology of finite groups of Lie type, I, Publ. Math. IHES 45 (1975), pp. 169–191.
- [C1] H. S. M. Coxeter, Twisted Honeycombs, Regional Conference Series in Mathematics 4, Amer. Math. Soc., (1970).
- [C2] H. S. M. Coxeter, *Regular Polytopes*, 3rd Edition, Dover, New York, (1973).
- [C3] H. S. M. Coxeter, Regular Complex Polytopes, Cambridge University Press, (1974).
- [Da] M. W. Davis, A Hyperbolic 4-Manifold, Proc. Amer. Math. Soc., 93 (1985), pp. 325-328.
- [DM1] D.A. Derevnin and A. D. Mednykh, Geometric properties of discrete groups acting with fixed points in Lobachevsky space, Soviet Math. Dokl. 37 No. 3 (1988), pp 614-617.
- [DM2] D. A. Derevnin and A. D. Mednykh, Discrete Extensions of the Lanner Groups, Soviet Math. Dokl. 58 No. 4 (1998), pp. 78-80.
- [DK] D. A. Derevnin and Chi Kim, *The Coxeter prisms in* \mathbb{H}^3 , Research and Exposition in Mathematics **27**, pp. 35-50.
- [Di] L. E. Dickson, Linear Groups with an exposition of the Galois Field Theory, Dover, (1958).
- [E1] B. Everitt, 3-Manifolds from Platonic Solids, Topology and its Applications, 138 (2004), pp. 253-263.
- [E2] B. Everitt, Coxeter Groups and Hyperbolic Manifolds, Math. Ann. 330 (2004), pp. 127-150.
- [EMc] B. Everitt and C. Maclachlan, *Constructing Hyperbolic Manifolds*, Computational and Geometric aspects of Modern Algebra, London Math. Soc. Lect. Notes, **275** (2000) pp. 78-86.

- [EEK] A. L. Edmonds, J. H. Ewing, R. S. Kulkarni, Torsion free subgroups of Fuchsian groups and tessellations of surfaces. Invent. Math. 69 (1982), pp. 331-346.
- [ERT] B. Everitt, J. Ratcliffe and S. Tschantz, The smallest hyperbolic 6-manifolds, Elec. Res. Ann. of the AMS, 11 (2005), pp 40-46
- [FT1] A. Felikson and P. Tumarkin, On finite index reflection subgroups of discrete reflection groups, preprint, arXiv.math.MG/0305093
- [FT2] A. Felikson and P. Tumarkin, Reflection subgroups of Euclidean reflection groups, SB MATH, 196 No. 9 (2005), pp. 1349-1369
- [FT3] A. Felikson and P. Tumarkin, *Euclidean simplices generating discrete reflection groups*, to appear in European J. of Combinatorics.
- [FT4] A. Felikson and P. Tumarkin, On the volume of a six-dimensional polytope, preprint, arXiv.math.MG/0502167
- [FT5] A. Felikson and P. Tumarkin, On simple ideal hyperbolic Coxeter polytopes, preprint, arXiv.math.MG/0502413
- [FT6] A. Felikson and P. Tumarkin, On compact hyperbolic Coxeter d-polytopes with d+4 facets, preprint, arXiv.math.MG/0510238
- [GAP] The GAP Group, GAP Groups, Algorithms, and Programming, Version 4.4.9; 2006, (http://www.gap-system.org).
- [GM1] F. W. Gehring, G. H. Martin, On the Minimal Volume Hyperbolic 3-Orbifold, Math. Research Letters, 1 (1994), pp 107-114.
- [GM2] F. W. Gehring, G. H. Martin, Minimal Co-Volume Hyperbolic Lattices, I: The spherical points of a Kleinian group, preprint.
- [Geom] The Geometry Center, University of Minnesota, (http://www.geom.uiuc.edu/)
- [GLR] C. McA. Gordon, D. Long, A. Reid, Surface subgroups of Coxeter and Artin groups, J. Pure and Applied Algebra 189 (2004), pp. 135-148.

- [Gor1] D. Gorenstein, *Finite Groups*, Harper's Series in Modern Mathematics, Harper and Row (1968).
- [Ha] R. Hamilton, Three-manifolds with positive Ricci curvature, J. Diff. Geom. 17, No. 2 (1982), pp. 255-306.
- [HEB] D. F. Holt, B. Eick, E. O'Brien, *Handbook of Computational Group Theory*, Chapman and Hall/CRC, (2005).
- [HW] R. Howlett *Miscellaneous facts about Coxeter groups*, Lectures given at the ANU Group Actions Workshop (1993).
- [HRT] Howlett, Rowley and Taylor, On Outer Automorphism groups of Coxeter Groups, Manuscripta Math. 93 No. 4 (1997), pp. 499-513.
- [Hu] J. E. Humphreys, Reflection Groups and Coxeter Groups, Cambridge Studies in Advanced Mathematics 29, Cambridge University Press, Cambridge (1997).
- [JKRT] N. W. Johnson, R. Kellerhals, J. G. Ratcliffe, S. T. Tschantz, The size of a hyperbolic Coxeter simplex, Trans. Groups, 4 No. 4 (1999), pp. 329-353.
- [JM] G. A. Jones and A. Mednykh, 3-Dimensional hyperbolic manifolds with a large isometry group, preprint.
- [JT] G.A. Jones, J.S. Thornton, Operations on maps, and outer automorphisms,
 J. Combin. Theory Ser. B 35 No. 2 (1983), pp. 93-103.
- [JR] K. N. Jones and A. W. Reid, Minimal index torsion-free subgroups of Kleinian groups, Math. Ann. 310 (1998), pp. 234-250.
- [JZ] G. A. Jones and A. Zvonkin, Orbits of Braid Groups on Cacti, Moscow Math. J., 2002, 2 No. 1, pp. 129-162
- [KZ] R. Kellerhals and T. Zehrt, The Gauss-Bonnet formula for hyperbolic manifolds of finite volume, Geom. Ded., 84 No. 1-3 (2001), pp. 49-62.
- [La] F. Lanner, On complexes with transitive groups of automorphisms, Comm. Sém. Math. Univ. Lund 11, (1950), pp. 1-71
- [Lo] P. Lorimer, Four dodecahedral spaces, Pacific J. Math. 156 (1992), pp. 329– 335.

- [Lx] P. Dobcsányi, Lowx version 0.11, (http://designtheory.org/~peter/software/lowx/)
- [Me] A. D. Mednykh, Automorphism groups of the three-dimensional hyperbolic manifolds, Soviet Math. Dokl., 32 No. 3 (1985), pp. 633-636.
- [Mo] J. M. Montesinos, Classical Tessellations and Three-Manifolds, Springer -Verlag, Berlin, (1987).
- [Ne] J. Newbüser, An elementary introduction to coset table methods in computational group theory, Groups - St. Andrews (1981), LMS Lecture Note Series 71.
- [Pe1] G. Perelman, The entropy formula for the Ricci flow and its geometric applications, preprint, arXiv:math.DG/0211159.
- [Pe2] G. Perelman, *Ricci flow with surgery on three-manifolds*, preprint, arXiv:math.DG/0303109.
- [Po] H. Poincaré, Second complément a l'analysis situs, Proc. Lond. Math. Soc. 32 (1900), pp. 277-308.
- [RT1] J. G. Ratcliffe and S. T. Tschantz, The Volume Spectrum of Hyperbolic 4-Manifolds, Experiment. Math. 9 No. 1 (2000), pp 101-125.
- [Si] C. C. Sims, Some algorithms based on coset enumeration, Unpublished notes,
- [ST1] H. Seifert and W. Threlfall, Lehrbuch der Topologie, AMS (2006).
- [ST2] H. Seifert and W. Threlfall, A Textbook of Topology, and Topology of 3dimensional Fibered Spaces, Academic Press, New York, (1980).
- [To] A. Torstensson, Algorithmic Methods in Combinatorial Algebra, PhD Thesis (2003).
- [Th1] W. Thurston, Three dimensional manifolds, Kleinian groups and hyperbolic geometry, Bull. Amer. Math. Soc. 6 (1982), pp. 357-381.
- [Th2] W. Thurston, Three-dimensional geometry and topology. Vol. 1, Edited by Silvio Levy. Princeton Mathematical Series, 35, Princeton University Press, Princeton, NJ, 1997.

- [Tu1] P. Tumarkin, Hyperbolic Coxeter n-polytopes with n+2 facets, Mathematical Notes, 75 Nos. 5-6, (2004), pp. 848-854.
- [Tu2] P. Tumarkin, Hyperbolic Coxeter n-polytopes with n+3 facets, Trans. Moscow Math. Soc. 65 (2004), pp. 235-250.
- [Tu3] P. Tumarkin, Compact hyperbolic Coxeter n-polytopes with n+3 facets, preprint, arXiv.math.MG/0406226.
- [V1] E. B. Vinberg, Discrete groups generated by reflections in Lobacevskii spaces, Math. USSR-Sbornik, 1 No. 3 (1967), pp. 429-444.
- [V2] E. B. Vinberg, Hyperbolic reflection groups, Russian Math. Surveys 40 No. 1, (1985), pp. 31-75.
- [VS] E. B. Vinberg and O. V. Shvartsman, Discrete Groups of Motions of Spaces of Constant Curvature, in Geometry II: Spaces of Constant Curvature, Encyclopedia of Mathematical Sciences, 29 (ed. E. B. Vinberg), Springer-Verlag, Berlin (1993), pp. 139–248.
- [W] R. A. Wilson, Notes on Finite Simple Groups, (http://www.maths.qmul.ac.uk/~raw/fsgs.html)
- [Wil] S.E. Wilson, Operators over regular maps, Pacific J. Math. 81 No. 2 (1979), pp. 559-568.
- [WS] C. Weber and H. Seifert, Die beiden Dodekaederäume, Math. Z. 37 (1933), pp. 237–253.
- [Z] B. Zimmermann, On a hyperbolic 3-manifold with some special properties, Math. Proc. Cambridge Philos. Soc., 113 No. 1 (1993), pp. 87-90.